

# CONTAINING ALL NOTICES PUBLISHED ONLINE BETWEEN 1 AND 3 APRIL 2022

PRINTED ON 4 APRIL 2022 | NUMBER 63658 PUBLISHED BY AUTHORITY | ESTABLISHED 1665 WWW.THEGAZETTE.CO.UK

# Contents

# State/6306\*

Royal family/

Parliament Assemblies & Government/

Honours & Awards/

Church/

# Environment & infrastructure/6308\*

Health & medicine/

Other Notices/6341\*

Money/

Companies/6342\*

People/6409\*

Terms & Conditions/6437\*

\* Containing all notices published online between 1 and 3 April 2022

# STATE

# **Honours & awards**

# **State Awards**

#### ORDER OF THE BRITISH EMPIRE

# **CENTRAL CHANCERY OF**

THE ORDERS OF KNIGHTHOOD

St. James's Palace, London S.W.1.

1 April 2022

#### KNIGHT BACHELOR

THE QUEEN has been graciously pleased to signify her intention of conferring the honour of Knighthood upon the undermentioned:

The Right Honourable Gavin Alexander WILLIAMSON, C.B.E., M.P. For Political Service.

(To be dated 3 March 2022)

#### **CENTRAL CHANCERY OF**

# THE ORDERS OF KNIGHTHOOD

St. James's Palace, London S.W.1.

1 April 2022

#### ORDER OF SAINT MICHAEL AND SAINT GEORGE

THE QUEEN has been graciously pleased to give orders for the Honorary appointment of the undermentioned to the Most Distinguished Order of Saint Michael and Saint George to be made

Theodor MERON, C.M.G. (To be dated 8 March 2022)

#### **CENTRAL CHANCERY OF**

#### THE ORDERS OF KNIGHTHOOD

St. James's Palace, London S.W.1.

1 April 2022

# **ROYAL VICTORIAN ORDER**

THE QUEEN has been graciously pleased to give make the following appointment to the Royal Victorian Order:

To be a Commander:

Rear Admiral James Norman MACLEOD, C.B., on his relinquishment of the appointment of Defence Services Secretary.

(To be dated 16 February 2022)

# **CENTRAL CHANCERY OF**

# THE ORDERS OF KNIGHTHOOD

St. James's Palace, London S.W.1.

1 April 2022

### ORDER OF THE BRITISH EMPIRE

THE QUEEN has been graciously pleased to give orders for the Honorary appointments of the undermentioned to the Civil Division of the Most Excellent Order of the British Empire to be made Substantive:

Professor Ronan LYONS, O.B.E. (To be dated 5 January 2022)

Brigitte Zelie SQUIRE-DEHOUCK, M.B.E. (To be dated 5 January

Ann Hutchinson GUEST, M.B.E. (To be dated 8 March 2022)

Diala KHLAT, M.B.E. (To be dated 5 January 2022)

### **CENTRAL CHANCERY OF**

## THE ORDERS OF KNIGHTHOOD

St. James's Palace, London S.W.1.

1 April 2022

Supplement No. 1 to The London Gazette dated 1 January 2022 (No. 63571)

#### ORDER OF THE BATH

# **OVERSEAS AND INTERNATIONAL LIST**

# Page N3 - C.B. (Civil Division)

Andrew Philip BIRD to read Adrian Philip BIRD, Director-General, Foreign, Commonwealth and Development Office. For services to British foreign policy.

# ORDER OF SAINT MICHAEL AND SAINT GEORGE

**OVERSEAS AND INTERNATIONAL LIST** 

Page N3 - C.M.G.

William BORRY to read William BLANCHARD, Director, Foreign, Commonwealth and Development Office. For services to British foreign policy and National Security.

#### ORDER OF THE BRITISH EMPIRE

#### Page N8 - C.B.E. (Civil Division)

Professor Phillip Thomas BLYTHE to read Professor Phillip Thomas BLYTHE, FREng., Chief Scientific Adviser, Department for Transport. For services to Science and Engineering in Transport and

#### Page N13 - O.B.E. (Civil Division)

Marie MALLON to read Marie GREENWOOD, M.B.E. (Marie Mallon), Chair, Labour Relations Agency. For services to Industrial and Employment Relations in Northern Ireland, particularly during the Covid-19 Pandemic.

Horace Michael McEVOY to read Horace Michael McEVOY, M.B.E. For services to the community in North Wales.

#### Page N16 - M.B.E. (Civil Division)

Dr. Wendy Jane Anne ANDERSON to read Dr. Wendy Jane Anne ANDERSON, Respiratory Consultant, Northern Health and Social Care Trust. For services to Healthcare, particularly during Covid-19.

#### Page N17 - M.B.E. (Civil Division)

Christine Rutherford CHISHOLM to read Christine Rutherford CHISHOLM, Founder Trustee, New Jumbulance Travel Trust. For services to Philanthropy and to Charitable Causes.

#### Page N18 - M.B.E. (Civil Division)

Paul Samuel CREELMAN to read Paul Samuel CREELMAN, Foster Carer and Support Worker, North Yorkshire. For services to Fostering and Youth Work.

#### Page N19 - M.B.E. (Civil Division)

Luke Berenger COLLETT-FENSON to read Luke Berenger COLLET-FENSON, Senior Private Secretary to the Chief Medical Officer, Department of Health and Social Care. For Public Service during Covid-19.

#### Page N20 - M.B.E. (Civil Division)

Dr. Adaeze IFEZULIKE to read Dr. Adaeze IFEZULIKE, General Practitioner. For services to Health Inequalities in Black, Asian and Minority Ethnic Communities in Scotland.

#### Page N21 - M.B.E. (Civil Division)

Hardip Parkesh Sigh LAWANA to read Hardip Parkash Singh LAWANA, Senior Officer, Border Force, Home Office. For services to Border Security and to Workplace Wellbeing.

# Page N22 - M.B.E. (Civil Division)

Carol Joan MOODY to read Carol Joan MOODY, Foster Carer, Kent. For services to Foster Care.

Derek Richard MOODY to read Derek Richard MOODY, Foster Carer, Kent. For services to Foster Care.

# Page N23 - M.B.E. (Civil Division)

Rosalind May PAUL to read Rosalind Mary PAUL, Chief Executive Officer and Artistic Director, Scene and Heard. For charitable services to Theatre, to Young People and to Families in Somers Town, London Borough of Camden.

### Page N24 - M.B.E. (Civil Division)

Louise Diane RHODES to read Louise Diane RHODES, British Sign Language Interpreter, University of Wolverhampton. For services to Education and to People with Hearing Impairments.

Dr. Amar Nath RUGHANI to read Dr. Amar RUGHANI, General Practitioner and Provost, Royal College of General Practitioners, South Yorkshire and North Trent. For services to General Practice.

#### **OVERSEAS AND INTERNATIONAL LIST**

#### Page N27 - O.B.E. (Civil Division)

Rebecca Ann HARRIS to read Rebecca Ann HARRIS, Team Leader, Foreign, Commonwealth and Development Office. For services to British foreign policy.

Paul Richard Peter KILLWORTH to read Paul Richard Peter KILLWORTH, Deputy Director, Foreign, Commonwealth and Development Office. For services to British foreign policy.

#### **BRITISH EMPIRE MEDAL**

# Page N29 - B.E.M. (Civil Division)

Nadia Nasreen AHMED to read Nadia Nasreen AHMED, lately Community Champion, Morrisons. For services to the community in Edinburgh.

Samuel Wesley AITCHISON to read Samuel Wesley ATCHISON, lately Editor, Tyrone Constitution and Strabane Weekly News. For services to Journalism and to the community in County Tyrone.

Jacqueline BAXTER to read Jacqueline BAXTER, Environmental Campaigns Coordinator Officer, London Borough of Bromley. For services to the community in Bromley, Kent.

#### Page N30 - B.E.M. (Civil Division)

Charlene Joanne BURNS to read Charlene Joanne BURNS, Co-Founder and Director, Real Education Empowering Lives. For services to Social Inclusion and to the community in Oldham, Greater Manchester.

#### Page N31 - B.E.M. (Civil Division)

Julie GOODWIN to read Julie GODWIN (Julie Harrison), Sergeant, West Midlands Police. For services to Policing.

#### Page N33 - B.E.M. (Civil Division)

Elizabeth Sandra KINNEAR to read Sandra Elizabeth KINNEAR, Health, Safety and Sustainability Advisor, University of Edinburgh. For services to the Environment and to Charity.

Catherine McKEE to read Catherine McKEE, Senior Playgroup and Family Project Manager. For services to Children and Families in North and West Belfast and Beyond.

#### Page N34 - B.E.M. (Civil Division)

Gita NATAJARAN to read Gita NATARAJAN. For services to the community in Rugby, Warwickshire.

Adria PITTOCK, Chartered Environmentalist and Chair, Suffolk Pride. For services to the Environment and to the LGBTQ+ community in Suffolk.

#### (To be deleted - published in error)

#### Page N35 - B.E.M. (Civil Division)

Irfran Hussain SHAH to read Irfan Hussain SHAH. For services to Young People and to the community in the London Borough of Redbridge, particularly during Covid-19.

The Reverend Keith Leslie OSMOND-SMITH to read The Reverend Keith Leslie OSMUND-SMITH. For services to the community in Telford, Shropshire, particularly during Covid-19.

# Page N36 - B.E.M. (Civil Division)

Mark Douglas Ashley STRACHAN to read Mark Douglas Ashley STRACHAN, Founder, The Choir of the Earth. For charitable services to Musicians during Covid-19.

Willam James Crawford WILSON to read William James Crawford WILSON. For services to the community in County Antrim.

#### **OVERSEAS AND INTERNATIONAL LIST**

#### Page N37 - B.E.M. (Civil Division)

Alexandra Mary HENNESSEY to read Alexandra Mary HENNESSY, Diary Manager to the Minister of State, Foreign, Commonwealth and Development Office. For services to British foreign policy.

Brian WARING to read Brian WARING, Team Leader, Foreign, Commonwealth and Development Office. For services to British foreign policy.

# **ERRATA**

The London Gazette dated 4 October 2021 (No. 63491 - Page 17512) to read:

# **CENTRAL CHANCERY OF**

### THE ORDERS OF KNIGHTHOOD

St. James's Palace, London S.W.1.

4 October 2021

# ORDER OF SAINT MICHAEL AND SAINT GEORGE

THE QUEEN has been graciously pleased to give orders for the Honorary appointments of the undermentioned to the Most Distinguished Order of Saint Michael and Saint George to be made Substantive:

Baron Peter Karel PIOT, K.C.M.G. (To be dated 30 July 2021)

Dr. Stefan Nicolaas DERCON, C.M.G. (To be dated 30 July 2021)

### ORDER OF THE BRITISH EMPIRE

THE QUEEN has been graciously pleased to give orders for the Honorary appointments of the undermentioned to the Civil Division of the Most Excellent Order of the British Empire to be made Substantive:

Professor Dame Uta FRITH, D.B.E. (To be dated 30 July 2021)

Catherine Ann Pollak MOSS, M.B.E. (To be dated 30 July 2021)

#### **BRITISH EMPIRE MEDAL**

THE QUEEN has been graciously pleased to give orders for the Honorary appointment of the undermentioned to the Civil Division of the British Empire Medal to be made Substantive:

Mary Caroline SENE, B.E.M. (To be dated 30 July 2021)

#### **ERRATA**

Supplement No. 1 to The London Gazette dated 10 October 2020 (No. 63135)

# BRITISH EMPIRE MEDAL

# Page B32 - B.E.M. (Civil Division)

Wendy Faye WELBERRY to read Wendy Fay WELBERRY, Assistant Private Secretary, Foreign, Commonwealth and Development Office. For services to National Security. (4033764)

#### STATE APPOINTMENTS

# DEPUTY LIEUTENANT COMMISSIONS LIEUTENANCY OF WEST MIDLANDS

Her Majesty's Lord-Lieutenant of West Midlands, Mr John Crabtree OBE, has appointed the following to be Deputy Lieutenants of West Midlands:

Mrs Pauline Black OBE

Mr Afzal Hussain

Professor Monder Ram OBE

Mr Jatinder Sharma OBE

Ms Jean Templeton

Mrs Dawn Patricia Ward CBE

Miss Kate Canty

Clerk to the West Midlands Lieutenancy

1st April 2022

(4037766)

# DEPUTY LIEUTENANT COMMISSIONS WEST YORKSHIRE LIEUTENANCY

Commissions have now been signed by Her Majesty's Lord-Lieutenant for the County of West Yorkshire appointing: -

Adeeba Malik CBE

John Robins QPM

Dr Kuldip Bharj OBE

His Honour Roger Thomas QC

Roohi Collins

Suzanne Watson

Kevin Sharp LVO, DL, Clerk to the West Yorkshire Lieutenancy

(4037767)

# ENVIRONMENT & INFRASTRUCTURE

#### **ENERGY**

# NOTICE OF REVISIONS TO THE NATIONAL TERMS OF CONNECTION

DCUSA Ltd whose registered office is situated at Northumberland House, 303-306 High Holborn, London, WC1V 7JZ, hereby gives notice of a revision to the National Terms of Connection (NTC). The NTC set out rights and duties in relation to the connection at which a Licensed Distribution Network Operator (LDNO) delivers electricity to, or accepts electricity from a home or a business.

The revisions will be made to Sections 3 and 4 of the NTC and will be implemented via version 14.1 of the Distribution Connections Use of System Agreement (DCUSA) which will become effective on 01 April 2022. The revisions being made to the NTC are simply the removal of two instances of gendered language used in the text and replacing it with gender neutral language to ensure that the drafting reflects a more diverse and inclusive approach.

If you want a copy of the NTC or have any questions about it, please write to Energy Networks Association, 1st Floor, 4 More London Riverside, London, SE1 2AU or see the website at www.connectionterms.co.uk. (4036505)

# **Communications**

#### **POSTAL SERVICES**

# THE ROYAL MAIL OVERSEAS POST SCHEME 06 April 2022

#### Article I. Contents

1	About This Scheme	2
2	What This Scheme Applies To	2
3	Definitions	2
4	Our Ability To Provide Services	2
5	What Can And Cannot Be Contained Within An Item	3
6	Size and Weight Limits and How To Package An Item	10
7	How To Address An Item	11
8	How To Pay For Postage And Other Services	11
9	How To Work Out How Much Postage To Pay	12
10	How To Show That An Item Has Had Postage Paid	13
11	Other Requirements Relating To The Address On The Cover Of An Item	14
12	How To Hand Over (or Post) An Item	15
13	How We Will Treat Items That Do Not Meet the Requirements Set Out In This Scheme	16

14	Additional Marks We May Add	17
15	Delivery Including Undeliverable and Re-Posted (Return To Sender) Items	17
16	Complaints handling process	19
17	Compensation – What We Are Liable For	20
18	Additional Terms and Conditions For Some Services / Items	25
19	International Signed and International Tracked & Signed	26
20	International Tracked	26
21	HM Forces Mail (with and without Special Delivery™ or Signed For™)	27
22	Articles For The Blind	28
23	Printed Papers	29
24	Customs Control	30
25	Your information	30
Annex A - Definitions used in this Scheme		32

#### 1 About This Scheme

1.1 This Scheme is a document that sets out the terms and conditions for some of the **items** that we handle. **We** (Royal Mail Group Ltd) are allowed to set out terms and conditions within a Scheme document by the Postal Services Act 2000 (and more specifically section 89 of that Act as amended by the Postal Services Act 2011 and in particular paragraph 24 of Schedule 12 of that Act).

1.2 Each time we alter the wording within a Scheme it has the effect of making a new Scheme. We publish our Schemes on our website (www.royalmail.com).

1.3 This Scheme is known as 'The Royal Mail Overseas Letter Post Scheme 25 May 2018' and is referred to in this document as 'this Scheme'. The date in the title is the date that the Scheme took (or will take) effect. This Scheme replaced an old Scheme which was called the Royal Mail Overseas Letter Post Scheme 30 January 2017 which is no longer in force. This Scheme complies with requirements as set out by the Universal Postal Union (www.upu.int).

#### 2 What This Scheme Applies To

2.1 This Scheme sets out the terms and conditions for:

a) the services provided by us to the person, **business** or organisation asking for the service (**you**) when we accept items within the United Kingdom for postal delivery to an **address** outside the United Kingdom, the **Channel Islands** and the Isle of Man. The terms and conditions for the specific services listed below are contained within this Scheme:

- International Standard
- International Economy
- International Signed
- International Tracked & Signed
- International Tracked
- $\bullet$  HM Forces Mail (with or without Special Delivery  $^{\text{\tiny TM}}$  or Signed For  $^{\text{\tiny TM}}$ )
- Articles for the Blind

All product names offered under this Scheme are shown in this Scheme in red text; and

#### b) incoming items.

#### 3 Definitions

3.1 You may not be familiar with some of the words or phrases we use in this document. When they are used for the first time, they are shown in **bold print** and explained in the relevant part of this document or in Annex A.

3.2 As mentioned in 1.3 above, this Scheme replaced a previous Scheme so any references in other documents to the Overseas Letters Scheme that was in force before this Scheme should now be interpreted as referring to this new Scheme.

4 Our Ability To Provide Services

- 4.1 We will provide the services set out in this Scheme where the relevant terms and conditions contained in this Scheme for the service are complied with. We will provide the services in the manner that we deem to be most appropriate according to our operational requirements and an overarching need to secure the sustainability of the Universal Service.
- 4.2 We may decide to withdraw the provision of any of the services set out in this Scheme if the terms and conditions contained within this Scheme are not abided by or in the event of misuse or if providing the service may cause us reputational damage.

#### 5 What Can And Cannot Be Contained Within An Item

5.1 There are some items that we cannot carry through our postal network for legal or safety reasons. Items that we cannot carry through our postal network are known as **prohibited items**. Some other items can only be carried under certain circumstances. These are known as **restricted items** and details of those are contained after the section on prohibited items.

#### Prohibited Items

- 5.2 We would like to make it very clear that we will not accept any liability for any item that contains prohibited items.
- 5.3 As the list of prohibited items can change from time to time we maintain a full, up-to-date list on our website (www.royalmail.com) which can be viewed at any time. The list quoted below was correct at the time of publication of this Scheme:
- 1) Any article or substance classified as dangerous goods (other than specified lithium batteries contained in equipment);
- 2) Aerosols
- 3) Alcoholic beverages with an alcohol content greater than 24% ABV;
  4) Ammunition (excluding lead pellets and other airgun and airsoft
- 4) Ammunition (excluding lead pellets and other airgun and airsoft projectiles);
- 5) Asbestos;
- 6) Batteries that are classed as dangerous goods by the latest edition of the Technical Instructions for Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO), when sent in mail including:
- i) non-spillable batteries that meet Special Provision A67 (e.g. sealed lead-acid, absorbed glass mat and gel cell batteries); and
- ii) spillable lead acid/lead alkaline batteries (e.g. car batteries), used alkaline and nickel metal hydride batteries, lithium batteries when not sent in equipment and damaged batteries of any type;
- 7) Balloons filled with non-flammable gas;
- 8) Biological substances including diagnostic specimens (e.g. blood, urine, faeces and animal remains). Biological substances, Category B (UN3373) as classified in the latest edition of the Technical Instructions for Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO);
- 9) Bladed items
- 10) Clinical and medical waste (e.g. contaminated dressings, bandages and needles);
- 11) Controlled drugs and narcotics (such as cannabis, cocaine, heroin, LSD, opium, and amyl nitrate). Those discovered in transit will be stopped and handed to Customs or the Police who may take legal action against the sender and/or recipient;
- 12) Corrosives (including dyes, acids, corrosive paint and rust removers, caustic soda, mercury and gallium metal);
- 13) Counterfeit currency, bank notes and stamps (including any false instrument, or copy of a false instrument within the meaning of section 5 of the Forgery and Counterfeiting Act 1981);
- 14) Dry ice (UN1845) when used as a coolant for UN3373 or for other perishable items;
- 15) Electronic items sent with lithium batteries of any kind (including laptops, mobile phones, digital cameras, MP3 players, portable DVD players and Sat Navs etc) when not installed in the electronic item;
- 16) Environmental waste (including used batteries and used engine oil):
- 17) Explosives (including fireworks, flares, blasting caps, and Christmas cracker snaps);
- 18) Flammable liquids (including petroleum, lighter fluid, certain adhesives, solvent based paints, wood varnish, enamels and acetones including nail varnish removers);
- 19) Flammable solids (including magnesium, phosphorous, potassium, sodium, zinc and fire lighters);
- 20) Foreign Lottery tickets and/or promotional materials for foreign lotteries, lottery syndicates or similar schemes;
- 21) Frozen water (e.g. packs of ice);

- 22) Gases including flammable, non-flammable, toxic and compressed gases new, used and empty gas cylinders, butane, ethane, refills for lighters, fire extinguishers and scuba tanks, life jackets, nitrogen dioxide and carbon dioxide canisters (culinary foaming devices & soda stream gas cylinders);
- 23) Goods made in foreign prisons, except those imported for a non-commercial purpose, of a kind not manufactured in the United Kingdom, or those in transit;
- 24) Hover boards or any type of rechargeable battery-powered: self-balancing scooter, mono-wheel, stand-up unicycle or electric skateboard:
- 25) Human and animal remains including ashes, other than as set out at 5.13 (h) below;
- 26) Obscene publications and unlawful indecent images and pornography, including pornography as detailed in Part 5 of the Criminal Justice and Immigration Act 2008 and indecent photographs or pseudo-photographs of a child as detailed in section 160 and 161 of the Criminal Justice Act 1988;
- 27) Infectious substances and pathogens (UN2814 or UN2900) as classified in the latest Edition of the Technical Instructions for Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO);
- 28) Lighters and refills containing flammable liquid or gas (including used butane and petrol cigar and cigarette lighters);
- 29) Liquids over 1 litre;
- 30) Live animals and reptiles (e.g. snakes, mice and rodents);
- 31) Live creatures, insects and invertebrates classified as dangerous within the Dangerous Wild Animals Act 1976 (e.g. venomous spiders);
- 32) Magnetised material with a magnetic field strength of 0.159A/ metre or more at a distance of 2.1 metres from the outside of the package (e.g. PA systems);
- 33) Matches;
- 34) Medicines and drugs classified as dangerous goods in the latest edition of the Technical Instructions for the Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO) including Cytotoxic medicines;
- 35) Nail varnish or polish;
- 36) Oxidising materials or organic peroxides (including disinfectants, nitrates, and hair dyes or colourants containing peroxide);
- 37) Perfumes and aftershaves (including eau de parfum and eau de toilette);
- 38) Perishables (including flowers, fresh fruit, vegetables and frozen or chilled foodstuffs);
- 39) Pesticides (including weed killer and any chemical used to kill pests and insects, including fly sprays);
- 40) Poisons, toxic liquids, solids and gases (including substances that are liable to cause death or injury if swallowed or inhaled or by skin contact such as arsenic, beryllium, cyanide, fluorine and rat poison);
- 41) Radioactive material and samples that are classified as dangerous goods in the latest edition of the Technical Instructions for the Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO) such as luminous dials from aircraft;
- 42) Scam Mail being mail that we reasonably consider to contain fraudulent, misleading or coercive, and/or which is otherwise in breach of UK law and which seeks to solicit payments or some other benefit from the recipient;
- 43) Solvent-based paints, wood varnishes and enamels;
- 44) Waste, dirt, filth or refuse (including household waste) Note soil samples sent for analysis are permitted provided they are packaged so that any leaks and spills are contained in the outer packaging; and
- 45) Weapons (including Section 5 firearms, CS gas and pepper sprays, flick knives and other knives that are banned knives under UK laws, taser and stun guns).
- 5.4 Any item which resembles a prohibited item may be subject to additional scrutiny which may cause delay and we may deal with such item in our sole and absolute discretion (without incurring any liability whatsoever to you or the **addressee**) including destroying or otherwise disposing of the relevant item (in whole or in part).
- 5.5 Failure to comply with these conditions could result in your prosecution which might result in you facing a fine or imprisonment.
- 5.6 If you **post** an item that contains a prohibited item or if the item is an incoming item that contains a prohibited item, we may deal with such item in our sole and absolute discretion (without incurring any liability whatsoever to you or the **addressee**) including destroying or otherwise disposing of the relevant item (in whole or in part).

- 5.7 The sender, or their representative, may be required to pay an amount (to be fixed by us) to cover the costs of returning, dealing with or disposing of the item.
- 5.8 The sender is responsible for checking whether an item is prohibited. We may also take appropriate action, including refusing to carry an item, if an item is banned by law (including under sanctions laws) or which, in our opinion, may be harmful or dangerous to our customers or employees (whether or not an item is prohibited).
- 5.9 All countries have their own rules regarding prohibitions on what they will allow in the post. These rules can vary from country to country, and can sometimes apply to items which you may think are ordinary. You are responsible for checking whether an item you send is prohibited in the country you are sending it to.
- 5.10 It is your responsibility not to send prohibited items in the post. If you are in any doubt you may seek information from our website (www.royalmail.com).

#### Restricted Items

- 5.11 There are some items, known as restricted items that we can, for legal or safety reasons, only carry through our postal network if you meet all our requirements for the acceptance of the item. Our requirements include packaging requirements as well as other requirements.
- 5.12 We will not accept any liability for any item that contains restricted items that we do carry if the requirements for the acceptance of those items are not met and we may refuse to carry and deliver those items.
- 5.13 As the list of restricted items and requirements for each can change from time to time we maintain a full, up-to-date list on our website (www.royalmail.com) (which can be viewed at any time). The list and details quoted below were correct at the time of publication of this Scheme.
- a) Alcoholic beverages with an alcohol content less than 24% ABV (e.g. wine and champagne): Volume per item should not exceed 1 litre per container. Wrap in polythene and seal with tape. Surround with absorbent material such as newspaper and sufficient cushioning material to protect each item from breakage. Mark as "FRAGILE" when sending glass bottles. The sender's name and return address must be clearly visible on the outer packaging.
- b) Batteries, specifically new and used lithium metal and lithium alloy when sent in equipment (e.g. non-rechargeable): Each cell and battery must be of a type proven to meet the requirements of each test in the UN Manual of Tests and Criteria, Part III, subsection 38.3. Batteries are subject to these tests irrespective of whether the cells of which they are composed have been so tested. Cells and batteries must be manufactured under a quality management programme as specified in the International Civil Aviation Organization (ICAO) Technical Instructions for the Safe Transport of Dangerous Goods by Air. Cells or batteries that are defective for safety reasons, or that have been damaged are prohibited. Any person preparing or offering cells or batteries in equipment for transport must receive adequate instruction on the requirements commensurate with responsibilities. Each package must contain no more than four cells or two batteries installed in equipment. The lithium content of a lithium metal/alloy cell or battery must not be more than 1g per cell or 2g per battery. For lithium ion/polymer cells and batteries, the watt-hour rating must not exceed 20Wh per cell or 100Wh per battery. Cells and batteries must be protected against short circuit. The equipment containing cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidental activation. The sender's name and return address must be clearly visible on the outer packaging. Each package is subject to the maximum weight limits of 2kg (see section 6.2)
- c) Batteries, specifically new alkaline, nickel metal hydride (NiMH) or nickel cadmium (NiCd): Must be new and sent unopened in their original retail packaging. Surround with cushioning material e.g. bubble wrap. The sender's name and return address must be clearly visible on the outer packaging.
- d) **Christmas crackers**: Can only be sent new in their made up form in their original retail packaging. The sender's name and return address must be clearly visible on the outer packaging.
- e) **Currency, bank notes and stamps**: Do not write or indicate in any way that cash or coins are enclosed within the package. The sender's name and return address must be clearly visible on the outer packaging.

- f) **Financial instruments** (e.g. cheques and postal orders): Do not write or indicate in any way that cheques or postal orders are enclosed within the package.
- g) **Guns for sporting use**: The sender's name and return address must be clearly visible on the outer packaging.
- h) **Human and animal ashes**: must have a volume per item not exceeding 50 grams and must be placed in a sift-proof container and securely closed. Items must be tightly packed in strong outer packaging and must be secured or cushioned to prevent any damage. The sender's name and return address must be clearly visible on the outer packaging.
- i) **Lighters (when new, empty and unused)**: Must be sent unopened in their original retail packaging. The sender's name and return address must be clearly visible on the outer packaging.
- j) Live creatures, insects and invertebrates (e.g. bees, caterpillars, and stick insects): Bees, caterpillars, cockroaches, crickets, destroyers of noxious pests, earthworms, fish fry and eggs, leeches and other parasites, lugworms, maggots, mealworms, pupae and chrysalides, rag worms, silkworms, spiders, stick insects and some other insects are allowed. Must be boxed and packaged to protect the creatures, our staff and our customers from harm. Use International Standard as the minimum service. Items must be clearly marked "URGENT LIVING CREATURES HANDLE WITH CARE". The sender's name and return address must be clearly visible on the outer packaging.
- k) Magnetised materials, other than those that are prohibited (including loud speakers): The magnetised material must have a magnetic field strength of less than 0.159A/m at a distance of 2.1m from the outside of the package. Wrap soft packing material at least 2cm thick around each item. The sender's name and return address must be clearly visible on the outer packaging.
- I) Prescription medicines and drugs sent for scientific or medical purposes that are not classified as dangerous goods in the latest edition of the Technical Instructions for the Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO) including asthma inhalers: May only be sent by, or at the specific request of, a qualified medical practitioner, registered dental practitioner, veterinary surgeon, registered nurse or a recognised laboratory or institution. The medicines must be securely closed and placed in a leak-proof container such as a sealed polythene bag (for liquids) or a sift-proof container (for solids). Must be tightly packed in strong outer packaging and must be secured or cushioned to prevent any damage. The sender's name and return address must be clearly visible on the outer packaging.
- m) Radioactive material and samples that are not classified as dangerous goods in the latest edition of the Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization's Technical Instructions (ICAO) e.g. samples of granite rock: Surround with cushioning material e.g. bubble wrap. The sender's name and return address must be clearly visible on the outer packaging.
- n) Sharp objects and instruments (which are not classified as prohibited items): Wrap heavy cardboard around sharp edges and points, strong enough to ensure that the contents do not pierce the outer packaging. Wrap each item with cushioning material. Place in a suitable outer container such as a padded envelope. The sender's name and return address must be clearly visible on the outer packaging.
- o) Vaccines that are not classified as dangerous goods in the latest edition of the Technical Instructions for the Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO): May only be sent by, or at the specific request of, a qualified medical practitioner, registered dental practitioner, veterinary surgeon, registered nurse or a recognised laboratory or institution. Must be tightly packed in strong outer packaging and must be secured or cushioned to contain any leakage and to prevent any damage to the individual items contained within the package. The sender's name and return address must be clearly visible on the outer packaging.
- p) Water-based paints, wood stains and enamels: Volume per item should not exceed 150ml. In the case of water-based paints, wood stains and enamels, there is no restriction on the number of items than can be sent in each package. Wrap in polythene and seal with tape. Surround with absorbent material such as newspaper and sufficient cushioning material to protect each item from damage. The sender's name and return address must be clearly visible on the outer packaging.

5.14 Any item which resembles a restricted item may be subject to additional scrutiny which may cause delays and we may deal with such item in our sole and absolute discretion (without incurring any liability whatsoever to you or the **addressee**) including destroying or otherwise disposing of the relevant item (in whole or in part).

5.15 If you post an item that contains a restricted item (and you do not comply with the relevant restrictions or requirements) or if the item is an incoming item and is discovered not to comply with the relevant restrictions or requirements, we may deal with such item in our sole and absolute discretion (without incurring any liability whatsoever to you or the addressee) including destroying or otherwise disposing of the relevant item (in whole or in part).

5.16 The sender, or their representative, may be required to pay an amount (to be fixed by us) to cover the costs of returning, dealing with or disposing of the item.

5.17 The sender is responsible for checking whether an item is restricted and, if it is, for making sure the requirements for that type of restricted item are met. We may also refuse to accept or deliver any item which is not a restricted item but which is banned by law or which in our opinion may be harmful or dangerous to our customers or employees.

5.18 All countries have their own rules regarding restrictions on what they will allow in the post. These rules can vary from country to country, and can sometimes apply to items which you may think are ordinary. You are responsible for checking whether an item is restricted in the country you are sending it to and for making sure it complies with any requirements in that country.

5.19 It is your responsibility to ensure that restricted items are sent correctly and that you meet the requirements for acceptance. If you are in any doubt as to whether an item is one of the restricted items you can seek guidance from our website (www.royalmail.com).

#### Sanctions Laws

5.20 You must make sure that any items you wish us to handle and deliver are not prohibited under applicable sanctions laws, for example because of their contents, their intended recipient or the country to or from which they are to be sent. Information about sanctions can be found on our website at www.royalmail.com/international-sanctions (or any replacement URL). Further information on trade controls and economic sanctions can be found on the UK Government's web pages at https://www.gov.uk/overview-of-export-control-legislation and https://www.gov.uk/sanctions-embargoes-and-restrictions.

5.21 If your items need a licence under applicable sanctions laws it is your responsibility to obtain it and (if we ask for it) you must provide us with acceptable evidence that you have it.

5.22 If we have reasonable suspicion that, because of its contents, an item does not comply with sanctions laws we may open that item or delay processing and delivery.

5.23 If you post an item which does not comply with sanctions laws we may deal with such item in our sole and absolute discretion (without incurring any liability whatsoever to you, your intended recipient and/or, to the extent not prohibited by applicable law, any third party), including destroying or otherwise disposing of such item in whole or in part, or returning the relevant item to you. We are entitled to charge you:

- the cost of disposal and/or destruction;
- the standard postage price;
- $\bullet$  all other costs reasonably incurred by us; and/or
- for any losses or damage we suffer or liability we incur as a result of you not complying with sanctions laws.

5.24 We will bear no responsibility in the event that you or your agents send an item with the wrong licence required under sanctions laws.

#### Ability to claim compensation

5.25 If you fail to comply with any of the conditions in this section 5, your ability to claim compensation for any item (in circumstances where you might ordinarily be entitled to compensation) may be lost, despite any of the other sections in this Scheme which deal with compensation.

#### Valuables

5.26 **Valuables** should only be sent using International Signed, International Tracked & Signed and International Tracked. Valuables sent under the HM Forces service shall be sent using the Special Deliverv™ service.

6 Size and Weight Limits and How To Package An Item

6.1 As well as the specific packaging and other requirements that relate to restricted items, there are some general rules set out below that must be followed for all items. The rules cover the weight, size, thickness and packaging of items.

6.2 Items sent using the International Standard, International Economy, International Signed, International Tracked & Signed and International Tracked services may weigh up to 2 kilograms unless the item is a letter or large letter, in which case section 6.3 applies. Items sent as Printed Papers may weigh up to 5 kilograms. Items sent using the Articles for the Blind service may weigh up to 7 kilograms.

6.3 The maximum size of any item with the length, width and depth combined must not exceed 900mm with the greatest dimension not exceeding 600mm. For a cylinder shaped item the length plus twice the diameter may not exceed 1040mm with the greatest dimension being no more than 900mm. To be sent as a letter the maximum weight must not exceed 100grams and size must not exceed 240 x 165mm with a maximum thickness of 5mm. To be sent as a large letter the maximum weight must not exceed 750g, and size must not exceed 353mm x250mm with a maximum thickness of 25mm. (please note that the letter and large letter sizes quoted here mirror standard UK letter and large letter dimensions). To be sent as a postcard the maximum size must not exceed 120 x 235mm.

 $6.4\,\mathrm{The}$  minimum thickness for any item is 0.25mm. We may refuse to accept or to deliver an item that is less than 0.25mm thick.

6.5 All items must be packaged so that, in our opinion, the item will not be likely to cause damage or harm to other items, equipment or individuals (including any recipient).

6.6 The **cover** and the protective material of the item must be appropriate to the contents so that we can convey it to its destination undamaged without having to take precautions over and above those normally taken by us with an item of mail. Packaging guidelines are available on our website (www.royalmail.com) and in booklets held by Post Office® branches.

6.7 Apart from items that are sent using the Articles for the Blind service (which is discussed in more detail in section 22) all items must be securely sealed or fastened.

6.8 As well as the requirement to provide sufficient protection both for the item itself and so as not to be likely to cause damage or harm to other items, equipment or individuals (as set out in sections 6.5 and 6.6), any item that contains anything breakable should be placed in a strong box filled with appropriate protective materials and must be clearly and legibly marked with the words 'FRAGILE – HANDLE WITH CARE' in capital letters above the address.

6.9 As well as the requirement to provide sufficient protection for the contents (as set out in sections 6.5 and 6.6) any item that has contents that may be damaged by bending must be clearly and legibly marked with the words 'DO NOT BEND' in capital letters above the address

6.10 The latest packaging guidelines which apply to all items can also be found on our website (www.royalmail.com).

# 7 How To Address An Item

7.1 Each item must be fully and correctly addressed. All the elements of the address (see section 7.2 for guidance) must be written clearly and legibly on the front or on a label securely attached to the front of the cover (or in another way so that the address is clearly visible in its entirety as if it had been written on the front of the cover) of the item.

7.2 You must ensure that addresses take the following format when added to an item. Please do not use commas or full stops and note that the LOCALITY NAME, **ZONE NUMBER** or POST OFFICE BOX NUMBER and the COUNTRY NAME should be in CAPITAL LETTERS in English. The text that makes up the address should be aligned to the left as it is in the example below. It should not be staggered or centred.

Example Name

House name or number and road name

LOCALITY NAME and ZONE NUMBER if one exists COUNTRY

The text set out in the example above and taken as a whole is known as the address block.

7.3 The address block must not be obscured or made difficult to read in any way including by the manner in which the contents are enclosed in the item.

7.4 For items sent using the International Standard service, our branded Airmail sticker (available free of charge from all Post Offices®) should be placed on the address side, in the top left hand corner. Alternatively the words "BY AIRMAIL – PAR AVION" should be written in capital letters in the same position.

7.5 Putting a correct address in the correct format in the correct location helps our sorting machines process the mail as quickly as possible. More information is available on our website (www.royalmail.com).

7.6 We recommend that you put a full and accurate return address on your item (in a position where it can't be confused with the destination address and preferably on the back) so that we are able to return the item to you, intact, should we need to. More information regarding Return to Sender can be found in section 15.

#### 8 How To Pay For Postage And Other Services

- 8.1 Of the services contained within this Scheme only the Articles for the Blind service is provided (as long as specific requirements are met) free of charge. The specific requirements for this service are set out in section 22 of this Scheme.
- 8.2 The other services that are contained in this Scheme require payment for **postage**. These services are:
- International Standard
- International Economy
- International Signed
- International Tracked & Signed
- International Tracked
- HM Forces Mail (with or without Special Delivery<sup>™</sup> or Signed For<sup>™</sup> add-ons);

The specific requirements for International Signed and International Tracked & Signed, International Tracked and HM Forces Mail are set out below in sections 19, 20 and 21 of this Scheme.

- a. The services listed above in 8.2 can be paid for by applying **postage stamps, postage labels** (which can be generated by our online postage applications or by Post Office® branches) or by use of a franking machine.
- b. All services listed in 8.2 can also be purchased using a credit account.
- 8.3 Stamps can be purchased from us online from our website (www.royalmail.com), from Post Office® branches or from a wide variety of other retailers. When postage is purchased from a Post Office® branch it often takes the form of a postage label if it is for anything but the first weight band. Similar postage labels are also available from our online postage tool and through our computer applications (e.g. SmartStamp®) which you may subscribe to or sign up for.
- 8.4 A franking machine allows money to be credited onto it. Postage for items is then paid for, from this credit, on an item-by-item basis. There is a wide range of franking machines that can either be leased or bought from franking manufacturers. If you use a franking machine you must abide by the terms and conditions that are contained within a separate Scheme called Royal Mail Scheme For Franking Letters And Parcels 2014 (which is published on our website (www.royalmail.com)) as well as the terms and conditions contained in this Scheme.
- 8.5 A credit account allows you to post items without pre-payment. You are sent an invoice periodically requesting payment for the items we have conveyed during the invoice period. If you set up a credit account with us you must abide by the terms and conditions that are contained within documents called the Royal Mail General Terms and Conditions and the Royal Mail Account Terms (both of which can be found on our website (www.royalmail.com)) as well as the terms and conditions contained in this Scheme.

### 9 How To Work Out How Much Postage To Pay

- 9.1 For the services listed in this Scheme that are not free, the amount of postage that you must pay is dependent on several factors. The first is the destination; the second is how quickly you would like the item to arrive and what options you would like for tracking and obtaining a signature on delivery (the service); the third is the shape and weight of the item (the weight and format); and the fourth is how much compensation is required in the event of loss or damage (the compensation). For the details of how we price, based on all of these factors, please see our website (www.royalmail.com) as well as immediately below for information in relation to the service, weight and format and compensation factors:
- a. The service: we offer the services set out in section 8.2. Details of the services, their availability and delivery aims (by service and destination) can be found on our website (www.royalmail.com). Not all services are available for all destinations. Further specific details about International Signed, International Tracked & Signed, International Tracked, HM Forces Mail and Articles for the Blind services, can be found in sections 19 22 below and on our website (www.royalmail.com).

- b. The weight and format: for pricing purposes items are classed as either letters, large letters or parcels. The requirements for an item to be classed as a letter or large letter are set out in section 6.3. All other items which exceed the weight or size limits for letters or large letters but do not exceed the maximum dimensions for an item under this Scheme (described within section 6 of this Scheme) are classed as parcels for pricing purposes.
- c. The compensation: International Standard and International Economy services include standard compensation for loss and damage on the basis of **actual loss**, up to a maximum of the **market value** of the item or  $\Sigma$ 20 (whichever is lower). International Signed, International Tracked & Signed and International Tracked include compensation for loss and damage on the basis of actual loss, up to a maximum of the market value of the item or  $\Sigma$ 50 (whichever is lower) as standard and are available with enhanced compensation options. Further specific details, including compensation and exemption information, can be found in section 17 below and on our website (www.rovalmail.com).
- 9.2 To help you find the best service for your needs and budget we provide a price finder tool on our website (www.royalmail.com). Prices and other pricing information can also be found in other locations on the same website (such as product specific pages) including in a tariff brochure with all pricing information for the services covered by this Scheme.
- 9.3 We may change the rates of postage and **service fees** from time to time. Changes to postage rates and service fees and the date on which the changed rates and fees will come into effect will be published at least one month (or whatever period of time agreed by us with our regulator, Ofcom) in advance on our website (www.royalmail.com).

#### 10 How To Show That An Item Has Had Postage Paid

- 10.1 You must show us that **postage** for an item has been paid for (where it applies) so that we can convey it without delay. Payment is demonstrated by the application of a **postage mark** to the item. The manner in which payment of postage can be shown depends on the method of payment used. Section 8 above set out the most common ways of applying a postage mark (**postage stamps**, **postage labels**, **service fee labels**, **franking marks**, marks created by online applications such as SmartStamps®, **Printed Postage Impressions** (**PPIs**) etc). In addition a postage mark may be fixed, printed, impressed, embossed or otherwise shown on the envelope or cover of an item as we may specify or approve.
- 10.2 Postage marks must be applied to the top right hand corner of the envelope or cover of the item unless we specify otherwise. If the item is spherical or without corners the postage mark should be placed above and to the right of the address.
- a. A postage mark that is damaged or imperfect, has been defaced or has anything written or printed over it becomes void and cannot be used to show that postage has been paid.
- b. You can use an adhesive postage stamp which has been perforated by means of a punch provided that the perforation holes are smaller than the holes dividing one stamp from another on a sheet of stamps.
- c. Postage marks can be used only once and then they become void and must not be re-used to pay postage on the same or another item.
- d. Any postage mark that carries an expiry time and/or date will become void from that time and/or date and cannot be used to show that postage has been paid after that time and/or date.
- 10.3 If you use a franking machine you must apply a franking mark to the cover of the item as set out in the Royal Mail Scheme For Franking Letters And Parcels 2014 (which is published on our website (www.royalmail.com)) and according to any guidelines that we publish on our Franking help centre website (www.royalmail.com).
- 10.4 If you make use of a credit account you must apply a Printed Postage Impression (PPI) in accordance with the PPI licence, the PPI design notes and any guidelines that we publish on our website (www.royalmail.com).

# 11 Other Requirements Relating To The Address On The Cover Of An Item

- 11.1 Section 7 set out how to address an item and section 10 set out how payment can be indicated. There are some further things that must be considered with regard to the cover of the item so that we can efficiently convey the item without delay.
- 11.2 An item must not have on its cover:
- a. anything which obscures the postage mark;
- b. anything which is likely, in our opinion, to make it difficult for us to apply a postmark to the cover;

- c. anything which, in our judgement, is likely to make the postmark illegible;
- d. any counterfeit or fake postage mark;
- e. any postage mark which we consider may have previously been used to pay postage;
- f. signs, words or marks, used without official permission, which could indicate that the item was sent on Her Majesty's Service or was conveyed and delivered by us; or
- g. any signs, words, images or designs which are offensive, obscene or indecent.
- 11.3 An item must not contain correspondence or letters other than between the sender and addressee of the item (or persons living with them). This does not include archived materials.
- 11.4 An item must not be made up in any way or have anything on its cover which, in our judgement makes it difficult or embarrassing to deliver the item.
- 11.5 A postage mark must not be attached to an item using staples, sticky tape, or any other means that we do not approve.

#### 12 How To Hand Over (or Post) An Item

- a. by handing it to an authorised member of staff at a Post Office® as long as considered feasible to do so;
- b. by Business Collection service (typically a paid for extra service where we collect the mail from a business customer's premises. Business Collections services are not covered by this Scheme, but the items that we collect under such services may be covered by this Scheme); or
- c. in any other manner which we may approve.
- 12.2 In addition, items sent under the following services:
- (i) International Standard:
- (ii) International Economy;
- (iii) HM Forces Mail (without Special Delivery or Signed For™ added or except those items that quality as free of charge items under BFPO); and
- (iv) Articles for the Blind, can also be posted in the following ways (unless the item is too large to do so):
- a. by placing it in a post box (typically a red post box on the street);
- b. by placing it in a **private post box** (typically a post box contained within a shop or other private premises).
- An item placed in a private post box will be deemed to have been posted when collected by us and not before.
- 12.3 An item that has had postage paid by a franking machine must be posted in accordance with the Royal Mail Scheme for Franking Letters and Parcels 2017 which is published on our website (www.royalmail.com).
- 12.4 Items sent through the International Standard, International Economy, International Signed, International Tracked & Signed and International Tracked services that have been paid for with a credit account must be presented to us in accordance with the guidance as outlined on our website (www.royalmail.com) and also in accordance with any conditions set out in the PPI licence.
- 12.5 Articles for the Blind sent by a business must present items separate to, but in the same manner (as required by section 12.4 above) and at the same time, as any items which are posted using a credit account.
- 12.6 All items must be posted by the latest acceptance time that we determine and advertise or agree with you (as appropriate to the method of posting). Items posted after that latest acceptance time will be considered to have been posted the following working day.

# 13 How We Will Treat Items That Do Not Meet the Requirements Set Out $\ln$ This Scheme

- 13.1 Once an item has been posted (whether as an **outgoing item** or an incoming item) we may carry out checks on that item.
- 13.2 Subject to sections 5.6 and 5.15 above (which set out how we treat items that contain prohibited items and/or restricted items where the relevant restrictions or requirements have not been complied with), if an item is found not to meet the conditions set out in this Scheme or an item is found that indicates that insufficient (or no) postage and/or service fee has been paid for the service requested, or for incoming items the item appears to satisfy the conditions of a **remail item**, we may decide at our discretion to do one of the following things set out below (in sections 13.3 and 13.4) within a reasonable period of time.
- 13.3 For outgoing items, we may:

- a. Convey the item to the relevant overseas postal operator for delivery to the address shown by the service requested. The overseas postal operator may require the addressee to collect the item from a specified location
- b. Convey the item to the relevant overseas postal operator for delivery to the address shown by an alternative service that we determine to be appropriate. In these situations the conditions of the service used to convey the item will apply.
- c. Treat the item as undeliverable. Details of how we may deal with undeliverable mail can be found in section 15.
- d. Return the item to you.
- e. Refuse to accept it.
- f. If none of the above courses of action are possible we may deal with or dispose of the item at our discretion.
- In all cases we may not accept any liability for that item. This section also applies to items which have been found either in a Post Office® branch or on Royal Mail premises and which appear to be intended for posting.
- 13.4 For incoming items, other than an incoming **registered item** (which shall be dealt with in accordance with section 13.5 below), we may:
- a. Convey the item to the addressee.
- b. Return the item to the country of origin.
- c. Treat the item as undeliverable. Details of how we may deal with undeliverable mail can be found in section 15.
- d. Otherwise deal with or dispose of the item at our discretion.
- 13.5 An incoming registered item shall be forwarded to the addressee without any additional charge.
- 13.6 In each case set out in sections 13.3 and 13.4 the addressee or you may be required to pay an amount (to be fixed by us or the relevant overseas postal operator) in order to cover additional costs before the item is delivered or released for collection:
- a. In the event of underpaid or unpaid postage and/or service fees the amount due may include a surcharge in addition to the underpaid or unpaid postage to cover administrative costs.
- b. Where the amount due is not paid the item may be dealt with or disposed of at our discretion.
- 13.7 In each case set out in sections 5.6, 5.15, 13.3 (d) to (f) and 13.4 (b) to (d) you or your **representative** or the addressee or their representative may be required to pay an amount (to be fixed by us or the relevant overseas postal operator) to cover the costs of returning, dealing with or disposing of the item.

### 14 Additional Marks We May Add

- 14.1 In the course of conveying an item we (or someone authorised by us) may alter the look of the cover of that item. Most usually this is by the addition of a postmark which shows that a stamp has been used and prevents it from being used again. A postmark may also contain an advertising or informative image. We may apply the postmark in any manner we consider appropriate. It is important to note that we may apply any form of postage mark (an image that denotes payment), impression, image, sticker, label or other device we consider suitable for any other postal purpose we consider appropriate.
- 14.2 We may also charge you a fee (to be fixed by us from time to time) and published on our website (www.royalmail.com) for cancelling postage marks with a special postmark including one that incorporates an advertisement, at your request.

# 15 Delivery Including Undeliverable and Re-Posted (Return To Sender) Items

- 15.1 Subject to sections 5.6 and 5.15, for outgoing items returned to us by an overseas postal operator as:
- a. Undeliverable, we will attempt to return the item to you. We will not charge any additional postage or fee for returning an undeliverable item but you or your representative may have to pay any other fees (e.g. surcharges or customs charges) that apply. If any applicable fee is not paid, the item may be dealt with or disposed of at our discretion.
- b. Undeliverable and the item was not originally posted in the United Kingdom, the Channel Islands or the Isle of Man, we may, at our discretion:
- Convey the item to the addressee.
- Return the item to the country of origin or to the postal administrator that forwarded the item to us.

- Otherwise deal with or dispose of the item at our discretion. In each case set out in this section 15.1(b) the addressee or you or the sender may be required to pay an amount (to be fixed by us) in order to cover underpaid or unpaid postage and/or service fees plus a surcharge to cover administrative costs before the item is delivered or released for collection.
- c. Return to Sender, we will attempt to return the item to you from a delivery address if all of the following conditions are met i.e. the item:
- was originally posted using a service set out in this Scheme; and
- is re-posted by the addressee or any other person for any reason; and
- the original address is crossed out; and
- a clear instruction to return the item to you is written on the cover; and
- the return address is written fully, correctly and legibly on the cover;
- the return address is in the United Kingdom, the Channel Islands or the Isle of Man: and
- the item is returned to us by the relevant overseas postal operator.
- 15.2 For incoming items, other than remail items (which shall be dealt with in accordance with section 13), we will attempt to deliver items that meet the conditions for the relevant service set out in this Scheme.
- 15.3 Some items require a signature at the point of delivery and some are too large to fit through a letter box. In these situations someone needs to be present at the address to take delivery of the item.
- a. If no one is available we may attempt to deliver the item to a **neighbour**. Please note that we will never attempt to deliver an item to a neighbour if it is a registered item or an **exprès item**.
- b. If a neighbour is available to take delivery of the item we will leave the item with that neighbour and leave a card at the address that the item was addressed to. That card will inform the addressee (or their representative) that an item is (or items are) being held by a neighbour for them and provide the address details for that neighbour and the date and the time that the item was left with the neighbour.
- c. If an addressee does not want their items to be delivered to a neighbour or does not want to take items for neighbours they can opt out of the Delivery to Neighbour service, free of charge, by registering their wish to opt out in such a way as Royal Mail may reasonably require from time to time. For more information on how to opt out please check our website (www.royalmail.com).
- d. If a neighbour is not available to take delivery of the item, or the addressee has opted-out and does not want their items to be delivered to a neighbour or to take items for neighbours, we will take the item back to the delivery office or an alternative location such as a Post Office® branch. We will leave a card at the address that the item was addressed to.
- e. The card that is left at the address will inform the addressee (or their representative) that we attempted to deliver an item or items and will offer the addressee (or their representative) one or more of the following opportunities:
- (i) to request that the item be redelivered to the same address;
- (ii) to request that the item be redelivered to an alternative local address ("local" in this context means an address covered by the delivery office to which the item was returned). Please note that this option is not available for registered items or exprès items;
- (iii) to collect (after the required period of time stated on the card has elapsed to allow for the item to reach that location and once identification that we deem to be suitable has been provided) the item from the delivery office or the alternative location such as a Post Office® branch to which the item was taken back to; or
- (iv) to request that the item be sent from the delivery office or other location (other than a Post Office® branch) to a Post Office® branch for collection. This service is called Local Collect (Social) and incurs an additional fee. When collecting the item the addressee (or their representative) must provide the card that was left at the address and identification that we deem acceptable. Further details including the current fee and identification requirements can be found on our website (www.royalmail.com).
- 15.4 We may decide not to deliver an incoming item which we consider impracticable or unreasonable to deliver. Such an item is referred to in this Scheme as undeliverable.
- 15.5 Where an incoming item is undeliverable the item may at our discretion be returned to the sender or otherwise may be disposed of as we may think fit.

- 15.6 Redirection for incoming items that have been redirected from an address outside of the United Kingdom the Channel Islands and Isle of Man, even if it was not originally posted in that country, to an address in the United Kingdom, we will attempt to convey the item to the addressee. In the event of underpaid or unpaid postage and/or service fees on such items the amount due may include a surcharge in addition to the underpaid or unpaid postage to cover administrative costs before the item is delivered or released for collection, in addition, the addressee may have to pay any other fees (e.g. surcharges or customs charges) that apply, before the item is delivered or released for collection. Where the amount due is not paid the item may be dealt with or disposed of at our discretion.
- 15.7 Forwarding unofficial redirections We are not under any obligation to do anything with an item that someone requests to be forwarded to another address other than the return to sender address on the cover. We consider these unpaid for, forwarding requests to be a form of unofficial redirections request. Typically in this situation the original address is crossed out and a written request ("forward to") is made on the cover to forward the item to another address. This is normally attempted in order to try to get the item to the addressee who has moved to another address. In these situations the person moving should take out a Redirection service with us or provide prepaid stationery. If we find an item that someone has requested be forwarded to another address then the item may be dealt with or disposed of at our discretion.

#### 16 Complaints Handling Process

- 16.1. We aim to provide you with a reliable and consistent service at all times. If, however, you are unhappy with the service we have given you, this section will help you to get in touch with us to discuss your complaint and if applicable refer your complaint to an external, independent Ombudsman style service for resolution.
- 16.2. There are 3 internal stages in our complaints handling process and then the final Ombudsman style stage. Further information and contact details for each stage are available on our website (www.royalmail.com).

### Stage 1: Contacting our Customer Services Advisors

16.2.1. Our Customer Services Advisors will make every effort to investigate and resolve your complaint providing you with an appropriate response. If you are not happy with the reply you can escalate your case to the Escalated Customer Resolution Team.

#### Stage 2: Contacting our Escalated Customer Resolution Team

16.2.2. Our Escalated Customer Resolution Team is trained in dealing with escalated complaints. They will endeavour to resolve your concerns. However, if after allowing them another chance to deal with your complaint you remain dissatisfied, you can contact the Postal Review Panel.

# Stage 3: The Postal Review Panel

16.2.3. The Postal Review Panel will take a fresh and impartial look at your case and provide a final response from us about your complaint. 16.2.4. If you are not satisfied after the Postal Review Panel has provided you with a final response then your case will be considered 'deadlocked'. Customer complaints in deadlock can be referred to the external, independent Ombudsman style service.

# Stage 4: The Postal Redress Service – an external, independent Ombudsman style service

- 16.2.5. If you remain dissatisfied with the response you have received after contacting the Postal Review Panel and you have received a deadlock letter, you may be able to escalate your complaint to the independent Ombudsman style scheme, the Postal Redress Service (POSTRS).
- 16.2.6. POSTRS's role is to resolve disputes between regulated postal operators (such as us) and their customers. POSTRS provide a free service and we are bound by their findings.
- 16.2.7. Depending on the nature of the specific complaint, POSTRS may be available. A list of products within the POSTRS scheme can be found on our website (www.royalmail.com). Generally, the service is available to you if:
- We have not been able to settle your complaint within the 90 day timescales set out by Royal Mail Customer Services.
- You have been referred to the scheme through your 'deadlock' letter.
- We have not followed our own complaints procedure in handling vour complaint.
- 16.2.8. For further information please refer to POSTRS's website www.cedr.com/postrs.

### 17 Compensation - What We Are Liable For

17.1 The vast majority of items arrive safely on time. In some instances items may be lost or damaged. In these situations we may refund postage and/or service fees and may award compensation. We do not pay compensation for delay. This section provides details as to what items qualify for refunds and compensation, what compensation is available and how and when claims need to be made.

17.2 We will accept liability, may refund postage and/or service fees and may pay compensation for the **loss** of, **damage** to, or **part loss** of an item posted under this Scheme, unless that item:

a. is considered by us or by an overseas postal operator to be undeliverable (as set out in section 15 above);

b. is one that has been successfully delivered but subsequently made the subject of a Return to Sender request (as set out in section 15 above):

c. is considered by us or by any overseas postal operator to be subject to a forwarding or unofficial redirection request (as set out in section 15.6 and 15.7 above);

d. has not fully meet all of the conditions set out in this Scheme (as mentioned in section 13.2);

e. is considered by us to indicate that insufficient (or no) postage and/or service fee has been paid for the service requested (also as mentioned in section 13.2);

f. is one which we have dealt with or disposed of at our discretion as mentioned under any section of this Scheme;

g. is one that contained prohibited items (as referred to in section 5.3 above);

h. is one that contains restricted items and the requirements for the acceptance of those items have not been met (as referred to in sections 5.12 and 5.13 above);

i. is one that is being sent by post in contravention of section 85 of the Postal Services Act 2000 (as amended), which essentially means any item that may damage another item or injure anyone handling that item. Also an item that contains or shows on its cover an indecent or obscene image;

j. is subject to circumstances where the event leading to the claim was caused by situations outside our control including exceptionally severe weather conditions, acts of terrorism and vandalism and acts of third parties with whom we have no contractual relationship;

k. is one where the loss or damage is due to a latent or inherent defect or natural deterioration;

I. is one that we do not believe has suffered damage or delay solely as a result of its transmission through the post. In particular where the cover does not appear to us to have suffered damage consistent with the damage caused to the contents of that item;

m. is one where the damage was pre-existing, that is, where the item was already damaged when it was posted; or

n. is one which has been re-directed to an address in the Channel Islands or the Isle of Man.

17.3 For clarity where an item falls into one of the categories listed in section 17.2 (a) to (n), then we will not accept liability, will not refund postage and/or service fees or pay compensation for the loss of or damage to that item.

17.4 In addition to section 17.3 we will only accept liability, may refund postage and/or service fees and may pay compensation where the loss or damage is due to any wrongful act done, or any neglect or default committed by a member of staff or agent of Royal Mail while performing or purporting to perform his functions as such in relation to the receipt, conveyance, delivery or other dealing with the item.

17.5 In addition to sections 17.3 and 17.4 we do not accept liability, will not refund postage and/or service fees or pay compensation for loss or damage to an item where the loss or damage is caused by your own act or omission in the course of either preparing to post or actually posting the item.

17.6 The amount of compensation that is payable depends on a number of factors including:

- The service used.
- The nature of compensation being sought i.e. whether for loss or damage,
- The evidence that can be provided, and
- The value of the item.

17.7 Compensation is available for items sent using the following services:

- International Standard;
- International Economy
- International Signed
- International Tracked & Signed
- International Tracked

• HM Forces Mail (with and without Special Delivery  $^{\text{TM}}$  or Signed For  $^{\text{TM}}$  added);

#### Claims and Evidence

17.8 We will only consider a claim for loss or damage compensation where the **required evidence** (as defined in section 17.9) is available and provided.

17.9 The required evidence is made up of all of the following:

a. the names and addresses of you, the addressee and, if different, the claimant;

b. the name of the service used;

c. evidence of posting and evidence of the value of the postage paid plus the method of postage, e.g. stamps, franking impression, Smartstamp®. Evidence of posting may include:

- an original Certificate of Posting/Posting Receipt (provided automatically for International Signed, International Tracked & Signed, International Tracked and available on request and free of charge at Post Office® branches for other items),
- an original Smartstamp® or on-line postage certificate of posting validated at a Post Office® branch,

d. for damage (or part loss) claims the item with cover and packaging, showing the postage paid as well as the original Certificate of Posting if provided as part of the service purchased (i.e. International Signed, International Tracked & Signed and International Tracked; the place of posting;

e. the date of posting;

f. for International Signed, International Tracked & Signed and International Tracked services with enhanced compensation, the product documentation provided to you when posting the item (i.e. certificate of posting or receipt);

g. a detailed description of the contents;

h. in addition, for damage (or part loss) claims, the date of delivery;

i. in addition for damage (or part loss) claims, a description of the packaging and condition of the mail item itself.

17.10 In order to claim loss or damage compensation for the actual loss of the item, additional evidence of the item's value is required (see section 17.11 for examples). Please note that, when claiming for actual loss, there is a cap on the amount of compensation that can be paid. For International Signed, International Tracked & Signed and International Tracked, it is the lower of market value or £50, or £250 if an enhanced compensation service has been purchased. Enhanced compensation is not available for mobile telephones (including Blackberrys and PDAs) and the maximum compensation available under the enhanced compensation service for loss or damage to cash, securities or instruments to the bearer is £100. There are restrictions to sending cash and other items to some destinations. Please see our website (www.royalmail.com) for further information on restrictions and prohibitions. For all other services listed in section 17.7 the maximum compensation we will pay for any item is the lower of market value or £20 (save where Special Delivery™ or Signed For™ is added for HM Forces Mail - see section 21). Where the item has no intrinsic value, compensation for the actual loss is not payable, however a postage refund may be considered for claims involving loss. Postage refunds are not available for any claims for damage and part loss. Details of compensation available in various circumstances are summarised in Tables 1 & 2 below.

17.11 Evidence of value can be:

- Original receipts;
- · Bank or credit card statements;
- PayPal records;
- Invoices;
- Manufacturing costs;
- Auctioneers valuations;
- Repair quotations.

This list is not exhaustive. It is provided for illustrative purposes. All evidence of value must be originals and not copies. You are advised to make copies before sending us originals.

17.12 In addition to information set out in sections 17.9 – 17.11, for eBay claims you must also include the following:

- item sale page and PayPal or bank/credit card statement; and
- eBay item number.

17.13 Please note that all of the packaging and contents of damaged items or items subject to part loss must be retained as we may need to inspect them. If they are not retained then compensation may not be paid.

Table 1 - Refunds and compensation available (assuming submission of sufficient evidence) for loss, damage and part loss for items conveyed using International Standard and International Economy and HM Forces Mail (without Special Delivery<sup>™</sup> or Signed For<sup>™</sup> added).

	Compensation for loss.	Compensation for damage and part loss.
Item has no intrinsic value	Postage refund.	No compensation or postage refund.
Item has intrinsic value (with required evidence and additional evidence)	Postage refund, plus compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £20.	Compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £20.

**Table 2** - Refunds and compensation available for loss (assuming submission of sufficient evidence), damage and part loss for items conveyed using International Signed, International Tracked & Signed and International Tracked

	Compensation for loss.	Compensation for damage and part loss.
Item has no intrinsic value	Postage refund.	No compensation or postage refund.
Item has intrinsic value (with required evidence and additional evidence)	Postage refund, plus compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £50 or, if enhanced compensation service has been purchased, the maximum payable is the lower of the market value of the item and £250	Compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £50 or, if enhanced compensation service has been purchased, the maximum payable is the lower of the market value of the item and £250

#### Making a Claim

17.14 This section sets out, for guidance, some details for making a compensation claim but not the full process. The full process for making a claim is set out on our website (www.royalmail.com). We may reject claims that do not follow that full process.

17.15 Claims should be made by the sender of the item. If the sender and the intended recipient both make a claim for the same item, then only the sender will have a right to compensation, unless a compensation payment has already been made to the recipient, in which case the sender will have no right to compensation.

17.16 Please note that loss claims can only be made if the item has not been delivered (or had delivery attempted) 20 or more working days after the **due delivery date** for EU destinations and 25 working days after the due delivery date for Rest of the World destinations. Please note that resolution of a claim may take some time, whilst we deal with overseas postal operators.

17.17 Any loss, damage (or part loss) claims must be made as soon as possible after the incident to help us investigate. All loss and damage (or part loss) claims must be made within 6 months of the date of posting for all services covered by this Scheme. No compensation or other payment will be paid by us for loss, damage (or part loss) claims not made within 6 months of the date of posting and not made in full compliance with the requirements of this Scheme (including the full claims procedure set out on our website (www.royalmail.com).

17.18 All claims should be made using the correct claim form which is available to download from our website (www.royalmail.com) and can be collected from Post Office® branches. By using the correct claim form we can process your claim as quickly and effectively as possible. Alternatively, claims may be submitted online via our website (www.royalmail.com).

17.19 The claim form needs to be completed as fully as possible: hard-copy claims form should be signed and dated by the claimant before being sent to us and all mandatory fields within the online claim form must be completed. Remember that in order to claim loss or damage compensation for actual loss of the item (if that compensation is available for the service you used) additional evidence of the item's value is required and should be submitted with the claim form (see section 17.11 for examples). Where evidence is submitted online, we reserve the right to request the originals of such documents in line with section 17.9 of this Scheme.

17.20 If additional evidence cannot be provided then only a postage refund can be considered.

17.21 We may at any time request more documentation and/or information from the claimant, sender or recipient to protect against unwarranted or duplicate claims. This includes, but is not limited to, a declaration of non-receipt (or proof of non-receipt in appropriate cases).

17.22 If compensation has been paid in respect of the loss of an item for a sum exceeding the real value of the contents, or if the item is subsequently found and delivered or returned, any compensation or postage refund awarded to you must be refunded to us.

#### 18 Additional Terms and Conditions For Some Services /Items

18.1 The terms and conditions set out in sections 5 to 17 apply to all items. These are referred to in this Scheme as the **common terms**. Some services that we offer have additional terms and conditions that are specific to those services. The additional terms are known as the **specific terms** in this Scheme and are contained in the sections below. There is a section each for:

- International Signed and International Tracked & Signed (Section 19)
- International Tracked (Section 20)
- HM Forces Mail (with and without Special Delivery<sup>™</sup> or Signed For<sup>™</sup>) (Section 21)
- Articles for the Blind (Section 22)
- Printed Papers (Section 23)

18.2 When using a service listed in Section 18.1 then the specific terms must be complied with in addition to the common terms. Where the specific terms conflict with the common terms then the specific terms should be followed as they take priority over the common terms.

18.3 In addition to the common and specific terms contained within this Scheme other terms and conditions may exist for some services. For example in section 10 above we mentioned that customers using a franking machine must abide by the Royal Mail Scheme For Franking Letters and Parcels 2017 and according to any guidelines that we publish on our Franking help centre website (www.royalmail.com). Also, customers who make use of a credit account must abide by the PPI licence, the PPI design notes and any guidelines that we publish on our website (www.royalmail.com). Such terms and conditions may be contained in application forms, licences, or any other document issued or authorised for issue by us and all may be updated from time to time.

### 19 International Signed and International Tracked & Signed

19.1 For International Signed and International Tracked & Signed, the item will only be delivered to an addressee or their representative once a signature or similar proof of delivery such as ID has been gained (in some countries, a signature is not obtained but a similar proof of delivery such as ID is the alternative). International Signed is tracked to the point it leaves the UK and takes a signature on delivery (or a similar proof of delivery). International Tracked & Signed is tracked in the UK and overseas and takes a signature on delivery (or a similar proof of delivery). In each destination country, either one or the other (but not both) of these services is available. For details of destinations that offer International Signed or International Tracked & Signed please see our website (www.royalmail.com). Please note that country details may change from time to time so you should check the website when necessary.

19.2 You must apply (or must ensure someone else applies) a fully completed International Signed or International Tracked & Signed label securely to the cover of the item for which the service has been purchased.

19.3 Upon delivery of an incoming item by International Signed or International Tracked & Signed services the addressee or their representative must sign (or otherwise provide some form of proof of delivery as we may specify). Where such a signature or proof of delivery is refused to be provided to us then the item may be dealt with or disposed of as we consider appropriate. Please note that the representative could be someone else at the delivery address or a neighbour.

19.4 We cannot provide you with a copy of the signature or a similar proof of delivery of the recipient of the International Signed or International Tracked & Signed item. We may however seek to obtain a copy of such signature or a similar proof of delivery in the event of a claim for loss or damage.

19.5 For International Tracked & Signed online confirmation of delivery (not including a copy of the signature of the recipient or a similar proof of delivery) in the majority of cases can be viewed on our website (www.royalmail.com) and will be available for up to 12 months after the item was posted. To access these details you will need to know the reference details on the barcoded label for the item sent.

#### 20 International Tracked

20.1 International Tracked is a fully tracked, service that can be sent to a number of destinations. Details of destinations can be found on our website (www.royalmail.com). Please note that country details may change from time to time so you should check the website when necessary.

20.2 International Tracked is fully tracked from despatch to delivery further details of which can be found on our website (www.royalmail.com).

20.3 International Tracked should not be confused with International Signed or International Tracked & Signed that takes a signature on delivery or a similar proof of delivery. International Tracked does not take a signature on delivery.

20.4 You must apply (or must ensure someone else applies) a fully completed International Tracked label securely to the cover of the item for which the service has been purchased.

20.5 If we find an item which has not met the conditions laid out in this Scheme for an International Tracked item but which either:

a) has International Tracked written on it, or anything which suggests that the item is intended to be sent via International Tracked or

b) is found to contain valuables, then we will treat it as an International Tracked item.

In either case, if the item is treated as an International Tracked item, you or the addressee will be charged the appropriate postage for that service. If the postage is not paid the item may be dealt with or disposed of at our discretion.

20.6 Confirmation of delivery can be viewed on our website (www.royalmail.com) and will be available for up to 12 months after the item was posted. To access these details you will need to know the reference details on the barcoded label for the item sent.

# 21 HM Forces Mail (with and without Special Delivery $^{\text{TM}}$ or Signed For $^{\text{TM}})$

21.1 Royal Mail works with the Ministry of Defence to provide a postage service to HM Forces overseas.

21.2 The weight limit for all HM Forces Mail is 2kg.

21.3 Prices for HM Forces Mail are set by the British Forces Post Office (BFPO). This currently follows domestic pricing, but may be subject to change. Details of pricing can be found on our website (www.royalmail.com).

21.4 In some cases, items may be sent to members of HM Forces overseas free of charge. Such concessions are determined by the BFPO details of which can be found on our website (www.royalmail.com).

21.5 The compensation available for HM Forces without Special Delivery™ or Signed For™ added is set out in section 17.10. Special Delivery™ may be bought with and added to HM Forces Mail when a greater level of compensation is required in the event of loss or damage to an item being sent to a member of HM Forces overseas.

21.6 Terms and conditions relating to Special Delivery™ sent through BFPO shall follow those as outlined by The Royal Mail United Kingdom Post Scheme (a copy of which may be found on our website (www.royalmail.com)) except that:

a. Compensation for delay or consequential loss is not available.

b. Money or jewellery must be sent in a Special Delivery™ pre-paid envelope if compensation is to apply in the event of loss or damage.

c. Compensation is available for loss and damage on the basis of actual loss up to the maximum of the market value or  $\mathfrak{L}500$  (or  $\mathfrak{L}1,000$  or  $\mathfrak{L}2,500$  if enhanced compensation is purchased) whichever is the lower of these.

d. Special Delivery™ items being sent via BFPO will take a signature on delivery but will not provide tracking or confirmation of delivery.

21.7 Terms and conditions relating to Signed For™ sent through BFPO shall follow those as outlined by The Royal Mail United Kingdom Post Scheme (a copy of which may be found on our website (www.royalmail.com)) except that Signed For™ items being sent via BFPO will take a signature on delivery but will not provide confirmation of delivery.

#### 22 Articles For The Blind

22.1 Articles for the Blind is a free of charge service for the conveyance of items that contain particular items that are of use to blind people.

22.2 In this Scheme blind people and the blind means:

a. persons registered as blind under the provisions of the National Assistance  $\mathop{\rm Act}\nolimits$  1948; or

b. persons whose standard of close-up vision, with spectacles, is N12 or less.

22.3 Articles for the Blind can be used to convey items to or from blind people that contain any of the following items:

a. books, papers and letters which are specifically prepared for use by blind people. Standard or large print items may only be sent using the Articles for the Blind service if, in our reasonable opinion, they fall within this definition and the font size is at least 16 points in size;

b. papers sent to anyone to be specially prepared or impressed so that blind people can use them;

c. relief maps;

d. machines, frames and attachments for making impressions for blind people to use;

e. writing frames and attachments; f. Braille instruction manuals; or g. any other item that we determine to be allowable as listed on our website (www.royalmail.com)).

22.4 Articles for the Blind can also be used to convey items between blind people and specific organisations or institutions (which are preapproved by us) that cater for blind people that contain any of the following items:

a. games (including card games);

b. mathematical appliances and attachments;

c. 'talking books' and 'talking newspapers', which are recordings of readings from printed sources, such as books, journals, newspapers, periodicals or similar publications;

d. equipment used to play talking books and newspapers;

e. metal plates impressed or sent for impressing for use by blind people;

f. supplies of covers, envelopes and labels for sending articles for use by blind people;

g. watches, clocks, timers, tools and measuring equipment designed for blind people to use;

h. walking sticks adapted for blind people;

i. harnesses for guide dogs;

j. computer disks and CDs which have been prepared for blind people and do not contain information which can be read without specially prepared encryption software; or

k. any other item that we determine to be allowable as listed on our website (www.royalmail.com).

22.5 Any item must meet the conditions set out below. We may open and inspect each item to ensure it meets the conditions set out below: a. it must weigh less than 7 kilograms;

b. it must clearly show the words 'ARTICLES FOR THE BLIND' on the front of the cover along with your name and address somewhere on the cover where it cannot be confused with the destination address (preferably on the back of the item). If sent through a Post Office® branch a label must also be applied which is known as a P4558;

c. it must be left un-sealed so the contents can be checked to make sure they are permissible as listed in sections 22.3 or 22.4. Alternatively they can be sealed in a manner that we can open and then re-seal or we may agree alternative arrangements with you in advance should you be unable to meet these requirements. For example you may be able to design your cover such that the contents are visible;

d. it must not contain any item or personal message which is not listed in sections 22.3 or 22.4, with the exception of a label showing the name and address of the sender, and instruction booklets, guarantees, technical documentation and other information provided for use with an item listed in sections 22.3 or 22.4:

e. it must be addressed to a named recipient. This can be the name of an individual or an organisation that provides services specifically for blind people. Where addressed to an individual that individual must be a blind person. It is not permissible to send an item using Articles for the Blind if it is addressed to 'The Occupier' or 'Dear Customer';

f. it must not contain any advertising literature; and

g. it must not contain any postage stamp or form of prepayment, whether cancelled or not, or any paper representing a monetary value, except in cases where the item contains as an enclosed card, envelope or wrapper bearing the printed address of the sender or his agent which is pre-paid for return.

22.6 If we find an item that has been posted with us as an Articles for the Blind item but which does not meet the requirements set out in these specific terms then we may convey it using a different service and we will charge you or the recipient the appropriate postage for that service along with a surcharge to cover our costs. If the postage and surcharge is not paid then we will deal with the item as we determine.

22.7 More information is available on our website (www.royalmail.com).

#### 23 Printed Papers

23.1 Printed Papers weighing no more than 5kg may be sent using International Standard, International Economy, International Signed, International Tracked & Signed and International Tracked services.

23.2 In such cases the item must be marked 'Printed Papers' on the left hand side of the address.

23.3 For many destinations, customs declarations are required for sending Printed Papers. It is therefore advisable to attach the relevant customs declaration. You are responsible for meeting all applicable customs regulations and we shall have no liability for any actions of any customs authority, even if those actions cause loss, damage (or part damage) or delay to the item.

23.4 Printed Papers may include the following: newspapers, periodicals, books and pamphlets, sheets of music, visiting or address cards, proofs of printing, engravings, photographs and albums containing photographs, pictures and drawings, plans, maps, patterns for cutting out, brochures, prospectuses, advertisements catalogues and notices.

23.5 The list in section 23.4 is an indicative list of items that can be sent under the Printed Papers service. For up to date information please refer to our website (www.royalmail.com). You are responsible for checking whether the contents of an item are Printed Papers at the time of posting.

23.6 Printed Papers must not contain any postage stamp or form of prepayment, whether cancelled or not, or any paper representing a monetary value, except in cases where the item contains as an enclosure a card, envelope or wrapper bearing the printed address of the sender of the item or his agent in the country of posting or destination of the original item, which is prepaid for return.

# 24 Customs Control

24.1 You may be required to pay customs duty and prepare customs documentation/complete customs processes when sending goods or merchandise to some overseas destinations.

24.2 Customs documentation is generally required when sending goods or merchandise outside of the UK. For up to date information on customs requirements, the forms and processes that should be completed, please refer to our website (www.rovalmail.com).

24.3 Customs documentation and requirements are available from all Post Offices® or can be downloaded from our website (www.royalmail.com). Such information may change from time to time and it is your responsibility to verify that any items you send meet relevant customs requirements at the time of sending. Please refer to our website (www.royalmail.com) when necessary for up to date information relating to customs requirements.

24.4 It is important that all customs requirements are complied with. Despatch of items without the necessary customs forms or completing the necessary processes will result in delay to delivery and the possibility that goods will be seized by customs authorities

24.5 For items entering the UK for delivery, where customs duties apply, Royal Mail may charge a fee for processing such items. The fee reflects the costs of handling such items. Where the amount due is not paid the item may be dealt with or disposed of at our discretion. Details of the fee payable can be found on our website (www.royalmail.com).

#### 25 Your information

25.1 Where we supply services to you under this Scheme, we are the controller of the personal data we process in providing services to you.

25.2 Where you supply personal data to us so we can provide services to you, and we process that personal data in the course of providing services to you, both you and we will comply with our obligations imposed by the Data Protection Legislation and you will not cause us to contravene the Data Protection Legislation. Where you have provided the personal data of a third party to us, you warrant that you have lawful grounds, such as their consent, to do so and that we are entitled to process that personal data to provide services.

25.3 From time to time we may be obliged by our Regulator to provide it with certain information about you, including your name and address.

25.4 Further information about how we use your personal data is set out in our Privacy Policy at www.royalmail.com/privacy-policy. Please read this Privacy Policy carefully.

25.5 The terms "personal data", "controller", "processing" and "supervisory authority" shall all have the same meaning as in the Data Protection Legislation and the term "process" shall be construed accordingly.

# Annex A - Definitions used in this Scheme

A1 The Interpretation Act 1978 applies for the interpretation of this Scheme and for this purpose this Scheme is to be treated as if it were an Act of Parliament.

A2 In this Scheme, the following words and terms have the following meanings:

#### actual loss

(a) Where an item is lost or damaged beyond repair then it is the amount it cost the customer to acquire, purchase or manufacture the item subject to adjustment to take account of condition, age and depreciation.

(b) Where an item is damaged (but not beyond repair) it is the cost of repair. No additional payment will be made for the reduced value of the repaired item.

#### address

means for any premises the address, including the name of the addressee, premises name or number, road name, locality name, zone number or post box number (if one exists) plus name of country of destination.

#### addressee

the person to whom an item is addressed.

#### Bladed Item

(Bladed Articles and Bladed Products as defined by Offensive Weapons Act 2019) including, but not limited to, any knife (including a folding pocket knife), any knife blade, cutlery knife, scissors with sharp edges or points, sporting equipment with a blade, tools with a blade, replica and antique knives including those used for reenactment purposes, handmade and bespoke knives, cut-throat razors, open razors where the blade is exposed, any axe, any sword, and any other article which has a blade - Excludes knives prohibited under UK law-see Weapons.

#### business

any entity engaged in commercial or economic activity, whether for profit or not, and irrespective of legal form.

# Channel Islands

the islands called Alderney, Guernsey, Jersey, Herm and Sark.

#### cove

any cover, envelope, or other packaging used to contain the contents of an item. In the case of a postcard or other card, any reference to a cover shall be construed as a reference to an outside face of the card.

#### damage

if something in an item of mail has suffered a degree of harm that reasonably impairs the material function or contents of the item.

### **Data Protection Legislation**

means (1) the Data Protection Act 1998; the Data Protection Directive (95/46/EC) and the Privacy and Electronic Communications Directive (2002/58/EC); (2) after 25 May 2018, Regulation (EU) 2016/679 of the European Parliament And Of The Council of 27 April 2016 on the protection of natural persons with regard to the processing of personal data and on the free movement of such data, and repealing Directive 95/46/EC (General Data Protection Regulation), (3) any guidance, directions, determinations, codes of practice, orders, notices or demands issued by any competent supervisory authority or other competent authority, any other applicable data protection laws or regulations and judgments of any court of law, tribunal or regulatory body as

amended, extended, re-enacted or replaced from time to time, and (4) the Regulation on Privacy and Electronic Communications, when in force.

#### due delivery date

means a date on which we or the relevant overseas postal operator will aim to deliver an item as more particularly described in the delivery aims section of our website (www.royalmail.com).

#### exprès items

means an incoming item sent using an overseas service equivalent to Royal Mail's International Tracked service.

#### franking mark

any mark or impression printed or otherwise made by a franking machine indicating the date and place of posting, the postage and fees paid or payable, the identity of the franking machine used and any other numbers, words, characters, symbols or marks be required by us.

#### incoming item

means (a) any item accepted by us from an overseas postal operator for delivery to an address within the United Kingdom the Channel Islands and the Isle of Man or (b) any item passing through the United Kingdom in transit to address outside the United Kingdom, the Channel Islands and the Isle of Man.

**intrinsic value** when used in relation to the content of an item means that it has an inherent monetary value relating to its essential nature.

#### item

means any letter, large letter, postcard, printed papers and every parcel or other article transmissible by post.

#### loss

an item shall be-deemed to be lost, (unless there is evidence to the contrary to demonstrate that it has not been lost) if it has not been delivered or had delivery attempted 20 or more working days after the due delivery date for EU destinations and 25 working days after the due delivery date for destinations outside the EU.

# market value

means what a willing buyer would pay a willing seller, both in a free market, for an item. When assessing the market value of an item, the value of any message or information it contains, or bears or refers to must be ignored. Market value is one of two caps upon the amount of compensation payable.

#### neighbour

for the purposes of delivering items, an appropriate person chosen at the discretion of the delivery officer on the basis of local knowledge and official guidance, who lives within close proximity of the address on the item. They may be a next-door neighbour or someone who lives sufficiently close by.

# outgoing item

means any item accepted by us within the United Kingdom for postal delivery to an address outside the United Kingdom, the Channel Islands and the Isle of Man.

#### part loss

Where an item is received and some or part of the content is missing. **post, posted** 

an item is posted if it has been entrusted to us for transmission by post in accordance with one of the approved methods set out in section 12 of this Scheme and related words shall be construed accordingly.

#### postage

the amount of money charged by us for delivery of an item.

#### postage mark

a collective term for a mark or impression (to include a franking mark, a Printed Postage Impression (PPI), any postage paid symbol, preprinted stationery and SmartStamp®) authorised for use by us to indicate payment of postage and/or service fees on an item or to indicate that you have entered into an arrangement with us to pay postage and/or service fees on that item. A postage mark may be non-adhesive (when it is embossed, impressed or printed on a cover) as we may from time to time permit.

#### postmark

any indicator, image or impression applied by us to a cover or envelope, to cancel a postage stamp or postage label, to indicate that we are the delivery agent of the item or for any other postal purpose approved and authorised by us.

#### **Printed Postage Impression (PPI)**

a postage mark indicating postage is payable to us and printed under a licence from us.

#### **Printed Papers**

means the Printed Papers service as described in section 23 of this Scheme or the items which may be sent under such service (depending on the context when used).

#### private post box

any post box owned and maintained by a person other than ourselves which has been authorised for use as a post box by either Post Office Limited or ourselves for the posting of items which are intended to be carried by us.

postage stamp means a stamp, authorised for use by us to indicate payment of postage and/or fees on an item, or to indicate that you have entered into an arrangement with us to pay postage and/or service fees on that item. A postage stamp may be adhesive in order for it to be affixed to a cover as we may from time to time permit.

#### postage label, service fee label

means a label, authorised for use by us to indicate payment of postage and/or service fees on an item. or to indicate that you have entered into an arrangement with us to pay postage and/or service fees on that item. A postage label may be adhesive in order for it to be affixed to a cover as we may from time to time permit.

#### registered item

means an incoming item sent using an overseas service equivalent to Royal Mail's International Signed service.

#### remail item

means incoming items addressed to persons within the UK and posted in a foreign jurisdiction other than the jurisdiction in which they were produced and/or made ready for sending.

#### representative

means a person who acts legitimately, with or without express permission, on behalf of the addressee.

# service fee

the amount of money charged by us for providing a service in connection with an item.

#### SmartStamp®

a postage mark used by customers posting items with the SmartStamp® indicia having entered into the relevant SmartStamp® terms and conditions.

# valuables (money and jewellery)

any item that is:

i) precious metal that has been manufactured in such a way as to add value to it, including coins used for ornament;

ii) ingots

iii) diamonds and precious stones;

iv) watches the cases of which are made wholly or mainly of precious metal:

v) articles similar to any of those referred to in i)- iii) above with an intrinsic value;

v) coins and bank notes of any currency that are legal tender at the time of posting;

vi) postal orders, cheques and dividend warrants uncrossed and payable to bearer;

vii) unused postage and revenue stamps and National Insurance stamps;

viii) Exchequer bills, bills of exchange, promissory notes and credit notes; Bonds, bond coupons and any other investment certificates; and

ix) coupons, vouchers, tickets, tokens, cards, stamps and other documents that can be exchanged in whole or in part for money, goods or services.

#### Zone Number

overseas equivalent of a UK postcode

Further information
(not part of the Scheme)

You can get more information on this Scheme, the services it covers and the details contained in the web pages that are linked in this document from our Sales Centre on 0345 740 740.

Lines are open from 8am to 6pm Monday to Friday and from 8am to 1pm on Saturdays.

# THE ROYAL MAIL UNITED KINGDOM POST SCHEME 06 April 2022

Contents		
1.	About this Scheme	3
2.	What this Scheme applies to	3
3.	Definitions	4
4.	Our ability to provide services	4
5.	What can and cannot be contained within an item	4
Prohibited items		4
Restricted items		7
6.	Size & weight limits and how to package an item	18
7.	How to address an item	18
8.	How to pay for postage and other services	19
9.	How to work out how much postage to pay	20
10.	How to show that an item has had postage paid	22
11.	Other requirements relating to the address on the cover of an item	22
12.	How to hand over (or post) an item	23
13.	How we will treat items that do not meet the requirements set out in this scheme	24
14.	Additional marks	25
15.	Delivery including undeliverable and re- posted (Return to Sender) items	25
16.	Complaints handling process	28
17.	Compensation – what we are liable for	29
	What determines compensation payable	31
	Claims and evidence	31
	Making a claim	35
18.	Additional terms & conditions for some services	37
19.	First Class, Second Class and delivery confirmation	37

20.	Royal Mail Signed For 1st Class, Royal Mail Signed For 2nd Class & Proof of Delivery	38
21.	Special Delivery	39
22.	Articles for the Blind	42
23.	Petitions and	43
	Addresses to the Sovereign	
24.	Petitions to Parliaments and Assemblies	44
25.	Poste Restante	44
26.	Items sent between the United Kingdom and the Channel Islands or the Isle of Man	45
27.	Sanctions Laws	45
28.	Your information	46
Annex A - Definitions (	used in this Scheme	47
1 About this Sahama		

1. About this Scheme

1.1 This Scheme is a document that sets out the terms & conditions for some 1 of the items that we handle. We (Royal Mail Group Ltd) are allowed to set out terms and conditions within a Scheme document by the Postal Services Act 2000 (and more specifically section 89 of that Act as amended by the Postal Services Act 2011 and in particular paragraph 24 of Schedule 12 of that Act).

1.2 Each time we alter the wording within a Scheme it has the effect of making a new Scheme. We publish our Schemes on our website .

1.3 This Scheme is known as 'The Royal Mail United Kingdom Post Scheme 27 April 2020' and is referred to in this document as 'this Scheme'. The date in the title is the date that the Scheme took (or will take) effect. This Scheme replaced the Royal Mail United Kingdom Post Scheme 25 May 2018. That old Scheme is no longer in force.

2. What this Scheme applies to

2.1 This Scheme sets out the terms & conditions for:

2.1.1 the services provided by us to the person, business or organisation asking for the service (you) when we accept items within the United Kingdom for postal delivery to an address in the United Kingdom, the Channel Islands and the Isle of Man. The terms and conditions for the specific services listed below are contained within this Scheme:

- · First Class.
- · Second Class,
- Royal Mail Signed For 1st Class
- Royal Mail Signed For 2nd Class
- Royal Mail Special Delivery Guaranteed by 1pm®2 ('Special Delivery'),
- · Articles for the Blind.
- Petitions and Addresses to the Sovereign and
- Petitions to Parliament & Assemblies.

All product names offered under this Scheme are shown, in this Scheme, in red text.

- 2.1.2 the service provided by us to accept an item sent from the Channel Islands or the Isle of Man for postal delivery to an address in the United Kingdom.
- 2.1.3 other services provided by us in connection with those mentioned in 2.1.1 and 2.1.2. Specifically these are:
- Return to Sender.
- Proof of Delivery,
- · Poste Restante,
- · Redelivery to Post Office, and
- Inflight Delivery Options.
- 3. Definitions
- 3.1 You may not be familiar with some of the words or phrases we use in this document. When they are used for the first time, they are shown in **bold print** and explained in the relevant part of this document or in Annex A.
- 3.2 As mentioned in 1.3 above this Scheme replaced a previous Scheme so any references in other documents to the previous Scheme that was in force before this Scheme should now be interpreted as referring to this new Scheme.
- 4. Our ability to provide services

- 4.1 We will provide the services set out in this Scheme where the relevant terms & conditions contained in this Scheme for the service are complied with. We will provide the services in the manner that we deem to be most appropriate according to our operational requirements and an overarching need to secure the sustainability of the Universal Service.
- 4.2 We may decide to withdraw the provision of any of the services set out in this Scheme if the terms & conditions contained within this Scheme are not abided by, in the event of misuse or if providing the service may cause us reputational damage.
- 5. What can and cannot be contained within an item
- 5.1 There are some items that we cannot carry through our postal network for legal or safety reasons. Items that we cannot carry through our postal network are known as **prohibited items**. Some other items can only be carried under certain circumstances. These are known as **restricted items** and details of those are contained after this section on prohibited items.

#### Prohibited items

- 5.2 We would like to make it very clear that as we cannot carry prohibited items through our network, we will not accept any liability for any item that contains prohibited items.
- 5.3 As the list of prohibited items can change from time to time (see 5.3.31), we maintain a full, up-to-date list on our website which can be viewed at any time. The list quoted below was correct at the time of publication:
- 5.3.1 Aerosols containing toxic, flammable or non-flammable compressed gas (including solvents, spray paints, air fresheners, polishes and other flammable or toxic materials),
- 5.3.2 Alcoholic beverages with an alcohol content greater than 70% ABV
- 5.3.3 Ammunition containing an explosive charge (excluding lead pellets and other airgun and airsoft projectiles)
- 5.3.4 Batteries that are classified as dangerous goods by the latest edition of the Technical Instructions for Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO) including wet spillable lead acid/lead alkaline batteries (such as car batteries), lithium batteries when not sent with or in equipment, damaged batteries of any type, together with used alkaline and nickel metal hydride (NiMH) batteries,

### 5.3.5 Bladed Items

- 5.3.6 Clinical and medical waste (e.g. contaminated dressings, bandages and needles),
- 5.3.7 Controlled drugs and narcotics (such as cannabis, cocaine, and heroin. LSD, opium, and amyl nitrate).
- 5.3.8 Corrosive substances which can cause severe damage to living tissue, other freight or transport by its chemical action (including dyes, acids, corrosive paint and rust removers, caustic soda, mercury and gallium metal).
- 5.3.9 Counterfeit currency, bank notes and **postage stamps** (including any false instrument or copy of a false instrument within the meaning of section 5 of the Forgery and Counterfeiting Act 1981) but excluding copies of old denominations or pre-decimalisation postage stamps which are now obsolete and worthless except for collectable value and cannot be passed as tender. Controlled drugs and narcotics (such as cannabis, cocaine, and heroin. LSD, opium, and amyl nitrate)
- 5.3.10 Dry ice (UN1845) when used as a coolant for biological substances (UN3373) or for other perishable items
- 5.3.11 Electronic items containing any batteries exceeding 100Wh (including some high performance laptops and power tools)
- 5.3.12 Environmental waste (including used batteries and used engine oil).
- 5.3.13 Explosives (including fireworks, flares, blasting caps, and Christmas cracker snaps),
- 5.3.14 Flammable liquids (including petroleum, lighter fluid, certain adhesives, solvent based paints, wood varnish, enamels and acetones including nail varnish removers),
- 5.3.15 Flammable solids (including magnesium, phosphorous, potassium, sodium, zinc and fire lighters),
- 5.3.16 Foreign Lottery tickets,
- 5.3.17 Frozen water e.g. packs of ice,
- 5.3.18 Gases including flammable, non-flammable, toxic and compressed gases, new, used and empty gas cylinders, butane, ethane, refills for lighters, fire extinguishers and scuba tanks, life jackets, nitrogen dioxide and carbon dioxide canisters (culinary foaming devices & soda stream gas cylinders),

- 5.3.19 Goods made in foreign prisons, except those imported for a non-commercial purpose, of a kind not manufactured in the United Kingdom, or those in transit,
- 5.3.20 Hover boards or any type of rechargeable battery-powered: self- balancing scooter, mono-wheel, stand-up unicycle or electric skateboard.
- 5.3.21 Human and animal remains including ashes other than as set out at 5.7.12 and 5.7.20 below,
- 5.3.22 Infectious substances and pathogens (UN2814 or UN2900) as classified in the latest Edition of the Technical Instructions for Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO),
- 5.3.23 Lighters and refills containing flammable liquid or gas (including used butane and petrol cigar and cigarette lighters),
- 5.3.24 Living animals and reptiles e.g. snakes, mice and rodents,
- 5.3.25 Live creatures, insects and invertebrates classified as dangerous within the Dangerous Wild Animals Act 1976 e.g. venomous spiders,
- 5.3.26 Magnetised material with a magnetic field strength of 0.159A/ metre or more at a distance of 2.1 metres from the outside of the package e.g. PA systems,
- 5.3.27 Matches (including safety matches)
- 5.3.28 Obscene publications and unlawful indecent images (including pornography as detailed in Part 5 of the Criminal Justice and Immigration Act 2008 and indecent photographs or pseudophotographs of a child as detailed in section 160 and 161 of the Criminal Justice Act 1988,
- 5.3.29 Oxidising materials or organic peroxides (including disinfectants, nitrates, and hair dyes or colourants containing peroxide),
- 5.3.30 Pesticides (including weed killer and any chemical used to kill pests and insects, including fly sprays),
- 5.3.31 Poisons, toxic liquids, solids and gases (including substances that are liable to cause death or injury if swallowed or inhaled or by skin contact such as arsenic, beryllium, cyanide, fluorine and rat poison),
- 5.3.32 Radioactive material and samples that are classified as dangerous goods in the latest edition of the Technical Instructions for the Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO) such as luminous dials from aircraft.
- 5.3.33 Paints, wood varnishes and enamels solvent-based (except nail varnish or nail polish with a volume of 30ml or less when sent to a UK destination)
- 5.3.34 Waste, dirt, filth or refuse (including household waste). Note-soil samples sent for analysis are permitted provided they are packaged so that any leaks and spills are contained in the outer packaging,
- 5.3.35 Weapons (including Section 5 firearms, CS gas and pepper sprays, flick knives, and other knives that are banned knives under UK laws, tasers and stun guns).
- $5.4~{\rm lt}$  is your responsibility not to send prohibited items in the post. If you are in any doubt you may seek advice from our website .
- 5.4.1 any item which resembles a prohibited item may be subject to additional scrutiny which may cause delays
- 5.4.2 any compensation claim for delay as a result of additional scrutiny is subject to the limits set out in Section 17 of the Scheme
- 5.4.3 failure to comply with these conditions could result in prosecution resulting in a fine or imprisonment
- 5.4.4 failure to comply with these conditions could affect your ability to claim compensation
- 5.4.5 if you post an item that contains a prohibited item we may deal with such item in our sole and absolute discretion (without incurring any liability whatsoever to you or the addressee) including disposing of the relevant item (in whole or in part)
- 5.4.6 the sender, or their representative, may be required to pay an amount (to be fixed by us) to cover the costs of returning, dealing with or disposing of the item
- 5.4.7 the sender is responsible for checking whether an item is prohibited.
- 5.4.8 we reserve the right to refuse any other item banned by law or that in our opinion may be harmful or dangerous to our customers or employees
- 5.4.9 any article or substance classified as dangerous goods is prohibited (other than specified in Restricted items below)
  Restricted items

- 5.5 There are some items, known as restricted items that we can, for legal or safety reasons, only carry through our postal network provided that you meet all our requirements for the acceptance of the item, including, but not limited to, packaging requirements.
- 5.6 We will not accept any liability for any item that contains restricted items if the requirements for the acceptance of those items are not met.
- 5.7 As the list of restricted items and requirements for each can change from time to time (see 5.7.15) we maintain a full, up-to-date list on our website (which can be viewed at any time). The list and detailsquoted below were correct at the time of publication:
- 5.7.1 Aerosols for personal grooming or medicinal purposes (including deodorants, body sprays, hair sprays, shaving and hair removal creams, medicinal aerosols for prevention or cure such as flea sprays, etc.)
- 5.7.1.1 Valves must be protected by a cap or other suitable means to prevent inadvertent release of the contents during transport. Aerosols must be tightly packed in strong outer packaging, and must be secured or cushioned to prevent any damage. Volume per item must not exceed 500ml. No more than two aerosols can be sent in any one package.
- 5.7.1.2 An ID8000 label must be applied. The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.2 Alcoholic beverages with an alcohol content less than 24% ABV (e.g. wine and champagne)
- 5.7.2.1 Volume per item should not exceed 1 litre per container, wrap in polythene and seal with tape. Surround with absorbent material such as newspaper and sufficient cushioning material to protect each item from breakage.
- 5.7.2.2 Mark as "FRAGILE" when sending glass bottles. The sender's name and return address must be clearly visible on the outer packaging
- 5.7.3 Alcoholic beverages with an alcohol content between 24% ABV 70% ABV (e.g. whisky, vodka, gin)
- 5.7.3.1 Volume per item should not exceed 1 litre per item. No more than two items to be sent in any one parcel. The items must be securely closed and placed in a leak-proof liner, such as a sealed polythene bag. Surround with absorbent material and sufficient cushioning material to protect each item from breakage.
- 5.7.3.2 Mark as 'FRAGILE' when sending glass bottles. The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.4 Asbestos
- 5.7.4.1 Samples of asbestos may be posted provided the sample is fixed within an inert material such as a glue or resin. Surround with cushioning material e.g. bubble wrap.
- 5.7.4.2 The sender's name and return address must be clearly visible on he outer packaging.
- 5.7.5 Balloons filled with non-flammable gas
- 5.7.5.1 Must be clearly marked on the outer packaging with the words 'BALLOONS FILLED WITH NON-FLAMMABLE GAS'.
- 5.7.6 Batteries new alkaline, nickel metal hydride (NiMH) or nickel cadmium (NiCd) including D, C, 9V, AA, AAA and AAAA alkaline batteries
- 5.7.6.1 Must be new and sent unopened in their original retail packaging. Surround with cushioning material e.g. bubble wrap.
- 5.7.6.2 The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.7 Batteries Lithium ion/polymer batteries sent with equipment (e.g. rechargeable batteries sent with electronic devices including mains powered battery chargers but excluding battery packs)
- 5.7.7.1 The maximum number of batteries allowed in each package is the number that may be connected to the equipment plus two spares. The maximum net quantity of cells or batteries is 5kg per package. Watt-hour rating must not exceed 20Wh per cell or 100Wh per battery. Cells or batteries that are defective for safety reasons, or that have been damaged, are forbidden. Cells and batteries must be protected against short circuit. Cells and batteries must be packed in inner packaging that completely encloses the cell or battery. Cells and batteries must be protected against short circuit, including protection against contact with conductive materials within the same packaging that could lead to a short circuit. The equipment sent with cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidental activation.

- 5.7.7.2 Lithium battery handling label to be applied. The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.8 Batteries lithium ion/polymer batteries contained in equipment (e.g. rechargeable batteries found in electronic devices)
- 5.7.8.1 Each package must contain no more than four cells or two batteries installed in equipment. The maximum net quantity of cells or batteries is 5kg per package. Watt-hour rating must not exceed 20Wh per cell or 100Wh per battery. Cells or batteries that are defective for safety reasons, or that have been damaged, are forbidden. Cells and batteries must be protected against short circuit. Cells and batteries must be protected against short circuit. The equipment containing cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidental activation.
- 5.7.8.2 The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.9 Batteries lithium metal/alloy batteries sent with equipment (e.g. non- rechargeable batteries found in electronic devices)
- 5.7.9.1 The maximum number of batteries allowed in each package is the minimum number required to power the equipment plus two spares. The maximum net quantity of cells or batteries is 5kg per package. The lithium content must not be more than 1g per cell or 2g per battery. Cells or batteries that are defective for safety reasons, or that have been damaged, are forbidden. Cells and batteries must be protected against short circuit. Cells and batteries must be packed in inner packaging that completely encloses the cell or battery. Cells and batteries must be protected against short circuit, including protection against contact with conductive materials within the same packaging that could lead to a short circuit. The equipment sent with cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidental activation.
- 5.7.9.2 Lithium battery handling label to be applied. The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.10 Batteries lithium metal/alloy batteries contained in equipment (e.g. non-rechargeable batteries found in electronic devices)
- 5.7.10.1 Each package must contain no more than four cells or two batteries installed in equipment. The maximum net quantity of cells or batteries is 5kg per package. The lithiumcontent must not be more than 1g per cell or 2g per battery. Cells or batteries that are defective for safety reasons, or that have been damaged, are forbidden. Cells and batteries must be protected against short circuit. The equipment containing cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidental activation.
- 5.7.11.1 The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.11 Batteries new wet, non-spillable (e.g. sealed lead acid batteries, absorbed glass mat and gel cell batteries)
- 5.7.11.1 Batteries must comply with Special Provision 238 of the UN Recommendations on the Transport of Dangerous Goods, Model Regulations (please check with the manufacturer or distributor). No more than one battery in any one package. Maximum weight 1.5kg. Item must be protected against short circuit (by the effective insulation of exposed terminals) and securely packaged.
- 5.7.11.2 Package must be marked "NOT RESTRICTED" and "SPA67 / SP238". The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.12 Biological substances (Diagnostic specimens including blood urine, faeces and animal remains. Category B (UN3373) as classified in the latest edition of the Technical Instructions for Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organization (ICAO))
- 5.7.12.1 May only be sent by, or at the specific request of, a qualified medical practitioner, registered dental practitioner, veterinary surgeon, registered nurse or a recognised laboratory or institution. The total sample volume/mass in any parcel must not exceed 50ml/50g. All biological substances must be posted in packaging that complies with Packaging Instruction 650, which provides guidance on suitable packaging for these items.
- 5.7.12.2 The sender's name and return address must be clearly visible on the outer packaging.
- 5.7.13 Christmas crackers
- 5.7.13.1 Can only be sent in their made up form in their original retail packaging.

5.7.13.2 The sender's name and return address must be clearly visible on the outer packaging. 5.7.14 Electronic items sent with new alkaline, nickel metal hydride (NiMH) or nickel cadmium (NiCd) batteries

5.7.14.1 Must be new and sent unopened in their original retail packaging. Surround with sufficient cushioning material to protect each item from damage. Wrap each item, including plugs, individually. Place item in a rigid container and cushion to avoid movement. Any equipment sent with batteries or cells must be secured against movement within the outer packaging and must be packed to prevent accidental activation.

5.7.14.2 The sender's name and return address must be clearly visible on the outer packaging.

5.7.15 Electronic items sent with new and used lithium ion/polymer batteries (e.g. rechargeable batteries sent with electronic devices including mains powered battery chargers but excluding battery packs)

5.7.15.1 The maximum number of batteries allowed in each package is the number that may be connected to the equipment plus two spares. The maximum net quantity of cells or batteries is 5kg per package. Watt-hour rating must not exceed 20Wh per cell or 100Wh per battery. Cells or batteries that are defective for safety reasons, or that have been damaged, are forbidden. Cells and batteries must be protected against short circuit. Cells and batteries must be packed in inner packaging that completely encloses the cell or battery. Cells and batteries must be cellprotected against short circuit, including protection against contact with conductive materials within the samepackaging that could lead to a short circuit. The equipmentsent with cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidentalactivation.

5.7.15.2 Lithium battery handling label to be applied. The sender's name and return address must be clearly visible on the outer packaging.

5.7.16 Electronic items sent with new and used lithium ion/polymer batteries contained in the device

5.7.16.1 Each package must contain no more than four cells or two batteries installed in equipment. The maximum net quantity of cells or batteries is 5kg per package. Watt- hour rating must not exceed 20Wh per cell or 100Wh perbattery. Cells or batteries that are defective for safety reasons, or that have been damaged, are forbidden. Cells and batteries must be protected against short circuit. Cellsand batteries must be protected against short circuit. The equipment containing cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidental activation.

5.7.16.1 The sender's name and return address must be clearly visible on the outer packaging.

5.7.17 Electronic items sent with new and used lithium metal/alloy batteries

5.7.17.1 The maximum number of batteries allowed in each package is the minimum number required to power the equipment plus two spares. The maximum net quantity of cells or batteries is 5kg per package. The lithium content must not be more than 1g per cell or 2g per battery. Cells or batteries that are defective for safety reasons, or that have been damaged, are forbidden. Cells and batteries must be protected against short circuit. Cells and batteries must be packed in inner packaging that completely encloses the cell or battery. Cells and batteries must be protected against short circuit, including protection against contact with conductive materials within the same packaging that could lead to a short circuit. The equipment sent with cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidental activation.

5.7.17.2 Lithium battery handling label to be applied. The sender's name and return address must be clearly visible on the outer packaging.

5.7.18 Electronic items sent with new and used lithium metal/alloy batteries contained in the device

5.7.18.1 Each package must contain no more than four cells or two batteries installed in equipment. The maximum net quantity of cells or batteries is 5kg per package. The lithiumcontent must not be more than 1g per cell or 2g per battery. Cells or batteries that are defective for safety reasons, or that have been damaged, are forbidden. Cells and batteries must be protected against short circuit. The equipment containing cells or batteries must be packed in strong rigid packaging and must be secured against movement within the outer packaging and packed to prevent accidental activation.

5.7.18.2 The sender's name and return address must be clearly visible on the outer packaging

5.7.19 Guns for sporting use (Guns intended for sporting purposes - including Section 1 and Section 2 firearms, low-powered air guns and their component parts - may be sent in compliance with UK law subject to any applicable controls on the possession of firearms

5.7.19.1 Use First Class as the minimum service.

5.7.19.2 The sender's name and address must be clearly visible on the outer packing.

5.7.20 Human and animal ashes:

5.7.20.1 Volume of human or animal ashes per item must not exceed 50 grams; and

5.7.20.2 Human and animal ashes must be placed in a sift-proof container and securely closed. Items must be tightly packed in strong outer packaging and must be secured or cushioned to prevent any damage. The sender's name and return address must be clearly visible on the outer packaging.

5.7.21 Lighters (new and unused empty lighters)

5.7.21.1 Must be new, empty and sent unopened in their original retail packaging.

5.7.21.2 A sender's name and return address must be clearly visible on the outer packaging

5.7.22 Liquids over 1 litre (containing liquids not classified as dangerous goods)

5.7.22.1 Some liquids, such as alcohol or paints, have their own additional restrictions. The items must be securely closed and placed in a leak-proof liner, such as a sealed polythene bag, so that any inadvertent leakage is contained within the outer packaging Surround with absorbent material such as newspaper and sufficient cushioning material to protect each item from damage.

5.7.22.2 Mark as "FRAGILE" when sending glass bottles. The sender's name and address must be clearly visible on the outer packaging.

5.7.23 Live creatures, insects and invertebrates (including bees, caterpillars, cockroaches, crickets, destroyers of noxious pests, earthworms, fish fry and eggs, leeches and other parasites, lugworms, maggots, mealworms, pupae and chrysalides, rag worms, silkworms, spiders and stick insects)

5.7.23.1 Must be boxed and packaged to protect the creatures, our staff and our customers from harm. Use First Class as the minimum service.

5.7.23.2 Items must be clearly marked 'URGENT - LIVING CREATURES - HANDLE WITH CARE'. The sender's name and address must be clearly visible on the outer packaging

5.7.24 Lottery tickets

5.7.24.1 UK lottery tickets are allowed in the domestic post.

5.7.25 Magnetised materials, other than those that are prohibited (including loud speakers)

5.7.25.1 The magnetised material must have a magnetic field strength of less than 0.159A/m at a distance of 2.1m from the outside of the package Wrap soft packing material at least 2cm thick around each item.

5.7.25.2 The sender's name and return address must be clearly visible on the outer packaging

5.7.26 Nail varnish and polish

5.7.26.1 Volume per item must not exceed 30ml. No more than four bottles of nail varnish can be sent in any one package. Bottles of nail varnish must be placed in strong outer packaging and be so packed, secured or cushioned in such a way that they cannot break, be punctured or leak their contents into the outer packaging.

5.7.26.2 An ID8000 label must be applied. The sender's name and return address must be clearly visible on the outer packaging.

5.7.27 Paints, wood stains and enamels - water-based

5.7.27.1 The items must be securely closed and placed in a leak-proof liner, such as a sealed polythene bag, so that any inadvertent leakage is contained within the outer packaging. In the case of water-based paints, woods stains and enamels, there is no limit on the number of items that can be sent in any one package. Surround with absorbent material such as newspaper and sufficient cushioning material to protect each item from damage. Volume per item should not exceed 150ml.

5.7.27.2 The sender's name and return address must be clearly visible on the outer packaging.

5.7.28 Perfume and aftershave (including eau de parfum and eau de toilette but excluding non-flammable perfumed creams, gels, oils or lotions)

5.7.28.1 Volume per item must not exceed 150ml. No more than four perfumes or aftershaves can be sent in any onepackage. The perfume or aftershave must be within its original retail packaging and then placed in strong outer packaging. The inner packagings must be packed, securedor cushioned to prevent breakage or leakage of their contents into the outer packaging.

5.7.28.2 An ID8000 label must be applied. The sender's name and return address must be clearly visible on the outer packaging.

5.7.29 Perishables (including flowers, fresh fruit, vegetables and frozen or chilled foodstuffs)

5.7.29.1 Packaging should be able to withstand a journey of up to 48 hours. You should use First Class as the minimum service. Must be suitably sealed to prevent leakage or tainting of other items such as in sealed vacuum packs.

5.7.29.2 The sender's name and return address must be clearly visible on the outer packaging.

5.7.30 Prescription medicines and drugs sent for scientific or medical purposes (non-toxic and non-flammable)

5.7.30.1 May only be sent by, or at the specific request of, a qualified medical practitioner, registered dental practitioner, veterinary surgeon, registered nurse or a recognised laboratory or institution. The medicines must be securely closed and placed in a leak-proof container such as a sealed polythene bag (for liquids) or a siftproof container (for solids). Must be tightly packed in strong outer packaging and must be secured or cushioned to prevent any damage. 5.7.30.2 The sender's name and return address must be clearly visible on the outer packaging.

5.7.31 Prescription medicines and drugs sent for scientific or medical purposes (toxic, flammable or toxic and flammable)

5.7.31.1 May only be sent by, or at the specific request of, a qualified medical practitioner, registered dental practitioner, veterinary surgeon, registered nurse or a recognised laboratory or institution. Volume/ mass per item must not exceed 50ml/50g. No more than eight items can be sent in any one parcel. The medicines must be securely closed and placed in a leak-proof container such as a sealed polythene bag (for liquids) or a sift-proof container (for solids). Must be tightly packed in strong outer packaging and must be secured or cushioned to prevent any damage.

5.7.31.2 ID8000 label must be applied. The sender's name and return address must be clearly visible on the outer packaging.

5.7.32 Radioactive material and samples that are not classified as dangerous goods in the latest edition of the Safe Transport of Dangerous Goods by Air published by the International Civil Aviation Organisation's Technical Instructions (ICAO) e.g. samples of granite rock

5.7.32.1 Surround with cushioning material e.g. bubble wrap.

5.7.32.2 The sender's name and return address must be clearly visible on the outer packaging.

5.7.33 Sharp objects and instruments (which are not classified as prohibited items)

5.7.33.1 Wrap heavy cardboard around sharp edges and points, strong enough to ensure that the contents do not pierce the outer packaging. Wrap each item with cushioning material. Place in a suitable outer container such as a padded envelope.

5.7.33.2. The sender's name and return address must be clearly visible on the outer packaging.

5.7.34 Vaccines that are not classified as dangerous goods

5.7.34.1 May only be sent by, or at the specific request of, a qualified medical practitioner, registered dental practitioner, veterinary surgeon, registered nurse or a recognised laboratory or institution. The vaccines must be securely closed and placed in a leak-proof liner such as a sealed polythene bag (for liquids) or a sift-proof container (for solids). Must be tightly packed in strong outer packaging and must be secured or cushioned to prevent any damage.

5.7.34.2 The sender's name and return address must be clearly visible on the outer packaging.

5.7.35 Human or animal samples

5.7.35.1 May only be sent by, or at the specific request of, a qualified medical practitioner, registered dental practitioner, veterinary surgeon, registered nurse or a recognised laboratory or institution.

5.7.35.2 The total sample volume/mass in any parcel must not exceed one kg. Solids only can be sent

5.7.35.3 All human or animal samples must be posted in packaging that complies with Packaging Instruction 650, such as our Safebox product.

5.7.35.4 The sender's name and return address must be clearly visible on the outer packaging.

5.7.36 Valuables can only be sent using the Special Delivery service.5.8 It is your responsibility to ensure that restricted items are sent

correctly and that you meet the requirements for acceptance. If you are in any doubt as to whether an item is one of the restricted items you can seek advice from our website.

5.8.1 any item which resembles a restricted item may be subject to additional scrutiny which may cause delays

5.8.2 any compensation claim for delay as a result of additional scrutiny is subject to the limits set out in Section 17 of the Scheme

5.8.3 failure to comply with these conditions could result in prosecution resulting in a fine or imprisonment

5.8.4 failure to comply with these conditions could affect your ability to claim compensation

5.8.5 if you post an item that contains a restricted item (and you do not comply with the relevant restrictions or requirements) we may deal with such item in our sole and absolute discretion (without incurring any liability whatsoever to you or the addressee) including disposing of the relevant item (in whole or in part)

5.8.6 the sender, or their representative, may be required to pay an amount (to be fixed by us) to cover the costs of returning, dealing with or disposing of the item

5.8.7 the sender is responsible for checking whether an item is restricted.

5.8.8 we reserve the right to refuse any other item banned by law or that in our opinion may be harmful or dangerous to our customers or employees

5.8.9 any article or substance classified as dangerous goods is prohibited (other than specified in Restricted items above)

6. Size & weight limits and how to package an item

6.1 In addition to the specific packaging and other requirements that relate to restricted items there are some general rules set out below that must be followed for all items as to weight, size, thickness and packaging.

6.2 Items sent using the First Class, Second Class, Royal Mail Signed For 1st Class or Royal Mail Signed for 2nd Class services may weigh up to 20 kilograms. Special Delivery can be used for items which weigh up to 10 kilograms or up to 20 kilograms where Special Delivery services are purchased at a Post Office® branch.

6.3 The maximum size of an item must not exceed 610mm in length by 460mm in width by 460mm in depth. For a cylinder shaped item the length plus twice the diameter may not exceed 1040mm with the greatest dimension being no more than 900mm. The maximum size for each format of item is set out in section 9.1 below.

6.4 The minimum thickness for an item is 0.25mm. We may refuse to accept or to deliver an item that is less than 0.25mm thick.

6.5 All items must be packaged such that, in our opinion, the item will not be likely to cause damage or harm to other items, equipment or individuals (including any recipient).

6.6 The **cover** and the protective material of the item must be appropriate to the contents such that we can convey it to its destination undamaged without having to take precautions over and above those normally taken by us with an item of mail. Packaging guidelines are available on our website and in booklets held by Post Office® branches.

6.7 With the exception of items that are sent using the Articles for the Blind service (which are discussed in more detail section 22) all items must be securely sealed or fastened.

6.8 In addition to the requirement to provide sufficient protection both for the item itself and so as not to be likely to cause damage or harm to other items, equipment or individuals as set out in sections 6.5 and 6.6, any item that contains anything breakable must be clearly and legibly marked with the words 'FRAGILE – HANDLE WITH CARE' in capital letters above the address.

6.9 In addition to the requirement to provide sufficient protection for the contents as set out in sections 6.5 and 6.6 any item that has contents that may be damaged by bending must be clearly and legibly marked with the words 'DO NOT BEND' in capital letters above the address.

6.10 In the same way that the list of restricted items can change from time to time so too can our packaging requirements for certain new or existing restricted items. You can review the full, up-to-date list of restricted items and associated packaging requirements on our website.

6.11 The latest packaging guidelines which apply to all items can also be found on our website.

7. How to address an item

7.1 Each item must be fully and correctly addressed. All the elements of the address (including the full postcode) must be written clearly and legibly on the front or on a label securely attached to the front of the cover (or otherwise in a manner such that the address is clearly visible in its entirety as if it had been written on the front of the cover) of the item.

7.2 You must ensure that addresses take the following format when added to an item. Please do not use commas or full stops and note that only the TOWN and POSTCODE should be in CAPITAL LETTERS. You don't need to include the county name. The postcode must be on a line of its own. The text that makes up the address should be aligned to the left as it is in the example below. It should not be staggered or centred.

**Example Name** 

House name or number and road name

Locality Name if one exists

TOWN POST CODE

The text set out in the example above and taken as a whole is known as the address block. The address block should appear in the centre of the front face of the item.

7.3 Postcodes are allocated by us at our discretion throughout the UnitedKingdom in accordance with the operational needs of our network. The postcodes are routing codes and we may amend them at our discretion fromtime to time. Where we do make changes we will ensure that all residents ofthe affected addresses are given sufficient notice and, in certain cases, as detailed in the PAF® File Code of Practice, we will follow a formal notification and consultation process before confirming changes.

7.4 The address must not be obscured or made difficult to read in any way including by the manner in which the contents are enclosed in the item.

7.5 Putting a correct address in the correct format in the correct location on an item helps our sorting machines process the mail as quickly as possible. More information is available on our website.

7.6 We recommend that you put a full and accurate return address on your item (in a position where it can't be confused with the destination address and preferably on the back) so that we are able to return the item to you, intact, should we need to. More information regarding Return to Sender can be found in section 15.

8. How to pay for postage and other services

8.1 Of the services contained within this Scheme the following are provided (as long as specific requirements are met) free of charge:

- Articles for the Blind.
- Petitions and Addresses to the Sovereign,
- Petitions to Parliament & Assemblies and
- Poste Restante.

The specific requirements for these services are set out in sections 22 to 26 below of this Scheme.

8.2 The other services that are contained in this Scheme require payment for **postage**. These services are:

- First Class,
- Second Class.
- Royal Mail Signed For 1st Class,
- Royal Mail Signed For 2nd Class and
- · Special Delivery.

The specific requirements for First Class, Second Class with delivery confirmation, Royal Mail Signed For 1st Class, Royal Mail Signed For 2nd Class and Special Delivery are set out below in sections 19 to 21 of this Scheme

8.2.1 The services listed above in 8.2 can be paid for by applying postage stamps, **postage labels** (which can be generated by our online postage applications or by Post Office® branches) or by use of a franking machine<sup>3</sup>.

8.2.2 First Class, Second Class, Royal Mail Signed For 1st Class, and Royal Mail Signed For 2nd Class services can also be purchased using a credit account.

8.3 Stamps can be purchased from us online from our website, from Post Office® branches or from a wide variety of other high street retailers. When postage is purchased from a Post Office® branch it often takes the form of a postage label if it is for anything but the first weight band. Similar postage labels are also available from our online postage tool and through our computer applications (e.g. SmartStamp®) which you may subscribe to or sign up for.

8.4 A franking machine allows money to be credited onto it. Postage for items is then paid for, from this credit, on an item by item basis. There is a wide range of franking machines which can either be leased or bought from franking manufacturers. If you use a franking machine you must abide by the terms and conditions that are contained within a separate Scheme document called Royal Mail Franking Letters And Parcels Scheme (which is publishedon our website) as well as the terms and conditions contained in this Scheme document.

8.5 A credit account allows you to **post** items without pre-payment. You are sent an invoice requesting payment for the items we have conveyed in the preceding period of time. If you set up a credit account with us you must abide by the terms and conditions that are contained within documents called the Royal Mail General Terms & Conditions and the Royal Mail Account Terms (both of which can be found on our website as well as the terms and conditions contained in this Scheme document.

9. How to work out how much postage to pay

9.1 For the services listed in this Scheme that are not free, the amount of postage that you must pay is dependent on several factors. The first is how quickly you would like the item to arrive (the service), the second is what weight the item is (the weight) and the third is how much compensation is required in the event of loss or damage. For items sent using the First Class, Second Class and Royal Mail Signed For 1st & 2nd Class services the amount of postage payable also varies by format (i.e. shape, size & weight):

9.1.1 items which do not exceed 100g in weight and 240mm by 165mm by 5mm in size are deemed to be **letters** for pricing purposes;

9.1.2 items which exceed 100g or 240mm by 165mm by 5mm (in any one or more of the three dimensions) but do not exceed 750g and 353mm by 250mm by 25mm are deemed to be **large letters** for pricing purposes;

9.1.3 items which exceed 353mm by 250mm by 25mm (in any one or more of the three dimensions) up to maximum dimensions of 450 mm by 350mm by 160mm and cylinder shaped items that do not exceed 450mm in length and has a diameter no greater than 160mm neither of which exceed 2 kilograms in weight are deemed to be **small parcels** for pricing purposes; and

9.1.4 items which exceed 450 mm by 350mm by 160mm (in any one or more of the three dimensions) up to maximum dimensions of 610mm by 460mm by 460mm which do not exceed 20 kilograms in weight are deemed to be **medium parcels** for pricing purposes. Exceptions apply (see 9.1.5).

9.2 We aim to deliver a Special Delivery item by 1pm the next working day after it has been posted. Please note that exemptions do apply in certain circumstances. All Special Delivery items are priced by service and by weight. Compensation for loss or damage is available on the basis of actual loss of up to the maximum of the market value of the item or £500 (whichever is the lower) comes as standard. Special Delivery is available with enhanced compensation options. Further add on services can also be purchased with Special Delivery such as a Saturday Guarantee or Consequential Loss for additional service fees. Further specific details, including compensation, exemption and suspension information, can be found in sections 17.7.4 & 21 below and on the Special Delivery website. 9.3 We aim to deliver:

- A First Class item the next working day after it has been posted.
- A Second Class item within three working days after it has been posted.

Compensation for loss or damage for any format of item sent using the First Class or Second Class services in accordance with this Scheme is available on the basis of actual loss up to the maximum of the market value of the item or £20 (whichever is the lower) comes as standard.

9.4 We aim to deliver:

• A Royal Mail Signed for 1st Class item the next working day after it has been posted.

• A Royal Mail Signed for 2nd Class item within three working days after it has been posted.

Compensation for loss or damage for any format of item sent using Royal Mail Signed for 1st Class or Royal Mail Signed For 2nd Class services in accordance with this Scheme is available on the basis of actual loss up to the maximum of the market value of the item or  $\mathfrak{L}50$  (whichever is the lower) comes as standard.

Further details of how the Royal Mail Signed For services work can be found in section 20 below.

9.5 To help you find the best service for your needs and budget we provide a price finder tool on our website. Prices and other pricinginformation can also be found in other locations on the same website (such as product specific pages) including in a Pricing document with all pricing information for the services mentioned in sections 9.1.1 to 9.1.4.

9.6 We may change the rates of postage and service fees from time to time. Changes to postage rates & service fees and the date on which the changed rates and fees will come into effect will be published at least one month (or whatever period of time agreed by us with our regulator, Ofcom) in advance on our website (www.royalmail.com).

10. How to show that an item has had postage paid

10.1 You must show us that postage for an item has been paid for (where it applies) so that we can convey it without delay. Payment is demonstrated by the application of a **postage mark** to the item. The manner in which payment of postage can be shown depends on the method of payment used. Section 8 above set out the most common ways of applying a postage mark (postage stamps, postage labels, **service fee labels, franking marks**, marks created by online applications such as SmartStamps®, **Printed Postage Impressions** (PPIs) etc). In addition, a postage mark may be fixed, printed, impressed, embossed or otherwise shown on the envelope or cover of an item as we may specify or approve.

10.2 Postage marks must be applied to the top right hand corner of the envelope or cover of the item unless we specify otherwise. If the item is spherical or without corners the postage mark should be placed above and to the right of the address.

10.2.1 A postage mark that is damaged or imperfect, has been defaced or has anything written or printed over it becomes void and cannot be used to show that postage has been paid.

10.2.2 You can use an adhesive postage stamp which has been perforated by means of a punch provided that the perforation holes are smaller than the holes dividing one stamp fromanother on a sheet of stamps.

10.2.3 Postage marks can be used only once and then they become void and must not be re-used to pay postage on the same or another item.

10.2.4 Any postage mark that carries an expiry time and/or date will become void from that time and/or date and cannot be used to show that postage has been paid after that time and/or date.

10.3 If you use a franking machine you must apply a franking mark to the cover of the item as set out in the Royal Mail Scheme For Franking Letters And Parcels (which is published on our website) and according to any guidelines that we publish on our Franking help centre website.

10.4 If you make use of a credit account you must apply a Printed Postage Impression (PPI) in accordance with the PPI Licence, the PPI design notes and any guidelines that we publish on our website.

11. Other requirements relating to the address on the cover of an item 11.1 Section 7 set out how to address an item and section 10 set out how payment can be indicated. There are some further things that must be considered with regard to the cover of the item so that we can efficiently convey the item without delay.

11.2 An item must not have on its cover:

11.2.1 anything which obscures the postage mark,

11.2.2 anything which is likely, in our opinion, to make it difficult for us to apply a postmark to the cover,

11.2.3 anything which, in our judgement, is likely to make the postmark illegible,

11.2.4 any counterfeit or fake postage mark,

11.2.5 any postage mark which we consider may have previously been used to pay postage,

11.2.6 signs, words or marks, used without official permission, which could indicate that the item was sent on Her Majesty's Service or was conveyed and delivered by us or

11.2.7 any signs, words, images or designs which are offensive, obscene or indecent.

11.3 An item must not be made up in any way or have anything on its cover which, in our judgement makes it difficult or embarrassing to deliver the item.

11.4 A postage mark must not be attached to an item using staples, sticky tape, or any other means that we do not approve.

12. How to hand over (or post) an item

12.1 All items sent using a service provided through this Scheme can be posted in the following ways.

12.1.1 By handing it to an authorised member of staff at a Royal Mail building (typically a Delivery Office, Customer Service Point or Mail Centre) as long as we consider it feasible to do so.

12.1.2 By handing it to an authorised member of staff at a Post Office® as long as considered feasible to do so.

12.1.3 By Business Collection (typically a paid for extra service where we collect the mail from a **business** customer's premises. Business Collections are not covered by this Scheme).

12.1.4 Or in any other manner which we may approve.

12.2 In addition the following services:

First Class.

· Second Class, and

Articles for the Blind.

can be posted in the following ways:

12.2.1 by placing it in a post box (typically a red (or very rarely gold) post box on the street) or

12.2.2 by placing it in a **private post box** (typically a post box contained with a shop or other private premises).

12.2.2.1 An item placed in a private post box will be deemed to have been posted when collected by us and not before.

Please note that delivery confirmation for small parcels and medium parcels sent by First Class or Second Class, requires the application of an appropriate postage label purchased online or from a Post Office. The labels are expected to be available online from January 2017<sup>4</sup> and at a Post Office® from April 2017<sup>5</sup>. Please see section 19 for further details of delivery confirmation.

12.3 An item that has had postage paid by a franking machine must be posted in accordance with the Royal Mail Scheme for Franking Letters and Parcels which is published on our website.

12.4 Items sent through the First Class, Second Class, Royal Mail Signed For 1st Class and Royal Mail Signed For 2nd Class services and via the Special Delivery service that have been paid for with a credit account must be separated out for posting by:
(i) class.

(ii) format (if appropriate to the service) and

(iii) service.

and presented to us in trays, bags, pouches or other containers supplied by us and also in accordance with any conditions set out in the PPI Licence.

12.5 Articles for the Blind sent by a business must present items separate to but in the same manner (as required by section 12.4 above) and at the same time as any items which are posted using a credit account

12.6 All items must be posted by the latest acceptance time that we determine and advertise or agree with you (as appropriate to the method of posting). Items posted after that latest acceptance time will be considered to have beenposted the following working day.

13. How we will treat items that do not meet the requirements set out in this scheme

13.1 Once an item has been posted we may carry out checks on that item.

13.2 Subject to sections 5.4 and 5.8 above (which set out how we treat items that contain prohibited items and/or restricted items where the relevant restrictions or requirements have not been complied with), if an item is found not to meet the conditions set out in this Scheme or an item is found that indicates that insufficient (or no) postage and/or service fee has been paid for the service requested, we may decide at our discretion to do one of the following things set out below within a reasonable period of time.

13.2.1 Deliver the item to the address shown by the service requested.

13.2.2 Deliver the item to the address shown by an alternative service that we determine to be appropriate. In these situations the conditions of the service used to convey the item will apply.

13.2.3 Convey the item part of the way and inform the **addressee** that an item can be collected from a specified location (normally a Royal Mail Customer Service Point) by the addressee or their **representative**.

13.2.4 Treat the item as undeliverable. Details of how we may deal with undeliverable mail can be found in section 15.

13.2.5 Return the item to you.

13.2.6 Refuse to accept it.

13.2.7 If none of the above courses of action are possible we may deal with or dispose of the item at our discretion.

In all cases we may not accept any liability for that item. This provision also applies to items which have been found either in a Post Office® branch or on Royal Mail premises and which appears intended for posting.

13.3 In each case set out in section 13.2 we may require the addressee or you to pay an amount (to be fixed by us in order to cover additional costs) before the item is delivered or released for collection.

13.3.1 In the event of underpaid or unpaid postage and/or service fees the amount due may include a surcharge in addition to the underpaid or unpaid postage to cover administrative costs.

13.3.2 Where the amount due is not paid the item may be dealt with or disposed of at our discretion.

13.4 In each case set out in sections 5.4, 5.8 and 13.2.4 to 13.2.7 we may additionally require you or your representative to pay an amount (to be fixed by us) to cover the costs of returning, dealing with or disposing of the item.

14. Additional marks

14.1 In the course of conveying an item we (or someone authorised by us) may alter the look of the cover of that item. Most usually this is by the addition of a postmark which shows that a stamp has been used and prevents it from being used again. A postmark may also contain an advertising or informative image. We may apply the postmark in any manner we consider appropriate. It is important to note that we may apply any form of postage mark (an image that denotes payment), impression, image, sticker, label or other device we consider suitable for any other postal purpose we consider appropriate.

14.2 We may also charge you a fee (to be fixed by us from time to time) and published on our website for cancelling postage marks with a special postmark including one that incorporates an advertisement, atyour request.

15. Delivery including undeliverable and re-posted (Return to Sender) items

15.1 Subject to sections 5.4 and 5.8, we will attempt to deliver items that meet the conditions for the relevant service set out in this Scheme.

15.2 Some items require a signature at the point of delivery and some are too large to fit through a letter box. In these situations someone needs to be present at the address to take delivery of the item.

15.2.1 If no one is available we may attempt to deliver the item to a **neighbour**. Please note that we will never attempt to deliver an item to a neighbour if it is a Special Delivery item or it is an item sent using

a service provided under this Scheme and is **Social Security post**<sup>6</sup>. 15.2.2 If a neighbour is available to take delivery of the item we will

leave the item with that neighbour and leave a card at the address that the item was addressed to. That card will inform the addressee (or their representative) that an item is (or items are) being held by a neighbour for them and provide the address details for that neighbour and the date and the time that the item was left with the neighbour.

15.2.3 If an addressee does not want their items to be delivered to a neighbour or does not want to take items for neighbours they can opt out of the Delivery to Neighbour service, free of charge, by registering their wish to opt out in such a way as Royal Mail may reasonably require from time to time.

15.2.4 If a neighbour is not available to take delivery of the item, or the addressee has opted-out and does not want their items to be delivered to a neighbour or to take items for neighbours, wewill take the item back to the Delivery Office or an alternative location such as a Post Office® branch. We will leave a card at the address that the item was addressed to.

15.2.5 The card that is left at the address will inform the addressee (or their representative) that we attempted to deliver an item or items and will offer the addressee (or their representative) one or more of the following opportunities:

15.2.5.1 to request that the item be redelivered to the same address,

15.2.5.2 to request that the item be redelivered to an alternative local<sup>7</sup> address. Please note that this option is not available with Special Delivery items or any item sent using a service provided under this Scheme and is Social Security post,

15.2.5.3 to collect (after the required period of time stated on the card has elapsed to allow for the item to reach that location and once identification has been provided that we deem to be suitable) the item from the Delivery Office or the alternative location such as a Post Office® branch to which the item was taken back to or

15.2.5.4 to request that the item be sent from the Delivery Office or other location (other than a Post Office® branch) to a Post Office® branch for collection. This service is called Redelivery to Post Office and incurs an additional fee. When collecting the item the addressee (or their representative) must provide the card that was left at the address and identification that we deem acceptable. Further details including the current fee and identification requirements can be found on our website.

15.3 We may decide not to deliver an item which we consider impracticable or unreasonable to deliver. Such an item is referred to in this Scheme as undeliverable. We may consider an item to be undeliverable if we are of the opinion that:

15.3.1 following a failed first delivery attempt, the addressee has not taken advantage of any of the opportunities set out in 15.2.5 within the advertised holding period,

15.3.2 following a failed first delivery attempt, we are required, by law to return an item immediately to you (e.g. Royal Mail's obligations in relation to Social Security Post),

15.3.3 the address is unsafe or unsecure or if no delivery point has been provided, or if access to the delivery point has been prevented or impeded.

15.3.4 the address to which the item is to be delivered is not permanently occupied $^8$ ,

15.3.5 the item is not fully and correctly addressed in a manner which includes all the elements of the address (including the full postcode), written clearly on the front or on a label securely attached to the front of the cover (or otherwise in a manner such that the address is clearly visible in its entirety as if it had been written on the front of the cover), 15.3.6 the address is illegible.

15.3.7 the health and safety of any individuals may be put at risk or 15.3.8 for any other reason, we form the opinion that it is impracticable or unreasonable to deliver the item.

15.4 Where an item is undeliverable and:

15.4.1 your name and your address (as long as it is in the United Kingdom, the Channel Islands of the Isle of Man) are legible on the cover, we may return the item to you unopened.

15.4.2 the item was originally paid for by franking machine or by using a credit account and your name and address are (a) not on the cover or (b) they are illegible or (c) your name and address are legible on the cover but the address is outside the United Kingdom, the Channel Islands or the Isle of Man, then the item may be dealt with or disposed of at our discretion.

15.4.3 the item was not originally posted as a franked item and your name and address are (a) not on the cover or (b) they are illegible, then the item may be opened. If your name and address are legible inside, the address is in the United Kingdom, the Channel Islands or Isle of Man, and the item does not consist solely of advertising material, newspapers or magazines, then the item may be returned to you. Otherwise the item may be disposed of at our discretion.

15.5 We will not charge any additional postage or fee for returning an undeliverable item but you or your representative may have to pay any other fees (e.g. surcharges or customs charges) that apply. If any applicable fee is not paid, the item may be dealt with or disposed of at our discretion.

15.6 Return to Sender - We will return an item to you from the delivery address, if the following conditions are all met – i.e. the item:

- was originally posted using a service set out in this Scheme,
- $\bullet$  is re-posted by the addressee or any other person for any reason,
- with the original address crossed out,
- with a clear instruction to return the item to you written on the cover,
- with the return address written fully, correctly and legibly on the cover and

• the return address is in the United Kingdom, the Channel Islands or the Isle of Man.

15.7 Forwarding – unofficial redirections – We are not under any obligation to do anything with an item that someone requests to be forwarded to another address other than the return to sender address on the cover. We consider these unpaid for, forwarding requests to be a form of unofficial redirections request. Typically in this situation the original address is crossed out and a written request (e.g. "please forward to") is made on the cover to forward the item to another address. This is normally attempted in order to try to getthe item to

the addressee who has moved to another address. In these situations the person moving should take out a Redirection service with us orprovide pre-paid stationery. If we find an item that someone has requested be forwarded to another address then the item may be dealt with or disposed at our discretion.

16. Complaints handling process

16.1. We aim to provide you with a reliable and consistent service at all times. If, however, you are unhappy with the service we have given you, this section will help you to get in touch with us to discuss your complaint and – if applicable – refer your complaint to an external, independent Ombudsman style service for resolution.

16.2. There are 3 internal stages in our complaints handling process and then the final Ombudsman style stage. Further information and contact details for each stage are available on our website.

#### Stage 1: Contacting our Customer Services Advisors

16.2.1. Our Customer Services Advisors will make every effort to investigate and resolve your complaint providing you with an appropriate response. If you are not happy with the reply you can escalate your case to the Escalated Customer Resolution Team.

#### Stage 2: Contacting our Escalated Customer Resolution Team

16.2.2. Our Escalated Customer Resolution Team is trained in dealing with escalated complaints. They will endeavour to resolve your concerns. However, if after allowing them another chance to deal with your complaint you remain dissatisfied, you can contact the Postal Review Panel.

#### Stage 3: The Postal Review Panel

16.2.3. The Postal Review Panel will take a fresh and impartial look at your case and provide a final response from us about your complaint. 16.2.4. If you are not satisfied after the Postal Review Panel has provided you with a final response then your case will be considered 'deadlocked'. Customer complaints in deadlock can be referred to the external, independent Ombudsman style service.

#### Stage 4: The Postal Redress Service – an external, independent Ombudsman style service

16.2.5. If you remain dissatisfied with the response you have received after contacting the Postal Review Panel and you have received a deadlock letter, you may be able to escalate your complaint to the independent Ombudsman style scheme, the Postal Redress Service (POSTRS).

16.2.6. POSTRS's role is to resolve disputes between regulated postal operators (such as us) and their customers. POSTRS provide a free service and we are bound by their findings.

16.2.7. Depending on the nature of the specific complaint, POSTRS may be available. A list of products within the POSTRS scheme can be found on our website. Generally, the service is available to you if:

- We have not been able to settle your complaint within the 90 day timescales set out by Royal Mail Customer Services.
- You have been referred to the scheme through your 'deadlock' letter.
- We have not followed our own complaints procedure in handling your complaint.

16.2.8. For further information please refer to POSTRS's website www.cedr.com/postrs.

17. Compensation - what we are liable for

17.1 The vast majority of items arrive safely on time. But, in a tiny fraction of cases items may be lost, damaged or delayed. In these situations we may refund postage and/or service fee and award compensation. Where we do not do what we were paid to do (e.g. gain a signature) a refund of the fee paid for that service may be payable by us. This section provides details as to:

- what items qualify for refunds and compensation,
- what compensation is available and
- how and when claims need to be made.

17.2 We will accept liability, may refund postage and/or service fees and may pay compensation for the **loss** of, **damage** to, **part loss** of or delay of an item posted under this Scheme, unless that item:

17.2.1 is considered by us to be undeliverable (as set out in section 15.3 above),

17.2.2 is one that has been successfully delivered but subsequently made the subject of a Return to Sender request (as set out in section 15.6 above),

17.2.3 is considered by us to be subject to a forwarding or unofficial redirection request (as set out in section 15.7 above).

17.2.4 has not fully meet all of the conditions set out in this Scheme (as mentioned in sections 5.3, 5.7 and 13.2),

17.2.5 is considered by us to indicate that insufficient (or no) postage and/or service fee has been paid for the service requested (also as mentioned in section 13.2 above),

17.2.6 is one which we have dealt with or disposed of at our discretion as mentioned under any section of this Scheme,

17.2.7 is one that contained prohibited items (as set out in section 5.3 above).

17.2.8 is one that contains restricted items and the requirements for the acceptance of those items have not been met (as set out in section 5.7 above).

17.2.9 is one that is being sent by post in contravention of section 85 of the Postal Services Act 2000 (as amended), which essentially means any item that may damage another item or injure anyone handling that item. Also an item that contains or shows on its cover an indecent or obscene image.

17.2.10 is subject to circumstances where the event leading to the claim was caused by situations outside our control including exceptionally severe weather conditions, acts of terrorism and vandalism and acts of third parties with whom we have no contractual relationship,

17.2.11 is one where the loss, damage or delay is due to a latent or inherent defect or natural deterioration,

17.2.12 is one that we do not believe has suffered damage or delay solely as a result of its transmission through the post. In particular where the cover does not appear to us to have suffered damage consistent with the damage caused to the contents of that item,

17.2.13 is one where the damage was pre-existing, that is, where the item was already damaged when it was posted, or

17.2.14 is one which has been re-directed to an address in the Channel Islands or the Isle of Man.

17.3 For clarity, where an item falls into one of the categories listed in sections 17.2.1 to 17.2.14 then we will not accept liability, will not refund postage and/or service fees or pay compensation for the loss of, damage to or delay of that item.

17.4 In addition to section 17.3 we will only accept liability, may refund postage and/or service fees and may pay compensation where the loss, damage or delay is due to any wrongful act done, or any neglect or default committed by a member of staff or agent of Royal Mail while performing or purporting to perform his functions as such in relation to the receipt, conveyance, delivery or other dealing with the item.

17.5 In addition to sections 17.3 and 17.4 we do not accept liability, will not refund postage and/or service fees or pay compensation for loss, damage or delay to an item where the loss, damage or delay is caused by your own act or omission in the course of either preparing to post or actually posting the item.

17.6 For the avoidance of doubt we will not accept liability, will not refund postage and/or service fees or pay compensation for loss, damage or delay to an item where the item was posted with, or handled by another postal operator or third party (other than a neighbour who has taken in an item under the delivery to neighbour arrangements mentioned in section 15.2 in which case the same compensation arrangements apply as they do for any item that was successfully delivered first time). What determines compensation payable

17.7 The amount of compensation that is payable depends on a number of factors including:

- The service used,
- The nature of compensation being sought (that is, is it for loss, damage or delay?),
- The evidence that can be provided and
- The value of the item

17.7.1 Compensation is only available for items sent using the following services:

- First Class,
- Second Class,
- Royal Mail Signed For 1st Class,
- Royal Mail Signed For 2nd Class,
- Special Delivery, and
- Articles for the Blind

17.7.2 In addition to the information contained in this Scheme we publish a policy for each type of compensation which can be found through the following links. These policies may be updated from time to time and the latest versions will always be published on our website: Royal Mail's retail compensation policies

Claims and evidence

17.7.3 We will only consider a claim for loss, damage or delay compensation where the minimum **basic evidence** is available and provided. Where only basic evidence is provided and the item is one which we accept liability for or has no **intrinsic value** we will refund the postage paid or issue six 1st class stamps<sup>9</sup> whichever is the higher in value for loss or damage (orpart loss).

17.7.4 To claim loss or damage compensation for the intrinsic value of the item **additional evidence** is required as well as the basic evidence. Details regarding what additional evidence is made up of are set out in 17.7.5 below.

Basic evidence is made up of all of the following:

17.7.4.1 the names and addresses of you, the addressee and, if different, the claimant,

17.7.4.2 the name of the service used,

17.7.4.3 the value of the postage paid and method of postage, e.g. stamps, franking impression, Smartstamp®. If the value of the postage paid is greater than the value of six 1st class, letter format, stamps (at the first weight step) then evidence of postage paid should be provided which could be product documentation, a certificate of posting or a receipt.

17.7.4.4 the place of posting,

17.7.4.5 the date of posting,

17.7.4.6 the basis for asserting the posting details and product used (such as date of postmark or certificate of posting (if available). For Special Delivery, Royal Mail Signed For 1st Class, Royal Mail Signed For 2nd Class and parcels sent by First Class or Second Class with delivery confirmation, the product documentation provided to you when posting the item (i.e. certificate of posting or receipt)

17.7.4.7 a detailed description of the contents,

17.7.4.8 in addition, for damage (or part loss) and delay claims the date of delivery, and

17.7.4.9 in addition, for damage (or part loss) claims a description of the packaging and condition of the mail item itself.

17.7.5 As mentioned in section 17.7.4 above in order to claim loss or damage compensation for the **actual loss** of the item additional evidence is required.

Please note that, when claiming for actual loss, there is a cap on the amount of compensation that can be paid. For Special Delivery it is the lower of market value or  $\mathfrak{L}500$  (or  $\mathfrak{L}1,000$  or  $\mathfrak{L}2,500$  if enhanced compensation has been purchased).

For Royal Mail Signed For 1st Class and Royal Mail Signed for 2nd Class it is the lower of the market value or  $\mathfrak{L}50$ . For items sent by the Articles for the Blind service it is the lower of the market value or  $\mathfrak{L}46$  For all other services listed in section 17.7.1 it is the lower of market value or  $\mathfrak{L}20$ .

Details are summarised in Tables 1 & 2 below section 17.7.7 of this Scheme.

17.7.5.1 Additional evidence is made up of evidence of posting and evidence of value.

17.7.5.1.1 Evidence of posting can be:

- Any original Certificate of Posting (provided automatically for Royal Mail Signed for 1st Class, Royal Mail Signed For 2nd Class, parcels sent by First Class or Second Class with delivery confirmation and Special Delivery and available on request and free of charge at Post Office® branches for other items),
- Original Smartstamp® or on-line postage certificate of posting validated at a Post Office® branch,
- Or for damage (or part loss) claims the item with cover and packaging, showing the postage paid as well as the original Certificate of Posting if provided as part of the service purchased (i.e. Royal Mail Signed For 1st Class, Royal Mail Signed For 2nd Class, parcels sent by First Class or Second Class with delivery confirmation and Special Delivery).

17.7.5.1.2 Evidence of value can be:

- · Original receipts,
- Bank or credit card statements,
- PayPal records,
- Invoices,
- Manufacturing costs,
- Auctioneers valuations,
- Repair quotations etc.

This list is not exhaustive. It is provided for illustrative purposes. All evidence of value must be originals and not copies. You are advised to make copies before sending us originals.

17.7.5.1.3 In addition to information set out in 17.7.5.1.1 - 2 for eBay claims you must also include the following:

• item sale page and PayPal or bank/credit card statement and

• eBay item number.

17.7.6 Please note that all of the packaging and contents of damaged items or items subject to part loss must be retained as we may need to inspect them. If they are not retained thencompensation may not be paid.

17.7.7 We may at any time request additional documentation and/or information from the claimant, sender or recipient to protect against unwarranted or duplicate claims. This includes, but is not limited to, a declaration of non-receipt (or proof of non receipt in appropriate cases).

Table 1 - Refunds and compensation available for loss, damage & part loss for item s conveyed using First Class & Second Class

	Loss	Damage and Part Loss
Item has no intrinsic value	Postage refund (a minimum 6 x First Class letter format stamps at their basic weight step.)	6 x First Class letter format stamps at their basic weight step.
Item has intrinsic value (with basic evidence only)	Postage refund (a minimum 6 x First Class letter format stamps at their basic weight step.)	6 x First Class letter format stamps at their basic weight step
Item has intrinsic value (with additional evidence)	Postage refund, plus compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and \$20.	Postage refund, plus compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £20.

Table 2 - Refunds and compensation available for loss, damage & part loss for items conveyed using Royal Mail Signed for 1st Class and Royal Mail Signed for 2nd Class

elace and riegal man elgited for zind elace			
	Loss	Damage and Part Loss	
Item has no intrinsic value	Postage refund (a minimum 6 x First Class letter format stamps at their basic weight step.)	6 x First Class letter format stamps at their basic weight step.	
Item has intrinsic value (with basic evidence only)	Postage refund (a minimum 6 x First Class letter format stamps at their basic weight step.)	6 x First Class letter format stamps at their basic weight step	
Item has intrinsic value (with additional evidence)	Postage refund, plus compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £50.	Postage refund, plus compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £50.	
Table 3 - Defunds and	componention availal	ole for loss, damage 8	

Table 3 - Refunds and compensation available for loss, damage & part loss for items conveyed using Articles for the Blind service

part loss for items conveyed using Articles for the Blind service		
	Loss	Damage and Part Loss
Item has no intrinsic value	Postage refund (a minimum 6 x First Class letter format stamps at their basic weight step.)	6 x First Class letter format stamps at their basic weight step.
Item has intrinsic value (with basic evidence only)	Postage refund (a minimum 6 x First Class letter format stamps at their basic weight step.)	6 x First Class letter format stamps at their basic weight step

Item has intrinsic value (with additional evidence)

compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £46.

Postage refund, plus Postage refund, plus compensation on basis of the customer's actual loss. This compensation is subject to the maximum payable being the lower of the market value of the item and £46.

Table 4 - Refunds and compensation available for delay for First Class, Second Class, Royal Mail Signed for 1st Class and Royal Mail Signed for 2nd Class, Articles for the Blind and Special Delivery

#### Delay

First Class, Second Class, Royal Mail Signed For 1st Class, Royal Mail Signed For 2nd Class & Articles for the Blind Special Delivery

#### Compensation payable

6 x First Class letter format stamps at their basic weight

A refund of your Special Delivery postage if your item arrives later than the due time of delivery. Compensation of £5 if the item is delivered more than 24 hours (Mon-Fri) after the guaranteed time. Or £10 if delivered 7 or more working days (Mon-Fri) after the guaranteed time. Additional evidence of posting is required

#### Making a claim

17.8 This section sets out some details for making a compensation claim. The full process for making a claim is set out on our website. We may reject claims that do not follow that process.

17.8.1 Claims should be made by you or the intended recipient of the item. However we will only make a payment once in respect of any

17.8.1.1 If you and the intended recipient both make a claimfor the same item, then only you will have a right to compensation, unless a compensation payment has already been made to the recipient, in which case you will have no rightto compensation.

17.8.2 Any loss, damage (or part loss) and delay claims must be made as soon as possible after the incident to help us investigate. Claims must be made by the deadlines set out below.

17.8.2.1 Please note that loss claims can only be made if the item has not been delivered (or had delivery attempted) 10 or more working days after the due date (5 working days for Special Delivery items).

17.8.2.2 All loss and damage (or part loss) claims must be made within 80 calendar days of the date of posting for all services covered by this Scheme.

17.8.2.3 Claims for delay relating to items posted using any of the First Class, Second Class, Royal Mail Signed For 1st Class, Royal Mail Signed For 2nd Class or Articles for the Blind services will only be considered valid by us if the item has not been delivered or no attempt has been made to deliver the item within 3 working days of the due date. An extra working day is added if items are posted during the Christmas and New Year period. If an item has been redirected using our Redirection  $^{\text{TM}}$  service then the period is 6 working days instead of 3; during the Christmas & New Year period this is extended to 8 days.

Table 5 - Due date and qualifying days to be used for determining when an item may be considered delayed outside of the Christmas and New Year period

Service	Due date	Delay if delivered
First Class, Royal Mail Signed For 1st Class & Articles for the Blind	Next working day after posting	3 or more working days after due date. 6 or more working days after due date if
Second Class, Royal Mail Signed For 2nd	3 working days after posting	redirected item.

17.8.2.4 Claims for delay relating to items sent using the Special Delivery service will only be considered valid by us if the item has not been delivered or no attempt has been made to deliver the item within 24 hours of the due date. This applies all year round. If an item sent using the Special Delivery service arrives 7 or more working days after the due date then we will make a payment for substantial delay in addition to a postage refund. Special Delivery items which have been redirected by our Redirection™ service are not eligible for delay compensation.

17.8.2.5 All delay claims must be made within 3 months of posting by you and within one month of delivery (or attempted delivery) by the addressee. We will not accept any liability or responsibility for claims made after this time periods. Please note that any claims for refunds relating to Special Delivery on the basis that the item sent using that service did not arrive by the specified time must be made within 14 days of posting.

17.8.2.6 In respect of claims for the loss and delay of items conveyed to addresses which have been exempted from a daily delivery obligation by our regulator, Ofcom, the assessment of lateness in paragraph 17.8.2.1 and the assessment of when an item is considered delayed in paragraph 17.8.2.3 and 17.8.2.4 shall apply on a case by case basis.

17.8.3 All claims should be made using the correct claim form which is available on our website and can be collected from Post Office® branches. By using the correct claim form we can process your claim as quickly and effectively as possible. Alternatively, claims may be submitted online via our website (www.royalmail.com).

17.8.4 The claim form needs to be completed as fully as possible: hard- copy claims form should be signed and dated by the claimant before being sent to us and all mandatory fields within the online claim form must be completed. Remember that in order to claim loss or damage compensation for actual loss of the item some additional evidence is required and should be sent in with the claim form. Where evidence is submitted online, we reserve the right to request the originals of such documents in line with section 17.7 of this Scheme.

17.8.4.1 If additional evidence cannot be provided then only a postage refund or stamps can be considered.

17.9 No compensation or other payment will be paid by us for loss or damage claims unless the claim is made within 80 days of the date of posting and unless the claim is made in full compliance with the requirements of this Scheme.

17.10 No compensation or other payment will be paid by us for delay claims unless the claim is made within 3 months of posting by the sender or 1 month of receipt by the recipient and unless the claim is made in full compliance with the requirements of this Scheme.

18. Additional terms & conditions for some services

18.1 The terms and conditions set out in sections 5 to 17 apply to all items. These are referred to in this Scheme as the common terms. Some services that we offer have additional terms and conditions that are specific to those services. The additional terms are known as the specific terms in this Scheme and are contained in the sections below. There is a section each for:

- First Class, Second Class and delivery confirmation (section 19),
- Royal Mail Signed For 1st Class, Royal Mail Signed For 2nd Class and Proof of Delivery (section 20),
- · Special Delivery (section 21),
- Articles for the Blind (section 22).
- Petitions and Addresses to the Sovereign (section 23),
- Petitions to Parliament & Assemblies (section 24),
- Poste Restante (section 25) and
- Items sent between the United Kingdom and the Channel Islands and the Isle of Man (section 26).

18.2 When using a service listed in section 18.1, the specific terms must be complied with in addition to the common terms. Where the specific terms conflict with the common terms, the specific terms should be followed as they take priority over the common terms.

18.3 In addition to the common & specific terms contained within this Scheme, other terms and conditions may exist for some services. For example in section 10 above we mentioned that customers using a franking machine must abide by the Royal Mail Scheme For Franking Letters And Parcels (which is published on our website) and according to any guidelines that we publish on our Franking help centre website. Also, customers who make use of a credit account must abide by the PPI Licence, the PPI design notes and any guidelines that we publish on our website. Such terms and conditions may be contained in application forms, licences, or any other document issued or authorised for issue by us and all may be updated from time to time.

19. First Class, Second Class and delivery confirmation

19.1. Delivery confirmation is available <sup>10</sup> for small parcels and medium parcels sent by First Class or Second Class services to addressee's in the UK excluding the Channel Islands and the Isle of Man. It is not available for letters or large letters.

19.2. Where delivery confirmation applies, a scan is taken by Royal Mail upon delivery or attempted delivery to the addressee or to another address such as a neighbour. Data from the scans is available online as described within this section 19<sup>11</sup>. Please note that delivery confirmation is not a tracked service: it simply provides a way of gaining confirmation of delivery or attempted delivery.

19.3. A Royal Mail barcoded postage label must be applied securely to the cover of the item for which First Class or Second Class has been purchased in a manner and position specified by us. The label is expected to be available online from January 2017<sup>12</sup> and in Post Offices from April 2017<sup>13</sup>.

19.4. You may obtain a copy of the data captured upon delivery or attempted delivery of the item free of charge from our website (www.royalmail.com) up to 12 months after the item was posted.

20. Royal Mail Signed For 1st Class, Royal Mail Signed For 2nd Class & Proof of Delivery

20.1. Royal Mail Signed For 1st Class and Royal Mail Signed For 2nd Class are services which can be bought on their own. Royal Mail Signed For 1st Class and Royal Mail Signed For 2nd Class items will only be delivered to an addressee or their representative once a signature or similar proof of delivery has been gained. Please note that Royal Mail Signed For 1st Class and Royal Mail Signed For 2nd Class are not tracked services; they simply provide a way of gaining the service called Proof of Delivery.

20.2. Royal Mail Signed For 1st Class and Royal Mail Signed For 2nd Class should not be confused with Special Delivery which is highly secure, fully tracked and offers next day delivery, a money back guarantee and is to be used for valuables.

20.3. You must apply (or must ensure someone else applies) a fully completed Royal Mail Signed For 1st Class or Royal Mail Signed for 2nd Class label securely to the cover of the item for which Royal Mail Signed For 1st Class or Royal Mail Signed for 2nd Class has been purchased in a manner and position specified by us.

20.4. Upon delivery of an item with Royal Mail Signed For 1st Class or Royal Mail Signed for 2nd Class added the addressee or their representative must sign (or otherwise provide some form of proof of delivery as we may specify). Where such a signature or proof of delivery is refused to be provided to us then the item may be dealt with or disposed of as we consider appropriate. Please note that the representative could be someone else at the delivery address or a neighbour.

20.5. You may request that our Customer Services provide you with a copy of the signature of the recipient of the Royal Mail Signed For 1st Class or Royal Mail Signed for 2nd Class item that was obtained when it was successfully delivered. This service is called Proof of Delivery and we may charge you an administration fee the value of which can be found on our website. This service is available up to 3 months after the datethat the item was posted. If we are unable to provide a copy of the signatureor proof of delivery (and cannot show that this was due to a refusal to provideone upon delivery) then we will refund an amount equal to the difference between the Royal Mail 1st Class Service fee and the Royal Mail Signed For 1stClass Service fee or Royal Mail 2nd Class Service fee and Royal Mail Signed for2nd Class Service fee, depending on the service purchased.

20.6. You may also obtain proof that the Royal Mail Signed For 1st Class or Royal Mail Signed for 2nd Class item was successfully delivered free of charge from our website up to 12 months after the item was posted. Please note that we will only refund an amount equal to the difference between the Royal Mail 1st Class Service fee and the Royal Mail Signed For 1st Class Service fee or Royal Mail 2nd Class Service fee and Royal Mail Signed for2nd Class Service fee, depending on the service purchased, if a claim is madewithin 3 months of the item being posted.

21. Special Delivery

21.1. Special Delivery <sup>14</sup> is a service that can be bought on its own. It offers next day delivery by 1pm to most of the UK with online tracking and proof of delivery. We will attempt to deliver a Special Delivery item by 1pm the next working day after it has been posted. If we do not succeed in attempting to deliver by this time we will refund your postage (see below). Please note that some destinations, redirected items, the Inflight Delivery Option set out at point 21.8.3 below and/or addresses where no one is available to accept the item are exempt from this money back guarantee. Please also note that we may suspend the money back guarantee from time to time for exceptional reasons (e.g. extreme weather). More information including an up to date list of UK destinations that aren't covered by the 1pm guarantee can be found on our website.

21.2. Special Delivery also offers a Saturday Guarantee add on for items posted on a Friday and the ability to purchase cover for Consequential Loss. Both are discussed further in sections below. The Special Delivery item will only be delivered to the delivery address once a signature or similar proof of delivery has been gained. Special Delivery is our only service with tracking that is offered under this Scheme. Royal Mail Signed For 1st Class and Royal Mail Signed For 2nd Class do not offer tracking; they simply provide a way of gaining proof that an item has been accepted at a Post Office® branch and delivered.

21.3. You must apply (or must ensure someone else applies) a fully completed Special Delivery label securely to the cover of a Special Delivery item in a manner and position specified by us.

21.4. Upon delivery of a Special Delivery item the addressee (or their representative) must sign their name (or otherwise provide some form of proof of delivery as we may specify). Please note that the person who signs for the item may not be the addressee; it could be anyone at the delivery address. Where such a signature or proof of delivery is not provided to us then the item may be dealt with or disposed of as we consider appropriate.

21.5. You may request that our Customer Services provide you with a copy of the signature of the recipient of the Special Delivery item that was obtained when it was successfully delivered. This service is called Proof of Delivery. This service is available up to 3 months after the date that the item was posted. If we are unable to provide a copy of the signature or proof of delivery (and cannot show that this was due to a refusal to provide one upon delivery) then we will refund the postage paid.

21.6. You may also obtain proof that a Special Delivery item was successfully delivered free of charge from our website up to 12 months after the item was posted. Please note that where we fail to give you Proof of Delivery we will only refund postage if a claim is made within 3 months of the Special Delivery item being posted.

21.7. Special Delivery items posted on a Friday are due for delivery the following Monday<sup>15</sup> (or next working day following a Bank Holiday). Saturday Guarantee can be purchased for an additional service fee at a Post Office® branch if delivery on the Saturday rather than the Monday is required. The exemptions and suspensions previously mentioned in 21.1 still apply<sup>16</sup>; information can be found in on the Special Delivery website.

21.8. Prior to delivery of the Special Delivery item, the addressee may (using the Special Delivery tracking number), request one of the following delivery options:

21.8.1. the item is held at the Customer Service Point to be collected by the addressee (or their representative);

21.8.2. the item is delivered to a local Post Office® branch nominated by the addressee so the item can be collected by the addressee (or their representative); or

21.8.3. the item is delivered to the address on a later day (such date to be no later than 6 working days after the original estimated delivery date), ("Inflight Delivery Options").

21.9. Collection of Special Delivery items as specified at 21.8.1 or 21.8.2 is subject to the addressee (or their representative) presenting identification of the addressee that we deem acceptable.

21.10. Please note that the Inflight Delivery Options at 21.8 are not available in respect of Channel Islands, Isle of Man or HM Forces (also known as BFPO) addresses.

21.11. Whilst we will make reasonable efforts to carry out the Inflight Delivery Options as set out at point 21.8 above, we will not be liable to you or the addressee if we choose not to accept a request or if we are otherwise unable to perform a request. Such circumstances include, but are not limited to, operational reasons, or where insufficient prior notice has been received from the addressee to process the request in good time.

21.12. As mentioned in section 17.7.5 Special Delivery comes with compensation of up to the lower of market value or £500 for loss or damage (or part loss). The purchase of enhanced compensation for an appropriate additional servicefee increases the maximum limit up to £1,000 or £2,500. These limits do not include cover for consequential loss. Consequential loss is the loss that occurs from failure of the Special Delivery service beyond the loss or delay of or damage to an item. For instance the loss of some legal contractual documents may lead to additional financial loss. Our Consequential Loss service is an add on service that will provide additional compensation if things go wrong. It is available for an additional service fee from a Post Office® branch. It is not available for items being sent to the Channel Islands, Isle of Man or HM Forces (also known as BFPO) addresses.

21.12.1. Claims for consequential loss must be made within 14 days of the day the item was posted.

21.12.2. Claims for consequential loss can only be made by you, as the sender of the item, for the loss you have incurred which is over & above and wholly distinct from the actual value of the contents of the Special Delivery item.

21.13. If we find an item which has not met the conditions laid out in this Scheme for a Special Delivery item but which either

21.14. has 'Special Delivery' written on it, or anything which suggests that the item is intended to be sent via Special Delivery or

21.15. is found to contain valuables,

21.15.1. then we will treat it as a Special Delivery item.

21.15.2. In either case, if the item is treated as a Special Delivery item, you or the addressee will be charged the appropriate postage for that service. If the fee is not paid the item may be dealt with or disposed of at our discretion.

22. Articles for the Blind

22.1. Articles for the Blind is a free of charge, next day service for the conveyance of items that contain particular items that are of use to blind people.

22.2. In this Scheme blind people and the blind means

22.2.1. persons registered as blind under the provisions of the National Assistance Act 1948 or

22.2.2. persons whose standard of close-up vision, with spectacles, is N12 or less.

22.3. Articles for the Blind can be used to convey items to or from blind people that contain any of the following items:

22.3.1. books, papers and letters which are specifically prepared for use by blind people. Standard or large print items may only be sent using the Articles for the Blind service if, in our reasonable opinion, they fall within this definition and the font size is at least 16 points in size.

22.3.2. papers sent to anyone to be specially prepared or impressed so that blind people can use them,

22.3.3. relief maps,

22.3.4. machines, frames and attachments for making impressions for blind people to use,

22.3.5. writing frames and attachments,

22.3.6. Braille instruction manuals orroyalmail

22.3.7. any other item that we determine to be allowable as listed on our website

22.4. Articles for the Blind can be also used to convey items between blind people and specific organisations or institutions (which are preapproved by us) that cater for blind people that contain any of the following items:

22.4.1. games (including card games),

22.4.2. mathematical appliances and attachments,

22.4.3. 'talking books' and 'talking newspapers', which are recordings of readings from printed sources, such as books, journals, newspapers, periodicals or similar publications,

22.4.4. equipment used to play talking books and newspapers,

22.4.5. metal plates impressed or sent for impressing for use by blind people,

22.4.6. supplies of covers, envelopes and labels for sending articles for use by blind people,

22.4.7. watches, clocks, timers, tools and measuring equipment designed for blind people to use,

22.4.8. walking sticks adapted for blind people,

22.4.9. harnesses for guide dogs,

22.4.10. computer disks and CDs which have been prepared for blind people and do not contain information which can be read without specially prepared encryption software or

22.4.11. any other item that we determine to be allowable as listed on our website.

22.5. Any item must meet all the conditions set out below and we may open and inspect each item to ensure it meets the conditions set out below:

22.5.1. It must weigh less than 7 kilograms.

22.5.2. It must clearly show the words 'ARTICLES FOR THE BLIND' on the front of the cover along with your name and address somewhere on the cover where it can't be confused with the destination address (preferably on the back of the item). If sent through a Post Office® branch a label must also be applied which is known as a P4558.

22.5.3. It must be left un-sealed so the contents can be checked to make sure they are permissible as listed in 22.3 or 22.4. Alternatively they can be sealed in a manner that we can open and then re-seal. Alternatively we may agree alternative arrangements with you in advance should you be unable to meet these requirements. For example you may be able to design your cover such that the contents are visible.

22.5.4. It must not contain any item or personal message which is not listed in 22.3 or 22.4, with the exception of a label showing the name and address of the sender, and instruction booklets, guarantees, technical documentation and other information provided for use with an item listed in 22.3 or 22.4. Please note that we may open and inspect items that are sent using the Articlesfor the Blind to ensure the service is not being abused.

22.5.5. It must be addressed to a named recipient. This can be the name of an individual or an organisation that provides services specifically for blind people. Where addressed to an individual thatindividual must be a blind person. It is not permissible to send an item using Articles for the Blind if it is addressed to 'The Occupier' or Dear Customer'.

22.5.6. It must not contain any advertising literature.

22.6. If we find an item that has been posted with us as an Articles for the Blind item but which does not meet the requirements set out in these specific terms then we may convey it using a different service and we will charge you or the recipient the appropriate postage for that service along with a surcharge to cover our costs. If the postage and surcharge is not paid then we will deal with the item as we determine.

22.7. More information is available on our website (www.royalmail.com).

23. Petitions and Addresses to the Sovereign

23.1. Special arrangements exist for the delivery of a petition or an address to the **Sovereign**, Her Majesty the Queen.

23.1.1. For the purposes of section 23.1 an **address** to the Sovereign is a formal signed written representation or appeal in respect of any concern or grievance for which there is no other constitutional remedy readily available.

23.1.2. For the purposes of section 23.1 a **petition** is a signed document intended to be presented to the Sovereign asking for action to be taken on a particular issue.

23.2. Petitions and Addresses to the Sovereign are free of charge as long as the special conditions set out below are all met:

23.2.1. it is a signed original and not a copy,

23.2.2. it is within the size limits set out in 6.3,

23.2.3. it does not weigh more than 2 kilograms,

23.2.4. it is packed so the contents can easily be inspected,

23.2.5. it clearly has 'ADDRESS TO HM THE QUEEN' or 'PETITION

TO HM THE QUEEN' marked on the cover and 23.2.6. it does not contain any other item

23.2.7. We will not accept any item for free delivery using Petitions and Addresses to the Sovereign unless we consider it complies with these conditions or if we do not consider the contents to be a genuine address or petition.

24. Petitions to Parliaments and Assemblies

24.1. Special arrangements exist for the delivery of a petition to a member of either House of Parliament (that is, the House of Commons or the House of Lords), the Scottish Parliament, the National Assembly for Wales or the Northern Ireland Assembly.

24.1.1. For the purposes of section

24.1 a petition is a signed document intended to be presented to a current member of either House of Parliament, the Scottish Parliament, the National Assembly for Wales or the Northern Ireland Assembly asking for action to be taken on a particular issue.

24.2. We will deliver a petition to either House of Parliament, the Scottish Parliament, the National Assembly for Wales or the Northern Ireland Assembly free of charge as long as the special conditions set out below are all met:

24.2.1. it is a signed original and not a copy,

24.2.2. it is within the size limits set out in section 6.3,

24.2.3. it does not weigh more than 2 kilograms,

24.2.4. is packed so the contents can easily be inspected.

24.2.5. It clearly has 'PETITION TO THE HOUSE OF COMMONS', 'PETITION TO THE HOUSE OF LORDS', 'PETITION TO THE SCOTTISH PARLIAMENT', 'PETITION TO THE NATIONAL ASSEMBLY FOR WALES' or 'PETITION TO THE NORTHERN IRELAND ASSEMBLY' marked on the cover.

24.2.6. it is marked for the attention of a current member of parliament, the assembly or parliamentary Lord and

24.2.7. it does not contain any other item.

24.3. We will not accept any item for free delivery using Petitions to Parliament and Assemblies unless we consider it complies with these conditions or if we do not consider the contents to be a genuine Petition.

25. Poste Restante

25.1. Poste Restante is a service that allows an item to be addressed to a person at a Post Office® branch where that person may, without being charged, collect that item during normal opening hours.

25.1.1. For the purposes of section

25.1 a person is a traveller with no fixed abode in the local area or any other person or class of person that we decide and publicise on our website. It is not available to businesses including sole traders.

25.2. The words 'POSTE RESTANTE - TO BE CALLED FOR' must be clearly marked on the cover of the item along with the full and correct surname of the addressee of the item.

25.3. The Post Office® branch staff may refuse to hand over a Poste Restante item if they are not satisfied as to the identity of the person collecting the item.

25.4. Post Office® branches will hold an item posted within the United Kingdom, the Channel Islands or the Isle of Man. They will hold an item for two weeks if posted within the United Kingdom, for one month if posted overseas and for two months if addressed to a branch at a sea port if you are arriving by ship.

25.4.1. items that are not collected within the time period set out in 25.4will be treated as if they were undeliverable (see section 15)

25.5. Please note that not all Post Offices® branches are capable of providing the Poste Restante service for operational or other reasons. The Post Office® Customer Service team can help set up this service. Their details are on their website (www.postoffice.co.uk).

25.6. We or the Post Office® will not provide the Poste Restante service where there is reason to believe that you or the addressee is mis-using the Poste Restante service; for example where the addressee already has a permanentresidential or business address in the area or where the addressee's name isabbreviated or in our reasonable opinion we believe it to be false.

26. Items sent between the United Kingdom and the Channel Islands or the Isle of Man

26.1. This section of this Scheme applies to an item posted in the United Kingdom for delivery to an address in the Channel Islands or the Isle of Man (referred to in this part of this Scheme as 'outgoing items') and to an item posted in the Channel Islands or Isle of Man for delivery to an address in the United Kingdom (referred to in this part of this Scheme as 'incoming items').

26.2. All incoming items are subject to HM Customs and Excise regulations. We may charge a handling fee for ensuring that an item passes through the proper HM Customs and Excise checks. In addition HM Customs and Excise may charge an import duty (customs charge). This handling fee and duty must be paid by the recipient on delivery. If the fee and/or duty are not paid we will deal with the item at our discretion. The fee we charge is published from time to time on our website.

26.3. Section 17 ('Compensation – what we are liable for') of this Scheme does not apply to incoming or outgoing items except as expressly provided for inthis section.

26.4. We may pay compensation for loss of or damage to (or part loss of) an incoming or outgoing item as set out in section 17 if the following criteria are all met:

26.4.1. we would have paid compensation if the item had been an item posted in and for delivery in the United Kingdom,

26.4.2. we are satisfied that the item was lost or damaged whilst in our custody and

26.4.3. we are satisfied that no compensation has been or will be paid in respect of that item by the postal administration of the Channel Islands or the Isle of Man.

27. Sanctions Laws

27.1. You must make sure that any items you wish us to handle and deliver are not prohibited under applicable sanctions laws, for example because of their contents, their intended recipient or the country to or from which they are to be sent. Information about sanctions can be found on our website at www.royalmail.com/international-sanctions (or any replacement URL). Further information on trade controls and economic sanctions can be found on the UK Government's web pages at www.gov.uk.

27.2. If your items need a licence under applicable sanctions laws it is your responsibility to obtain it and (if we ask for it) you must provide us with acceptable evidence that you have it.

27.3. If we have reasonable suspicion that, because of its contents, an item does not comply with sanctions laws we may open that item or delay processing and delivery.

27.4. If you post an item which does not comply with sanctions laws we may deal with such item in our sole and absolute discretion (without incurring any liability whatsoever to you, your intended recipient and/or, to the extent not prohibited by applicable law, any third party), including destroying or otherwise disposing of such item in whole or in part, or returning the relevant item to you. We are entitled to charge you:

- the cost of disposal and/or destruction;
- the standard postage price;
- all other costs reasonably incurred by us; and/or
- for any losses or damage we suffer or liability we incur as a result of you not complying with sanctions laws.

27.5. We will bear no responsibility in the event that you or your agents send an item with the wrong licence required under sanctions laws

28. Your information

28.1 Where we supply services to you under this Scheme, we are the controller of the personal data we process in providing services to you.

28.2 Where you supply personal data to us so we can provide services to you, and we process that personal data in the course of providing services to you, both you and we will comply with our obligations imposed by the Data Protection Legislation and you will not cause us to contravene the Data Protection Legislation. Where you have provided the personal data of a third party to us, you warrant that you have lawful grounds, such as their consent, to do so and that we are entitled to process that personal data to provide services.

28.3 From time to time we may be obliged by our Regulator to provide it with certain information about you, including your name and address

28.4 Further information about how we use your personal data is set out in our Privacy Policy at www.royalmail.com/privacy-policy. Please read this Privacy Policy carefully.

28.5 The terms "personal data", "controller", "processing" and "supervisory authority" shall all have the same meaning as in the Data Protection Legislation and the term "process" shall be construed accordingly.

Annex A - Definitions used in this Scheme

A1 The Interpretation Act 1978 applies for the interpretation of this Scheme and for this purpose this Scheme is to be treated as if it were an Act of Parliament.

A2 In this Scheme, the following words and terms have the following meanings:

#### **Actual Loss**

(a) Where an item is lost or damaged beyond repair then it is the amount it cost the customer to acquire, purchase or manufacture the item subject to adjustment to take account of condition, age and depreciation.

(b) Where an item is damaged it is the cost of repair. No additional payment will be made for the reduced value of the repaired item.

Address

means for any premises the address, including the postcode, maintained by us from time to time as corresponding to those premises in the Postcode Address File

#### Addresses

The person to whom an item is addressed.

#### **Antiques**

Items that are at least one hundred years old.

#### **Bladed Items**

(Bladed Articles and Bladed Products as defined by Offensive Weapons Act 2019) including, but not limited to, any knife (including a folding pocket knife), any knife blade, cutlery knife, scissors with sharp edges or points, sporting equipment with a blade, tools with a blade, replica and antique knives including those used for reenactment purposes, handmade and bespoke knives, cut-throat razors, open razors where the blade is exposed, any axe, any sword, and any other article which has a blade - Excludes knives prohibited under UK law-see Weapons.

#### **Business**

Any entity engaged in commercial or economic activity, whether for profit or not, and irrespective of legal form.

#### Channel Islands

The islands called Guernsey, Jersey, Herm and Sark.

#### Christmas and New Year period

The period commencing on the first Monday in December in any year and ending at the start of the first working day following the New Year public holiday(s)

#### Cover

Any Cover, envelope, or other packaging used to contain the contents of an item. In the case of a postcard or other card, any reference to a Cover shall be construed as a reference to an outside face of the card.

# Damage

If something in an item of mail has suffered a degree of harm that reasonably impairs the material function or contents of the item.

#### **Data Protection Legislation**

means (1) the Data Protection Act 1998; the Data Protection Directive (95/46/EC) and the Privacy and Electronic Communications Directive (2002/58/EC); (2) after 25 May 2018, Regulation (EU) 2016/679 of the European Parliament And Of The Council of 27 April 2016 on the protection of natural persons with regard to the processing of personal data and on the free movement of such data, and repealing Directive 95/46/EC (General Data Protection Regulation), (3) any guidance, directions, determinations, codes of practice, orders, notices or demands issued by any competent supervisory authority or other competent authority, any other applicable data protection laws or regulations and judgments of any court of law, tribunal or regulatory body as amended, extended, re- enacted or replaced from time to time, and (4) the Regulation on Privacy and Electronic Communications, when in force.

#### Due date

#### Means:

(a) for First Class, Royal Mail Signed For 1st Class and Special Delivery, the next working day following the date of posting;

(b) for Second Class and Royal Mail Signed For 2nd Class services the third working day following the date of posting.

# Franking Mark

Any mark or impression printed or otherwise made by a franking machine indicating the date and place of posting, the postage and fees paid or payable, the identity of the franking machine used and any other numbers, words, characters, symbols or marks be required by us.

# Intrinsic Value

When used in relation to the content of an item means that it has an inherent monetary value relating to its essential nature.

#### Item

means any letter, postcard, reply postcard, newspaper, printed packet, sample packet or parcel and every packet or other article transmissible by post.

#### Loss

An item shall be deemed to be lost, (unless there is evidence to the contrary to demonstrate that it has not been lost) if it has not been delivered by us by the end of the tenth working day after its **due date**, or the fifth working day after its due date for Special Delivery.

### Neighbour

For the purposes of delivering items, an appropriate person chosen at the discretion of the delivery officer on the basis of local knowledge and official guidance, who lives within close proximity of the address on the item. They may be a next-door neighbour or someone who lives sufficiently close by.

#### **Market Value**

means what a willing buyer would pay a willing seller, both in a free market, for an item. When assessing the market value of an item, the value of any message or information it contains, or bears or refers to must be ignored. Market value is one of two caps upon the amount of compensation payable.

#### Part Loss

Where a letter is received and some or part of the content is missing. **Post, Posted** 

An item is posted if it has been entrusted to us for transmission by post in accordance with one of the approved methods set out in section 12 of this Scheme and related words shall be construed accordingly.

**Postage** The amount of money charged by us for delivery of an item. **Postage Mark** 

A collective term for a mark or impression (to include a Franking Mark, a Printed Postage Impression (PPI), any Postage paid symbol, preprinted stationery and SmartStamp®) authorised for use by us to indicate payment of postage and/or service fees on an item or to indicate that you have entered into an arrangement with us to pay postage and/or service fees on that item. A postage mark may be non-adhesive (when it is embossed, impressed or printed on a cover) as we may from time to time permit.

#### **Postmark**

Any indicator, image or impression applied by us to a cover or envelope, to cancel a postage stamp or postage label, to indicate that we are the delivery agent of the item or for any other postal purpose approved and authorised by us.

# Printed Postage Impression (PPI)

A Postage mark indicating postage is payable to us and printed under a Licence from us.

#### **Private Post Box**

Any post box owned and maintained by a person other than ourselves which has been authorised for use as a post box by either Post Office Limited or ourselves for the posting of items which are intended to be carried by us.

#### Postage Stamp

means a stamp, authorised for use by us to indicate payment of postage and/or fees on an item or to indicate that you have entered into an arrangement with us to pay postage and/or service fees on that item. A postage stamp may be adhesive in order for it to be affixed to a cover as we may from time to time permit.

# Postage Label, Service fee Label

means a label, authorised for use by us to indicate payment of postage and/or service fees on an item. or to indicate that you have entered into an arrangement with us to pay postage and/or service fees on that item. A postage label may be adhesive in order for it to be affixed to a cover as we may from time to time permit.

Representative means a person who acts legitimately, with or without express permission, on behalf of the addressee.

#### Service fee

The amount of money charged by us for providing a service in connection with an item.

#### **SmartStamp®**

A postage mark used by customers posting items with the SmartStamp® indicia having entered into the relevant SmartStamp® terms and conditions.

#### **Small Parcel Box**

A 15cm³ specifically designed mailing box with the "Royal Mail Delivered By" symbol on it as well as reference to "Small Parcel Box". Customers sending items with a height greater than 8cm weighing no more than 2kg must use one of these boxes to take advantage of the Small Parcel price.

# Social Security post

Social security post means items whose contents relate to any benefit, contributions or national insurance number or to any other matter relating to social security. Social security post is further defined in Section 182A of the Social Security Administration Act 1992 and Section 158A of the Social Security Administration (Northern Ireland) Act 1992.

## Sovereign

The reigning Monarch of the United Kingdom.

#### Valuables (money and jewellery)

Any item that is:

- i) precious metal that has been manufactured in such a way as to add value to it, including coins used for ornament;
- ii) ingots;
- iii) diamonds and precious stones;
- iv) watches the cases of which are made wholly or mainly of precious metal: and
- v) articles similar to any of those referred to in i) iii) above with an intrinsic value.
- v) Coins & Bank notes of any currency that are legal tender at the time of posting;
- vi) Postal Orders, cheques and dividend warrants uncrossed and payable to bearer;
- vii) Unused postage and revenue stamps and National Insurance stamps; Exchequer bills, bills of exchange, promissory notes and credit notes; Bonds, bond coupons and any other investment certificates; and
- viii) Coupons, vouchers, tickets, tokens, cards, stamps and other documents that can be exchanged in whole or in part for money, goods or services.

#### **Working Day**

For any item posted using (or otherwise treated by us as being posted using) a service other than Special Delivery, working days are Monday to Saturday inclusive, excluding any day which is (in the part of the United Kingdom concerned) a public holiday and any other day that Ofcom has directed us to treat as a public holiday.

For items that have been posted using (or otherwise treated by us as being posted using) the Special Delivery service to which Saturday Guarantee has not been added, working days are Monday to Friday inclusive excluding any day which is (in the part of the United Kingdom concerned) a public holiday and any other day that Ofcom has directed us to treat as a public holiday. For items posted using (or otherwise treated by us as being posted using) the Special Delivery service to which Saturday Guarantee has been added, working days are Monday to Saturday inclusive, excluding any day which is (in the part of the United Kingdom concerned) a public holiday and any other day that Ofcom has directed us to treat as a public holiday.

Further information (not part of the Scheme)

You can get more information on this Scheme, the services it covers and the details contained in the web pages that are linked in this document from our Sales Centre on 0345 740 740.

Lines are open from 8am to 6pm Monday to Friday and from 8am to 1pm on Saturdays.

- <sup>1</sup> We also offer some services under contracts which have their own terms and conditions. Some of those contracts incorporate this Scheme and so some of the clauses contained within this Scheme apply to services provided under contract.
- <sup>2</sup> Special Delivery may also be provided under a contract.
- <sup>3</sup> Special Delivery Next Day bought with Consequential Loss or with a Saturday Guarantee attract Value Added Tax (VAT) and so can only be purchased with a 'smart' franking machine that is, one that (amongst other features) accounts correctly for VAT. Please note that delivery confirmation is not currently available where First Class or Second Class services are purchased by use of a franking machine or stamps, or for items sent to the Channel Islands or Isle of Man. Postage labels for delivery confirmation will be available for purchase online from January 2017 and from Post Offices from April 2017. Please note that the dates for launch of delivery confirmation may change. Please check our website (www.royalmail.com) for updates.
- <sup>4</sup> The date may vary. Please check our website (www.royalmail.com) for details of the launch of delivery confirmation.
- <sup>5</sup> The date may vary. Please check our website (www.royalmail.com) for details of the launch of delivery confirmation.
- <sup>6</sup> NOTE: Not part of the Scheme. We will not attempt to deliver to a neighbour an international inbound item that requires a signature or any other item that we are contractually not permitted to deliver to a neighbour
- <sup>7</sup> Local in this context means an address covered by the Delivery Office to which the item was returned.
- <sup>8</sup> For instance the landlord boards the property up or it becomes physically impossible to put any more mail through the letterbox or a relative makes contact with us to let us know that mail is no longer required for that address.
- <sup>9</sup> Valid for a Letter format item weighing up to 100g

- <sup>10</sup> Delivery confirmation is expected to be available online from January 2017 and from Post Offices from April 2017. Please note that the dates for launch of delivery confirmation may change. Please check our website (www.royalmail.com) for updates.
- <sup>11</sup> Please note that in certain circumstances, for example where the label has become damaged or because of the shape of the packaging, it may not be possible for us to successfully scan the item and provide the data.
- 12 The date may vary. Please check our website (www.royalmail.com) for details on the launch of this service.
- 13 The date may vary. Please check our website (www.royalmail.com) for details on the launch of this service.
- <sup>14</sup> Royal Mail also offers Special Delivery 9:00am. This service is offered under contract rather than through this Scheme. Details can be found on our website
- <sup>15</sup> Please note that we may, on occasion, deliver items on a Saturday if it is operationally convenient for us to do so.
- <sup>16</sup> In addition we will not deliver items on a Saturday if (i) the addressee has requested for items not to be delivered on a Saturday under a retention arrangement with their local Delivery Office, or (ii) the addressee has selected an alternative delivery date under the Inflight Delivery Option at point 21.8.3 above. (4034847)

# **Planning**

# **TOWN PLANNING**

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of the proposal to make an Order under section 247 of the above Act to authorise the stopping up of a length of footpath adjoining the northern boundary of the former Dukes Head public house as lies off Wretton Road at Stoke Ferry, Norfolk in the Borough of King's Lynn and West Norfolk.

If made, the Order would authorise the stopping up only to enable development as permitted by the Borough Council of King's Lynn and West Norfolk, under reference 21/01415/F.

Copies of the draft Order and relevant plan will be available for inspection during normal opening hours at King's Lynn & West Norfolk Borough Council, King's Court, Chapel Street, King's Lynn Norfolk PE30 1EX in the 28 days commencing on 1 April 2022, and may be obtained, free of charge, from the addresses stated below quoting NATTRAN/E/S247/5017.

Any person may object to the making of the proposed order by stating their reasons in writing to the Secretary of State at nationalcasework@dft.gov.uk or National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle upon Tyne NE4 7AR, quoting the above reference. Objections should be received by midnight on 29 April 2022. Any person submitting any correspondence is advised that your personal data and correspondence will be passed to the applicant/agent to be considered. If you do not wish your personal data to be forwarded, please state your reasons when submitting your correspondence.

S Zamenzadeh, Casework Manager (40377

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of an Order made under Section 247 of the above Act entitled "The Stopping up of Highways (East) (No.9) Order 2022" authorising the stopping up of two irregular shaped western part widths of Rectory Lane at Worlingham in the District of Waveney to enable development as permitted by Waveney District Council under reference DC/15/0213/FUL.

Copies of the Order may be obtained, free of charge, from the Secretary of State, National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle Business Park, Newcastle upon Tyne NE4 7AR or nationalcasework@dft.gov.uk (quoting NATTRAN/E/S247/4953. They may also be inspected during normal opening hours at Worlingham Post Office, 27 Hillside Ave, Worlingham, Beccles, NR34 7AJ.

Any person aggrieved by or desiring to question the validity of or any provision within the Order, on the grounds that it is not within the powers of the above Act or that any requirement or regulation made has not been complied with, may, within 6 weeks of 1 April 2022 apply to the High Court for the suspension or quashing of the Order or of any provision included.

G Patrick, Casework Manager

(4037745)

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of an Order made under Section 247 of the above Act entitled "The Stopping up of Highway (East) (No.10) Order 2022" authorising the stopping up of a western part width of Southend Road consisting highway verge, at Hunstanton in the Borough of King's Lynn and West Norfolk to enable development as permitted by King's Lynn and West Norfolk Council, under reference 21/00243/FM.

Copies of the Order may be obtained, free of charge, from the Secretary of State, National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle Business Park, Newcastle upon Tyne NE4 7AR or nationalcasework@dft.gov.uk (quoting NATTRAN/E/S247/4888. They may also be inspected during normal opening hours at Hunstanton Town Hall, The Green, Hunstanton, Norfolk PE36 6BQ. Any person aggrieved by or desiring to question the validity of or any provision within the Order, on the grounds that it is not within the powers of the above Act or that any requirement or regulation made has not been complied with, may, within 6 weeks of 01 April 2022 apply to the High Court for the suspension or quashing of the Order or of any provision included.

G Patrick, Casework Manager

(4037746)

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of an Order made under Section 247 of the above Act entitled "The Stopping up of Highway (Yorkshire & Humber) (No.6) Order 2022" authorising the stopping up of a western part width of Lumb Lane consisting footway at Manningham in the City of Bradford, to enable development as permitted by City of Bradford Metropolitan District Council, under reference 20/04441/FIII

Copies of the Order may be obtained, free of charge, from the Secretary of State, National Transport Casework Team, at nationalcasework@dft.gov.uk (quoting NATTRAN/Y&H/S247/4578).

Any person aggrieved by or desiring to question the validity of or any provision within the Order, on the grounds that it is not within the powers of the above Act or that any requirement or regulation made has not been complied with, may, within 6 weeks of 01 April 2022 apply to the High Court for the suspension or quashing of the Order or of any provision included.

D Hoggins, Casework Manager

(4037747)

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of an Order made under Section 247 of the above Act entitled "The Stopping up of Highway (South East) (No.10) Order 2022" authorising the stopping up of a southern part width of Sovereign Harbour North, in Eastbourne to enable development as permitted by Eastbourne Borough Council, under reference PC/210952.

Copies of the Order may be obtained, free of charge, from the Secretary of State, National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle Business Park, Newcastle upon Tyne NE4 7AR or nationalcasework@dft.gov.uk (quoting NATTRAN/SE/S247/4962. They may also be inspected during normal opening hours at Eastbourne Library, Grove Road, Eastbourne BN21 4TI

Any person aggrieved by or desiring to question the validity of or any provision within the Order, on the grounds that it is not within the powers of the above Act or that any requirement or regulation made has not been complied with, may, within 6 weeks of 01 April 2022 apply to the High Court for the suspension or quashing of the Order or of any provision included.

G Patrick, Casework Manager

(4037748)

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of the proposal to make an Order under section 247 of the above Act to authorise the stopping up of an irregular shaped area of highway as lies off Heighington Lane comprising its lay-by at Aycliffe Business Park, Newton Aycliffe in the County of Durham.

If made, the Order would authorise the stopping up only to enable development as permitted by Durham County Council, under reference DM/17/01621/FPA.

Copies of the draft Order and relevant plan will be available for inspection during normal opening hours at Great Aycliffe Town Council, Council Offices, School Aycliffe Lane, School Aycliffe, Newton Aycliffe, DL5 6QF in the 28 days commencing on 1 April 2022, and may be obtained, free of charge, from the addresses stated below quoting NATTRAN/NE/S247/4795.

Any person may object to the making of the proposed order by stating their reasons in writing to the Secretary of State at nationalcasework@dft.gov.uk or National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle upon Tyne NE4 7AR, quoting the above reference. Objections should be received by midnight on 29 April 2022. Any person submitting any correspondence is advised that your personal data and correspondence will be passed to the applicant/agent to be considered. If you do not wish your personal data to be forwarded, please state your reasons when submitting your correspondence.

S Zamenzadeh, Casework Manager

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of an Order made under Section 247 of the above Act entitled "The Stopping up of Highways (North East) (No.5) Order 2022" authorising the stopping up of the whole of the network of highways bounded by the rear of Newbottle Street Quarry service station, Sunrise Lane, the field to the north and the east of Newburn Crescent to enable development as permitted by Sunderland City Council, under reference 20/01591/FU4.

Copies of the Order may be obtained, free of charge, from the Secretary of State, National Transport Casework Team, at nationalcasework@dft.gov.uk (quoting NATTRAN/NE/S247/4400).

Any person aggrieved by or desiring to question the validity of or any provision within the Order, on the grounds that it is not within the powers of the above Act or that any requirement or regulation made has not been complied with, may, within 6 weeks of 1 April 2022 apply to the High Court for the suspension or quashing of the Order or of any provision included.

S Zamenzadeh, Casework Manager

(4037750)

(4037749)

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of an Order made under Section 247 of the above Act entitled "The Stopping up of Highway (South East) (No.11) Order 2022" authorising the stopping up of a southern part width of Sovereign Harbour North, abutting the rear boundary of Nos. 19 and 20 Silver Strand West, at Eastbourne, in the Borough of Eastbourne, to enable development as permitted by Eastbourne Borough Council, under reference PC/210954.

Copies of the Order may be obtained, free of charge, from the Secretary of State, National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle Business Park, Newcastle upon Tyne NE4 7AR or nationalcasework@dft.gov.uk (quoting NATTRAN/SE/S247/4963. They may also be inspected during normal opening hours at Eastbourne Library, Grove Road, Eastbourne, East Sussex BN21 4TL.

Any person aggrieved by or desiring to question the validity of or any provision within the Order, on the grounds that it is not within the powers of the above Act or that any requirement or regulation made has not been complied with, may, within 6 weeks of 01 April 2022 apply to the High Court for the suspension or quashing of the Order or of any provision included.

G Patrick, Casework Manager

(4037751)

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of an Order made under Section 247 of the above Act entitled "The Stopping up of Highway (North East) (No.6) Order 2022" authorising the stopping up of an irregular shaped area of highway known as Times Square at the site of the International Centre for Life at Newcastle upon Tyne in the City of Newcastle upon Tyne to enable development as permitted by Newcastle City Council under reference 2019/0688/01/DET.

Copies of the Order may be obtained, free of charge, from the Secretary of State, National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle Business Park, Newcastle upon Tyne NE4 7AR or nationalcasework@dft.gov.uk (quoting NATTRAN/NE/S247/3841. They may also be inspected during normal opening hours at Newcastle City Council, Civic Centre, Newcastle upon Tyne NE1 8PD.

Any person aggrieved by or desiring to question the validity of or any provision within the Order, on the grounds that it is not within the powers of the above Act or that any requirement or regulation made has not been complied with, may, within 6 weeks of 01 April 2022 apply to the High Court for the suspension or quashing of the Order or of any provision included.

D Hoggins, Casework Manager

(4037752)

# DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of an Order made under Section 247 of the above Act entitled "The Stopping up of Highway (South East) (No.06) Order 2022" authorising the stopping up of a part-width of London Street, adjoining number 81 London Street, at Chertsey in the of Borough of Runnymede to enable development as permitted by Runnymede Borough Council, under reference RU.21/2061.

Copies of the Order may be obtained, free of charge, from the Secretary of State, National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle Business Park, Newcastle upon Tyne NE4 7AR or nationalcasework@dft.gov.uk (quoting NATTRAN/SE/S247/4336).

Any person aggrieved by or desiring to question the validity of or any provision within the Order, on the grounds that it is not within the powers of the above Act or that any requirement or regulation made has not been complied with, may, within 6 weeks of 01 April 2022 apply to the High Court for the suspension or quashing of the Order or of any provision included.

C Moody, Casework Manager

(4037753)

### ZOO LICENSING ACT 1981 NOTICE OF APPLICATION FOR A ZOO LICENCE

Mark Brattle on behalf of Leeds Castle hereby gives notice that not less than 2 months after publication of this notice, I intend to make an application to Maidstone Borough Council for a licence to operate a proposed Zoo (Bird of Prey Centre) at: Leeds Castle, Maidstone, Kent, ME17 1PL

A written notice of the intention to make an application for a Zoo Licence, giving details about the proposed Zoo, has been sent to Maidstone Borough Council.

This may be inspected, free of charge, at the offices of Maidstone Borough Council, Maidstone House, King Street, Maidstone, Kent, ME15 6JQ. Opening hours 8.30am – 5pm

Please note that the application is for operational and lawful reasons, Leeds Castle will not be changing or adding to the current Bird of Prey Centre. (4037768)

# **Property & land**

# **PROPERTY DISCLAIMERS**

T S Ref: BV21920405/77/JZM

NOTICE OF DISCLAIMER UNDER SECTION 1013 OF THE COMPANIES ACT 2006

DISCLAIMER OF WHOLE OF THE PROPERTY

1. In this Notice the following shall apply:

Company Name: PARK PREWETT ESTATE MANAGEMENT COMPANY LIMITED

Company Number: 06035922 Title Number: HP723366

Property: All rights, benefit, provisions and interest whatsoever in the Restriction dated 19 January 2010 referred to at point B3 of the Proprietorship Register of title number HP723366.

Treasury Solicitor: The Solicitor for the Affairs of Her Majesty's Treasury of PO Box 2119, Croydon (DX 325801 Croydon 51).

2. In pursuance of the powers granted by section 1013 of the Companies Act 2006 the Treasury Solicitor as nominee for the Crown (in whom the property and rights of the company vested when the Company was dissolved) hereby disclaims the Crown's title (if any) in the Property the vesting of the Property having come to her notice on 17 March 2022.

Dated 28th March 2022 Assistant Treasury Solicitor (Section 3 Treasury Solicitor Act 1876)

(4035174)

# NOTICE OF DISCLAIMER UNDER SECTION 1013 OF THE COMPANIES ACT 2006

#### **DISCLAIMER OF WHOLE OF THE PROPERTY**

T S ref: BV22201961/1/BA

1 In this notice the following shall apply:

Company Name: NORTHUMBERLAND COURT (STREATHAM)

Company Number: 00546650

Interest: freehold Title number: NGL455487

Property: The Property situated at 19 Rectory Road being the land

comprised in the above mentioned title

Treasury Solicitor: The Solicitor for the Affairs of Her Majesty's Treasury of PO Box 70165, London WC1A 9HG (DX 123240 Kingsway).

In pursuance of the powers granted by Section 1013 of the Companies Act 2006, the Treasury Solicitor as nominee for the Crown (in whom the property and rights of the Company vested when the Company was dissolved) hereby disclaims the Crown's title (if any) in the property, the vesting of the property having come to his notice on 2 February 2022.

Assistant Treasury Solicitor

29 March 2022

(4032246)

# NOTICE OF DISCLAIMER UNDER SECTION 1013 OF THE COMPANIES ACT 2006

# DISCLAIMER OF WHOLE OF THE PROPERTY

T S ref: BV22204026/1/JZM

1 In this notice the following shall apply:

Company Name: MAGNA LIGHTING LIMITED

Company Number: 10079979 Interest: leasehold

Title number: BGL109598

Property: The Property situated at Ground Floor and basement being 118 Fulham Road, London (SW3 6HU) being the land comprised in the above mentioned title

Treasury Solicitor: The Solicitor for the Affairs of Her Majesty's Treasury of PO Box 70165, London WC1A 9HG (DX 123240 Kingsway).

In pursuance of the powers granted by Section 1013 of the Companies Act 2006, the Treasury Solicitor as nominee for the Crown (in whom the property and rights of the Company vested when the Company was dissolved) hereby disclaims the Crown's title (if any) in the property, the vesting of the property having come to his notice on 8 March 2022.

Assistant Treasury Solicitor

29 March 2022

(4032245)

NOTICE OF DISCLAIMER UNDER SECTION 1013 OF THE COMPANIES ACT 2006

DISCLAIMER OF WHOLE OF THE PROPERTY

T S ref: BV22204036/1/MO

1 In this notice the following shall apply:

Company Name: DARLINGTON BREWERS COLLECTIVE LTD

Company Number: 11857119

Interest: leasehold

Lease: Lease dated 25 April 2019 and made between Stenprop Industrials 6 Limited(1) and Darlington Brewers Collective Ltd(2)

Property: The Property situated at Unit 1, Cleveland Trading Estate, Darlington being the land comprised in and demised by the above mentioned Lease

Treasury Solicitor: The Solicitor for the Affairs of Her Majesty's Treasury of PO Box 70165, London WC1A 9HG (DX 123240 Kingsway).

In pursuance of the powers granted by Section 1013 of the Companies Act 2006, the Treasury Solicitor as nominee for the Crown (in whom the property and rights of the Company vested when the Company was dissolved) hereby disclaims the Crown's title (if any) in the property, the vesting of the property having come to his notice on 11 March 2022.

Assistant Treasury Solicitor

29 March 2022

(4032247)

# **Roads & highways**

# **ROAD RESTRICTIONS**

LONDON BOROUGH OF ISLINGTON
LOW TRAFFIC NEIGHBOURHOODS
THE ISLINGTON (PRESCRIBED ROUTES) (NO. 13)
EXPERIMENTAL TRAFFIC ORDER 2022
THE ISLINGTON (WAITING AND LOADING RESTRICTION)
(AMENDMENT NO. 14) EXPERIMENTAL TRAFFIC ORDER 2022
THE ISLINGTON (PARKING PLACES) (ZONE A) (NO. 55)
EXPERIMENTAL TRAFFIC ORDER 2022

THE ISLINGTON (FREE PARKING PLACES) (DISABLED PERSONS) (DEDICATED PERMIT PARKING PLACES) (NO. 4) EXPERIMENTAL TRAFFIC ORDER 2022

- 1. NOTICE IS HEREBY GIVEN that the Council of the London Borough of Islington ('the Council') on 30th March 2022 made the above-mentioned Orders under sections 9, 10 and 124 of and Part IV of Schedule 9 to of the Road Traffic Regulation Act 1984, as amended.
- 2. The general effect of the Orders will be to introduce a new experimental traffic management scheme. The effect of the new experimental scheme will be to:

[Note: the measurements in this Notice are given to the nearest 0.1 metres!

- (a) prohibit the passage of all motor vehicles except 'AMW' permit holders past a point at the following locations:
- (i) Lloyd Square, the north-western north-east to south-west arm, at a point in line with the common boundary of Nos. 17 and 18 Lloyd Square.:
- (ii) Lloyd Square, the south-eastern north-east to south-west arm, at a point in line with the common boundary of Nos. 6 and 7 Lloyd Square. (b) in Margery Street, prohibit the passage of all motor vehicles except 'AMW' permit holders past a point in line with the south-western kerbline of Wilmington Street;

[NOTE: an additional exemption will apply here for vehicles accessing premises situated on Margery Street between Farringdon Road and Wilmington Street!

- (c) in Great Percy Street, prohibit the passage of all motor vehicles, except 'AMW' permit holders and local buses, past a point in line with the common boundary of Nos. 33 and 35 Great Percy Street;
- 'AMW' permits allowing vehicles to pass the points listed in paragraphs (a) (b) and (c) above will be available to disabled person's badge holders who are residents of premises listed in the Schedule to this Notice, by application to the Council.
- (d) require all vehicles, except pedal cycles and authorised vehicles ('AMW' permit holders), proceeding in Margery Street in a generally south-westward direction, to turn left into Wilmington Street;
- (e) introduce 'at any time' waiting restrictions in the following locations:
- (i) Great Percy Street, on the north-west side:
- (A) between a point 16.0 metres south-west of the common boundary of Nos. 35 and 37 Great Percy Street and a point 3.0 metres northeast of that common boundary:

the south-east side

(B) between a point 4.4 metres north-east of the north-eastern kerb-line of Cumberland Gardens and a point 27.0 metres north-east of that kerb-line:

- (ii) Lloyd Square, the north-west arm:
- (A) on the south-east side, between a point 20.9 metres north-east of the north-eastern kerb-line of Lloyd Square (south-west arm) and a point 40.8 metres north-east of that kerb-line;
- (B) on the south-east side, between a point 1 metre south-west of the south-western kerb-line of Lloyd Square (north-east arm) and that kerb-line:
- (C) on the north-west side, between a point 1.6 metres north-east of the common boundary of Nos. 19 and 20 Lloyd Square and a point 1.5 metres north-east of the common boundary of Nos. 16 and 17 Lloyd Square.
- (iii) Lloyd Square, the south-east arm
- (A) on the south-east side, from a point 2.5 metres south-west of the common boundary of Nos. 6 and 7 Lloyd Square and a point 1.0 metre south-west of the common boundary of Nos. 9 and 10 Lloyd Square:
- (B) on the north-west side, between its junction with the south-west arm of Lloyd Square and its junction with Lloyd Street
- (iv) Wilmington Street, on the south-west side between its junction with Margery Street and a point 1.5 metres south-east of the north-western wall of No. 25 Wilmington Square.
- (e) formalise 'CPZ A' waiting restrictions to apply between 8:30am and 6:30pm, Monday to Friday and between 8:30am and 1:30pm on Saturday in Lloyd Square, the north-west arm, on the north-western side:
- (i) between its junction with Wharton Street and a point 1.6 metres north-east of the common boundary of Nos. 19 and 20 Lloyd Square; and
- (ii) between a point 1.5 metres north-east of the common boundary of Nos. 16 and 17 Lloyd Square and a point 6 metres south-west of the south-western kerb-line of Lloyd Street.
- (f) amend parking places as follows:
- (i) in Great Percy Street, shorten the shared use (CPZ 'A' permit holders and pay-by-phone) parking place on the north-west side, outside No. 31 Great Percy Street by 5.5 metres;
- (ii) in Great Percy Street, shorten the CPZ 'A' residents permit holders parking place on the north-west side, outside No. 37 Great Percy Street by 5.5 metres;
- (iii) in Great Percy Street, remove the shared use (CPZ 'A' permit holders and pay-by-phone) parking place on the south-east side, outside Nos. 28 and 30 Great Percy Street by 9.1 metres;
- (iv) in Great Percy Street, shorten the CPZ 'A' residents permit holders parking place on the south-east side, outside Nos. 34 and 36 Great Percy Street by 5.5 metres;
- (v) in Great Percy Street, extend the CPZ 'A' residents permit holders parking place on the north-west side, outside No. 69 Great Percy Street by 8.2 metres;
- (vi) in Great Percy Street, extend the shared use (CPZ 'A' permit holders and pay-by-phone) parking place on the south-east side, outside No. 70 Great Percy Street by 6.1 metres;
- (vii) in Lloyd Square (north-western arm), shorten the shared use (CPZ 'A' permit holders and pay-by-phone) parking place on the south-east side, opposite Nos. 16 and 17 Lloyd Square by 7.6 metres;
- (viii) in Lloyd Square (north-western arm), shorten the CPZ 'A' residents permit holders parking place on the south-east side, opposite Nos. 18 and 19 Lloyd Square by 10.2 metres;
- (ix) in Lloyd Square (south-eastern arm), remove 23.8 metres of CPZ 'A' residents permit holders parking place on the south-east side, outside Nos. 5 to 8 Lloyd Square;
- (ix) in Lloyd Square (south-eastern arm), amend the CPZ 'A' residents permit holders parking place on the south-east side, outside Nos. 3 to 5 Lloyd Square to allow the dedicated disabled bay in item (i) below:
- (x) in Wilmington Street, remove the CPZ 'A' residents permit holders parking place on the south-west side, at the junction with Margery Street
- (g) suspend the operation of the width restriction in Great Percy Street between its junctions with Lloyd Street and Amwell Street and on Lloyd Baker Street between its junctions with Lloyd Street and Amwell Street.
- (h) suspend the operation of the one way system in Lloyd Baker Street between its junctions with Amwell Street and the south-western north-west to south-east arm of Lloyd Square
- (j) Introduce a 6.6 metre length dedicated disabled bay outside No 4 Lloyd Square

- 3. The Order provides that in pursuance of section 10(2) of the Road Traffic Regulation Act 1984, the Director of Environment and Commercial Operations at the London Borough of Islington, or some person authorised by him, may, if it appears to him or that person essential in the interests of the expeditious, convenient and safe movement of traffic or for preserving or improving the amenities of the area through which any road affected by the Orders extends, modify or suspend any of the Orders or any provision thereof.
- 4. A copy of the Orders, which will come into force on 8th April 2022, and other documents giving more detailed particulars of the Orders, are available for inspection by the public until the Orders cease to have effect, during normal office hours at 222 Upper Street, London, N1 1XR. Copies of these documents may also be obtained either by emailing publicrealm@islington.gov.uk, by telephoning Public Realm on 020 7527 2000 or by sending a postal request to Public Realm, 1 Cottage Road, London N7 8TP. You can arrange inspection in person by emailing publicrealm@islington.gov.uk or calling 020 7527 2000.
- 5. The Council will be considering in due course whether the provisions of the experimental Orders should be continued in force indefinitely by means of permanent Orders made under sections 6, 45, 46 and 49 of the Road Traffic Regulation Act 1984. Any person may object to the making of the permanent Orders within a period of six months beginning with the day on which the experimental Orders come into force or, if the Orders are varied by another Order or modified pursuant to section 10(2) of the 1984 Act, beginning with the day on which the variation or modification or the latest variation or modification comes into force. Any such objection must be in writing and must state the grounds on which it is made and be sent to Public Cottage Road, London N7 publicrealm@islington.gov.uk. Any objection may be communicated to, or be seen by, other persons who may be affected.
- 6. Any person who wishes to question the validity of the Orders, or of any provisions contained therein on the grounds that it or they are not within the relevant powers of the Road Traffic Regulation Act 1984 or that any requirement thereof or any relevant regulations made thereunder has not been complied with in relation to the Orders, may make application for the purpose to the High Court within six weeks from the date on which the Orders are made.

Dated 1st April 2022

Tony Ralph

Director of Environment and Commercial Operations

#### **SCHEDULE**

Roads and parts of roads where disabled person's badge holders will be eligible for 'AMW' permits (all properties are eligible unless otherwise specified)

Amwell Street, Nos. 1-83 (odd only); Attneave Street; Claremont Square, Nos. 2-17; Cruikshank Street; Cumberland Gardens; Easton Street; Farringdon Road, Nos. 114-164 (even only); Fernsbury Street; Granville Square; Great Percy Street; Hardwicke Mews; Holford Yard; Kings Cross Road, Nos. 2-8, 10-42, 44-100 (even only); Lloyd Baker Street; Lloyd Square; Lloyd Street; Margery Street; Naoroji Street; Penton Rise, Nos. 1-27 (odd only); Pentonville Road, Nos. 79-123 (odd only); Percy Circus; Prideaux Place; Rosebery Avenue, Nos. 49-119 (odd only) and Greenwood House; Soley Mews; Tysoe Street, Nos. 8-20; Vernon Rise; Vernon Square; Wharton Cottages; Wharton Street; Wilmington Square; Yardley Street. (4037761)

# LONDON BOROUGH OF LAMBETH PROPOSED CHANGES TO THE LAYOUT OF PARKING PLACES IN ELMS CRESCENT AND WYATT PARK ROAD

[NOTE: This Notice is about Orders to formally remove certain parking spaces in Elms Crescent and Wyatt Park Road to accommodate the installation of new vehicle crossovers (dropped kerbs).

- 1. NOTICE IS HEREBY GIVEN that the Council of the London Borough of Lambeth (the "Council") on the 30th March 2022 made The Lambeth (Clapham "L") (Parking Places) (No.2) Order 2022 and The Lambeth (Streatham Hill East "M") (Parking Places) (No. 2) Order 2022 under sections 6, 45, 46, 49 and 124 of and Part IV of Schedule 9 to the Road Traffic Regulation Act 1984.
- 2. The general effect of the Orders which will come into force on 4th April 2022 will be to remove a length of parking space at each of the following locations and replace it with single yellow line waiting restrictions that will apply at the same time as the controlled parking zone operates in that road (the type of parking space is specified in brackets):-

- (1) **Elms Crescent**, the south-east side, outside No. 161 (resident's parking space):
- (2) Elms Crescent, the south-east side, outside No. 165 (resident's parking space);
- (3) **Wyatt Park Road**, the south-west side, outside No. 62 (shared use parking space).
- 3. The proposed removal of parking spaces at the locations described in paragraph 2 above is necessary to allow vehicle crossovers (dropped kerbs) to be installed, so to enable vehicular access to properties.
- 4. If you have any enquiries about this matter, please email: cpzconsultation@lambeth.gov.uk (quoting reference VC-1000).
- 5. A copy of each of the Orders and documents giving more details about them are available for inspection online at: <a href="https://www.lambeth.gov.uk/traffic-management-orders">www.lambeth.gov.uk/traffic-management-orders</a> or from 9.30 am until 4.30 pm on Mondays to Fridays inclusive (except bank/public holidays) until the last day of a period of six weeks beginning with the date on which the Orders are made, at the offices of Lambeth Council's Highways Team, 3rd Floor, Civic Centre, 6 Brixton Hill, London SW2 1EG. To arrange inspection please telephone 020 7926 0209 or email: Trafficorders@lambeth.gov.uk.
- 6. If any person wishes to question the validity of the Orders or of any of their provisions on the grounds that it or they are not within the powers conferred by the Road Traffic Regulation Act 1984, or that any requirement of that Act or of any instrument made under that Act has not been complied with, that person may, within 6 weeks from the date on which the Orders are made, apply for the purpose to the High Court.

Dated 30th March 2022

Ben Stevens

Highway Network Manager

(4037763)

# LONDON BOROUGH OF LAMBETH INTRODUCTION OF ELECTRIC VEHICLE CHARGING PARKING PLACES IN VARIOUS ROADS

[Note: This notice is about the introduction of new electric vehicle charging parking places, some of which will be for certain CPZ permit holders only, in various locations as specified in paragraph 2 and the Schedules to this Notice. If there is any existing parking space at the locations specified in Schedule 2 or 3, then this will be removed to accommodate the new electric vehicle charging parking places. controlled hours waiting restrictions (single yellow lines) will also be removed to accommodate the electric vehicle charging place outside No. 33 Killyon Road and a blue badge disabled bay will be relocated by 2 metres south-eastwards to accommodate the electric vehicle charging place outside No. 12 Englewood Road.]

- 1. NOTICE IS HEREBY GIVEN that the Council of the London Borough of Lambeth (the "Council") on the 30th March 2022 made The Lambeth (Free Parking Places) (Disabled Persons) (No. 5) Order 2022, The Lambeth (Free Parking Places) (Electric Vehicles) (No. 2) Order 2022, The Lambeth (Camberwell) (Parking Places) (No. 2) Order 2022; The Lambeth (Brixton "B") (Parking Places) (No. 3) Order 2022; The Lambeth (Clapham "C") (Parking Places) (No.2) Order 2022; The Lambeth (Brixton 'E') (Parking Places) (No.2) Order 2022; The Lambeth (Streatham Hill West "G") (Parking Places) (No.2) Order 2022; The Lambeth (Tulse Hill) (Parking Places) (No.1) Order 2022; The Lambeth (Kennington) (Parking Places) (No. 5) Order 2022; The Lambeth (Clapham "L") (Parking Places) (No. 3) Order 2022; The Lambeth (Streatham Hill East "M") (Parking Places) (No. 3) Order 2022; The Lambeth (Herne Hill) (Parking Places) (No. 2) Order 2022; The Lambeth (Poets' Corner) (Parking Places) (No.4) Order 2022; The Lambeth (Brixton Hill East) (Parking Places) (No. 2) Order 2022; The Lambeth (Thornton) (Parking Places) (No. 2) Order 2022; The Lambeth (Stockwell) (Parking Places) (No. 3) Order 2022; The Lambeth (Vassall "V") (Parking Places) (No. 2) Order 2022; under sections 6, 45, 46, 49 and 124 of and Part IV of Schedule 9 to the Road Traffic Regulation Act 1984, as amended.
- 2. The general effect of the Orders which will come into force on 11th April 2022, will be:-

(a) to introduce:

(i) electric vehicle charging parking places at the locations specified in Schedule 1 to this Notice. These parking places, the length of which is specified in brackets next to the locations, will operate at any time and during these hours and will be available to any vehicle whilst charging, for a maximum stay period of 24 hours, with no return for 2 hours;

(ii) resident permit holders electric vehicle charging parking places at the locations specified in Schedule 2 to this Notice. These parking places, the length of which is specified in brackets next to the locations, will operate at any time and will be available to resident permit holders of the controlled parking zone ("CPZ") shown in brackets next to the locations, whilst charging, for a maximum stay period of 24 hours, with no return for 2 hours. [Note: other types of permits that can be used in resident permit holders parking places within those CPZs will also be permitted to use these parking places]; and

(iii) permit holders (resident and business permits) electric vehicle charging parking places at the locations specified in Schedule 3 to this Notice. These parking places, the length of which is specified in brackets next to the locations, will operate at any time and will be available to resident permit holders or business permit holders of the controlled parking zone ("CPZ") shown in brackets next to the locations, whilst charging, for a maximum stay period of 24 hours, with no return for 2 hours. [Note: other types of permits that can be used in permit holders (resident and business permits) parking places within those CPZs will also be permitted to use these parking places]; (b) in the locations specified in Schedules 2 and 3 to this Notice, to remove or reduce in length existing parking space to accommodate the resident permit holders (resident and business permits) electric vehicle charging parking places;

(c) to remove 2.5 metres of controlled hours waiting restrictions (single yellow lines) outside No. 33 Killyon Road, to accommodate the electric vehicle charging place in that approximate location referred to in paragraph (a)(iii) above; and

(d) to relocate south-eastwards by approximately 2 metres, the blue badge disabled bay outside No. 12 Englewood Road, to accommodate the electric vehicle charging place in that approximate location referred to in paragraph (a)(iii) above.

- 3. The Orders are necessary to provide electric vehicle charging parking places to meet demand and aid the transition to electric vehicles (Lambeth Council is currently committed to install over 200 charging points by 2022). Electric vehicles contribute to improvements in air quality, specifically nitrogen dioxide (NO2) levels and reduce greenhouse gas emissions and so the Council wishes to encourage their use. The current lack of charging facilities is a barrier people wishing to buy electric vehicles and so by providing more charging places, the aim is to remove that barrier and give potential users more confidence that they will be able to access charging facilities when necessary.
- 4. If you have any enquiries, please telephone Lambeth Council's Parking Design Team on 020 7926 6799 or email: cpzconsultation@lambeth.gov.uk.
- 5. A copy of each of the Orders and documents giving more details about them are available for inspection online at: <a href="https://www.lambeth.gov.uk/traffic-management-orders">www.lambeth.gov.uk/traffic-management-orders</a> or from 9.30 am until 4.30 pm on Mondays to Fridays inclusive (except bank/public holidays) until the last day of a period of six weeks beginning with the date on which the Orders are made, at the offices of Lambeth Council's Highways Team, 3rd Floor, Civic Centre, 6 Brixton Hill, London SW2 1EG. To arrange inspection please telephone 020 7926 0209 or email: Trafficorders@lambeth.gov.uk.
- 6. If any person wishes to question the validity of the Orders or of any of their provisions on the grounds that it or they are not within the powers conferred by the Road Traffic Regulation Act 1984, or that any requirement of that Act or of any instrument made under that Act has not been complied with, that person may, within 6 weeks from the date on which the Orders are made, apply for the purpose to the High Court.

Dated 30th March 2022

Ben Stevens

Highway Network Manager

SCHEDULE 1 - locations of electric vehicle charging parking places Angles Road outside No. 7 (length 6 metres); Arragon Gardens outside No. 30 (length 6 metres); Bloomhall Road outside No. 42 (length 6 metres) [2 wheels on the footway]; Carnac Street outside No. 11 (length 6 metres); Casewick Road outside No. 121 (length 6 metres); Dodbrooke Road adjacent to No. 15 Selsdon Road (length 6 metres); Harborough Road outside Nos. 94/96 (length 6 metres);

Heathdene Road adjacent to No. 26 Fontaine Road (length 6 metres); Linton Grove adjacent to No. 38 Ladas Road (length 6 metres); Madiera Road outside Nos. 1 to 12 St. Anselm's Court (length 6 metres); Mountearl Gardens outside Nos. 13 to 27 (length 6 metres); and Towton Road outside No. 5 (length 6 metres).

SCHEDULE 2 - locations of resident permit holders electric vehicle charging parking places

Abbeville Road adjacent to No. 1 Crescent Lane (length 6 metres) (CPZ L); Barcombe Avenue outside No. 73 (length 6 metres) (CPZ M); Burnbury Road outside No. 14 (length 6.1 metres) (CPZ R); Cambray Road outside No. 78 (length 6 metres) (CPZ R); Clapham Manor Street outside No. 21 (length 6 metres) (CPZ C); Dorchester Drive outside No. 5 (length 6 metres) (CPZ N); Elms Crescent outside No. 28 (length 6 metres) (CPZ L); Elms Crescent outside No. 35 (length 6 metres) (CPZ L); Elms Crescent outside No. 73 (length 6 metres) (CPZ L); Flaxman Road outside Nos. 101 and 103 (length 6 metres) (CPZ A); Gauden Road outside No. 102 (length 6 metres) (CPZ S); Gubyon Avenue opposite No. 50 (length 6 metres) (CPZ N); Guildford Road outside No. 46 (length 6 metres) (CPZ S); Kennington Oval outside The Cricketers PH (length 6 metres) (CPZ K and CPZ KS); Kimberley Road outside No. 21 (length 6 metres) (CPZ S); Larkhall Lane outside No. 42 (length 6 metres) (CPZ S); Lydon Road outside No. 21 (length 6 metres) (CPZ C); Offerton Road outside No. 23 (length 6 metres) (CPZ C); Parkthorne Road opposite No. 2 (length 6 metres) (CPZ R); Prideaux Road adjacent to No. 13 Tregothnan Road (length 6 metres) (CPZ S): Shakespeare Road outside Nos. 277 and 279 (length 6 metres) (CPZ P); Shandon Road adjacent to No. 116 Elms Crescent (length 6 metres) (CPZ L); Shandon Road outside Nos. 3 and 5 (length 6 metres) (CPZ L); Tierney Road outside No. 75 (length 6 metres) (CPZ G); Tremadoc Road outside No. 33 (length 6 metres) (CPZ C); and Walcot Square adjacent to No. 9 Walcot Square (length 6 metres) (CPZ K and CPZ KS).

SCHEDULE 3 - locations of permit holders (resident and business permits) electric vehicle charging parking places

Berwyn Road outside No. 2 (length 6 metres) (CPZ H); Bonham Road outside Nos. 1 to 24 Beatrice House (length 6 metres) (CPZ E); Burnbury Road adjacent to No. 50 Haverhill Road (length 6.5 metres) (CPZ R); Calais Street opposite No. 16 (length 6 metres) (CPZ V); Claverdale Road outside Nos. 67 and 69 (length 5.9 metres) (CPZ Q); Combermere Road opposite No. 20 (length 6 metres) (CPZ B); Craster Road adjacent to No. 119 Elm Park (length 6 metres) (CPZ Q); Dalyell Road outside No. 42 (length 6 metres) (CPZ B); Elms Crescent outside No. 58 (length 6 metres) (CPZ L); Englewood Road outside Nos. 10 and 12 (length 6 metres) (CPZ L); Faygate Road adjacent to No. 99 Cricklade Avenue (length 5 metres) (CPZ M); Flaxman Road opposite No. 77 (length 6 metres) (CPZ A); Flaxman Road outside No. 64 (length 5.9 metres) (CPZ A); Honeybrook Road adjacent to No. 147 Cavendish Avenue (length 6 metres) (CPZ L); Killieser Avenue outside Nos. 30 and 32 (length 7.2 metres) (CPZ G); Killyon Road outside No. 33 (length 6 metres) (CPZ C); Killyon Road outside Nos. 22 and 24 (length 6 metres) (CPZ C); Kimberley Road adjacent to No. 82 Grantham Road (length 5.5 metres) (CPZ S); Klea Avenue opposite No. 33 (length 6 metres) (CPZ L); Klea Avenue outside No. 61 (length 6 metres) (CPZ L); Lambourn Road outside No. 40 (length 6 metres) (CPZ C); Larkhall Rise outside No. 78 (length 6 metres) (CPZ C); Larkhall Rise outside No. 97 (length 6 metres) (CPZ C); Methley Street outside No. 35 (length 6 metres) (CPZ KS); Moorland Road outside No. 50 (length 6.6 metres) (CPZ B and CPZ BI); Peardon Street outside No. 16 (length 6 metres) (CPZ C); Rattray Road adjacent to No. 33 Mervan Road (length 5 metres) (CPZ B and CPZ BI); Rozel Road adjacent to No. 12 Turret Grove (length 6 metres) (CPZ C); Scholars Road outside No. 2 and adjacent to No. 14 Hydelthorpe Road (length 6 metres) (CPZ R); Stockwell Park Crescent outside No. 17 (length 6.1 metres) (CPZ S); Stockwell Park Crescent outside Nos. 48 and 50 (length 5.9 metres) (CPZ S); The Chase outside No. 23 (length 6 metres) (CPZ C); Union Grove adjacent to No. 80 Courland Grove (length 5.4 metres) (CPZ S); Victoria Rise outside No. 126 (length 6 metres) (CPZ C); and Walberswick Street outside Nos. 5 to 11 (length 6 metres) (CPZ S). (4037762)

# OTHER NOTICES

#### **COMPANY LAW SUPPLEMENT**

The Company Law Supplement details information notified to, or by, the Registrar of Companies. The Company Law Supplement to *The London, Belfast and Edinburgh Gazette* is published weekly on a Tuesday.

These supplements are available to view at https://www.thegazette.co.uk/browse-publications.

Alternatively use the search and filter feature which can be found here https://www.thegazette.co.uk/all-notices on the company number and/or name. (3966852)

In the In the High Court of Justice

Business & Property Courts, Bristol District Registry Court Number: PT-2022-BRS-000006

#### **EDWARD JAMES STEVENSON**

In the matter of an application for a declaration that EDWARD JAMES STEVENSON be presumed dead

A claim has been issued in the High Court of Justice for a declaration that EDWARD JAMES STEVENSON whose last known place of residence was outside of the jurisdiction, in Uganda, is presumed to be dead. Any person having an interest may apply to the court to intervene in the matter. The Court has listed a hearing to take place at 12pm on 19 April 2022 at which the declaration will be sought. If you wish to apply to the court, you should do so at the Royal Courts of Justice as soon as possible, and if possible within 21 days of the date of this notice or at the hearing at 12pm on 19 April 2022. Any delay in doing so may harm your prospects of being able to intervene.

Hugh James, Two Central Square, Cardiff, CF10 1FS, SABOL/STE1330.1 and sarah.bolt@hughjames.com

Ag OH123097 (4035349)

# **COMPANIES**

# Corporate insolvency

#### **NOTICES OF DIVIDENDS**

In the County Court at Portsmouth No 75 of 2015

#### **COLOURWORX GARAGE SERVICES LTD**

07140706

Registered office: Unit 15, Quay Lane, GOSPORT, Hampshire, PO12 4LJ

Notice is hereby given that I intend to declare an Interim Dividend of  $4p/\Sigma$  to unsecured creditors within a period of 2 months from the last date of proving.

Creditors who have not proved their debts must do so by 3 May 2022 otherwise they will be excluded from the dividend.

The required proof of debt form, which must be lodged with me at the address below, is available on the Insolvency Service website (www.bis.gov.uk/insolvency, select "Forms" and then form 4.25). Alternatively, you can contact my office at the address below to supply a form.

M Commins, Official Receiver, Official Receiver's Office, Insolvency Service, PO Box 16653, Birmingham B2 2NG.

Tel: 0303 003 1741, Email: RTLU.lpswich@insolvency.gov.uk (LQD4343547)

Capacity: Liquidator (4037776)

#### **AQUEL CONSULTING LIMITED**

08736879

Registered office: Stamford House, Northenden Road, Sale, Cheshire, M33 2DH Previous registered office Bennett Brooks & Co Limited Suite 345, 50 Eastcastle Street, Fitzrovia, London, W1W 8EA

NOTICE IS HEREBY GIVEN, pursuant to Rule 14.28 of The Insolvency (England & Wales) Rules 2016 that the Joint Liquidators intend to declare a first and final dividend to creditors within two months of the last date for proving which is 27 April 2022.

Creditors who have not already proved their debts must send their full names and addresses (and those of their Solicitors, if any), together with full particulars of their debts or claims to the Joint Liquidators at KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH by 27 April 2022.

The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: It is anticipated that all known Creditors will be paid in full.

Joint Liquidator: Steve Kenny (IP number 24030) of KBL Advisory

Joint Liquidator: Steve Kenny (IP number 24030) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH.

Joint Liquidator: *Richard Cole* (IP number 26070) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33

Date of Appointment: 30 March 2022

For further details contact Adam Horan on 0161 637 8100 or at Adam@KBL-Advisory.com (4037859)

#### **BAR FIBRE LIMITED**

04059051

Trading Name: Fibre

Registered office: The Old Bank, 187a Ashley Road, Hale, Cheshire, WA15 9SQ

Principal trading address: 168 Lower Briggate, Leeds, LS1 6LY

Notice is hereby given that I, Darren Brookes, the Liquidator of the above-named Company, intend declaring a final dividend to the non-preferential unsecured creditors within two months of the last date for proving specified below.

Creditors who have not already proved are required, on or before 28 April 2022, the last date for proving, to submit a proof of debt to me at Milner Boardman & Partners, The Old Bank, 187a Ashley Road, Hale, Cheshire, WA15 9SQ, and, if so requested by me, to provide such further details or produce such documentary or other evidence as may appear to be necessary. A creditor who has not proved his debt before the date specified above is not entitled to disturb the distribution of the prescribed part because he has not participated in it.

Date of Appointment: 23 July 2021

Office Holder Details: *Darren Brookes* (IP No. 9297) of Milner Boardman & Partners, The Old Bank, 187a Ashley Road, Hale, Cheshire. WA15 9SQ

Further details contact: Natasha Baldwin, Tel: 0161 927 7788, Email: natashab@milnerboardman.co.uk

Darren Brookes, Liquidator

30 March 2022

Ag OH123313

(4035625)

#### **CLEAR LETTINGS LTD**

06275759

Registered office: 3rd Floor, Vintry Building, Wine Street, Bristol, BS1 2BD

Principal trading address: 5 Southernhay West, Exeter, Devon, EX1 1.IG

Notice is hereby given by the Joint Liquidators, Louise Durkan and Richard Easterby of Quantuma Advisory Limited, 3rd Floor, Vintry Building, Wine Street, Bristol, BS1 2BD, (telephone number 01174 287 408), pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND & WALES) RULES 2016, that we intend to declare a first and final dividend to the unsecured creditors of the Company within two months of the last date for proving set out below.

The unsecured creditors of the Company are required, on or before 30 April 2022 ("the last date for proving"), to prove their debts by sending to the undersigned, Louise Durkan of Quantuma Advisory Limited, 3rd Floor, Vintry Building, Wine Street, Bristol, BS1 2BD, the Joint Liquidator of the Company, written statements of the amounts they claim to be due to them from the Company and, if so requested, to provide such further details or produce such documentary evidence as may appear to the liquidator to be necessary.

A creditor who has not proved this debt before the declaration of any dividend is not entitled to disturb, by reason that he has not participated in it, the distribution of that dividend or any other dividend declared before his debt was proved.

Date of Appointment: 5 April 2019

Office Holder Details: Louise Durkan (IP No. 25010) and Richard Easterby (IP No. 18034) both of Quantuma Advisory Limited, 3rd Floor, Vintry Building, Wine Street, Bristol, BS1 1BD

Further details contact: Marius Pitulac, Email: Marius.Pitulac@quantuma.com, Tel: 0117 428 7408.

Louise Durkan, Joint Liquidator

29 March 2022

Ag OH123253

(4035540)

# COUNTERPARTY RISK SOLUTIONS LIMITED

08239036

Registered office: c/o Absolute Recovery Limited, Unit 2, Railway Court, Ten Pound Walk, Doncaster, DN4 5FB

Principal trading address: 23 Warminster Road, London, SE25 4DL Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Liquidator intends to declare a first and final dividend to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Liquidator at Absolute Recovery Limited, Unit 2, Railway Court, Ten Pound Walk, Doncaster, DN4 5FB by no later than 25 April 2022 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

Date of Appointment: 21 December 2016

Office Holder Details: Stephen Penn (IP No. 6899) of Absolute Recovery Limited, Unit 2 Railway Court, Ten Pound Walk, Doncaster, DN4 5FB

Further details contact: Stephen Penn, Email: info@absrecovery.co.uk, Tel: 01302 572701. Alternative email: D.jones@absrecovery.co.uk

Stephen Penn, Liquidator

30 March 2022 Ag OH123337

(4035402)

#### **CRAWSHAW BUTCHERS LIMITED**

0239968

Registered office: c/o Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR

Principal trading address: Unit 4, Sandbeck Way, Hellaby Industrial Estate, Rotherham, South Yorkshire, S66 8QL

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Liquidators intend to declare a first and final dividend to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Liquidators at Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR or via email to crawshawgroup@uk.ey.com by no later than 22 April 2022 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

In accordance with the provision of Part 14 of the Insolvency (England and Wales) Rules 2016, a creditor whose debt is a 'small debt' not exceeding £1,000 is deemed to have proved for the purposes of determination and payment of this dividend and is not required to prove their debt as described by this notice unless advised otherwise by the Joint Liquidators.

Date of Appointment: 30 October 2019

Office Holder Details: Robert Hunter Kelly (IP No. 8582) and Charles Graham John King (IP No. 8985) both of Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR

Further details contact: The Joint Liquidators, Email: crawshawgroup@uk.ey.com, Alternative contact: Amanda Scuffham. Robert Hunter Kelly, Joint Liquidator

31 March 2022

Ag OH123487 (4036159)

#### **CRAWSHAW GROUP PLC**

04755803

Registered office: c/o Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR

Principal trading address: Unit 4, Sandbeck Way, Hellaby Industrial Estate, Rotherham, South Yorkshire, S66 8QL

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Liquidators intend to declare a first and final dividend to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Liquidators at Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR or via email to crawshawgroup@uk.ey.com by no later than 22 April 2022 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

In accordance with the provision of Part 14 of the Insolvency (England and Wales) Rules 2016, a creditor whose debt is a 'small debt' not exceeding £1,000 is deemed to have proved for the purposes of determination and payment of this dividend and is not required to prove their debt as described by this notice unless advised otherwise by the Joint Liquidators.

Date of Appointment: 30 October 2019

Office Holder Details: Robert Hunter Kelly (IP No. 8582) and Charles Graham John King (IP No. 8985) both of Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR

Further details contact: The Joint Liquidators, Email: crawshawgroup@uk.ey.com Alternative contact: Amanda Scuffham Robert Hunter Kelly, Joint Liquidator

31 March 2022

Ag OH123488 (4036157)

#### **EAST YORKSHIRE BEEF LIMITED**

04390139

Trading Name: WF Burton Butchers

Registered office: c/o Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR

Principal trading address: Unit 4, Sandbeck Way, Hellaby Industrial Estate, Rotherham, South Yorkshire, S66 8QL

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Liquidators intend to declare a first and final dividend to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Liquidators at Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR or via email to crawshawgroup@uk.ey.com by no later than 22 April 2022 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

In accordance with the provision of Part 14 of the Insolvency (England and Wales) Rules 2016, a creditor whose debt is a 'small debt' not exceeding £1,000 is deemed to have proved for the purposes of determination and payment of this dividend and is not required to prove their debt as described by this notice unless advised otherwise by the Joint Liquidators.

Date of Appointment: 30 October 2019

Office Holder Details: Robert Hunter Kelly (IP No. 8582) and Charles Graham John King (IP No. 8985) both of Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR

Further details contact: The Joint Liquidators, Email: crawshawgroup@uk.ey.com, Alternative contact: Amanda Scuffham Robert Hunter Kelly, Joint Liquidator

31 March 2022

Ag OH123489 (4036160)

#### GABBOTTS FARM (RETAIL) LIMITED

03953569

Registered office: c/o Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR

Principal trading address: Unit 4, Sandbeck Way, Hellaby Industrial Estate, Rotherham, South Yorkshire, S66 8QL

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Liquidators intend to declare a first and final dividend to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Liquidators at Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR or via email to crawshawgroup@uk.ey.com by no later than 22 April 2022 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

In accordance with the provision of Part 14 of the Insolvency (England and Wales) Rules 2016, a creditor whose debt is a 'small debt' not exceeding  $\mathfrak{L}1,000$  is deemed to have proved for the purposes of determination and payment of this dividend and is not required to prove their debt as described by this notice unless advised otherwise by the Joint Liquidators.

Date of Appointment: 30 October 2019

Office Holder Details: Robert Hunter Kelly (IP No. 8582) and Charles Graham John King (IP No. 8985) both of Ernst & Young LLP, 1 Bridgewater Place, Water Lane, Leeds, LS11 5QR

Further details contact: The Joint Liquidators, Email crawshawgroup@uk.ey.com, Alternative contact: Amanda Scuffham Robert Hunter Kelly, Joint Liquidator

31 March 2022

Ag OH123491 (4036158)

# GREEN ENERGY GEOTHERMAL INTERNATIONAL LIMITED

09512859

Trading Name: Green Energy Geothermal International Limited and GEG International

Registered office: Cavendish House, 39-41 Waterloo Street, Birmingham B2 5PP

Principal trading address: Niddry Lodge, 51 Holland Street, London W8.7.IB

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016 (AS AMENDED), that the Joint Liquidators intend to declare a second and final dividend to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Liquidators at Cavendish House, 39-41 Waterloo Street, Birmingham B2 5PP by no later than 27 April 2022 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

Date of Appointment: 24 July 2019

Office Holder Details: *David Griffiths* (IP No. 22930) and *Conrad Beighton* (IP No. 9556) both of Leonard Curtis, Cavendish House, 39-41 Waterloo Street, Birmingham, B2 5PP

For further details contact: The Joint Liquidators, Tel: 0121 200 2111. Alternative contact Abbie Lenihan

David Griffiths, Joint Liquidator

30 March 2022

Ag OH123231 (4035501)

**MARQUEES BY STUART LIMITED** 

06635586

Trading Name: Funky Tents

Registered office: Kroll Advisory Ltd., 4B Cornerblock, 2 Cornwall Street, Birmingham, B3 2DX

Principal trading address: Glaston Road, Uppingham, Oakham, LE15

A first and final dividend to preferential creditors is intended to be declared in the above matter within 2 months of 29 April 2022 (the last date of proving).

Any creditor who has not yet lodged a proof of debt, with full supporting documentation, must do so by 29 April 2022. Creditors should send their claims to Craig.Draper@kroll.com. A creditor who has not proved their debt by this date will be excluded from this dividend.

In order to allow payment of the dividend via BACS transfer, creditors should confirm their bank details by sending an email to Craig.Draper@kroll.com with the account name, account number and sort code

Date of Appointment: 4 March 2020

Office Holder Details: *Allan Graham* (IP No. 8719) and *Matthew Ingram* (IP No. 10790) both of Kroll Advisory Ltd, 4B Cornerblock, 2 Cornwall Street, Birmingham, B3 2DX

Further details contact: The Joint Liquidators, Tel: +44 (0) 121 214 1120. Alternative contact: Craig Draper, Email: craig.draper@kroll.com *Allan Graham,* Joint Liquidator

30 March 2022

Ag OH123236 (4035382)

**VIADUCT LEISURE LIMITED** 

05331780

Trading Name: Viaduct Showbar

Registered office: The Old Bank, 187a Ashley Road, Hale, Cheshire, WA15 9SO

Principal trading address: 11 Lower Briggate, Leeds, LS1 6ER

Notice is hereby given that I, Darren Brookes, the Liquidator of the above-named Company, intend declaring a final dividend to the non-preferential unsecured creditors within two months of the last date for proving specified below.

Creditors who have not already proved are required, on or before 28 April 2022, the last date for proving, to submit a proof of debt to me at Milner Boardman & Partners, The Old Bank, 187a Ashley Road, Hale, Cheshire, WA15 9SQ, and, if so requested by me, to provide such further details or produce such documentary or other evidence as may appear to be necessary. A creditor who has not proved his debt before the date specified above is not entitled to disturb the distribution of the prescribed part because he has not participated in it.

Date of Appointment: 2 February 2022

Office Holder Details: *Darren Brookes* (IP No. 9297) of Milner Boardman & Partners, The Old Bank, 187a Ashley Road, Hale, Cheshire. WA15 9SQ

Further details contact: Natasha Baldwin, Tel: 0161 927 7788, Email: natashab@milnerboardman.co.uk

Darren Brookes, Liquidator

30 March 2022

Ag OH123314

(4035452)

#### **QUALIFYING DECISION PROCEDURE**

**4TK INTERNATIONAL LTD** 

Company Number: (10782526) previously G P Mechanical Ltd

Registered office: BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP

Principal trading address: 28 Giles Close, Hedge End, Southampton, Hampshire, SO30 2TH

Notice is hereby given, pursuant to Rule 15.13 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Liquidator of the above-named Company is seeking a decision from creditors on the basis of the Liquidator's remuneration by way of correspondence.

A creditor's vote by correspondence must be received by no later than 23.59 hours on 14 April 2022 (the 'decision date'). Details of how to cast a vote are included in the correspondence delivered to creditors. If any creditor has not received this correspondence or requires further information, please contact the Liquidator using the details below.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A creditor's vote will be disregarded if their proof in respect of their claim is not received on or before the decision date. A creditor who has opted out from receiving notices may nevertheless vote if the creditor provides a proof of debt in the requisite time frame. Proofs may be delivered to Beacon, BizSpace, Steel House, 4300 Parkway, Solent Business Park, Whiteley, Hampshire PO15 7FP.

Date of Appointment: 28 February 2022

Office Holder Details: *Matthew Fox* (IP No. 9325) of Beacon, Spaces 4500 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO15 7AZ

Contact details for Liquidator: Tel: 023 8065 1441; Email: mfox@beaconllp.com. Alternative contact: Chanise Scullion.

Matthew Fox, Liquidator

30 March 2022

Ag OH123276

(4035490)

#### TMB TRANSPORT SERVICES LTD

Company Number: (09946106)

Registered office: BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP

Principal trading address: 74 Manor Road, Rugby CV21 2TQ

Notice is hereby given, pursuant to Rule 15.13 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Liquidator of the above-named Company is seeking a decision from creditors on the basis of the liquidator's remuneration by way of correspondence.

A creditor's vote by correspondence must be received by no later than 23.59 hours on 19 April 2022 (the 'decision date'). Details of how to cast a vote are included in the correspondence delivered to creditors. If any creditor has not received this correspondence or requires further information, please contact the Liquidator using the details below.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A creditor's vote will be disregarded if their proof in respect of their claim is not received on or before the decision date. A creditor who has opted out from receiving notices may nevertheless vote if the creditor provides a proof of debt in the requisite time frame. Proofs may be delivered to Beacon, BizSpace, Steel House, 4300 Parkway, Solent Business Park, Whiteley, Hampshire PO15 7FP.

Date of Appointment: 7 January 2022

Office Holder Details: *Matthew Fox* (IP No. 9325) of Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP

Contact details for Liquidators: Tel: 023 8065 1441; Email: mfox@beaconllp.com. Alternative contact: chanise@beaconllp.com *Matthew Fox*, Liquidator

30 March 2022

Ag OH123275

(4035359)

#### **RE-USE OF A PROHIBITED NAME**

#### **BELLO DEVELOPMENTS LTD**

11907050

Registered office: 25 Abbey Drive, Swinton, Manchester, M27 6AS Principal trading address: 25 Abbey Drive, Swinton, Manchester, M27 6AS

Notice to the creditors of an insolvent company of the re-use of a prohibited name - Rule 22.4 of the Insolvency (England and Wales) Rules 2016

On 30 March 2022 the above-named Company went into insolvent liquidation.

I, David Hall of 25 Abbey Drive, Swinton, Manchester, M27 6AS was a Director of the above-named Company during the 12 months ending with the day before it went into liquidation.

I give notice that it is my intention to act in one or more of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent Company under the following name: Bello Doors Ltd

Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2): Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are - a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company; (b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or (c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above.

This notice is given in pursuance of Rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name.

Ag OH123257 (4035527)

# NOTICE TO THE CREDITORS OF AN INSOLVENT COMPANY OF THE RE-USE OF A PROHIBITED NAME - RULE 22.4 OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016 BODY AND FACE OPTIONS LIMITED

04515029

Registered office: Waen Barn, Waen Lane, Maesbury Marsh, Oswestry, Shropshire, SY10 8JD

Trading Name: Body and Face Options

On 24 February 2022 the above-named company went into insolvent liquidation.

I, Jane Millington Baker, Waen Barn, Waen Marsh, Maesbury Marsh, Oswestry, Shropshire, SY10 8JD was a director of the above-named company during the 12 months ending with the day before it went into liquidation.

I give notice that it is my intention to act in one or more of the ways specified in section 216(3) of the INSOLVENCY ACT 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name:

Body and Face Options Clinic

# Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2):

Section 216(3) of the INSOLVENCY ACT 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the INSOLVENCY ACT 1986. (This includes the exceptions in Part 22 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016). These activities are-

 a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;

b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or

c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above.

This notice is given in pursuance of Rule 22.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts. Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name.

(4035177

#### **LONDON JAMES LTD**

09421511

Registered office: Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE (formerly at Unit 8 Langford Hall, Witham Road, Maldon, Essex, CM9 4ST

Principal trading address: Unit 8 Langford Hall, Witham Road, Maldon, Essex, CM9 4ST

Notice to the creditors of an insolvent company of the re-use of a prohibited name - Rule 22.4 of the Insolvency (England and Wales) Rules 2016

On 3 September 2022 the above-named company went into insolvent liquidation.

I, Sean Nutley of Unit 8 Langford Hall, Witham Road, Maldon, Essex, CM9 4ST was a director of the above-named company during the 12 months ending with the day before it went into liquidation.

I give notice that it is my intention to act in one or more of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name: Tavern Auto Ltd.

Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2): "Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This

includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are-(a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;(b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or(c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under aname of the kind mentioned in (a) above. This notice is given in pursuance of Rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the businessof a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company. The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts. Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name".

Ag OH123254 (4035467)

#### STONE RANGE LTD

13421674

Trading Name: Stone Range

Registered office: 16 High Road, Potters Bar, England, EN6 5AF

Notice to the creditors of an insolvent company of the re-use of a prohibited name - Rule 22.4 of the Insolvency (England and Wales) **Rules 2016** 

On 4 April 2022, the above-named company will go into insolvent liquidation.

I, Cezar Altan Ahmet of 20-22 Wenlock Road, London N1 7GU, was a director of the above-named company during the 12 months ending with the day before it will go into liquidation.

I give notice that it is my intention to act in one or more of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name: Stone Range Group Ltd, company number 13955114, trading as Stone Range.

I would not otherwise be permitted to undertake these activities without leave of the court or the application of an exception created by the Insolvency (England and Wales) Rules 2016. A breach of the prohibition created by section 216 of the Insolvency Act 1986 is a criminal offence.

Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2): Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are- (a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company; (b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or (c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above.

This notice is given in pursuance of Rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name.

Ag PH10003 (4036315)

## Administration

#### APPOINTMENT OF ADMINISTRATORS

In the High Court of Justice Court Number: CR-2022-000132

**FOUNDA LIMITED** 

(Company Number 06693566)

Nature of Business: Test drilling and boring

Registered office: c/o FRP Advisory Trading Limited, Ashcroft House,

Meridian Business Park, Leicester, LE191WL

Principal trading address: 16 Commerce Square, Lace Market,

Nottingham, NG1 1HS

Date of Appointment: 30 March 2022

Names and Address of Administrators: John Lowe (IP No. 009513) and Yasmin Bhikha (IP No. 19972) both of FRP Advisory Trading Limited, Ashcroft House, Ervington Court, Meridian Business Park, Leicester, LE19 1WL

Further details contact: The Joint Administrators, Tel: 0116 303 333. Alternative contact: Samantha Wetwood, Email:

Samantha.Wetwood@frpadvisory.com

(4035526) Aa OH123311

In the High Court of Justice Business and Property Courts of Manchester Insolvency and Companies List (ChD)

No CR-2022-MAN-000119. (Company Number 07022488)

JAYFEX LIMITED

Trading Name: Aura Business Centres Nature of Business: Serviced Offices

Registered office: 412 Stretford Road, Old Trafford, Manchester, M15

4AF

Principal trading address: 412 Stretford Road, Old Trafford, Manchester, M15 4AE

Joint Administrator: Edward M Avery-Gee (IP number 12410) of CG&Co, Greg's Building, 1 Booth Street, Manchester, M2 4DU. Joint Administrator: Daniel Richardson (IP number 12650) of CG&Co,

Greg's Building, 1 Booth Street, Manchester, M2 4DU.

Date of Appointment: 29 March 2022

For further details contact Chris Brennan on 0161 358 0210 or at info@cg-recovery.com (4037910)

#### Appointment of Administrators

In the High Court of Justice Business & Property Courts of Liverpool Insolvency & Companies List (ChD),

No 000066 of 2022

## **MICHAEL ROSE & BAYLIS LIMITED**

(Company Number 09207292)

Registered office: The Royal Liver Building, Pier Head, Liverpool, L3 1HU

Principal trading address: The Royal Liver Building, Pier Head, Liverpool, L3 1HU

Nature of Business: Solicitors Date of Appointment: 22 March 2022

Joint Administrator's Name and Address: Jason Mark Elliott (IP No. 9496) of Cowgill Holloway Business Recovery LLP, Regency House, 45-53 Chorley New Road, Bolton, BL1 4QR. Telephone: 0161 827 1200.

Joint Administrator's Name and Address: Craig Johns (IP No. 013152) of Cowgill Holloway Business Recovery LLP, Regency House, 45-53 Chorley New Road, Bolton, BL1 4QR. Telephone: 0161 827 1200.

For further information contact Ben Gray at the offices of Cowgill Holloway Business Recovery LLP on 0161 672 5767, Ben.Gray@cowgills.co.uk. Address: Regency House, 45-53 Chorley New Road, Bolton, BL1 4QR.

29 March 2022 (4036389)

#### **MEETINGS OF CREDITORS**

In the High Court of Justice The Business and Property Court, Leeds No CR-2022-LDS-000038.

#### **LUXURY HOME FURNITURE LIMITED**

(Company Number 11955828) Trading Name: Shabby Store

Registered office: c/o Inquesta Corporate Recovery & Insolvency St John's Terrace 11-15 New Road Manchester M26 1LS

Principal trading address: Ashfield Mills, 298 Leeds Road, Bradford, BD10 9AD

Notice to Creditors of Decision by Correspondence

Notice is hereby given by the Administrator to the creditors of Luxury Home Furniture Limited t/a Shabby Store that a decision of creditors are required by correspondence in accordance with Rule 3.38, 15.3 and 15.13 of The Insolvency (England and Wales) Rules 2016 (as amended) and Paragraph 51 Schedule B1 of the Insolvency Act 1986 (as amended).

The purpose of the decision procedure is to consider the following resolutions:

- 1. To approve the Administrator's Proposals for achieving the purpose of the Administration
- 2. Approval of the basis of the Administrator's remuneration
- 3. Approval of the basis of the Administrator's Category 2 Disbursements
- 4. Approval of the timing of the Administrator's discharge from liability
- 5. Approval of payment of the Administrator's pre-appointment fees and expenses
- 6. Establishing a Creditors Committee if sufficient creditors willing to

The final date for votes is:

Date: 27 April 2022 - The Decision Date

In order to be entitled to vote, a creditor, including those whose debts are less than  $\mathfrak{L}1,000$ , must lodge a statement of claim in writing on or before the Decision Date, failing which the vote will be disregarded. Secured creditors (unless they surrender their security) must also give particulars of their security, the date on which it was granted and its estimated value if they wish to vote.

Nominations for the appointment of a Committee member must be delivered to the Convenor by the date specified for the lodgement of votes in the notice and can only be accepted if the Convenor is satisfied as to the nominee's eligibility under Rule 17.4.

Creditors who have opted out of receiving information may vote by submitting a claim form in writing and a voting form on or before the Decision Date.

All claims and voting forms must be delivered to Inquesta, St John's Terrace, 11-15 New Road, Manchester M26 1 LS on or before the Decision Date

Creditors may require a physical meeting to be convened to consider these Decisions provided that the request is made before the Decision Date and within 5 business days of the deemed delivery of this notice by delivering an authenticated and dated Notice of the objection to the Convener Clo Inquesta at the address below.

The threshold for a request to the use of this procedure and for the convening of a physical meeting is 10% of creditors by value, 10% by number or 10 creditors. Creditors means all creditors.

If sufficient requests for a physical meeting are received the decision procedure will not be held and no decisions taken. The Convener will then write to creditors to give Notice of a physical meeting to enable creditors to consider these decisions.

Creditors also have the right to appeal the decision(s) made by the resolution(s) by appealing to Court under Rule 15.35 of the Rules within 21 days of the Decision Date.

Administrator: Steven Wiseglass (IP number 9525) of Inquesta Corporate Recovery & Insolvency,, St John's Terrace, 11-15 New Road, Manchester, M261LS..

Date of Appointment: 02 February 2022

For further details contact Kat Davies on 0333 005 0080 or at kat.davies@inquesta.co.uk

Steven Wiseglass, ADMINISTRATOR (4036348)

Name of Company: OH ELECTRICAL SERVICES LTD

Company Number: 09907526
Company Type: Registered Company
Nature of the business: Electricians
Type of Liquidation: Creditors' Voluntary

Registered office: 38 Stonyhill Avenue, Blackpool FY4 1PR
Principal trading address: 38 Stonyhill Avenue, Blackpool FY4 1PR
Office Holder/s: Stuart Rathmell, of STUART RATHMELL
INSOLVENCY and, Egyptian Mill, Egyptian Street, Bolton BL1 2HS,
Telephone: 01204867615, Email address:

stuart.rathmell.insolvency@outlook.com
Office Holder Number/s: 10050
Date of appointment: 17 March 2022
By whom Appointed: Members and Creditors

Thursday 17 March 2022 (4037769)

Name of Company: PERFECT DAY CATERING LTD

Company Number: 13323530 Company Type: Registered Company Nature of the business: Licensed restaurants

Trading as: The Playa

Type of Liquidation: Creditors' Voluntary

Registered office: 21 Highfield Road, Dartford DA1 2JS

Principal trading address: The Playa, The Leas, Sheerness ME12 2NL Office Holder/s: Isobel Susan Brett, of Bretts Business Recovery Limited and, 21 Highfield Road, Dartford DA1 2JS, Telephone: 01474532862, Email address: ellieraycraft@brettsbr.co.uk

Office Holder Number/s: 9643

Date of appointment: 30 March 2022

By whom Appointed: Members and Creditors (4036591)

Name of Company: TOTAL VEHICLE LEASING LIMITED

Company Number: 09564032 Company Type: Registered Company Nature of the business: Vehicle Leasing Broker Type of Liquidation: Creditors' Voluntary

Registered office: St John's Terrace, 11-15 New Road, Radcliffe,

Manchester M26 1LS

Principal trading address: St. Annes House, Suite 5, 1 Old Market

Place, Knutsford WA16 6PD

Office Holder/s: Steven Wiseglass, of Inquesta Corporate Recovery and, St John's Terrace, 11-15 New Road, Radcliffe, Manchester M26 1LS, Telephone: 0333 005 0080, Email address: info@inquesta.co.uk (A1541)

Office Holder Number/s: 9525
Date of appointment: 30 March 2022

By whom Appointed: Members and Creditors (4037821)

Name of Company: S & Y CATERING LIMITED

Company Number: 11445334 Company Type: Registered Company Nature of the business: Chinese Takeaway

Trading as: Silver Hill

Type of Liquidation: Creditors' Voluntary

Registered office: 79a High Road, Willesden, London NW10 2SU Principal trading address: 13 High Street, Brierley Hill, Dudley DY5

3AA

Office Holder/s: Kian Seng Tan, of 79a High Road, Willesden, London NW10 2SU, Telephone: 02075861280, Email address: mail@kstan.co.uk (KST/IY/L1898)

Office Holder Number/s: 8032 Date of appointment: 31 March 2022

By whom Appointed: Members and Creditors (4037865)

# Creditors' voluntary liquidation

#### APPOINTMENT OF LIQUIDATORS

Name of Company: 1 TO WEAR LIMITED

Company Number: 07019382 Nature of Business: Ladies outerwear

Registered office: Mountview Court, 1148 High Road, Whetstone,

London N20 0RA

Type of Liquidation: Creditors
Date of Appointment: 25 March 2022

Liquidator's name and address: Kikis Kallis (IP No. 4692) of Kallis & Company, Mountview Court, 1148 High Road, Whetstone, London,

N20 0RA

By whom Appointed: Members and Creditors

Ag OH123292 (4035609)

Pursuant to section 109 of the Insolvency Act 1986

Name of Company: Al LIFTING LTD Company Number: 10326297

Registered office: 25 Brixham Building, 8 Artillery Place, London SE18

4AB

Principal trading address: 25 Brixham Building, 8 Artillery Place,

London SE18 4AB

Type of Liquidation: Creditors' Voluntary Liquidation

Further Contact Details Viera Navratilova Tel No. 01455 555 444 Liquidator's name and address: *Richard Frank Simms*, Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17

5FB Office Holder Number: 9252.

Capacity: Liquidator

Date of appointment: 25 March 2022

By whom appointed: The Members and Creditors

Richard Frank Simms Liquidator

25 March 2022 (4037771)

Name of Company: ALGAR SIGNCRAFT SERVICES (CAMBRIDGE) LIMITED

CIIVII I ED

Company Number: 01196266 Nature of Business: Sign Writers

Registered office: Salisbury House, Station Road, Cambridge, CB1

2LA

Type of Liquidation: Creditors
Date of Appointment: 29 March 2022

Liquidator's name and address: Kate Merry (IP No. 9439) and Ben Dyer (IP No. 18016) both of PEM, Salisbury House, Station Road,

Cambridge, CB1 2LA By whom Appointed: Creditors

Ag OH123319 (4035380)

Company Number: 09969070

Name of Company: ASL BOSTON LIMITED

Nature of Business: Legal Support

Registered office: Office 10 Ground Floor The Old,

Tannery, Eastgate, Accrington, BB5 6PW

Principal trading address: Office 10 Ground Floor The Old,

Tannery, Eastgate, Accrington, BB5 6PW

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: *Helen Whitehouse* (IP number 9680) of McAlister & Co Insolvency Practitioners Limited,

10 St Helens Road, Swansea, SA1 4AW.

Liquidator's name and address: Joint Liquidator: Sandra McAlister (IP number 9375) of McAlister & Co Insolvency Practitioners Limited, 10  $\,$ 

St Helens Road, Swansea, SA1 4AW.
Date of Appointment: 23 March 2022
By whom Appointed: Members and Creditors

For further details contact Jonathan Waller on 03300 563600 or at creditors@mcalisterco.co.uk (4037901)

Company Number: 12342634

Name of Company: **AVIANCE SERVICES LTD**Nature of Business: Freight transport by road

Registered office: Unit 8, Riverside Court, Pride Park, Derby, DE24

8JN

Principal trading address: 9 Timberlands, Crawley, West Sussex,

RH11 9PH

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: Simon Gwinnutt (IP number 8877) of Greenfield Recovery Limited, Trinity House, 28-30

Blucher Street, Birmingham, B1 1QH.

Liquidator's name and address: Joint Liquidator: Sajid Sattar (IP number 15590) of Greenfield Recovery Limited, Trinity House, 28-30

Blucher Street, Birmingham, B1 1QH. Date of Appointment: 24 March 2022 By whom Appointed: Members and Creditors

For further details contact Lee Adams on 0121 201 1720 or at la@greenfieldrecovery.co.uk (4037883)

Company Number: 12488946

Name of Company: BASIL FRESH PIZZA LTD

Trading Name: Basil Pizza

Nature of Business: Pizza shop & fast food delivery

Registered office: C/o B&C Associates Limited, Concorde House,

Grenville Place, Mil! Hill, London, NW7 3SA

Principal trading address: 73 Barlby Road, London, W10 6AW

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Liquidator: Jeffrey Mark Brenner (IP number 9301) of B&C Associates Limited,, Concorde House, Grenville

Place, Mill Hill, London, NW7 35A.

Date of Appointment: 30 March 2022

By whom Appointed: Members and Creditors

For further details contact Besa Mustafa on 020 8906 7730 or at besa@bcassociates.uk.com (4037890)

Name of Company: BBG KITCHEN LTD

Company Number: 11575737

Trading Name: The Broadway Bar And Grill

Nature of Business: Unlicensed restaurants and cafes

Registered office: The Broadway Bar And Grill, Birmingham Road,

Walsall, West Midlands, WS5 3AB
Type of Liquidation: Creditors
Date of Appointment: 30 March 2022

Liquidator's name and address: Laura Anne Walshe (IP No. 26250) of Keywood Group Limited, Gill House, 140 Holyhead Road,

Birmingham, West Midlands, B21 0AF By whom Appointed: Members and Creditors

Ag OH123342 (4035455)

Name of Company: BRITANICO LIMITED

Company Number: 08297933 Trading Name: Britanico Coffee House Nature of Business: Coffee House and Cafe Registered office: 4a Roman Road, London, E6 3RX

Type of Liquidation: Creditors
Date of Appointment: 29 March 2022

Liquidator's name and address: Jane Hardy (IP No. 9384) and Gary Birchall (IP No. 9725) both of Focus Insolvency Group, Skull House

Lane, Appley Bridge, Wigan, WN6 9EU By whom Appointed: Members and Creditors

Ag OH123383 (4035552)

Name of Company: CAMBRIDGE STRATUM LIMITED

Company Number: 09687862

Registered office: c/o Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, 30 Churchgate, Bolton, BL1

1HL

Principal trading address: None

Nature of Business: Sale of medical aesthetic equipment to

specialised clinics

Type of Liquidation: Creditors' Voluntary Date of Appointment: 29 March 2022

By whom Appointed: Members and Creditors

Liquidator's name and address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, Bolton, BL1 1HL. Telephone: 01204 255 051. For further information contact Jasmine Baxter at the offices of

Anderson Brookes Insolvency Practitioners Limited on 01204 255

051, or jasmine@andersonbrookes.co.uk.

31 March 2022 (4036403)

Name of Company: CAREMARS LTD. Company Number: 12450963 Nature of Business: Residential Care Activities

Registered office: 77 Cavendish Avenue, Harrow, HA1 3RF

Type of Liquidation: Creditors Date of Appointment: 28 March 2022

Liquidator's name and address: Gemma Louise Roberts (IP No. 9701) and Robert Neil Dymond (IP No. 10430) both of Wilson Field Limited, The Manor House, 260 Ecclesall Road South, Sheffield, S11 9PS

By whom Appointed: Members and Creditors

(4035607) Ag OH123381

Name of Company: CHEON LTD Company Number: 10737120 Nature of Business: Wholesale of textiles

Previous Name of Company: Holnic Ltd

Registered office: A2, Seedbed Centre, Vanguard Way, Southend-on-

Sea, SS3 9QY

Type of Liquidation: Creditors Date of Appointment: 24 March 2022

Liquidator's name and address: Wayne MacPherson (IP No. 9445) and Louise Donna Baxter (IP No. 009123) both of Begbies Traynor (Central) LLP, The Old Exchange, 234 Southchurch Road, Southend-

on-Sea, Essex, SS1 2EG

By whom Appointed: Members and Creditors

Ag OH123346 (4035554)

Company Number: 12188790

Name of Company: CHESHIRE ESTATES (MARY MILL) LIMITED

Nature of Business: Development of building projects

Registered office: Repton Manor, Repton Avenue, Ashford, TN23 3GP Principal trading address: Repton Manor, Repton Avenue, Ashford,

**TN23 3GP** 

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Liquidator: Philippa Smith (IP number 18670) of Smith & Barnes Insolvency Practitioners Ltd, 22A Main

Street, Garforth, Leeds, LS25 1AA. Date of Appointment: 30 March 2022 By whom Appointed: Members and Creditors

For further details contact Muzamal Ahmad on 0113 532 3278

(4036928)

Company Number: 10522000

Name of Company: CHESHIRE ESTATES (MERE APARTHOTEL) LIMITED

Nature of Business: Development of building projects

Registered office: Repton Manor, Repton Avenue, Ashford, TN23 3GP Principal trading address: Repton Manor, Repton Avenue, Ashford,

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Liquidator: Philippa Smith (IP number 18670) of Smith & Barnes Insolvency Practitioners Ltd, 22A Main Street, Garforth, Leeds, LS25 1AA.

Date of Appointment: 30 March 2022 By whom Appointed: Members and Creditors

For further details contact Muzamal Ahmad on 0113 532 3278

(4037723)

Name of Company: CLAIRE GRACE INTERIORS LTD

Company Number: 09160339

Nature of Business: Kitchen sales and installer

Registered office: 22 Regent Street, Nottingham, NG1 5BQ

Type of Liquidation: Creditors Date of Appointment: 24 March 2022

Liquidator's name and address: Joseph Gordon Maurice Sadler (IP No. 9048) and Andrew J Cordon (IP No. 009687) both of Elwell Watchorn & Saxton LLP, 22 Regent Street, Nottingham, NG1 5BQ

By whom Appointed: Creditors

(4035347)Ag OH123269

Company Number: 06648975

Name of Company: CLEAN 4 YOU LONDON LIMITED

Nature of Business: Cleaning Services

Registered office: Suite 4 Aus-Bore House, 19-25 Manchester Road,

Wilmslow, SK9 1BQ

Principal trading address: 383 London Road, Grays, RM20 4AA

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: James Kaye (IP number 9373) of NTF Corporate Solutions Limited, Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow, Cheshire, SK9 1BQ. Liquidator's name and address: Joint Liquidator: Nick Morgan (IP number 9078) of NTF Corporate Solutions Limited, Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow, Cheshire, SK9 1BQ.

Date of Appointment: 30 March 2022 By whom Appointed: Members and Creditors

For further details contact Rebecca Dale on 01625 540744 or at info@ntffs.co.uk (4037289)

Name of Company: CONSTELLATION STYLING LTD

Company Number: 11989231 Nature of Business: Hair Salon

Registered office: XL Business Solutions, Premier House, Bradford

Road, Cleckheaton, BD19 3TT Type of Liquidation: Creditors Date of Appointment: 29 March 2022

Liquidator's name and address: J N Bleazard (IP No. 09354) of XL Business Solutions, Premier House, Bradford Road, Cleckheaton,

**BD193TT** 

By whom Appointed: Members and Creditors

Ag OH123290 (4035424)

Name of Company: COSMIN BUILDING SOLUTIONS LIMITED

In Creditors' Voluntary Liquidation ("the Company")

Company Number: 11791668

Nature of Business: Construction of domestic buildings

Type of Liquidation: Creditors

Registered office: 13c Saint Germans Road, London, Greater London, SE23 1RH

Principal trading address: 13c Saint Germans Road, London, Greater

London, SE23 1RH

Date of Appointment: 30 March 2022

Liquidator's name and address: Brendan P. Hogan (IP No 13030) of Cromwell & Co Insolvency Practitioners, 5 Mercia Business Village,

Torwood Close, Coventry, CV4 8HX.

(4035185)By whom Appointed: Members and Creditors

Name of Company: CREATIVE AUTOMATION LIMITED

Company Number: 06456004

Nature of Business: Engineering services to the Contact Lens industry

Previous Name of Company: Gorman Dynamics Ltd

Registered office: Unit F10, Mills House, Mills Way, Amesbury, SP4

7RX

Type of Liquidation: Creditors Date of Appointment: 29 March 2022

Liquidator's name and address: Hugh Francis Jesseman (IP No. 9480) and Claire Howell (IP No. 23610) both of Antony Batty & Company

LLP, 3 Field Court, Gray's Inn, London, WC1R 5EF By whom Appointed: Members and Creditors

Ag OH123343 (4035410) Company Number: 09518062

Name of Company: CYGNET SPIRITS LIMITED

Nature of Business: Distilling of gin

Registered office: 10 St Helen's Road Swansea SA1 4AW Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: Gareth Bishop (IP number 17870) of McAlister & Co Insolvency Practitioners Limited, 10

St Helens Road, Swansea, SA1 4AW.

Liquidator's name and address: Joint Liquidator: Helen Whitehouse (IP number 9680) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1 4AW.

Date of Appointment: 29 March 2022
By whom Appointed: Members and Creditors

For further details contact Gareth Bishop on 03300 563600 or at Gareth@mcalisterco.co.uk (4037351)

Name of Company: **DANNIC LTD** Company Number: 09468701

Nature of Business: Construction of railways and underground

railways

Registered office: Redheugh House, Teesdale South, Thornaby Place,

Stockton on Tees, TS17 6SG Type of Liquidation: Creditors Date of Appointment: 15 March 2022

Liquidator's name and address: *Kenneth Wilson Pattullo* (IP No. 8368) of Begbies Traynor (Central) LLP, Finlay House, 10-14 West Nile Street, Glasgow G1 2PP and *Ian James Royle* (IP No. 18934) of Begbies Traynor (Central) LLP, Redheugh House, Teesdale South,

Thornaby Place, Stockton on Tees, TS17 6SG By whom Appointed: Members and Creditors

Ag OH123299 (4035412)

Name of Company: ENKO JOINERY LTD

Company Number: 08423086

Nature of Business: Joinery Installation

Previous Name of Company: Enko Building Services Ltd

Registered office: Unit 9 Mead Park Industrial Estate, Harlow, CM20 2SE and it is in the process of being changed to Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE

Type of Liquidation: Creditors

Date of Appointment: 10 March 2022

Liquidator's name and address: *Martin Weller* (IP No. 9419) and *Glyn Mummery* (IP No. 8996) both of FRP Advisory Trading Limited, Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex,

CM13 3BE

By whom Appointed: Members and Creditors

Ag OH123272 (4035431)

Name of Company: FITNESS DATA SCIENCE LIMITED

Company Number: 11149327

Nature of Business: Providing fitness software

Registered office: Rownhams House, Betteridge Drive, Rownhams,

Southampton, SO16 8LS Type of Liquidation: Creditors Date of Appointment: 28 March 2022

Liquidator's name and address: Timothy Frank Corfield (IP No. 8202) of Griffin & King, 26/28 Goodall Street, Walsall, West Midlands, WS1

1QL

By whom Appointed: Creditors

Ag OH123241 (4035537)

Name of Company: FLEET ASSISTANCE SERVICES LTD

Company Number: 12110115

Registered office: c/o Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, 30 Churchgate, Bolton, BL1

1HL

Principal trading address: None

Nature of Business: Maintenance and repair of motor vehicles

Type of Liquidation: Creditors' Voluntary Date of Appointment: 30 March 2022 By whom Appointed: Members and Creditors Liquidator's name and address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, Bolton, BL1 1HL. Telephone: 01204 255 051. For further information contact Jasmine Baxter at the offices of

Anderson Brookes Insolvency Practitioners Limited on 01204 255 051, or jasmine@andersonbrookes.co.uk.

31 March 2022 (4036400)

Name of Company: FOURMAN LTD

Company Number: 05942718

Nature of Business: Construction of commercial buildings;

Construction of domestic buildings; Architectural activities

Registered office: Unit 1 Grange Farm, Doctor Hill, Wheatley, Halifax HX2 0TG (to be changed to XL Business Solutions, Premier House,

Bradford Road, Cleckheaton, BD19 3TT)
Type of Liquidation: Creditors
Date of Appointment: 28 March 2022

Liquidator's name and address: J N Bleazard (IP No. 009354) of XL Business Solutions Limited, Premier House, Bradford Road,

Cleckheaton, BD19 3TT

By whom Appointed: Members and Creditors

Ag OH123295 (4035471)

Name of Company: GUARDIAN HOMES & DRIVEWAYS LTD

Company Number: 11751748

Nature of Business: Recreational vehicle parks, trailer parks and

camping grounds

Registered office: Glencote Office Station Road, Cheddleton, Leek, ST13 7EE (to be changed to c/o FRP, Ashcroft House, Ervington

Court, Meridian Business Park, Leicester, LE19 1WL)

Type of Liquidation: Creditors
Date of Appointment: 28 March 2022

Liquidator's name and address: Yasmin Bhikha (IP No. 19972) and John Anthony Lowe (IP No. 009513) both of FRP Advisory Trading Limited, Ashcroft House, Ervington Court, Meridian Business Park, Leicester, LE19 1WL

By whom Appointed: Creditors

Ag OH123247 (4035420)

Name of Company: HOLIDAY DEALERS LIMITED

Company Number: 11762558 Nature of Business: Travel Agency

Registered office: Rose Villa, 63 Bank Street, Ossett, WF5 8PR

Type of Liquidation: Creditors
Date of Appointment: 25 March 2022

Liquidator's name and address: *Nicola Baker* (IP No. 15852) of Rushtons Insolvency Limited, 3 Merchant's Quay, Ashley Lane,

Shipley, BD17 7DB By whom Appointed: Members and Creditors

Ag OH123232 (4035435)

Name of Company: HUNSBURY TARMACADAM AND PAVING LTD

Company Number: 09982993

Registered office: 100 St James Road, Northampton, NN5 5LF Principal trading address: 8 Gresham Drive, Northampton, NN4 9SB

Nature of Business: Civil engineering projects Type of Liquidation: Creditors' Voluntary Date of Appointment: 28 March 2022 By whom Appointed: Members and Creditors

Joint Liquidator's Name and Address: Lauren Louise Auburn(IP No. 20894) of BRI Business Recovery and Insolvency, 100 St James Road, Northampton, NN5 5LFTelephone: 01604 754352. Joint Liquidator's Name and Address: John William Rimmer(IP No. 13836) of BRI Business Recovery and Insolvency,

100 St James Road, Northampton, NN5

5LFTelephone: 01604 754352.

For further information contact Rebecca Boston at the offices of BRI Business Recovery and Insolvency on 01604595613, or rboston@briuk.co.uk.

29 March 2022 (4036373)

Name of Company: I VISION CONSTRUCTION LTD

Company Number: 13004536

Registered office: c/o Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, 30 Churchgate, Bolton, BL1

Principal trading address: None

Nature of Business: Construction of domestic buildings

Type of Liquidation: Creditors' Voluntary Date of Appointment: 30 March 2022 By whom Appointed: Members and Creditors

Liquidator's name and address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, Bolton, BL1 1HL. Telephone: 01204 255 051. For further information contact Laura Wright at the offices of

Anderson Brookes Insolvency Practitioners Limited on 01204255051, or laura@andersonbrookes.co.uk.

(4036396) 31 March 2022

Company Number: 09114316

Name of Company: INTEX SITE SOLUTIONS LTD

Nature of Business: Specialised Construction

Registered office: 36 Ruby Street, Wakefield, West Yorkshire, WF1

2GP

Principal trading address: 36 Ruby Street, Wakefield, West Yorkshire,

WF1 2GP

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: Philip David Nunney (IP number 9507) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House 60 Charter Row Sheffield S1 3FZ.

Liquidator's name and address: Joint Liquidator: Joanne Louise Hammond (IP number 17030) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House 60 Charter Row Sheffield S1 3FZ.

Date of Appointment: 29 March 2022

By whom Appointed: Members and Creditors

For further details contact Shannon Jones on 0114 275 5033 or at sheffield.north@btguk.com (4036417)

Pursuant to section 109 of the Insolvency Act 1986 Name of Company: INVESTABLY LIMITED

Company Number: 11707395

Registered office: c/o Begbies Traynor (Central) LLP St James Court

St James Parade Bristol BS1 3LH

Nature of Business: 74909 Other professional, scientific and technical

activities not elsewhere classified

Type of Liquidation: Creditors' Voluntary Liquidation

Liquidator's name and address: Paul Wood , Begbies Traynor (Central) LLP, St James Court, St James Parade, Bristol BS1 3LH and Simon Robert Haskew of Begbies Traynor (Central) LLP, St James

Court, St James Parade, Bristol BS1 3LH Office Holder Number: 009872 and 008988. Date of Appointment: 24 March 2022 By whom Appointed: Members and Creditors

Further Details: Any person who requires further information may contact the joint liquidators by e-mail at bristol@btguk.com or by telephone on 0117 937 7130. Alternatively, enquiries can be made to Clive Hobbs by e-mail at Clive.Hobbs@btguk.com or by telephone on 0117 937 7148.

24 March 2022

Liquidator (4037772) Name of Company: IODETAIL LIMITED

Company Number: 11749343

Nature of Business: Valeting and Detailing Supplies

Registered office: The Old Exchange, 234 Southchurch Road,

Southend-on-Sea, Essex, SS1 2EG Type of Liquidation: Creditors Date of Appointment: 24 March 2022

Liquidator's name and address: Wayne MacPherson (IP No. 9445) and Louise Donna Baxter (IP No. 009123) both of Begbies Traynor (Central) LLP, The Old Exchange, 234 Southchurch Road, Southend-

on-Sea, Essex, SS1 2EG

By whom Appointed: Members and Creditors

Ag OH123349 (4035427)

Name of Company: JKC CONTRACTORS LIMITED

Company Number: 10952395

Nature of Business: Construction - Building Construction

Registered office: C/o Begbies Traynor, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle,

Staffordshire, ST5 9QF Type of Liquidation: Creditors Date of Appointment: 29 March 2022

Liquidator's name and address: Michaela Daly (IP No. 17570) of Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF and David Adam Broadbent (IP No. 009458) of Begbies Traynor (Central) LLP, 11 Clifton Moor Business Village, James Nicolson Link,

Clifton Moor, York, YO30 4XG

By whom Appointed: Members and Creditors Ag OH123259

(4035418)

Name of Company: KEITH STEVENSON AND ASSOCIATES **LIMITED** 

Company Number: 05929481

Nature of Business: Other business support service activities not

elsewhere classified

Registered office: Redheugh House, Teesdale South, Thornaby Place,

Stockton on Tees, TS17 6SG Type of Liquidation: Creditors Date of Appointment: 24 March 2022

Liquidator's name and address: Ian Royle (IP No. 18934) and David Adam Broadbent (IP No. 9458) both of Begbies Traynor (Central) LLP, Redheugh House, Teesdale South, Thornaby Place, Stockton on Tees,

By whom Appointed: Members and Creditors

Ag OH123256 (4035335)

Name of Company: LANGDON TRANSPORT LIMITED

Company Number: 04661798

Nature of Business: Freight transport by road

Registered office: 5 Barnfield Crescent, Exeter, EX1 1QT

Principal trading address: 38 Central Avenue, Lee Mill Industrial

Estate, Ivybridge, PL21 9PE Type of Liquidation: Creditors Date of Appointment: 28 March 2022

Liquidator's name and address: David Gerard Kirk (IP No. 8830) and Daniel Robert Jeeves (IP No. 26032) both of Kirks, 5 Barnfield

Crescent, Exeter, EX1 1QT

By whom Appointed: Members and Creditors

Ag OH123274 (4035572)

Name of Company: LB ALLOYS LIMITED

Company Number: 12274496

Nature of Business: Maintenance and repair of motor vehicles Registered office: 1st Floor, Enterprise House, 202-206 Linthorpe

Road, Middlesbrough, Cleveland, TS1 3QW

Type of Liquidation: Creditors Date of Appointment: 29 March 2022

Liquidator's name and address: David James Robson (IP No. 11130) of Gaines Robson Insolvency Ltd, Carrwood Park, Selby Road, Leeds,

LS15 4LG

By whom Appointed: Creditors

Ag OH123239 (4035601) Name of Company: LITTLEGOLD CONSULTING LIMITED

Company Number: 09458931

Nature of Business: Management consultancy activities other than

financial management

Registered office: White Hart House, Silwood Road, Ascot, SL5 0PY

Type of Liquidation: Creditors Date of Appointment: 29 March 2022

Liquidator's name and address: Ashok Bhardwaj (IP No. 4640) of Bhardwaj Limited, 47/49 Green Lane, Northwood, Middlesex, HA6

By whom Appointed: Creditors

Ag OH123291

(4035422)

Name of Company: LK BUILDS LTD Company Number: 12340206

Nature of Business: Development of building projects

Registered office: Fortis Insolvency Limited, 683-693 Wilmslow Road,

Manchester, M20 6RE Type of Liquidation: Creditors Date of Appointment: 30 March 2022

Liquidator's name and address: Daniel Taylor (IP No. 21050) of Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE

By whom Appointed: Members and Creditors

(4035394)Ag OH123324

Name of Company: M&F PROPERTY MAINTENANCE LTD

Company Number: 11946584

Nature of Business: General building work

Registered office: Fortis Insolvency Limited, 683-693 Wilmslow Road,

Manchester, M20 6RE Type of Liquidation: Creditors Date of Appointment: 29 March 2022

Liquidator's name and address: Daniel Taylor (IP No. 21050) of Fortis

Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE

By whom Appointed: Members and Creditors

(4035539) Ag OH123336

Name of Company: MAK POWER EUROPE LIMITED

Company Number: 09622554

Nature of Business: Agents specialised in the sale of other particular

Registered office: Basepoint Business Centre c/o Meon Accounting Limited, Premier Way, Abbey Park Industrial Estate, Romsey, Hampshire SO51 9AQ (to be changed to: The Old Town Hall, 71

Christchurch Road, Ringwood, BH24 1DH) Type of Liquidation: Creditors Date of Appointment: 29 March 2022

Liquidator's name and address: David Meany (IP No. 9453) and Tauseef A Rashid (IP No. 9718) both of Quantuma Advisory Limited,

The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH

By whom Appointed: The Company

Ag OH123318 (4035495)

Name of Company: MAVENS PROPERTY SERVICES LTD

In Creditors' Voluntary Liquidation ("the Company")

Company Number: 10819507

Previous Name of Company: XPM LIMITED

Nature of Business: Management of real estate on a fee or contract

basis

Type of Liquidation: Creditors

Registered office: 83 Galaxy Buildings Crews Street, London, E14

Principal trading address: 83 Galaxy Buildings Crews Street, London,

E14 3SP

Date of Appointment: 28 March 2022

Liquidator's name and address: Brendan P. Hogan (IP No 13030) of Cromwell & Co Insolvency Practitioners, 5 Mercia Business Village,

Torwood Close, Coventry, CV4 8HX.

(4037800)By whom Appointed: Members and Creditors

Company Number: 07799624

Name of Company: MICK WHEELER CONSTRUCTION LIMITED

Nature of Business: Development of building projects

Registered office: Matrix House, 12-16 Lionel Road, Canvey Island,

Essex, SS8 9DF

Principal trading address: 61 Brentwood Road, Ingrave, Brentwood,

Essex, CM13 3RE

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: Philip David Nunney (IP number 9507) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield

House 60 Charter Row Sheffield S1 3FZ.

Liquidator's name and address: Joint Liquidator: Joanne Louise Hammond (IP number 17030) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House 60 Charter Row Sheffield S1 3FZ.

Date of Appointment: 29 March 2022 By whom Appointed: Members and Creditors

For further details contact Shannon Jones on 0114 275 5033 or at sheffield.north@btguk.com (4036564)

Name of Company: OTI'S RUBBISH COLLECTION LIMITED

Company Number: 11073215

Nature of Business: Collection of waste

Registered office: Olympia House, Armitage Road, London, NW11

Type of Liquidation: Creditors Date of Appointment: 29 March 2022

Liquidator's name and address: Jason Callender (IP No. 22650) of Panos Eliades Franklin & Co, Olympia House, Armitage Road,

London, NW11 8RQ

By whom Appointed: Creditors

Ag OH123258 (4035548)

Name of Company: PINEOAKIO FURNITURE LIMITED

Company Number: 07399873

Nature of Business: Furniture Maker / Kitchen Fitter

Registered office: M.R. Insolvency, 95 Pobox, Heckmondwike, West

Yorkshire WF16 6AU Type of Liquidation: Creditors Date of Appointment: 30 March 2022

Liquidator's name and address: Michael Royce (IP No. 9692) of M.R.

Insolvency, PO Box 95, Heckmondwike, WF16 6AU By whom Appointed: Members and Creditors

Ag OH123359 (4035473)

Name of Company: PROTECTION OF HUMAN RIGHTS IN PUBLIC

LAW Company Number: 07068071

Nature of Business: Solicitors

Registered office: City View House, 5 Union St, Manchester, M12 4JD to be changed to c/o Quantuma Advisory Limited, 14 Derby Road,

Stapleford, Nottingham, NG9 7AA Type of Liquidation: Creditors Date of Appointment: 28 March 2022

Liquidator's name and address: Rehan Ahmed (IP No. 20390) and Tauseef A Rashid (IP No. 9718) both of Quantuma Advisory Limited,

14 Derby Road, Stapleford, Nottingham, NG9 7AA

By whom Appointed: The Company

Ag OH123238 (4035376)

Name of Company: QUINT SOLUTIONS LIMITED

Company Number: 08719104

Nature of Business: Employment placement agencies, recruitment Registered office: Caci House, 9 Spring Villa Park, Spring Villa Road,

Edgware, HA8 7EB Type of Liquidation: Creditors Date of Appointment: 28 March 2022

Liquidator's name and address: Natasha Brodie (IP No. 19490) and Adam Harris (IP No. 15454) both of Mazars LLP, Tower Bridge House,

St Katharine's Way, London, E1W 1DD By whom Appointed: Creditors

Ag OH123244 (4035440) Name of Company: RAVENBROOK (HOLDINGS) LIMITED

Company Number: 13089087

Nature of Business: Activities of distribution holding companies Registered office: Units 8 - 10 Eyston Way, Abingdon, OX14 1TR and it is in the process of being changed to Office D, Beresford House,

Town Quay, Southampton, SO14 2AQ Type of Liquidation: Creditors Date of Appointment: 24 March 2022

Liquidator's name and address: *Andrew Watling* (IP No. 15910) and *Simon Campbell* (IP No. 10150) both of Quantuma Advisory Limited, Office D, Beresford House, Town Quay, Southampton, SO14 2AQ

By whom Appointed: The Company

Ag OH123344 (4035528)

Name of Company: REMGENT LTD

Company Number: 11187146

Nature of Business: Other retail sale not in stores, stalls or markets Registered office: c/o Poppleton & Appleby, The Media Centre, 7

Northumberland Street, Huddersfield, HD1 1RL

Type of Liquidation: Creditors
Date of Appointment: 29 March 2022

Liquidator's name and address: Charles Michael Brook (IP No. 9157) and Allan Christopher Cadman (IP No. 9522) both of Poppleton & Appleby, The Media Centre, 7 Northumberland Street, Huddersfield,

HD1 1RL

By whom Appointed: Members and Creditors

Ag OH123255 (4035500)

Name of Company: RENTONIT LIMITED

Company Number: 08709485

Registered office: 29 Headington Road, London, SW18 3PR Principal trading address: 29 Headington Road, London, SW18 3PR Nature of Business: Information technology consultancy activities

Type of Liquidation: Creditors' Voluntary Date of Appointment: 28 March 2022 By whom Appointed: members and creditors

Liquidator's name and address: Clive Morris (IP No. 8820) of Marshall Peters, Heskin Hall Farm, Wood Lane, Heskin, Preston, PR7 5PA.

Telephone: 01257 452021.

For further information contact Heather Dolan at the offices of Marshall Peters on 01257 452021, or

heatherdolan@marshallpeters.co.uk.

28 March 2022 (4036375)

Name of Company: RISING SUN HOTEL (UK) LTD

Company Number: 11212134 Trading Name: The Rising Sun

Nature of Business: Restaurant and hotel

Registered office: 138 Greenford Road, Sudbury Hill, Harrow, HA1

3QL

Type of Liquidation: Creditors
Date of Appointment: 28 March 2022

Liquidator's name and address: Umang Patel (IP No. 18790) of Neum Insolvency, Suite 9, Amba House, 15 College Road, Harrow,

Middlesex, HA1 1BA

By whom Appointed: Creditors and Members

Ag OH123356 (4035357)

Name of Company: SDI (OXFORD STREET HOF) LIMITED

Company Number: 09798784

Nature of Business: Property holding company

Registered office: Unit A, Brook Park East, Shirebrook, NG20 8RY

Type of Liquidation: Creditors
Date of Appointment: 23 March 2022

Liquidator's name and address: *Michael Lennon* (IP No. 24650) of Kroll Advisory Ltd, The Chancery, 58 Spring Gardens, Manchester, M2 1EW and *Benjamin John Wiles* (IP No. 10670) of Kroll Advisory Ltd,

The Shard, 32 London Bridge Street, London, SE1 9SG

By whom Appointed: Members and Creditors

Ag OH123386 (4035336)

Name of Company: SENTINEL ENTERPRISES LIMITED

Company Number: 07523567

Nature of Business: Other sports activities

Registered office: The Greenhouse, Barnards Way, Lowestoft, NR32

2HD

Type of Liquidation: Creditors
Date of Appointment: 25 March 2022

Liquidator's name and address: Mark Christopher Ford (IP No. 9521) and Kevin Ley (IP No. 25090) both of Smith & Williamson LLP, 25

Moorgate, London, EC2R 6AY By whom Appointed: Creditors

Ag OH123363 (4035386)

Name of Company: **SENTINEL LEISURE TRUST** 

Company Number: 07521701

Nature of Business: Other sports activities

Registered office: The Greenhouse, Barnards Way, Lowestoft, NR32

2HD

Type of Liquidation: Creditors
Date of Appointment: 25 March 2022

Liquidator's name and address: Adam Henry Stephens (IP No. 9748) and Kevin Ley (IP No. 25090) both of Smith & Williamson LLP, 25

Moorgate, London, EC2R 6AY By whom Appointed: Creditors

Ag OH123360 (4035468)

Name of Company: SLT ENTERPRISES LIMITED

Company Number: 11116583

Nature of Business: Other sports activities

Registered office: The Greenhouse, Barnards Way, Lowestoft, NR32

2HD

Type of Liquidation: Creditors
Date of Appointment: 25 March 2022

Liquidator's name and address: Adam Henry Stephens (IP No. 9748) and Mark Christopher Ford (IP No. 9521) both of Smith & Williamson

LLP, 25 Moorgate, London, EC2R 6AY By whom Appointed: Creditors

Ag OH123362 (4035608)

Name of Company: **SLT GROUP** Company Number: 11116012

Nature of Business: Other sports activities

Registered office: The Greenhouse, Barnards Way, Lowestoft, NR32

2HD

Type of Liquidation: Creditors
Date of Appointment: 25 March 2022

Liquidator's name and address: Adam Henry Stephens (IP No. 9748) and Mark Christopher Ford (IP No. 9521) both of Smith & Williamson

LLP, 25 Moorgate, London, EC2R 6AY By whom Appointed: Creditors

Ag OH123361 (4035425)

Company Number: 10901372

Name of Company: SOUTHGATE CATERING LTD

Trading Name: Southgate Kebabs Nature of Business: Takeaway Kebab

Registered office: 601 High Road Leytonstone, London, E11 4PA formerly, Southgate Kebabs, 8a Station Parade, London, N14 5BJ Principal trading address: Southgate Kebabs, 8a Station Parade,

London, N14 5BJ

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: George Michael (IP number 9230) of Ashcrofts, 601 High Road Leytonstone, London, E11 4PA

Liquidator's name and address: Joint Liquidator: *Harjinder Johal* (IP number 9175) of Ashcrofts, 601 High Road Leytonstone, London, E11 4PA

Date of Appointment: 30 March 2022

By whom Appointed: The Members and Creditors

For further details contact Amrit Johal on 020 8556 2888 or at info@ashcrofts.co.uk (4037875)

Pursuant to section 109 of the Insolvency Act 1986 Name of Company: **SPOTLESS CLEAN AND SHINE LTD** 

Company Number: 10866620

Registered office: Rotunda Buildings, Montpellier Exchange,

Cheltenham GL50 1SX

Principal trading address: 5 Oswald Drive, Longford GL2 9FW

Type of Liquidation: Creditors' Voluntary Liquidation

Further Contact Details Maria Navratilova Tel No. 01455 555 444 Liquidator's name and address: *Richard Frank Simms*, Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB

Office Holder Number: 9252. Capacity: Liquidator

Date of appointment: 28 March 2022

By whom appointed: The Members and Creditors

Richard Frank Simms Liquidator

28 March 2022 (4037773)

Name of Company: TAPASWINI LIMITED

Company Number: 08423910 Trading Name: Chowki

Nature of Business: Licensed Restaurants Previous Name of Company: Tapaswani Limited

Registered office: 24 Barnehurst Avenue, Erith, Kent DA8 3NF

Type of Liquidation: Creditors
Date of Appointment: 23 March 2022

Liquidator's name and address: Andrew Mackenzie (IP No. 009581) and Laura Baxter (IP No. 25350) both of Begbies Traynor (Central)

LLP, Unit 8B, Marina Court, Castle Street, Hull, HU1 1TJ

By whom Appointed: Members and Creditors

Ag OH123285 (4035517)

Name of Company: **TH GROUP LTD**Trading Name: T/A Taylor Higson

Company Number: 11997415

Registered office: Reedham House, 31 King Street West, Manchester,

M3 2PJ

Principal trading address: Centurion House, 129 Deansgate,

Manchester, M3 3WR

Nature of Business: Recruitment Type of Liquidation: Creditors' Voluntary Date of Appointment: 22 March 2022 By whom Appointed: Members and creditors

Joint Liquidator's Name and Address: James Fish(IP No. 021390) of Cowgill Holloway Business Recovery LLP, Regency House, 45-53 Chorley New Road, Bolton, BL1 4QRTelephone: 0161 827 1200. Joint Liquidator's Name and Address: Alan Coleman(IP No. 009402) of Cowgill Holloway Business Recovery LLP, Regency House, 45-53 Chorley New Road, Bolton, BL1 4QRTelephone: 0161 827 1200.

For further information contact Ben Gray at the offices of Cowgill Holloway Business Recovery LLP on 0161 672 5767, or Ben.Gray@cowgills.co.uk.

29 March 2022 (4036381)

RULE 4.106A OF THE INSOLVENCY RULES 1986 (AS AMENDED) Name of Company: **THE BEAR HOTEL AT WANTAGE LTD** 

Company Number: 07431339
Trading Name: The Bear Hotel

Registered office: The Bear Hotel, Wantage, Oxfordshire OX12 8AB Principal trading address: The Bear Hotel, Wantage, Oxfordshire

OX12 8AB

Name of Office Holder: Peter Richard James Frost

Office Holder IP Number: 008935

Postal Address of Office Holder: Staverton Court, Staverton,

Cheltenham, GL51 0UX

Office holder's telephone no and email address 01242 680000 and

creditors@hazlewoods.co.uk

Capacity of office holder(s) Liquidator Date of Appointment: 22 March 2022

Alternative contact for enquiries on proceedings Gina Clare

Peter Richard James Frost was appointed Liquidator of The Bear Hotel at Wantage Ltd Trading As The Bear Hotel on 22 March 2022 by Members and Creditors. (4037770)

Name of Company: THE BUSINESS PORTFOLIO UK LTD

Company Number: 06145179

Nature of Business: Training & Development

Type of Liquidation: Creditors

Registered office: Saxon House, Saxon Way, Cheltenham GL52 6QX Principal trading address: 15 Poplar Road, Oxton, Prenton,

Merseyside, CH43 5TB

Liquidator's name and address: Alisdair J Findlay, of Findlay James,

Saxon House, Saxon Way, Cheltenham GL52 6QX.

Office Holder Number: 008744.

For further details contact: Alisdair J Findlay Email:

info@findlayjames.co.uk Tel: 01242 576555 Date of Appointment: 30 March 2022

By whom Appointed: Members and Creditors (4036177)

Name of Company: THE CASPI PUB COMPANY LIMITED

Company Number: 08316478

Registered office: 59 Dexter Close, Luton, LU3 4DY Principal trading address: 59 Dexter Close, Luton, LU3 4DY

Nature of Business: Public House Type of Liquidation: Creditors' Voluntary Date of Appointment: 28 March 2022 By whom Appointed: Members & Creditors

Joint Liquidator's Name and Address: Alessandro Sidoli(IP No. 14270) of Kay Johnson Gee Corporate Recovery Limited, 1 City Road East, Manchester,

M15 4PNTelephone: 0161 832 6221. Joint

Liquidator's Name and Address: Peter Anderson(IP No. 15336) of Kay Johnson Gee Corporate Recovery Limited, 1 City Road East, Manchester, M15

4PNTelephone: 0161 832 6221. Joint Liquidator's Name and Address: Alan Fallows(IP No. 9567) of Kay Johnson Gee Corporate Recovery Limited, 1 City Road East, Manchester, M15 4PNEmail: alanfallows@kjqcr.com. Telephone: 0161 832 6221.

For further information contact Lindsay Moore at the offices of Kay Johnson Gee Corporate Recovery Limited on 0161 212 8388, or lindsaymoore@kjgcr.com.

29 March 2022 (4036363)

# Name of Company: THE CLOSED CIRCUIT TELEVISION COMPANY LIMITED

Company Number: 03264312

Nature of Business: Installation of industrial machinery and equipment Registered office: 75 Haverstock Hill, Hampstead, London, NW3 4SL

Type of Liquidation: Creditors
Date of Appointment: 30 March 2022

Liquidator's name and address: Mark Reynolds (IP No. 8838) of Valentine & Co, 1st Floor Galley House, Moon Lane, Barnet, EN5 5YL

By whom Appointed: Members and Creditors

Ag OH123329 (4035489)

Name of Company: THE CURTAIN LADY INTERIORS LIMITED

Company Number: 07356098

Henwood, Ashford, TN24 8DH

Nature of Business: Retail of Furniture & Lighting

Registered office: Gateway House, Highpoint Business Village,

Type of Liquidation: Creditors
Date of Appointment: 29 March 2022

Liquidator's name and address: Ian Yerrill (IP No. 8924) of Yerrill Murphy, Gateway House, Highpoint Business Village, Henwood,

Ashford, Kent, TN24 8DH

By whom Appointed: Members and Creditors

Ag OH123230 (4035345)

Name of Company: THE EDGBASTON HOTEL LIMITED

Company Number: 08151595 Nature of Business: Hotel

Registered office: Highdown House, 11 Highdown Road, Leamington

Spa, CV31 1XT

Type of Liquidation: Creditors
Date of Appointment: 23 March 2022

Liquidator's name and address: Office Holder Details: Gavin Bates (IP No. 8983) of Smart Business Recovery Limited, The Old Rectory, Main

Street, Glenfield, Leicester, LE3 8DG By whom Appointed: Members and Creditors

Aq OH123385 (4035406)

Name of Company: THE EPSOM EXPLORE COACH CO LTD.

Company Number: 10717959 Trading Name: Epsom Coaches

Nature of Business: Chartered coach transport operator Previous Name of Company: Epsom Coaches Limited

Registered office: c/o Harveys Insolvency & Turnaround, 2 Old Bath

Road, Newbury, Berkshire RG14 1QL Type of Liquidation: Creditors Date of Appointment: 30 March 2022

Liquidator's name and address: Debi Harvey (IP No. 12150) of Harveys Insolvency & Turnaround Limited, 2 Old Bath Road, Newbury,

Berkshire, RG14 1QL

By whom Appointed: Members and Creditors

Ag OH123300 (4035498)

Company Number: 07241477

Name of Company: THE GRAINSTORE BREWERY LIMITED

Nature of Business: Manufacture of beer

Registered office: 110 Regent Road, Leicester LE1 7LT

Principal trading address: Station Approach, Oakham, Rutland, LE15

6RE

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Liquidator: Neil Charles Money (IP number 8900) of CBA Business Solutions, 126 New Walk, Leicester,

LE1 7JA.

Date of Appointment: 30 March 2022

By whom Appointed: The Members and Creditors

For further details contact CBA Business Solutions on 0116 262 6804

(4037886) Princ

Name of Company: THE RETREAT NEWPORT LTD

Company Number: 09411765

Trading Name: The Retreat Cafe (until January 2020) Nature of Business: Unlicensed restaurants and cafes

Registered office: 41 St Michaels Road, St Helens, Ryde, Isle of

Wight, PO33 1YJ
Type of Liquidation:

Type of Liquidation: Creditors
Date of Appointment: 29 March 2022

Liquidator's name and address: Chris Tate (IP No. 22910) and Duncan Swift (IP No. 8093) both of Azets, Secure House, Lulworth Close,

Chandlers Ford, Southampton, SO53 3TL By whom Appointed: Members and Creditors

Ag OH123327 (4035606)

Name of Company: THE TEACHING ASSISTANT COLLEGE LIMITED

Company Number: 05840111

Nature of Business: Other education not elsewhere classified Registered office: c/o Begbies Traynor (Central) LLP, Lymedale

Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF

Type of Liquidation: Creditors
Date of Appointment: 30 March 2022

Liquidator's name and address: *Michaela Daly* (IP No. 17570) of Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF and *Louise Longley* (IP No. 23874) of Begbies Traynor (Central) LLP, Fourth Floor, Toronto Square, Toronto Street, Leeds, LS1 2HJ

By whom Appointed: Members and Creditors

Ag OH123376 (4035514)

Company Number: 09916737

Name of Company: THOMPSON & SHELL TRANSPORT LIMITED

Nature of Business: Road haulage, shipping containers Registered office: 10 St Helen's Road Swansea SA1 4AW

Principal trading address: 1 Mariners Cottages South Shields NE33

2NG

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: Gareth Bishop (IP number 17870) of McAlister & Co Insolvency Practitioners Limited, 10

St Helens Road, Swansea, SA1 4AW.

Liquidator's name and address: Joint Liquidator:  $Helen\ Whitehouse$  (IP number 9680) of McAlister & Co Insolvency Practitioners Limited,

10 St Helens Road, Swansea, SA1 4AW. Date of Appointment: 29 March 2022 By whom Appointed: Members and Creditors

For further details contact Gareth Bishop on 03300 563600 or at gareth@mcalisterco.co.uk (4036576)

Name of Company: **UNDER THE ROCK LTD** 

Company Number: 12360606 Nature of Business: Bar Management

Registered office: 33 Harrison Road, Halifax, HX1 2AF (to be changed to XL Business Solutions, Premier House, Bradford Road,

Cleckheaton, BD19 3TT)
Type of Liquidation: Creditors
Date of Appointment: 28 March 2022

Liquidator's name and address: J N Bleazard (IP No. 009354) of XL Business Solutions Limited, Premier House, Bradford Road,

Cleckheaton, BD19 3TT

By whom Appointed: Members and Creditors

Ag OH123296 (4035565)

Name of Company: VIRTUAL DISTRIBUTION LIMITED

Company Number: 07848768

Registered office: C/O Ideal Corporate Solutions Limited, Lancaster

House, 171 Chorley New Road, Bolton, BL1 4QZ

Principal trading address: Unit 1 Magnesium Way, Hapton, Burnley,

BB12 7BF

Type of Liquidation: Creditors' Voluntary Date of Appointment: 23 March 2022 By whom Appointed: Members and Creditors

Joint Liquidator's Name and Address: Andrew David Rosler(IP No. 9151) of Ideal Corporate Solutions Limited, Lancaster House, 171 Chorley New Road,

Bolton, BL1 4QZJoint Liquidator's Name and Address: Tom Bowes(IP No. 17010) of Ideal

Corporate Solutions Limited, Lancaster House, 171 Chorley New Road, Bolton, BL1 4QZ

For further information contact Michael Walsh at the offices of Ideal Corporate Solutions Limited on 01204 663000, or michael.walsh@idealcs.co.uk.

25 March 2022 (4036367)

Name of Company: XANTIVE LIMITED

Company Number: 03891544

Nature of Business: Software Service & Support

Previous Name of Company: PTCD Limited; Planthinkcheckdo

Limited; Identinet Limited

Registered office: Lifford Hall, Lifford Lane, Kings Norton,

Birmingham, West Midlands B30 3JN Type of Liquidation: Creditors Date of Appointment: 28 March 2022

Liquidator's name and address: Adam Peter Jordan (IP No. 9616) and Nickolas Garth Rimes (IP No. 009533) both of Rimes & Co, 3 The Courtyard, Harris Business Park, Hanbury Road, Stoke Prior,

Bromsgrove B60 4DJ

By whom Appointed: Members and Creditors

Ag OH123248 (4035409)

#### **DEEMED CONSENT (CVL)**

#### **COUNTY CAPITAL WEALTH MANAGEMENT LIMITED**

Company Number: (06049063)

Registered office: 79 Caroline Street, Birmingham B3 1UP

Principal trading address: Prosperity House, 1 Barnfield Road,

Spennymoor, DL16 6EB

Notice is given by Roderick Graham Butcher to the creditors of County Capital Wealth Management Limited that the following decision is sought from them by way of a Deemed Consent Procedure, for the appointment of Roderick Graham Butcher of Butcher Woods Limited as Liquidator.

To object to this decision, creditors must send to me at my offices at Butcher Woods Limited, 79 Caroline Street, Birmingham, B3 1UP a duly completed notice of objection by no later than 23.59 hours on 14 April 2022, the Decision Date, together with proof of their debt, (if not already been lodged) without which objections will be invalid.

Unless 10% of the creditors who would be entitled to vote at a qualifying decision procedure object to this decision, then it will be automatically approved on 14 April 2022, the Decision Date.

If no person is nominated by the creditors, then the existing liquidator will be the liquidator in the creditors' voluntary winding up.

Date of Appointment: 14 April 2022

Office Holder Details: Roderick Graham Butcher (IP No. 8834) of Butcher Woods, 79 Caroline Street, Birmingham, B3 1UP

For further details contact Jim Goudie by telephone on 0121 236 6001, or by email at jim.goudie@butcher-woods.co.uk

R G Butcher, Liquidator

29 March 2022 Ag OH123277

(4035432)

# **MEETINGS OF CREDITORS**

## 22M LIMITED

(Company Number 04582550)

Registered office: Huxley House, 11 William Street, Redditch, B97 4AJ Principal trading address: Block A, Unit 1 Imex Space Business Centre, Oxleasow Road, East Moons Moat, Redditch, B98 0RE

NOTICE IS HEREBY GIVEN that the directors of the company are convening a virtual meeting of creditors to be held on 11 April 2022 at 10.30am, for the purpose of deciding on the nomination of a liquidator. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information, please contact Tim Heaselgrave at The Timothy James Partnership Limited on 01527 314050 or email tim@timothyjamespartnership.co.uk.

In order to be entitled to vote creditors must deliver proofs by 4 pm on the business day prior to the meeting to The Timothy James Partnership, Darwin House, 7 Kidderminster Road, Bromsgrove, B61 7JJ. Proxies must be delivered to the convenor or chair before they may be used at the meeting.

Creditors are to note, resolutions regarding the Liquidator's fees will be voted on at the meeting.

NOTE: the meeting may be suspended or adjourned by the chair of the meeting (and must be adjourned if it is so resolved at the meeting)

Nominated Liquidator: *Timothy James Heaselgrave* (IP number 9193) of The Timothy James Partnership Limited, Darwin House, 7 Kidderminster Road, Bromsgrove, B61 7JJ.

For further details contact Paul Childs on 01527 314050 or at

paul@timothyjamespartnership.co.uk

Mark Richard Webb, Director (4037937)

#### 3F LTD

(Company Number 10948302)
Trading Name: Creams Cafe

Registered office: Unit 83 The Stratford Centre, London, E15 1XA Principal trading address: Unit 83 The Stratford Centre, London, E15 1XA

NOTICE IS HEREBY GIVEN under Section 100 of the Insolvency Act 1986 and Rule 6.14 and 15.8 of the Insolvency (England & Wales) Rules 2016 that a virtual meeting of the creditors of the above-named company is being proposed by Ahmed Furquan, the Director of the Company in accordance with resolutions passed by the Board of Directors

The virtual meeting will be held as follows:

Time: 11:15am Date: 11 April 2022

To access the virtual meeting, which will be held via telephone conferencing platform, contact the convener - details below.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ("proof"), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be delivered by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, a resolution specifying the terms on which the Liquidators are to be remunerated, and the meeting may receive information about, or be called upon to approve, the costs of convening the meeting and assisting in the preparation of associated documentation. If there are no non-associate votes received in relation to the approval of Liquidator's fees and expenses then associate or connected party votes admitted for voting purposes will be taken into account.

Engin Faik (IP No. 9635) is qualified to act as an Insolvency Practitioner in relation to the above company and a list of names and addresses of the Company's creditors will be available for inspection at Cornerstone Business Recovery and Turnaround Limited, 136 Hertford Road, Enfield, Middlesex EN3 5AX on the two business days preceding the meeting.

For further details contact Eren Faik on 020 3793 3338 or at eren@cornerstonerecoverv.co.uk

Ahmed Furquan, Director and Convener

(4036503)

#### **BUSY BISHTONS LTD**

(Company Number 08987874)

Registered office: 18 Galloway Road, Hamworthy, Poole, Dorset, BH15 4JX

Principal trading address: 18 Galloway Road, Hamworthy, Poole, Dorset, BH15 4JX

Notice is hereby given, pursuant to RULE 15.13 of the Insolvency (England and Wales) Rules 2016, that the Director of the abovenamed Company (the 'convener') is seeking a decision from creditors on the nomination of a Liquidator by way of a virtual meeting. A resolution to wind up the Company is to be considered on 12 April 2022

The meeting will be held as a virtual meeting by telephone conference on 12 April 2022 at 2.15 pm. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated liquidator using the details below.

Natalie Hughes of Simply Corporate Ltd, The Town Hall, 83 Burnley Road, Padiham, BB12 8BS will provide reasonable information to creditors during business hours on the two business days prior to the meeting date upon receipt of requests for such information.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to Simply Corporate Limited, The Town Hall, 83 Burnley Road, Padiham, BB12 8BS.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4pm on the business day before the meeting date (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to Simply Corporate Limited, The Town Hall, 83 Burnley Road, Padiham, BB12 8BS or emailed to: info@simplycorporate.co.uk

Name and address of nominated Liquidator: Natalie Hughes (IP No. 14336) of Simply Corporate Ltd, The Town Hall, 83 Burnley Road, Padiham, BB12 8BS.

Further details contact: Natalie Hughes, Tel: 01282 222420, Email: info@simplycorporate.co.uk. Alternative contact: Catherine Unsworth. Stephen Leslie Bishton, Director

30 March 2022

Ag OH123250

(4035415)

#### **CARLTON ENGINEERING WIRRAL LIMITED**

(Company Number 08372250)

Registered office: 3 Platt Grove, Rock Ferry, Wirral, CH42 1QH

Principal trading address: 3 Platt Grove, Rock Ferry, Wirral, CH42

Nature of Business: Engineering design activities for industrial process and production.

Type of Liquidation: Creditors' Voluntary.

Place of meeting: Kay Johnson Gee Corporate Recovery Limited, 1

City Road East, Manchester, M15 4PN.

Date of meeting: 7 April 2022. Time of meeting: 11:00 am.

Notice is hereby given under Section 100 of the Insolvency Act 1986 and Rules 6.14 and 15.8 of the Insolvency (England and Wales) Rules 2016 that a physical meeting of the creditors of the above Company is being convened.

The prescribed proportion of creditors have objected to decisions proposed by deemed consent and therefore the deemed consent has been superseded. The physical meeting will be held at the place, date and time specified in this notice.

A meeting of shareholders has been called and will be held prior to the physical meeting of creditors passing a resolution for the voluntary winding up of the Company.

Creditors wishing to vote at the meeting must lodge their proxy, together with a full statement of account at the registered office, or with the Insolvency Practitioners not later than 4pm on the last business day preceding the meeting. For the purposes of voting, any secured creditors are required (unless they surrender their security) to lodge a statement with the Insolvency Practitioners prior to the meeting giving particulars of their security, the date when it was given and its assessed value.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of Joint Liquidator, a resolution specifying the terms on which the Joint Liquidators are to be remunerated, and the meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting.

By Order of the Board

Kevin McCann, Director

Joint Insolvency Practitioner's Name and Address: *Alan Fallows* (IP No. 9567) of Kay Johnson Gee Corporate Recovery Limited, 1 City Road East, Manchester, M15 4PN. Telephone: 0161 832 6221.

Joint Insolvency Practitioner's Name and Address: *Alessandro Sidoli* (IP No. 14270) of Kay Johnson Gee Corporate Recovery Limited, 1 City Road East, Manchester, M15 4PN. Telephone: 0161 832 6221. Joint Insolvency Practitioner's Name and Address: *Peter Anderson* (IP No. 15336) of Kay Johnson Gee Corporate Recovery Limited, 1 City

Road East, Manchester, M15 4PN. Email: peteranderson@kjgcr.com. Telephone: 0161 832 6221.

For further information contact Nicola Melling at the offices of Kay Johnson Gee Corporate Recovery Limited on 0161 212 8421, or nicolamelling@kjgcr.com.

29 March 2022 (4036390)

#### **CLEAR COAT NORTH LTD**

(Company Number 11492696)

Trading Name: Mobile Outfitters

Registered office: 17 Spring Avenue, Ashby-de-la-Zouch,

Leicestershire, LE65 2RB

Principal trading address: 17 Spring Avenue, Ashby-de-la-Zouch, Leicestershire, LE65 2RB

NOTICE IS HEREBY GIVEN that the directors of the company are convening a virtual meeting of creditors to be held on 13 April 2022 at 10.15 am, for the purpose of deciding on the nomination of a liquidator.

Creditors can access the virtual meeting as follows: Please contact Thomas Bowen of MB Insolvency on the below contact details at least one business day prior to the virtual meeting which will be held via Zoom.

In order to be entitled to vote creditors must deliver proxies and proofs to MB Insolvency, 11 Roman Way, Berry Hill, Droitwich, WR9 9AJ. Proofs must be received by 4pm on the business day before the virtual meeting. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy form with the Liquidator before they may be used at the meeting.

NOTE: the meeting may be suspended or adjourned by the chair of the meeting (and must be adjourned if it is so resolved at the meeting).

For further details contact Thomas Bowen on 01905 776 771 or at thomasbowen@mb-i.co.uk (4037864)

#### **ELKADOR FINANCE LIMITED**

(Company Number 03057410)

Registered office: Griffins, Tavistock House South, Tavistock Square, London, WC1H 9LG

Principal trading address: Top Floor 14 Dunbar Road, Bournemouth, Dorset, BH3 7BA

Notice is hereby given, pursuant to Rule 15.13 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Liquidator of the above-named Company (the 'convener') is seeking a decision from creditors on the formation of a Liquidation Committee or/and increasing the basis of the Liquidator's remuneration by way of a virtual meeting.

The meeting will be held as a virtual meeting by Microsoft Teams on 14 April 2022 at 12.00 noon. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the Liquidator using the details below.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. A continuing proxy must be delivered to the Liquidator may be exercised at any meeting which begins after the proxy is delivered. Proxies may be delivered to Griffins, Tavistock House South, Tavistock Square, London, WC1H 9LG.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4.00 pm on 13 April 2022 (unless the chair of the meeting is content to accept the proof later). A creditor who has opted out from receiving notices may nevertheless vote if the creditor provides a proof of debt in the requisite time frame. Proofs may be delivered to Griffins, Tavistock House South, Tavistock Square, London, WC1H 9LG.

Date of Appointment: 6 March 2018

Office Holder Details: Stephen Hunt (IP No. 9183) of Griffins, Tavistock House South, Tavistock Square, London WC1H 9LG

For further details contact: Vasiliki Nikoletopoulou, Email: Vasiliki.nikoletopoulou@griffins.net

Stephen Hunt, Liquidator

30 March 2022

Ag OH123377

(4035600)

#### **FORBES JOINERY LIMITED**

(Company Number 11492121)

Registered office: 17 Burnville Road, Sunderland, SR4 7LY

Principal trading address: Unit 1, 147 Stockton Street, Middlesbrough, Teesside TS2 1BU

Notice is hereby given, pursuant to Section 100 of the INSOLVENCY ACT 1986 ('IA 1986') and Rules 6.14 and 15.8 of the Insolvency (England and Wales) Rules 2016 that the Director (the 'convener') is seeking a decision by way of a virtual meeting of the creditors of the above-named Company.

This virtual meeting will be held on 13 April 2022 at 11.15 am by video conference platform. Details of how to attend the virtual meeting is included in the notice delivered to creditors. If any creditor has not received this notice or requires further information including how to attend, please contact Stephen O'Hara of O'Haras Limited using the contact details below.

The decisions being sought from creditors at the meeting will include considering resolutions for the appointment by creditors of a liquidator and a resolution specifying the terms on which the liquidator is to be remunerated. The meeting will receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting.

A meeting of shareholders has been called and will be held immediately prior to the virtual meeting of creditors to consider passing resolutions for the voluntary winding up of the Company and appointment of the proposed liquidator, Jeremy Bennett.

A creditor is entitled to attend and/or vote at the virtual creditors meeting in person or by proxy. Creditors wishing to attend and/or vote at the meeting must lodge their proxy at O'Haras Limited, Moorend House, Snelsins Lane, Cleckheaton, BD19 3UE or electronically at soh@oharas.co prior to the meeting. A proof of debt (unless it has already been delivered) must also be lodged not later than 4.00pm on the business day prior to the date of the meeting at the aforementioned address (note if not received in time, vote by creditor will be disregarded).

Notice is further given that a list of the names and addresses of the Company's creditors may be inspected, free of charge, on the two business days prior to the virtual meeting at O'Haras Limited, Moorend House, Snelsins Lane, Cleckheaton, West Yorkshire, BD19 3UE.

If necessary please contact Jeremy Bennett, (IP No. 8776), who is licensed to act as Insolvency Practitoner in relation to the above Company, by email at soh@oharas.co or telephone on 01274 800380. Ross Doyle, Director

29 March 2022

Ag OH123303 (4035470)

#### **GOLDEN ROSE ENTERPRISES LIMITED**

(Company Number 12270340)

Registered in England and Wales

Registered office: 27 The Fountains Ballards Lane, London N3 1NL Principal trading address: 27 The Fountains Ballards Lane, London N3 1NL

IN THE MATTER OF GOLDEN ROSE ENTERPRISES LIMITED AND IN THE MATTER OF THE INSOLVENCY ACT 1986

NOTICE IS HEREBY GIVEN under Section 100 of the Insolvency Act 1986 and RULE 6.14 AND 15.8 OF THE INSOLVENCY (ENGLAND & WALES) RULES 2016 that a virtual meeting of the creditors of the above-named company have been convened by Margareta Mardare , the director of the company in accordance with resolutions passed by the Board of Directors.

The virtual meeting will be held as follows:

Time: 11.15 am Date: 19 April 2022

To access the virtual meeting, which will be held via skype conferencing platform, please contact Mansoor Mubarik on behalf of the convener - detail below:

Name: Mansoor Mubarik of Capital Books (UK) Limited Address: 66 Earl Street, Maidstone, Kent ME14 1PS

Contact Number: 01622 754927 Email ID: mmubarik@capital-books.co.uk This virtual meeting of creditors may be recorded (video and/or audio) in order to establish and maintain records of the existence of relevant facts, or decisions that are taken at such meetings. By attending this meeting, you consent to being recorded, including possible recordings of your facial image. Where any recording of a meeting also entails the processing of personal data, such personal data shall be treated in accordance with the Data Protection Act 2018.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be delivered by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting.

Mr Mansoor Mubarik ACA FCCA FABRP (office holder no 009667) is qualified to act as Insolvency Practitioner in relation to the above company and a list of names and addresses of the company's creditors will be available for inspection at 66 Earl Street, Maidstone, Kent, ME14 1PS on the two business days preceding the meeting.

In case of queries, please contact Hassan Kaleem on 02088523707 or email mail@capital-books.co.uk

Dated 21 March 2022

Margareta Mardare Director and Convener

(4037731)

#### INTERNATIONAL MEDICAL RESCUE LIMITED

(Company Number 08344325)

Previous Name of Company: Just Good Limited

Registered office: 60 Wensleydale Road, Hampton, TW12 2LX

Principal trading address: East Links, 15 Tollgate, Eastleigh, SO53 3TG

Notice is hereby given under Section 100 of the Insolvency Act 1986 and Rules 6.14 and 15.13 of the Insolvency (England and Wales) Rules 2016 that the director of the Company ('the convener') is seeking a decision from creditors on the nomination of joint liquidators by way of a virtual meeting. In addition, the meeting may be asked to consider resolutions regarding the joint liquidator's pre and post appointment remuneration. The virtual meeting will be held on 7 April 2022 at 10.30 am.

For details of how to join the meeting, please contact the nominated liquidator's office using the details below.

In order to vote at the virtual meeting, creditors must deliver a form of proxy to the chair before the commencement of the virtual meeting. For any vote to be counted it must be accompanied by a proof of debt in respect of the creditors claim (unless one has already been submitted) and received by the chair by 4pm on the business day before the virtual meeting. Proxy forms and proof of debt forms can be delivered to 67 Grosvenor Street, Mayfair, London, W1K 3JN or by email using the details below.

Further details contact: Tel: 0207 769 6831, Email: team@voscab.co.uk

Richard John Gildea, Director

7 March 2022 Ag OH123345

(4035509)

# IR35 SPECIALISTS UK LTD

(Company Number 09365012)

Registered office: 4 Admiral Way, Doxford International Business Park, Sunderland. SR3 3XW

Principal trading address: 4 Admiral Way, Doxford International Business Park, Sunderland, SR3 3XW

Notice is hereby given, pursuant to Rules 6.14, 6.19, 15.13 & 15.8 of the Insolvency (England and Wales) Rules 2016, that the Board of Directors of the above-named Company (the 'conveners') are seeking a decision from creditors on the nomination of a Liquidator and whether to establish a Liquidation Committee by way of a virtual meeting. A resolution to wind up the Company is to be considered on 12 April 2022.

In the event no Liquidation Committee is established, other resolutions will be put to the meeting to determine the basis of the nominated Liquidator 's remuneration, drawing of pre appointment fees and expenses and authorisation of the drawing of category 2 disbursements.

The meeting will be held as a virtual meeting by Telephone Conference on 12 April 2022 at 11:00 am. If a creditor requires access details to the telephone conference or requires any other information, please contact the nominated Liquidator using the details below.

The nominated Liquidator is a person qualified to act as an insolvency practitioner in relation to the company who, during the period before the meeting date, will furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require. Their contact details are below.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to c/o Fortis Insolvency Limited, 683-693 Wilmslow Road, Didsbury, Manchester M20 6RE.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4pm on the business day before the meeting date (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to the nominated Liquidator using the details below.

Name and address of nominated Liquidator: Daniel Taylor (IP No. 21050) of Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE

Further details contact: Daniel Taylor, Tel: 0161 694 9955. Alternative contact: Matthew Bannon.

Ryan Duncan Angus, Director 30 March 2022

30 March 2022 Ag OH123328

(4035551)

# ITC TAXIS LIMITED

(Company Number 02190331)

Registered office: Mountview Court, 1148 High Road, Whetstone, London N20 0RA

Principal trading address: 2a Three Colts Lane, London, E2 6JL

Notice is hereby given that a virtual meeting of the creditors of the above-named Company is being convened by Mr Ian Currans, to be held on 7 April 2022 at 11.30 am for the purpose provided for in section 100 of the Insolvency Act 1986.

Creditors entitled to attend and vote at the virtual meeting may do so personally or by proxy. A creditor can attend the virtual meeting in person and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4 p.m. on the business day before the meeting. If a creditor cannot attend in person, or does not wish to attend but still wishes to vote at the meeting, they can either nominate a person to attend on their behalf, or they may nominate the Chair of the meeting, who will be a director of the Company, to vote on their behalf. Creditors must deliver their proxy by no later than the commencement of the meeting. Creditors must deliver all proofs of their debt and proxies to Kallis Insolvency Practitioners, Mountview Court, 1148 High Road, Whetstone, London, N20 ORA. Creditors failing to lodge a proof of their debt or proxy as indicated will lead to their vote(s) being disregarded.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

At the meeting, creditors may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting of creditors.

Kikis Kallis (IP No. 004692) of Kallis Insolvency Practitioners, is qualified to act as an Insolvency Practitioner in relation to the Company and will, during the period before the day of the meeting, furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require. Creditors can contact Kallis Insolvency Practitioners on 020 8446 6699 or by email at eleni@kallis.co.uk.

lan Currans, Director 14 March 2022 Ag OH123228

(4035369)

#### JOHNSONS DEVELOPMENTS LTD

(Company Number 08598440)

Registered office: Nova House Hall Farm Way, Smalley, Ilkeston, DE7 6.IS

Principal trading address: Nova House Hall Farm Way, Smalley, Ilkeston, DE7 6JS

NOTICE IS HEREBY GIVEN that the directors of the Company are convening a meeting of creditors to be held at Trinity House, 28 - 30 Blucher Street, Birmingham, B1 1QH on 8 April 2022 at 11.15 am, for the purpose of deciding on the nomination of a liquidator. Decisions regarding the Joint Liquidators remuneration and the formation of a liquidation committee may also be sought at the meeting.

In order to be entitled to vote creditors must deliver proxies and proofs to Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham, B1 1QH by 16:00 on the business day before the meeting

Nominated Liquidator: *Sajid Sattar* (IP number 15590) of Greenfield Recovery Limited,, Trinity House, 28-30 Blucher Street, Birmingham, B1 1QH.

Nominated Liquidator: Simon Gwinnutt (IP number 8877) of Greenfield Recovery Limited,, Trinity House, 28-30 Blucher Street, Birmingham, B1 1QH.

For further details contact Glenda Hassall on 0121 201 1720 or at g.h@greenfieldrecovery.co.uk

Jeremy Robert Johnson, Convener (4036309)

#### JUMBO SKIPS AND FABRICATIONS LIMITED

(Company Number 08873459)

Registered office: 25 Ford Street, Silverdale, Newcastle, ST5 6LT Principal trading address: Jumbo Yard, Turner Crescent, Newcastle-under-Lyme, Staffordshire, ST5 7JZ

Notice is hereby given pursuant to Section 100 OF THE INSOLVENCY ACT 1986 and Rules 6.14 and 15.8 of the Insolvency (England & Wales) Rules 2016 ("the Rules") that a Virtual Meeting of the Creditors of the above named Company is being proposed by Jonathan David Bowden the director of the Company in accordance with resolutions passed by the Board of Directors. The virtual meeting will be held on 8 April 2022 at 12.15 pm. A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company. The Virtual meeting will be held using ZOOM. Contact Vincent McBrien for password and logon details. The meeting may be suspended or adjourned (and must be adjourned if so resolved at the meeting). The proposed liquidator during the period before the decision date, will furnish creditors free of charge with such information concerning the company's affairs as they may reasonably require.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be delivered to the convener no later than 4.00 pm the business day before the meeting. Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a Liquidator, a resolution specifying the terms on which the Liquidator is to be remunerated, and the meeting may receive information about, or be called up to approve, the costs of preparing the statement of affairs and convening the meeting.

Name and address of Insolvency Practitioner calling the meeting: *Jonathan Mark Taylor* (IP No. 10570) of T H Financial Recovery, Suites 101 & 102, Empire Way Business Park, Liverpool Road, Burnley, BB12 6HH

Further details contact: Vincent McBrien, Email: vmcbrien@thfr.co.uk, Tel: 01282 332222.

Jonathan David Bowden, Director 29 March 2022 Ag OH123252

(4035482)

#### L & O (MANCHESTER) LIMITED

(Company Number 11762630)

Registered office: 883 Belmont Road, Bolton, BL1 7BT

Principal trading address: 883 Belmont Road, Bolton, BL1 7BT

NOTICE IS HEREBY GIVEN THAT Pursuant to Section 100 of the Insolvency Act 1986, a virtual meeting of L & O (Manchester) Limited creditors has been summoned by the director(s) of the Company in accordance with resolutions passed by the Board of Directors. The virtual meeting will be held on 11 April 2022 at 11:15 am.

The purpose of the virtual meeting is to:

- Seek agreement from the creditors of the appointment of the members' Joint Liquidators;
- Invite creditors to appoint a creditors' committee and if no committee is appointed;
- To consider a resolution fixing the basis of the Joint Liquidators' remuneration and authorising payment of Category 2 expenses to be charged in accordance with the firm's policy;
- To consider a resolution to confirm that the Joint Liquidators act jointly and separately; and
- The attendees of the virtual meeting may also receive information about, and be called upon to approve, the costs of preparing the Statement of Affairs and convening the meeting for KBL Advisory Limited and Xeinadin Group.

Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting. The convener can be contacted care of Adam Horan, KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH or Adam@kbl-advisory.com / 0161 637 8100.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless their written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be lodged by 4.00 pm on the business day before the meeting. Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

Steve Kenny (IP no 24030) and Richard Cole (IP no 26070) are qualified to act as insolvency practitioners in relation to the Company, and a list of names and addresses of the Company's creditors will be available for inspection at the offices of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH on the two business days preceding the meeting.

Miguel Estevao, Director / Convener of Meeting (4037279)

# MAXHAN DEVELOPMENTS LIMITED

(Company Number 07509439)

Registered office: c/o AJC Accounting, Moorend House, Snelsins Road, Cleckheaton, West Yorkshire, BD19 3UE

Principal trading address: 29 Staincliffe Hall Road, Batley, West Yorkshire, WF17 7QX

Notice is hereby given, pursuant to Section 100 of the INSOLVENCY ACT 1986 ('IA 1986') and Rules 6.14 and 15.8 of the Insolvency (England and Wales) Rules 2016 that the Director (the 'convener') is seeking a decision by way of a virtual meeting of the creditors of the above-named Company.

This virtual meeting will be held on 14 April 2022 (the 'Decision Date') at 10.15 am by video conference platform. Details of how to attend the virtual meeting is included in the notice delivered to creditors. If any creditor has not received this notice or requires further information including how to attend, please contact Stephen O'Hara of O'Haras Limited using the contact details below.

The decisions being sought from creditors at the meeting will include considering resolutions for the appointment by creditors of a liquidator and a resolution specifying the terms on which the liquidator is to be remunerated. The meeting will receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting.

A meeting of shareholders has been called and will be held immediately prior to the virtual meeting of creditors to consider passing resolutions for the voluntary winding up of the Company and appointment of the proposed liquidator, Jeremy Bennett.

A creditor is entitled to attend and/or vote at the virtual creditors meeting in person or by proxy. Creditors wishing to attend and/or vote at the meeting must lodge their proxy at O'Haras Limited, Moorend House, Snelsins Lane, Cleckheaton, BD19 3UE or electronically at soh@oharas.co prior to the meeting. A proof of debt (unless it has already been delivered) must also be lodged not later than 4.00pm on the business day prior to the date of the meeting at the aforementioned address (note if not received in time, vote by creditor will be disregarded).

Notice is further given that a list of the names and addresses of the Company's creditors may be inspected, free of charge, on the two business days prior to the virtual meeting at O'Haras Limited, Moorend House, Snelsins Lane, Cleckheaton, West Yorkshire, BD19

If necessary please contact Jeremy Bennett (IP No. 8776), who is licensed to act as Insolvency Practitoner in relation to the above Company, by email at soh@oharas.co or telephone on 01274 800380. Hadrian Clarke, Director

31 March 2022

Ag OH123392 (4035494)

### MEDWAY INTERNATIONAL LTD

(Company Number 10031496)

Registered office: Hammerain House, Hookstone Avenue, Harrogate, HG2 8ER

Principal trading address: Hammerain House, Hookstone Avenue, Harrogate, HG2 8ER

Notice is hereby given, pursuant to Rule 15.13 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016 that the Director of the abovenamed Company (the 'convener') is seeking a decision from creditors on the nomination of a Liquidator by way of a virtual meeting. A resolution to wind up the Company is to be considered on 28 April 2022

The meeting will be held as a virtual meeting by telephone conference on 28 April 2022 at 11.30 am. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated Liquidator using the details below.

Nicola Baker of Rushtons Insolvency Limited, 3 Merchant's Quay, Ashley Lane, Shipley, West Yorkshire, BD17 7DB is a person qualified to act as an insolvency practitioner in relation to the company who, during the period before the meeting date, will furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to Rushtons Insolvency Limited, 3 Merchant's Quay, Ashley Lane, Shipley, West Yorkshire, BD17 7DB.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4pm on 27 April 2022 (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to Rushtons Insolvency Limited, 3 Merchant's Quay, Ashley Lane, Shipley, West Yorkshire, BD17 7DB.

The Director of the Company, before the meeting date and before the end of the period of seven days beginning with the day after the day on which the company passed a resolution for winding up, is required by Section 99 of the Insolvency Act 1986: (i) to make out a statement in the prescribed form as to the affairs of the company, and (ii) send the statement to the Company's creditors.

Name and address of nominated Liquidator: Nicola Baker (IP No. 15852) of Rushtons Insolvency Limited, 3 Merchant's Quay, Ashley Lane, Shipley, West Yorkshire, BD17 7DB

Further details contact: The nominated Liquidator. Tel: 01274 598 585. Alternative contact: Rhys Wordsworth. Email: rwordsworth@rushtonsifs.co.uk

Martin Greenwood. Director

30 March 2022

Ag OH123293 (4035488)

#### **MEDWAY MARINE LTD**

(Company Number 10350632)

Registered office: 2nd Floor Offices, 26-28 West Street, Market Square, Rochford, Essex SS4 1AJ

Principal trading address: Unit 5, Millers Business Park, Church Street, Sittingbourne ME10 2WS

Notice is hereby given, under Section 100 of the INSOLVENCY ACT 1986 and Rule 6.14 and 15.8 of the INSOLVENCY RULES (ENGLAND & WALES) 2016 that a virtual meeting of the creditors of the abovenamed company has been convened by Michael Lawrence Palmer, the director of the company in accordance with resolutions passed by the Board of Directors.

The virtual meeting will be held on 11 April 2022 at 11.15 am. To access the virtual meeting, which will be held virtually via a telephone / an online conferencing platform, contact Restructuring LLP, on behalf of the convener - details below.

This virtual meeting will be recorded in order to establish and maintain records of the existence of relevant facts or decisions that are taken at the meeting. By attending this meeting, you consent to being recorded including recordings of your facial image. Where any recording of the meeting also entails the processing of personal data, such personal data shall be treated in accordance with the Data Protection Act 2018.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be delivered by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, a resolution specifying the terms on which the Liquidators are to be remunerated, and the meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the procedure to seek a decision from creditors on the nomination of a liquidator.

Adrian Paul Dante and Joanne Kim Rolls (IP Nos. 9600 and 8867) are qualified to act as Insolvency Practitioners in relation to the above company and during the period before the decision date they will furnish creditors free of charge with such information concerning the company's affairs as they may reasonably require.

In case of queries, please contact Ashleigh Smith on 01908 087220, Email: ashleigh.smith@opusllp.com

Michael Lawrence Palmer, Director/Convener

25 March 2022

Ag OH122803

**N&G FARCAL LTD** 

(Company Number 11074780) Registered in England and Wales

Registered office: The Barn, Crumps Butts, Bicester, OX26 6EB Principal trading address: The Barn, Crumps Butts, Bicester, OX26

IN THE MATTER OF N&G Farcal Ltd

AND IN THE MATTER OF THE INSOLVENCY ACT 1986

NOTICE IS HEREBY GIVEN under Section 100 of the Insolvency Act 1986 and RULE 6.14 AND 15.8 OF THE INSOLVENCY (ENGLAND & WALES) RULES 2016 that a virtual meeting of the creditors of the above-named company have been convened by Constantin Farcal, the director of the company in accordance with resolutions passed by the Board of Directors.

The virtual meeting will be held as follows:

Time: 12.15 pm Date: 20 April 2022

To access the virtual meeting, which will be held via skype conferencing platform, please contact Mansoor Mubarik on behalf of the convener - detail below:

Name: Mansoor Mubarik of Capital Books (UK) Limited Address: 66 Earl Street, Maidstone, Kent ME14 1PS

Contact Number: 01622 754927

Email ID: mmubarik@capital-books.co.uk

This virtual meeting of creditors may be recorded (video and/or audio) in order to establish and maintain records of the existence of relevant facts, or decisions that are taken at such meetings. By attending this meeting, you consent to being recorded, including possible recordings of your facial image. Where any recording of a meeting also entails the processing of personal data, such personal data shall be treated in accordance with the Data Protection Act 2018.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim. ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be delivered by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting.

Mr Mansoor Mubarik ACA FCCA FABRP (office holder no 009667) is qualified to act as Insolvency Practitioner in relation to the above company and a list of names and addresses of the company's creditors will be available for inspection at 66 Earl Street, Maidstone, Kent, ME14 1PS on the two business days preceding the meeting.

In case of queries, please contact Hassan Kaleem on 02088523707 or email mail@capital-books.co.uk

Dated 22 March 2022

Constantin Farcal Director and Convener

(4037732)

# PUREX ICT SOLUTIONS LIMITED

(Company Number 10550817)

Registered office: 19 Crescent Avenue, Farnworth, Bolton, BL4 9NH Principal trading address: 19 Crescent Avenue, Farnworth, Bolton, BI 4 9NH

Notice is hereby given that a virtual meeting of the creditors of the above-named Company will be held on 27 April 2022 at 2:30 pm for the purpose provided for in Section 100 of the INSOLVENCY ACT

Creditors entitled to attend and vote at the meeting may do so either in person or by proxy. A creditor can attend the meeting in person and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4 pm. on the business day before the meeting. If a creditor cannot attend in person, or do not wish to attend but still

(4035596)

wish to vote at the meeting, they can either nominate a person to attend on their behalf, or they may nominate the Chair of the meeting, who will be a director of the Company, to vote on their behalf. Creditors must deliver their proxy, together with proof of their debt, by no later than the commencement of the meeting. Creditors must deliver all proofs of their debt and proxies to Findlay James (Insolvency Practitioners) Limited, Saxon House, Saxon Way, Cheltenham, GL52 6QX.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

At the meeting, creditors may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting of creditors, and may be requested to consider a resolution specifying the terms on which the Liquidator is to be remunerated.

Alisdair J Findlay of Findlay James (Insolvency Practitioners) Limited, Saxon House, Saxon Way, Cheltenham, GL52 6QX, is qualified to act as an Insolvency Practitioner in relation to the Company and will, during the period before the day of the meeting, furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require. Creditors can contact him on 01242 576555 or by email at info@findlayjames.co.uk

Joseph Lagbenro, Director

25 March 2022 (4035083)

#### REVEL EXPORT LIMITED

(Company Number 01687250) Trading Name: Louth Vision Centre

Registered office: 33 Mill Ridge, Edgware, HA8 7PE

Principal trading address: 4 Market Place Louth LN11 9NB

Principal trading address: 4 Market Place, Louth, LN11 9NR Notice is hereby given that a virtual meeting of the creditors of the

above-named Company is being convened by Mr Paul Avraam, to be held on 12 April 2022 at 11:00 am for the purpose provided for in section 100 of the Insolvency Act 1986.

Creditors entitled to attend and vote at the virtual meeting may do so personally or by proxy. A creditor can attend the virtual meeting in person and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4 p.m. on the business day before the meeting. If a creditor cannot attend in person, or does not wish to attend but still wishes to vote at the meeting, they can either nominate a person to attend on their behalf, or they may nominate the Chair of the meeting, who will be a director of the Company, to vote on their behalf. Creditors must deliver their proxy by no later than the commencement of the meeting. Creditors must deliver all proofs of their debt and proxies to Kallis & Company, Mountview Court, 1148 High Road, Whetstone, London, N20 0RA. Creditors failing to lodge a proof of their debt or proxy as indicated will lead to their vote(s) being disrenarded.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

At the meeting, creditors may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting of creditors.

Peter Charalambous (IP No. 20512) of Kallis & Company, is qualified to act as an Insolvency Practitioner in relation to the Company and will, during the period before the day of the meeting, furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require.

Creditors can contact Kallis & Company on 020 8446 6699 or by email at mike@kallis.co.uk.

Paul Avraam, Director

30 March 2022

Ag OH123320 (4035377)

# RIKI DALAL GROUP UK LTD

(Company Number 11124833) Trading Name: Riki Dalal

Registered office: 34 Mayfair Point, South Molton Street, London,

W1K 5RG

Principal trading address: 8 Duke Street, Marylebone, London, W1U 3EN

This notice is given pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016.

Notice is hereby given that a virtual meeting of the creditors of the above named Company is being convened and will be held on 11 April 2022 at 11.00 am. Menahem Geber in his capacity as the chair of the Company's Board of Directors is the convener of the meeting ("the Convener").

The primary purpose of the virtual meeting is for creditors to consider a decision on the nomination of liquidators, and if the creditors think fit, to appoint a liquidation committee.

Creditors participating in the virtual meeting or attending by proxy, (unless a committee is formed), will also be asked to consider approving the basis of the joint liquidators remuneration and whether they also approve the joint liquidators drawing their disbursements, details of which have been provided in a report to creditors.

In order to be entitled to vote at the meeting (either in person or by proxy), creditors must have submitted a proof in respect of their claim (unless a proof has already been submitted) to the Convener via the contact details below, by no later than 4pm on 8 April 2022.

Any creditor entitled to attend and vote at the meeting is entitled to do so either in person or by proxy. If you cannot attend and wish to be represented at the meeting, a completed proxy form must be delivered to the Convener via the contact details below, prior to the meeting.

Log-in details for the virtual meeting will be supplied to creditors on request.

Any person who requires further information may contact Robert Cook by telephone on 020 7400 7900 or alternatively by e-mail at RC-Team@btguk.com

Menahem Geber, Director/Convener

28 March 2022

Ag OH123309 (4035479)

#### **RJ DECORATORS LTD**

(Company Number 11441951)

Registered office: Unit 8d Hepworth Business Park, Pontyclun, CF72

Principal trading address: None

Type of Liquidation: Creditors' Voluntary.

Date of meeting: 7 April 2022. Time of meeting: 10:00 am.

NOTICE IS HEREBY GIVEN pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016, that a virtual meeting of the creditors of the above named Company will be held on the date and time specified in this notice for the purposes mentioned in Section 100 of the Insolvency Act 1986. The Insolvency Practitioner named below is qualified to act in this matter. A list of names and addresses of the Company's creditors may be inspected, free of charge, at the address given below, between 10.00am and 4.00pm on the two business days preceding the date of the meeting.

Resolutions to be taken at the meeting may include a resolution specifying the terms on which the liquidator is to be remunerated, and the meeting may receive information about, or be called upon to approve the costs of preparing the statement of affairs and convening the meeting.

Creditors entitled to attend and vote at the virtual meeting may participate either in person or by proxy. A creditor can attend the virtual meeting and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4 pm on the business day before the meeting. If a creditor cannot attend, or does not wish to attend, but still wishes to vote at the virtual meeting, they can ether nominate a person to attend on their behalf, or they may nominate the Chairman of the virtual meeting, who will be a director of the Company, to vote on their behalf. Such creditors must still submit the proof of their claim by no later than 4 pm on the business day before the meeting but proxies will be accepted by the Chairman up to the commencement of the virtual meeting. For the purposes of voting, any secured creditors are required (unless they surrender their security) to lodge a statement with the Insolvency Practitioners prior to the Meeting, giving particulars of their security, the date when it was given and its assessed value. Creditors must deliver proof of their claim and their proxy using the details provided below.

By Order of the Board Robert Jenkins. Director Insolvency Practitioner's Name and Address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, Bolton, BL1 1HL. Telephone: 01204 255 051.

For further information contact Jasmine Baxter at the offices of Anderson Brookes Insolvency Practitioners Limited on 01204 255 051, or jasmine@andersonbrookes.co.uk.

31 March 2022 (4036399)

#### **BJ SPRAY LTD**

(Company Number 12638210)

Registered office: First Floor, Hill House 23-25 Spur Road, Cosham,

Portsmouth, PO6 3DY Principal trading address: None Nature of Business: Painting.

Type of Liquidation: Creditors' Voluntary.

Date of meeting: 7 April 2022. Time of meeting: 9:30 am.

NOTICE IS HEREBY GIVEN pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016, that a virtual meeting of the creditors of the above named Company will be held on the date and time specified in this notice for the purposes mentioned in Section 100 of the Insolvency Act 1986. The Insolvency Practitioner named below is qualified to act in this matter. A list of names and addresses of the Company's creditors may be inspected, free of charge, at the address given below, between 10.00am and 4.00pm on the two business days preceding the date of the meeting.

Resolutions to be taken at the meeting may include a resolution specifying the terms on which the liquidator is to be remunerated, and the meeting may receive information about, or be called upon to approve the costs of preparing the statement of affairs and convening the meeting.

Creditors entitled to attend and vote at the virtual meeting may participate either in person or by proxy. A creditor can attend the virtual meeting and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4 pm on the business day before the meeting. If a creditor cannot attend, or does not wish to attend, but still wishes to vote at the virtual meeting, they can ether nominate a person to attend on their behalf, or they may nominate the Chairman of the virtual meeting, who will be a director of the Company, to vote on their behalf. Such creditors must still submit the proof of their claim by no later than 4 pm on the business day before the meeting but proxies will be accepted by the Chairman up to the commencement of the virtual meeting. For the purposes of voting, any secured creditors are required (unless they surrender their security) to lodge a statement with the Insolvency Practitioners prior to the Meeting, giving particulars of their security, the date when it was given and its assessed value. Creditors must deliver proof of their claim and their proxy using the details provided below.

By Order of the Board

Robert Jenkins, Director

Insolvency Practitioner's Name and Address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, Bolton, BL1 1HL. Telephone: 01204 255 051

For further information contact Jasmine Baxter at the offices of Anderson Brookes Insolvency Practitioners Limited on 01204 255 051, or jasmine@andersonbrookes.co.uk.

31 March 2022 (4036398)

# SES AGENCY SERVICES LTD

(Company Number 12873680)

Registered in England and Wales

Registered office: 26 Hathaway Gardens, Romford, England, RM6 5TL Principal trading address: 26 Hathaway Gardens, Romford, England, RM6 5TI

IN THE MATTER OF SES AGENCY SERVICES LTD

AND IN THE MATTER OF THE INSOLVENCY ACT 1986

NOTICE IS HEREBY GIVEN under Section 100 of the Insolvency Act 1986 and RULE 6.14 AND 15.8 OF THE INSOLVENCY (ENGLAND & WALES) RULES 2016 that a virtual meeting of the creditors of the above-named company have been convened by Hafiz Sohaib Ijaz Cheema , the director of the company in accordance with resolutions passed by the Board of Directors.

The virtual meeting will be held as follows:

Time: 11.15 am Date: 22 April 2022

To access the virtual meeting, which will be held via skype conferencing platform, please contact Mansoor Mubarik on behalf of the convener - detail below:

Name: Mansoor Mubarik of Capital Books (UK) Limited Address: 66 Earl Street, Maidstone, Kent ME14 1PS

Contact Number: 01622 754927

Email ID: mmubarik@capital-books.co.uk

This virtual meeting of creditors may be recorded (video and/or audio) in order to establish and maintain records of the existence of relevant facts, or decisions that are taken at such meetings. By attending this meeting, you consent to being recorded, including possible recordings of your facial image. Where any recording of a meeting also entails the processing of personal data, such personal data shall be treated in accordance with the Data Protection Act 2018.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be delivered by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting.

Mr Mansoor Mubarik ACA FCCA FABRP (office holder no 009667) is qualified to act as Insolvency Practitioner in relation to the above company and a list of names and addresses of the company's creditors will be available for inspection at 66 Earl Street, Maidstone, Kent, ME14 1PS on the two business days preceding the meeting.

In case of queries, please contact Hassan Kaleem on 02088523707 or email mail@capital-books.co.uk

Dated 28 March 2022

Hafiz Sohaib Ijaz Cheema Director and Convener (4037734)

#### **SUTTON DINING LTD**

(Company Number 11031971)

Registered in England and Wales

Registered office: 358 Birmingham Road Sutton Coldfield,

Birmingham, West Midlands, B72 1YH

Principal trading address: 358 Birmingham Road Sutton Coldfield,

Birmingham, West Midlands, B72 1YH IN THE MATTER OF SUTTON DINING LTD

AND IN THE MATTER OF THE INSOLVENCY ACT 1986

NOTICE IS HEREBY GIVEN under Section 100 of the Insolvency Act 1986 and RULE 6.14 AND 15.8 OF THE INSOLVENCY (ENGLAND & WALES) RULES 2016 that a virtual meeting of the creditors of the above-named company have been convened by Taslim Kausar , the director of the company in accordance with resolutions passed by the Board of Directors.

The virtual meeting will be held as follows:

Time: 01.00 pm Date: 20 April 2022

To access the virtual meeting, which will be held via skype conferencing platform, please contact Mansoor Mubarik on behalf of the convener - detail below:

Name: Mansoor Mubarik of Capital Books (UK) Limited Address: 66 Earl Street, Maidstone, Kent ME14 1PS

Contact Number: 01622 754927 Email ID: mmubarik@capital-books.co.uk This virtual meeting of creditors may be recorded (video and/or audio) in order to establish and maintain records of the existence of relevant facts, or decisions that are taken at such meetings. By attending this meeting, you consent to being recorded, including possible recordings of your facial image. Where any recording of a meeting also entails the processing of personal data, such personal data shall be treated in accordance with the Data Protection Act 2018.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be delivered by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting.

Mr Mansoor Mubarik ACA FCCA FABRP (office holder no 009667) is qualified to act as Insolvency Practitioner in relation to the above company and a list of names and addresses of the company's creditors will be available for inspection at 66 Earl Street, Maidstone, Kent, ME14 1PS on the two business days preceding the meeting.

In case of queries, please contact Hassan Kaleem on 02088523707 or email mail@capital-books.co.uk

Dated 17 March 2022

8 April 2022.

Taslim Kausar Director and Convener

(4037733)

#### THE VILLAGE BALTI & PIZZA LTD

(Company Number 09721192)

Trading Name: Take-away food shops and mobile food stands Registered office: 45 St Martin's Drive, Desford, LE9 9GX Principal trading address: 45 St Martin's Drive, Desford, LE9 9GX Notice is hereby given, pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016, that the Directors of the abovenamed Company (the 'convener(s)') are seeking a decision from creditors on the nomination of a Joint Liquidators by way of a virtual meeting. A resolution to wind up the Company is to be considered on

The meeting will be held as a virtual meeting by telephone conference, on 08/04/2022 at 04:15 pm. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated Joint Liquidators using the details below.

Sajid Sattar and Simon Gwinnutt of Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham B1 1QH are persons qualified to act as an insolvency practitioner in relation to the company who, during the period before the meeting date, will furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require. To participate in the call creditors will need to dial 0333 0112 319 and enter room number: 30939335, followed by pin code: 9086.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham B1 1OH.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4pm on 07/04/2022 (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham B1 1QH.

The Directors of the Company, before the meeting date and before the end of the period of seven days beginning with the day after the day on which the company passed a resolution for winding up, are required by Section 99 of the Insolvency Act 1986: (i) to make out a statement in the prescribed form as to the affairs of the company, and (ii) send the statement to the Company's creditors.

Nominated Liquidator: *Sajid Sattar* (IP number 15590) of Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham, B1 10H

Nominated Liquidator: Simon Gwinnutt (IP number 8877) of Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham, R1 10H

For further details contact Glenda Hassall on 0121 201 1720 or at g.h@greenfieldrecovery.co.uk

Mohammed Abdul Khalique, Director

(4036303)

#### **WORLD OF RESIN LIMITED**

(Company Number 11848144)

Registered office: The Old Court House, New Road Avenue, Chatham, Kent, ME4 6BE

Principal trading address: 11 Wouldham Road, Rochester, Kent, ME1 3.IY

Notice is hereby given, under Section 100 of the INSOLVENCY ACT 1986 and Rule 6.14 and 15.8 of the INSOLVENCY RULES (ENGLAND & WALES) 2016 that a virtual meeting of the creditors of the abovenamed company has been convened by Kevin Tester, the director of the company in accordance with resolutions passed by the Board of Directors.

The virtual meeting will be held on 19 April 2022 at 10.15 am. To access the virtual meeting, which will be held via Microsoft Teams, contact Opus Restructuring LLP, on behalf of the convener – details below.

This virtual meeting will be recorded in order to establish and maintain records of the existence of relevant facts or decisions that are taken at the meeting. By attending this meeting, you consent to being recorded including recordings of your facial image. Where any recording of the meeting also entails the processing of personal data, such personal data shall be treated in accordance with the Data Protection Act 2018.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be delivered by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, a resolution specifying the terms on which the Liquidators are to be remunerated, and the meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the procedure to seek a decision from creditors on the nomination of a liquidator.

Adrian Paul Dante and Joanne Kim Rolls (IP Nos 9600 and 8867) are qualified to act as Insolvency Practitioners in relation to the above company and during the period before the decision date they will furnish creditors free of charge with such information concerning the company's affairs as they may reasonably require.

In case of queries please contact: Sam Knight, Email: cvlteam@opusllp.com or Tel: 01908 087220.

Kevin Tester, Director/Convener

29 March 2022

Ag OH123322

(4035581)

(4035459)

#### **NOTICES TO CREDITORS**

#### **INVESTABLY LIMITED**

(Company Number 11707395)

("the Company")

Registered office: c/o Begbies Traynor (Central) LLP, St James Court, St James Parade, Bristol, BS1 3LH

Principal trading address: Kings Parade, Lower Coombe Street, Croydon, Surrey, CR0 1AA

Paul Wood (IP Number: 009872) and Simon Robert Haskew (IP Number: 008988), both of Begbies Traynor (Central) LLP of St James Court, St James Parade, Bristol, BS1 3LH were appointed as Joint Liquidators of the Company on 24 March 2022.

Creditors of the Company are required on or before the 30 April 2022 to send their names and addresses and particulars of their debts or claims and the names and addresses of the solicitors (if any) to the joint liquidators, at Begbies Traynor (Central) LLP, St James Court, St James Parade, Bristol, BS1 3LH and, if so required by notice in writing from the joint liquidators, by their solicitors or personally, to come in and prove their said debts or claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.

Any person who requires further information may contact the Joint Liquidator by e-mail at bristol@btguk.com or by telephone on 0117 937 7130. Alternatively, enquiries can be made to Clive Hobbs by e-mail at clive.hobbs@btguk.com or by telephone on 0117 937 7148.

Joint Liquidator

Dated: 29 March 2022 (4037774)

Notice to Creditors to Submit Claim – Creditors' Voluntary Liquidation THE INSOLVENCY RULES 1986 (AS AMENDED) – RULE 4.73 THE BEAR HOTEL AT WANTAGE LTD

(Company Number 07431339)

Trading Name: The Bear Hotel Registered office: The Bear Hotel, Wantage, Oxfordshire OX12 8AB Principal trading address: The Bear Hotel, Wantage, Oxfordshire

OX12 8AB Notice is hereby given, pursuant to Rule 4.73 of THE INSOLVENCY RULES 1986 (AS AMENDED), that the creditors of the above named company, which is being voluntarily wound up, are required on or before 25 April 2022, to send their names and addresses along with descriptions and full particulars of their debts or claims and the names and addresses of their solicitors (if any), to Peter Richard James Frost at Staverton Court, Staverton, Cheltenham, GL51 0UX and, if so required by notice in writing from the Liquidator of the company or by the Solicitors of the Liquidator, to come in and prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the benefit of any dividend paid before such debts/claims are proved

Name of office holder 1: Peter Richard James Frost

Office holder 1 IP number: 008935

Postal address of office holder(s): Staverton Court, Staverton, Cheltenham, GL51 0UX

Office holder's telephone no and email address: 01242 680000 and creditors@hazlewoods.co.uk

Date of Appointment: 22 March 2022 Capacity of office holder(s): Liquidator

Alternative contact for enquiries on proceedings: Gina Clare (4037775)

# ALGAR SIGNCRAFT SERVICES (CAMBRIDGE) LIMITED

(Company Number 01196266)

Registered office: Salisbury House, Station Road, Cambridge, CB1 2LA

Principal trading address: Unit 2, Coldhams Road, Cambridge, CB1 3FW

Notice is hereby given that creditors of the Company are required, on or before 13 May 2022, to prove their debts by delivering their proofs (in the format specified in RULE 14.4 of the Insolvency (England and Wales) Rules 2016) to the Joint Liquidators at PEM, Salisbury House, Station Road, Cambridge, CB1 2LA.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim.

Date of Appointment: 29 March 2022

Office Holder Details: *Kate Merry* (IP No. 9439) and *Ben Dyer* (IP No. 18016) both of PEM, Salisbury House, Station Road, Cambridge, CB1 2I A

Further details contact: The Joint Liquidators, Tel: 01223 728 222 or Email: bdyer@pem.co.uk

Kate Merry, Joint Liquidator

30 March 2022

Ag OH123319

#### **ASL BOSTON LIMITED**

(Company Number 09969070)

Registered office: Office 10 Ground Floor The Old, Tannery, Eastgate, Accrington, BB5 6PW

Principal trading address: Office 10 Ground Floor The Old, Tannery, Eastgate, Accrington, BB5 6PW

NOTICE IS HEREBY GIVEN that the creditors of the above named Company, which is being voluntarily wound up, are required to prove their debts on or before 23rd June 2022 by sending their names and addresses along with descriptions and full particulars of their debts or claims and the names and addresses of their solicitors (if any), to McAlister & Co Insolvency Practitioners Ltd at 10, St Helens Road, Swansea SA1 4AW and, if so required by notice in writing from the liquidator of the Company or by the Solicitors of the Liquidator, to come in and prove their debts or claims, or in default thereof they will be excluded from the benefit of any distribution made before such debts or claims are proved.

Joint Liquidator: *Helen Whitehouse* (IP number 9680) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1 4AW.

Joint Liquidator: Sandra McAlister (IP number 9375) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1 4AW.

Date of Appointment: 23 March 2022

For further details contact Jonathan Waller on 03300 563600 or at creditors@mcalisterco.co.uk (4037902)

#### **LANGDON TRANSPORT LIMITED**

(Company Number 04661798)

Registered office: 5 Barnfield Crescent, Exeter, Devon, EX1 1QT

Principal trading address: 38 Central Avenue, Lee Mill Industrial Estate, Ivybridge, PL21 9PE

Notice is given that under RULE 6.23 of the Insolvency (England and Wales) Rules 2016 ("the Rules") that the company was placed into Creditors' Voluntary Liquidation (insolvent liquidation) and *David Gerard Kirk* (IP No. 8830) and *Daniel Robert Jeeves* (IP No. 26032) both of Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT were appointed Joint Liquidators by the creditors on 28 March 2022.

Notice is further given that the Creditors are required to prove their debts on or before 27 April 2022 by sending full details of their claims to the Liquidators at Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT. Creditors must also, if so requested by the Liquidators, provide such further details and documentary evidence to support their claims as may appear to the Liquidators to be necessary.

Please note that no further public notice will be made and therefore the Liquidators shall be entitled to make any distribution without regard to any claims not proved in the manner required by statute.

For further details contact Daniel Jeeves, Tel: 01392 474303 or Email: nathan@kirks.co.uk.

David Gerard Kirk, Joint Liquidator 30 March 2022

Ag OH123274

(4035371)

#### **TOWER BUILDING LTD**

(Company Number 08733754)

Registered office: C/O Quantuma Advisory Limited, High Holborn House, 52 – 54 High Holborn, London, WC1V 6RL

Principal trading address: Tower Building, Marine Parade, Great Yarmouth, NR30 2EW

Notice is hereby given that under Rule 6.23 of the Insolvency (England and Wales) Rules 2016 ('the Rules') that the Company was placed into creditors' voluntary liquidation (insolvent liquidation) and *Andrew Andronikou* (IP No. 8806) and *Michael Kiely* (IP No. 9617) both of Quantuma Advisory Limited, High Holborn House, 52-54 High Holborn, London, WC1V 6RL were appointed Joint Liquidators by the creditors on 25 February 2022.

Notice is further given that the creditors are required to prove their debts on or before 29 April 2022 by sending full details of their claims to the Liquidators at Quantuma Advisory Limited, High Holborn House, 52-54 High Holborn, London, WC1V 6RL. Creditors must also, if so requested by the Liquidators, provide such further details and documentary evidence to support their claims as may appear to the Liquidators to be necessary.

Please note that no further public notice will be made and therefore the Liquidators shall be entitled to make any distribution without regard to any claims not proved in the manner required by statute.

In case of queries, please contact Selina Rai on 020 8059 8926 or email selina.rai@quantuma.com.

Andrew Andronikou, Joint Liquidator 30 March 2022 Ag OH123341

(4035381)

#### **RESOLUTION FOR WINDING-UP**

#### PERFECT DAY CATERING LTD

(Company Number: 13323530)

trading as The Playa

Registered Office: 21 Highfield Road, Dartford DA1 2JS

Principal Trading Address: The Playa, The Leas, Sheerness ME12 2NL  $\,$ 

Nature of Business: Licensed restaurants

At a General Meeting of the Members of the above-named Company, duly convened, and held by correspondence on Wednesday 30 March 2022, the following Resolution/s was/were duly passed:

- 1. (Special Resolution) That it has been resolved by special resolution that the company be wound up voluntarily
- 2. (Ordinary Resolution) That Isobel Susan Brett of Bretts Business Recovery Limited, 21 Highfield Road, Dartford, Kent, DA1 2JS be appointed liquidator of the company for the purposes of the winding-up

For further details, please contact: Isobel Susan Brett, (9643), Bretts Business Recovery Limited, 21 Highfield Road, Dartford DA1 2JS, Telephone: 01474532862, Email address: ellieraycraft@brettsbr.co.uk. Scott I'Anson, Chairman (4036910)

#### OH ELECTRICAL SERVICES LTD

(Company Number: 09907526)

trading as OH ELECTRICAL SERVICES LTD

Registered Office: 38 Stonyhill Avenue, Blackpool FY4 1PR

Principal Trading Address: 38 Stonyhill Avenue, Blackpool FY4 1PR

Nature of Business: Electrcian

- At a Extraordinary Meeting of the Members of the above-named Company, duly convened, and held remotely on Thursday 17 March 2022, the following Resolution/s was/were duly passed:
- 1. (Special Resolution) THAT it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily
- 2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is hereby appointed

For further details, please contact: Stuart Rathmell, (10050), STUART RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1 2HS, Telephone: 01204867615, Email address: stuart.rathmell@icloud.com.

Alexander Hutchinson, Chairman

Thursday 17 March 2022 (4037756)

# TOTAL VEHICLE LEASING LIMITED

(Company Number: 09564032)

trading as Total Vehicle Leasing Limited

Registered Office: St John's Terrace, 11-15 New Road, Radcliffe,

Manchester M26 1LS

Principal Trading Address: St. Annes House, Suite 5, 1 Old Market Place, Knutsford WA16 6PD

Nature of Business: Vehicle Leasing Broker

At a General Meeting of the Members of the above-named Company, duly convened, and held at St John's Terrace, 11-15 New Road, Radcliffe, Manchester M26 1LS on Wednesday 30 March 2022, the following Resolution/s was/were duly passed:

- 1. (Special Resolution) That the Company be wound up voluntarily.
- 2. (Ordinary Resolution) That Steven Wiseglass of Inquesta Corporate Recovery & Insolvency, St John's Terrace, 11- 15 New Road, Manchester, M26 1LS, be appointed Liquidator of the Company.

For further details, please contact: Steven Wiseglass, (9525), Inquesta Corporate Recovery & Insolvency, St John's Terrace, 11-15 New Road, Radcliffe, Manchester M26 1LS, Telephone: 0333 005 0080, Email address: info@inquesta.co.uk, (A1541).

Kevin Anthony Devine, Director

(4037796)

#### 1 TO WEAR LIMITED

(Company Number 07019382)

Registered office: Mountview Court, 1148 High Road, Whetstone, London N20 0RA

Principal trading address: N/A

At a general meeting of the members of the above named Company, duly convened and held at Mountview Court, 1148 High Road, Whestone, London N20 0RA on 25 March 2022 the following resolutions were passed by the Members, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Kikis Kallis* (IP No. 4692) of Kallis & Company, Mountview Court, 1148 High Road, Whetstone, London, N20 0RA be appointed Liquidator of the Company."

Further details contact: Kikis Kallis, Tel: 020 8446 6699

Mehmet Kazim, Chair 25 March 2022 Ag OH123292

(4035365)

#### AI LIFTING LTD

(Company Number 10326297)

Registered office: 25 Brixham Building, 8 Artillery Place, London, SE18 4AB

Principal trading address: 25 Brixham Building, 8 Artillery Place, London, SE18 4AB

At a General Meeting of the members of the above named company, duly convened and held at 25 Brixham Building, 8 Artillery Place, London, SE18 4AB on 25 March 2022 the following special resolution was duly passed:

"That the Company be wound up voluntarily and that Richard Frank Simms of F A Simms & Partners Limited, Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB is hereby appointed liquidator for the purpose of the winding up"

Further Details: Viera Navratilova Tel No. 01455 555 444

Name of Office Holder: Richard Frank Simms

Office Holder Number: 9252

Address of Office Holder: Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB

Aleksander Ivanov (4037804)

#### ALGAR SIGNCRAFT SERVICES (CAMBRIDGE) LIMITED

(Company Number 01196266)

Registered office: Salisbury House, Station Road, Cambridge, CB1 2LA

Principal trading address: Unit 2, Coldhams Road, Cambridge, CB1 3EW

Notice is hereby given that the following written resolutions were passed on 29 March 2022 as a special resolution and an ordinary resolution respectively:

"That the company be wound up voluntarily and that *Kate Merry* (IP No. 9439) and *Ben Dyer* (IP No. 18016) both of PEM, Salisbury House, Station Road, Cambridge, CB1 2LA be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: The Joint Liquidators, Tel: 01223 728 222 or

Email: bdyer@pem.co.uk Nicolas Algar, Director 29 March 2022 Ag OH123319 (4035504)

#### **ASL BOSTON LIMITED**

(Company Number 09969070)

Registered office: Office 10 Ground Floor The Old, Tannery, Eastgate, Accrington, BB5 6PW

Principal trading address: Office 10 Ground Floor The Old, Tannery, Eastgate, Accrington, BB5 6PW

At a meeting of the above named company duly convened and held on 23 March 2022, the following resolutions were passed:

That it has been proved to the satisfaction of this meeting that the company cannot, by reason of its current and/or impending liabilities, continue its business, and that it is advisable to wind up the same under a Creditors Voluntary Liquidation.

That Helen Whitehouse and Sandra McAlister of McAlister & Co Insolvency Practitioners Ltd, 10 St Helen's Road, Swansea, SA1 4AW be appointed for the purposes of the voluntary winding up of the Company and that they be authorised to act jointly and separately.

The appointment of Helen Whitehouse and Sandra McAlister as joint liquidators was confirmed at the subsequent virtual meeting of creditors on 21 March 2022.

Joint Liquidator: Helen Whitehouse (IP number 9680) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1 4AW

Joint Liquidator: Sandra McAlister (IP number 9375) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1

Date of Appointment: 23 March 2022

For further details contact Jonathan Waller on 03300 563600 or at creditors@mcalisterco.co.uk

Bushra Bostan, Director (4037903)

That Jeffrey Mark Brenner of B&C Associates Limited, Concorde House, Grenville Place, Mill Hill, London, NW7 3SA, be appointed liquidator of the Company for the purposes of the voluntary winding-

Liquidator: *Jeffrey Mark Brenner* (IP number 9301) of B&C Associates Limited,, Concorde House, Grenville Place, Mill Hill, London, NW7 35A.

Date of Appointment: 30 March 2022

For further details contact Besa Mustafa on 020 8906 7730 or at besa@bcassociates.uk.com

Saeid Maleki, Director (4037891)

#### **BBG KITCHEN LTD**

(Company Number 11575737)

Trading Name: The Broadway Bar And Grill

Registered office: The Broadway Bar And Grill, Birmingham Road, Walsall, West Midlands, WS5 3AB

Principal trading address: The Broadway Bar And Grill, Birmingham Road, Walsall, West Midlands, WS5 3AB

Notice is hereby given that the following resolutions were passed on 30 March 2022 as a Special resolution and an Ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Laura Anne Walshe* (IP No. 26250) of Keywood Group Limited, Gill House, 140 Holyhead Road, Birmingham, West Midlands, B21 0AF be and is hereby appointed Liquidator for the purpose of such winding up."

Contact details for Liquidator: Email: team@keywoodgroup.co.uk Alternative contact: Harry Chahal.

Ajay Subedi, Director

30 March 2022

Ag OH123342 (4035461)

#### **AVIANCE SERVICES LTD**

(Company Number 12342634)

Registered office: Unit 8, Riverside Court, Pride Park, Derby, DE24 8 IN

Principal trading address: 9 Timberlands, Crawley, West Sussex, RH11 9PH

At a General Meeting of the Members of the above-named company, duly convened, and held on 24/03/2022 the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution

That the Company be wound up voluntarily.

And

That Simon Gwinnutt and Sajid Sattar of Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham, B1 1QH be appointed Joint Liquidators for the purposes of the voluntary winding up of the company and that they may act jointly or severally in this regard.

Joint Liquidator: Simon Gwinnutt (IP number 8877) of Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham, B1 1QH.

Joint Liquidator: *Sajid Sattar* (IP number 15590) of Greenfield Recovery Limited, Trinity House, 28-30 Blucher Street, Birmingham, B1 1OH.

Date of Appointment: 24 March 2022

For further details contact Lee Adams on 0121 201 1720 or at la@greenfieldrecovery.co.uk

Iulian Octavian Melnic, Director (4037884)

# BASIL FRESH PIZZA LTD

(Company Number 12488946)
Trading Name: Basil Pizza

Registered office: C/o B&C Associates Limited, Concorde House, Grenville Place, Mil! Hill, London, NW7 3SA

Principal trading address: 73 Barlby Road, London, W10 6AW

Notice is hereby given that the following resolutions were passed on 30 March 2022, as a special resolution and an ordinary resolution respectively:

That the Company be wound up voluntarily; and

#### **BRITANICO LIMITED**

(Company Number 08297933)

Trading Name: Britanico Coffee House

Registered office: 4a Roman Road, London, E6 3RX

Principal trading address: 359 High Street, Newham, London, E15 4QZ

Notice is hereby given that the following resolutions were passed on 29 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that *Jane Hardy* (IP No. 9384) and *Gary Birchall* (IP No. 9725) both of Focus Insolvency Group, Skull House Lane, Appley Bridge, Wigan, WN6 9EU be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: The Joint Liquidators, Tel: 01257 257 030, Email: n.taylor@focusinsolvencygroup.co.uk. Alternative contact: Email: votes@focusinsolvencygroup.co.uk

(4035423)

Rimantas Zinis, Director

30 March 2022

Ag OH123383

#### **CAMBRIDGE STRATUM LIMITED**

(Company Number 09687862)

Registered office: c/o Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, 30 Churchgate, Bolton, BL1 1HL

Principal trading address: None

Nature of Business: Sale of medical aesthetic equipment to specialised clinics.

Type of Liquidation: Creditors' Voluntary.

Place of meeting: 4th Floor Churchgate House, Bolton, BL1 1HL.

Date of meeting: 29 March 2022.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 29 March 2022

Liquidator's Name and Address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, Bolton, BL1 1HL. Telephone: 01204 255 051.

For further information contact Jasmine Baxter at the offices of Anderson Brookes Insolvency Practitioners Limited on 01204 255 051, or jasmine@andersonbrookes.co.uk.

31 March 2022 (4036404)

#### CAREMARS LTD.

(Company Number 12450963)

Registered office: 77 Cavendish Avenue, Harrow, HA1 3RF

Principal trading address: 77 Cavendish Avenue, Harrow, HA1 3RF

Notice is hereby given that the following resolutions were passed on 28 March 2022, as a special resolution and an ordinary resolution:

"That the Company be wound up voluntarily and that *Gemma Louise Roberts* (IP No. 9701) and *Robert Neil Dymond* (IP No. 10430) both of Wilson Field Limited, The Manor House, 260 Ecclesall Road South, Sheffield, S11 9PS be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: The Joint Liquidators, Tel: 0114 2356780. Alternative contact: Kamal Chopra, Email: k.chopra@wilsonfield.co.uk *Mirela Coioc.* Director

30 March 2022

Ag OH123381 (4035450)

#### **CHEON LTD**

(Company Number 10737120)

Registered office: A2, Seedbed Centre, Vanguard Way, Southend-on-Sea. SS3 9QY

Principal trading address: (Previously) A2, Seedbed Centre, Vanguard Way, Southend-on-Sea, SS3 9QY

At a general meeting of the members of the above named Company, duly convened and held at The Old Exchange, 234 Southchurch Road, Southend on Sea, Essex SS1 2EG on 24 March 2022 the following resolutions were duly passed as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Wayne MacPherson* (IP No. 9445) and *Louise Donna Baxter* (IP No. 099123) both of Begbies Traynor (Central) LLP, The Old Exchange, 234 Southchurch Road, Southend-on-Sea, Essex, SS1 2EG be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 01702 467255. Alternatively enquiries can be made to lan Goodhew by e-mail at southendadmin@btguk.com or by telephone on 01702 443923.

Victoria Morris Chapman, Chair

24 March 2022

Ag OH123346 (4035582)

#### **CHESHIRE ESTATES (MARY MILL) LIMITED**

(Company Number 12188790)

Registered office: Repton Manor, Repton Avenue, Ashford, TN23 3GP Principal trading address: Repton Manor, Repton Avenue, Ashford, TN23 3GP

At a general meeting of the members of the above named company, duly convened and held at Repton Manor, Repton Avenue, Ashford, Kent, United Kingdom, TN23 3GP on 30 March 2022 the following special resolution was duly passed:

That the Company be wound-up voluntarily and that Philippa Smith of Smith & Barnes Insolvency Practitioners Ltd, 22A Main Street, Garforth, Leeds, LS25 1AA is hereby appointed liquidator for the purpose of the winding up.

Liquidator: *Philippa Smith* (IP number 18670) of Smith & Barnes Insolvency Practitioners Ltd, 22A Main Street, Garforth, Leeds, LS25 1AA.

Date of Appointment: 30 March 2022

For further details contact Muzamal Ahmad on 0113 532 3278

(4036929)

#### CHESHIRE ESTATES (MERE APARTHOTEL) LIMITED

(Company Number 10522000)

Registered office: Repton Manor, Repton Avenue, Ashford, TN23 3GP Principal trading address: Repton Manor, Repton Avenue, Ashford, TN23 3GP

At a general meeting of the members of the above named company, duly convened and held at Repton Manor, Repton Avenue, Ashford, Kent, United Kingdom, TN23 3GP on 30 March 2022 the following special resolution was duly passed:

That the Company be wound-up voluntarily and that Philippa Smith of Smith & Barnes Insolvency Practitioners Ltd, 22A Main Street, Garforth, Leeds, LS25 1AA is hereby appointed for the purpose of the winding up.

Liquidator: *Philippa Smith* (IP number 18670) of Smith & Barnes Insolvency Practitioners Ltd, 22A Main Street, Garforth, Leeds, LS25 1AA

Date of Appointment: 30 March 2022

For further details contact Muzamal Ahmad on 0113 532 3278

(4037724)

#### **CLAIRE GRACE INTERIORS LTD**

(Company Number 09160339)

Registered office: 22 Regent Street, Nottingham, NG1 5BQ

Principal trading address: 20 Market Street, Bingham, Nottinghamshire, NG13 8AB

At a General Meeting of the members of the above named Company, duly convened and held at 22 Regent Street, Nottingham, NG1 5BQ on 24 March 2022 the following resolutions were duly passed; as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that Joseph Gordon Maurice Sadler (IP No. 9048) and Andrew J Cordon (IP No. 009687) both of Elwell Watchorn & Saxton LLP, 22 Regent Street, Nottingham, NG1 5BQ be and are hereby appointed Joint Liquidators of the Company for the purpose of the voluntary winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 0115 8387330. Alternatively enquiries can be made to Joseph Sadler by email at joe@cfs-llp.com.

Claire Grace, Chair 24 March 2022

Ag OH123269

(4035466)

# **CLEAN 4 YOU LONDON LIMITED**

(Company Number 06648975)

Registered office: Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow, SK9 1BQ

Principal trading address: 383 London Road, Grays, RM20 4AA

At a General Meeting of the Members of the above-named company, duly convened, and held on 30 March 2022 the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution:

That the Company be wound up voluntarily

That Nick Morgan and James Kaye, of NTF Corporate Solutions Limited, Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow, Cheshire, SK9 1BQ, be appointed Joint Liquidators of the Company and that they be authorised to act either jointly or separately.

Joint Liquidator: *James Kaye* (IP number 9373) of NTF Corporate Solutions Limited, Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow, Cheshire, SK9 1BQ.

Joint Liquidator: *Nick Morgan* (IP number 9078) of NTF Corporate Solutions Limited, Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow. Cheshire. SK9 1BQ.

Date of Appointment: 30 March 2022

For further details contact Rebecca Dale on 01625 540744 or at info@ntffs.co.uk

Snezhana Nikolova, Director / Chairman

(4037288)

# CONSTELLATION STYLING LTD

(Company Number 11989231)

Registered office: XL Business Solutions, Premier House, Bradford Road, Cleckheaton, BD19 3TT

Principal trading address: 3 Sea Road, Sunderland, SR6 9BP

Notice is hereby given that the following resolutions were passed on 29 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *J N Bleazard* (IP No. 09354) of XL Business Solutions, Premier House, Bradford Road, Cleckheaton, BD19 3TT be appointed as Liquidator for the purposes of such voluntary winding up."

Further details contact: Jeremy Bleazard, Tel: 01274 870101. Alternative contact: Graham Harsley.

A Deighton, Director 30 March 2022 Ag OH123290

(4035534)

#### **COSMIN BUILDING SOLUTIONS LIMITED**

(Company Number 11791668)

In Creditors' Voluntary Liquidation ("the Company")

Registered office: 13c Saint Germans Road, London, Greater London, SE23 1RH

Principal trading address: 13c Saint Germans Road, London, Greater London, SE23 1RH

At a General Meeting of the members of the above named Company, duly convened and held at 10G Highland Road, Bromley BR1 4AD on 30 March 2022 the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound-up voluntarily" and "that *Brendan P. Hogan*, of Cromwell & Co Insolvency Practitioners, 5 Mercia Business Village, Torwood Close, Coventry, CV4 8HX, be appointed Liquidator of the Company."

Office Holder Details: Brendan P. Hogan (IP No 13030) of Cromwell & Co Insolvency Practitioners, 5 Mercia Business Village, Torwood Close, Coventry, CV4 8HX. Date of Appointment: 30 March 2022. For further details contact Brendan P. Hogan on 0800 061 4002 or by Email: info@cromwellinsolvency.co.uk

Vasile Cosmin Pislariu, Chairperson

(4035184)

(4035483)

## **CREATIVE AUTOMATION LIMITED**

(Company Number 06456004)

Previous Name of Company: Gorman Dynamics Ltd

Registered office: Unit F10, Mills House, Mills Way, Amesbury, SP4 7RX

Principal trading address: Unit 10 Avro Business Park, Mosquito Way, Christchurch. BH23 4FN

Notice is hereby given that the following resolutions were passed on 29 March 2022, as a Special resolution and an Ordinary resolution respectively:

"That the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the company be wound up voluntarily and that *Hugh Francis Jesseman* (IP No. 9480) and *Claire Howell* (IP No. 23610) both of Antony Batty & Company LLP, 3 Field Court, Gray's Inn, London, WC1R 5EF be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: The Liquidator, Tel: 01722 685101. Alternative contact: Alison Gates.

Peter Gorman, Director

29 March 2022

Ag OH123343

#### **CYGNET SPIRITS LIMITED**

(Company Number 09518062)

Registered office: 10 St Helen's Road Swansea SA1 4AW

At a meeting of the above named company duly convened and held on 29 March 2022, the following resolutions were passed:

That it has been proved to the satisfaction of this meeting that the company cannot, by reason of its current and/or impending liabilities, continue its business, and that the company be wound up the under a Creditors Voluntary Liquidation.

That Gareth Bishop and Helen Whitehouse of McAlister & Co Insolvency Practitioners Limited, 10 St Helen's Road, Swansea, SA1 4AW be appointed Joint Liquidators for the purposes of the voluntary winding up of the company and that they be authorised to act jointly and separately. The appointment of Gareth Bishop and Helen Whitehouse was confirmed by a deemed consent process which was completed on 29 March 2022.

Joint Liquidator: Gareth Bishop (IP number 17870) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1 4AW

Joint Liquidator: *Helen Whitehouse* (IP number 9680) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1 4AW.

Date of Appointment: 29 March 2022

For further details contact Gareth Bishop on 03300 563600 or at

Gareth@mcalisterco.co.uk

Joanne Begley, Director (4037350)

#### DANNIC LTD

(Company Number 09468701)

Registered office: 6 Northcraft Court, 61 Becklow Road, Shepherds Bush, London W12 9HE

Principal trading address: N/A

At a General Meeting of the Members of the above-named Company, duly convened and held at Redheugh House, Teesdale South, Thornaby Place, Stockton on Tees, TS17 6SG on 15 March 2022 the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Kenneth Pattullo* (IP No. 8368) and *Ian James Royle* (IP No. 18934) both of Begbies Traynor (Central) LLP, Redheugh House, Teesdale South, Thornaby Place, Stockton on Tees, TS17 6SG be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 01642 796 640. Alternatively enquiries can be made to Stephanie Breckon by email at Stephanie.Breckon@bgtuk.com or by telephone on 01642 796 640. *Kevin Manley*, Chair

15 March 2022

Ag OH123299 (4035589)

# **ENKO JOINERY LTD**

(Company Number 08423086)

Previous Name of Company: Enko Building Services Ltd

Registered office: Unit 9 Mead Park Industrial Estate, Harlow, CM20 2SE and it is in the process of being changed to Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE Principal trading address: N/A

At a General Meeting of the above named Company duly convened and held at Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE on 10 March 2022 at 3.00 pm, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution:

"That the Company be wound up voluntarily and that *Martin Weller* (IP No. 9419) and *Glyn Mummery* (IP No. 8996) both of FRP Advisory Trading Limited, Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE be and are hereby appointed Liquidators for the purposes of such winding up."

Further details contact: The Joint Liquidators, Email: cp.brentwood@FRPadvisory.com

Georghe Rusu, Chair

10 March 2022

Ag OH123272

(4035485)

#### FITNESS DATA SCIENCE LIMITED

(Company Number 11149327)

Registered office: Rownhams House, Betteridge Drive, Rownhams, Southampton, SO16 8LS

Principal trading address: Rownhams House, Betteridge Drive, Rownhams, Southampton, SO16 8LS

Notice is hereby given that the following written resolutions were passed on 28 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Timothy Frank Corfield* (IP No. 8202) of Griffin & King, 26/28 Goodall Street, Walsall, West Midlands, WS1 1QL be nominated Liquidator for the purposes of the voluntary winding up."

Further details contact: Timothy Frank Corfield, Email: enquiries@griffinandking.co.uk or by Tel: 01922 722205.

Adrian Murphy, Director

30 March 2022

Ag OH123241

(4035597)

#### FLEET ASSISTANCE SERVICES LTD

(Company Number 12110115)

Registered office: c/o Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, 30 Churchgate, Bolton, BL1 1HI

Principal trading address: None

Nature of Business: Maintenance and repair of motor vehicles.

Type of Liquidation: Creditors' Voluntary.

Place of meeting: 4th Floor Churchgate House, Bolton, BL1 1HL.

Date of meeting: 30 March 2022.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 30 March 2022

Liquidator's Name and Address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, Bolton, BL1 1HL. Telephone: 01204 255 051.

For further information contact Jasmine Baxter at the offices of Anderson Brookes Insolvency Practitioners Limited on 01204 255 051, or jasmine@andersonbrookes.co.uk.

31 March 2022 (4036402)

#### **FOURMAN LTD**

(Company Number 05942718)

Registered office: Unit 1 Grange Farm, Doctor Hill, Wheatley, Halifax HX2 0TG

Principal trading address: Unit 1 Grange Farm, Doctor Hill, Wheatley, Halifax HX2 0TG

Notice is hereby given that the following resolutions were passed on 28 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *J N Bleazard* (IP No. 09354) of XL Business Solutions, Premier House, Bradford Road, Cleckheaton, BD19 3TT be appointed as Liquidator for the purposes of such voluntary winding up."

Further details contact: Jeremy Bleazard, Tel: 01274 870101. Alternative contact: G Harsley.

W Durkin, Director 30 March 2022 Ag OH123295

(4035543)

#### **GUARDIAN HOMES & DRIVEWAYS LTD**

(Company Number 11751748)

Registered office: Glencote Office Station Road, Cheddleton, Leek, ST13 7EE (to be changed to c/o FRP, Ashcroft House, Ervington Court, Meridian Business Park, Leicester, LE19 1WL)

Principal trading address: Glencote Office Station Road, Cheddleton, Leek, ST13 7EE

At a General Meeting of the above named Company, duly convened, and held at FRP Advisory Trading Limited, Ashcroft House, Ervington Court, Meridian Business Park, Leicester, LE19 1WL on 28 March 2022 at 4.00 pm, the following resolutions were duly passed as Special and Ordinary Resolutions:

"That the Company be wound up voluntarily and that Yasmin Bhikha (IP No. 19972) and John Anthony Lowe (IP No. 009513) both of FRP Advisory Trading Limited, Ashcroft House, Ervington Court, Meridian Business Park, Leicester, LE19 1WL be and are hereby appointed Liquidators for the purposes of such winding up and that anything required or authorised to be done by the Liquidators be done by both or either of them."

Further details contact: The Joint Liquidators, Email: cp.leicester@frpadvisory.com

Louis Burnside, Chair

28 March 2022

Ag OH123247 (4035588)

#### **HOLIDAY DEALERS LIMITED**

(Company Number 11762558)

Registered office: Rose Villa, 63 Bank Street, Ossett, WF5 8PR

Principal trading address: Rose Villa, 63 Bank Street, Ossett, WF5 8PR

Notice is hereby given that the following resolutions were passed on 25 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Nicola Baker* (IP No. 15852) of Rushtons Insolvency Limited, 3 Merchant's Quay, Ashley Lane, Shipley, BD17 7DB be appointed Liquidator for the purposes of such voluntary winding up."

Further details contact: Nicola Baker, Tel: 01274 598585. Alternative contact: Email: rwordsworth@rushtonsifs.co.uk

Paul McCann, Director

25 March 2022

Ag OH123232

(4035475)

#### **HUNSBURY TARMACADAM AND PAVING LTD**

(Company Number 09982993)

Registered office: 100 St James Road, Northampton, NN5 5LF Principal trading address: 8 Gresham Drive, Northampton, NN4 9SB

Nature of Business: Civil engineering projects. Type of Liquidation: Creditors' Voluntary.

Place of meeting: 100 St James Road, Northampton, NN5 5LF.

Date of meeting: 28 March 2022.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Joint Liquidators for the purposes of the winding-up.

Date of Appointment: 28 March 2022

Joint Liquidator's Name and Address: Lauren Louise Auburn (IP No. 20894) of BRI Business Recovery and Insolvency, 100 St James Road, Northampton, NN5 5LF. Telephone: 01604 754352.

Joint Liquidator's Name and Address: *John William Rimmer* (IP No. 13836) of BRI Business Recovery and Insolvency, 100 St James Road, Northampton, NN5 5LF. Telephone: 01604 754352.

For further information contact Rebecca Boston at the offices of BRI Business Recovery and Insolvency on 01604595613, or rboston@briuk.co.uk.

29 March 2022 (4036374)

#### I VISION CONSTRUCTION LTD

(Company Number 13004536)

Registered office: c/o Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, 30 Churchgate, Bolton, BL1 1HL

Principal trading address: None

Nature of Business: Construction of domestic buildings.

Type of Liquidation: Creditors' Voluntary.

Place of meeting: 4th Floor Churchgate House, Bolton, BL1 1HL.

Date of meeting: 30 March 2022.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 30 March 2022

Liquidator's Name and Address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 4th Floor Churchgate House, Bolton, BL1 1HL. Telephone: 01204 255 051.

For further information contact Laura Wright at the offices of Anderson Brookes Insolvency Practitioners Limited on 01204255051, or laura@andersonbrookes.co.uk.

31 March 2022 (4036397)

(4035505)

#### INTEX SITE SOLUTIONS LTD

(Company Number 09114316)

Registered office: 36 Ruby Street, Wakefield, West Yorkshire, WF1 2GP

Principal trading address: 36 Ruby Street, Wakefield, West Yorkshire, WF1 2GP

At a General Meeting of the members of the above named company, duly convened and held at 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ on 29 March 2022 the following resolutions were duly passed; as a Special Resolution and as an Ordinary Resolution respectively:

1. That the Company be wound up voluntarily.

2. That Philip David Nunney and Joanne Louise Hammond of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter Row, Sheffield. S1 3FZ, be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of liquidator from time to time.

Joint Liquidator: *Philip David Nunney* (IP number 9507) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House 60 Charter Row Sheffield S1 3FZ.

Joint Liquidator: *Joanne Louise Hammond* (IP number 17030 ) of Begbies Traynor (SY) LLP , 3rd Floor, Westfield House 60 Charter Row Sheffield S1 3FZ .

Date of Appointment: 29 March 2022

For further details contact Shannon Jones on 0114 275 5033 or at sheffield.north@btguk.com

Daniel Barrett, Chair (4036418)

# NOTICE OF WRITTEN RESOLUTIONS (SPECIAL AND ORDINARY) PURSUANT TO PART 13 OF THE COMPANIES ACT 2006 AND SECTIONS 84(1)(B) AND 85(1) OF THE INSOLVENCY ACT 1986 INVESTABLY LIMITED

(Company Number 11707395)

(Registered in England and Wales) ("the Company")

Registered office: Kings Parade, Lower Coombe Street, Croydon, Surrey, CR0 1AA.

Notification of written resolutions of the above-named Company proposed by the sole director and having effect as a special resolution and as an ordinary resolution respectively pursuant to the provisions of Part 13 of the Companies Act 2006.

Circulation Date: 14 March 2022 Effective Date: 24 March 2022

I, the undersigned, being the sole director of the Company hereby certify that the following written resolutions were circulated to all eligible members of the Company on the Circulation Date and that the written resolutions were passed on the Effective Date:

"That the Company be wound up voluntarily and that Paul Wood and Simon Robert Haskew both of Begbies Traynor (Central) LLP of St James Court, St James Parade, Bristol, BS1 3LH be and are hereby appointed as joint liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Paul Wood (IP Number: 009872) and Simon Robert Haskew (IP Number: 008988).

Any person who requires further information may contact the Joint Liquidator either by e-mail at bristol@btguk.com or by telephone on 0117 937 7130. Alternatively, enquiries can be made to Clive Hobbs by e-mail at clive.hobbs@btguk.com or by telephone on 0117 937 7148.

Colum Jeffrey Arment Smith

Director

Dated: 24 March 2022 (4037802)

#### **IODETAIL LIMITED**

(Company Number 11749343)

Registered office: Unit D6, Seedbed Centre, Vanguard Way, Shoeburyness, Southend-on-Sea, SS3 9QY

Principal trading address: (Previously) Unit D6, Seedbed Centre, Vanguard Way, Shoeburyness, Southend-on-Sea, SS3 9QY

At a general meeting of the members of the above named Company, duly convened and held at The Old Exchange, 234 Southchurch Road, Southend on Sea, Essex SS1 2EG on 24 March 2022 the following resolutions were duly passed as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Wayne MacPherson* (IP No. 9445) and *Louise Donna Baxter* (IP No. 009123) both of Begbies Traynor (Central) LLP, The Old Exchange, 234 Southchurch Road, Southend-on-Sea, Essex, SS1 2EG be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 01702 467255. Alternatively enquiries can be made to lan Goodhew by e-mail at southendadmin@btguk.com or by telephone on 01702 443 923.

Victoria Morris Chapman, Chair

24 March 2022

Ag OH123349

#### JKC CONTRACTORS LIMITED

(Company Number 10952395)

Registered office: 10a High Street, Market Deeping, Peterborough, PE6 8EB

Principal trading address: (Formerly) Unit 8, Casterton Business Park, Casterton Road, Stamford, PE9 4EJ

At a general meeting of the members of the above named Company, duly convened and held at Unit 11, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF on 29 March 2022 the following resolutions were duly passed as a special resolution and as an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Michaela Samantha Daly* (IP No. 17570) and *David Adam Broadbent* (IP No. 009458) both of Begbies Traynor (Central) LLP, Unit 11, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF be and are hereby appointed Joint Liquidators of the Company for the purpose of the voluntary winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the person holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 01782 569 515. Alternatively, enquiries can be made to Helen Lowe by e-mail at Helen.Lowe@btguk.com or by telephone on 01782 569 515.

James Clipston, Chair

29 March 2022

Ag OH123259

(4035531)

# KEITH STEVENSON AND ASSOCIATES LIMITED

(Company Number 05929481)

Registered office: 16 Calthwaite Close, Hylton Castle Estate, Sunderland, SR5 3QR

Principal trading address: (Formerly) 3 Haswell Court, Quaywest Business Village, Sunderland, SR5 2AR

At a General Meeting of the Members of the above-named Company, duly convened and held at Redheugh House, Teesdale South, Thornaby Place, Stockton on Tees, TS17 6SG on 24 March 2022 the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Ian James Royle* (IP No. 18934) and *David Adam Broadbent* (IP No. 9458) both of Begbies Traynor (Central) LLP, Redheugh House, Teesdale South, Thornaby Place, Stockton on Tees, TS17 6SG be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 01642 796 640. Alternatively enquiries made to Stephanie Breckon by email stephanie.breckon@bgtuk.com or by telephone on 01642 796 640. Keith Stevenson, Chair

24 March 2022

Ag OH123256 (4035404)

#### **LANGDON TRANSPORT LIMITED**

(Company Number 04661798)

Registered office: 38 Central Avenue, Lee Mill Industrial Estate, Ivybridge, PL21 9PE

Principal trading address: 38 Central Avenue, Lee Mill Industrial Estate, Ivybridge, PL21 9PE

At a General Meeting of the above-named Company, duly convened, and held at 5 Barnfield Crescent, Exeter, Devon, EX1 1QT on 28 March 2022, at 10.00 am, the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily, and that David Gerard Kirk (IP No. 8830) and Daniel Robert Jeeves (IP No. 26032) both of Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT be and are hereby appointed Joint Liquidators for the purpose of such winding-up."

For further details contact: Nathan Jeeves, Tel: 01392 474303 or Email: nathan@kirks.co.uk.

G G Witts-Davies, Chair

28 March 2022

Ag OH123274

(4035578)

#### **LB ALLOYS LIMITED**

(Company Number 12274496)

Registered office: First Floor, Enterprise House, 202-206 Linthorpe Road, Middlesbrough, Cleveland, TS1 3QW

Principal trading address: Todd Point Road, Warrenby, TS10 5AU Notice is hereby given that the following resolutions were passed on 29 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that David James Robson (IP No. 11130) of Gaines Robson Insolvency Ltd, Carrwood Park, Selby Road, Leeds, LS15 4LG be appointed as Liquidator for the purposes of such voluntary winding up.

Contact details for Liquidator, Email: David@gainesrobson.co.uk. Alternative contact: Lynsey Haigh.

Lee Blackburn, Director

30 March 2022

Ag OH123239 (4035413)

#### LITTLEGOLD CONSULTING LIMITED

(Company Number 09458931)

Registered office: White Hart House, Silwood Road, Ascot, SL5 0PY Principal trading address: White Hart House, Silwood Road, Ascot, SL5 0PY

At a General Meeting of the Members of the above-named company, duly convened and held on 29 March 2022 at 12.00 noon the following resolutions were passed as a special resolution and as an ordinary resolution respectively:

"That the Company be wound up voluntarily and that Ashok Bhardwaj (IP No. 4640) of Bhardwaj Limited, 47/49 Green Lane, Northwood, Middlesex, HA6 3AE be appointed Liquidator of the Company.

Further details contact: Linda Tuci, Tel: 01923 820966 or by email at linda@bhardwaj.co.uk.

Sally Nuza, Director 16 March 2022 Ag OH123291

(4035416)

#### LK BUILDS LTD

(Company Number 12340206)

Registered office: 72 Lansdown Road, Swindon, SN1 3NF Principal trading address: 72 Lansdown Road, Swindon, SN1 3NF At a general meeting of the members of the above named Company held at 72 Lansdown Road, Swindon, SN1 3NF on 30 March 2022, the following resolutions were passed as a special resolution and an ordinary resolution:

"That the Company be wound up voluntarily and that Daniel Taylor (IP No. 21050) of Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE be and is hereby appointed Liquidator for the purposes of such winding up. Any act required or authorised under any enactment to be done by a liquidator may be done by them."

For further details contact: Daniel Taylor, Tel: 0161 694 9955. Alternative contact: Jennifer Rayner.

Lyubomir Tomov, Director

30 March 2022

Ag OH123324

(4035577)

#### **M&F PROPERTY MAINTENANCE LTD**

(Company Number 11946584)

Registered office: 57 Havering Gardens, Romford, RM6 5BH

Principal trading address: 57 Havering Gardens, Romford, RM6 5BH At a meeting of the members of the above named Company duly convened and held at c/o Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE on 29 March 2022, the following resolutions were passed as a special resolution and an

ordinary resolution: "That the Company be wound up voluntarily and that Daniel Taylor (IP No. 21050) of Fortis Insolvency Limited, 683-693 Wilmslow Road,

Manchester, M20 6RE be and is hereby appointed as Liquidator of the Company for the purposes of such winding up."

For further details contact: Liam O'Connor, Tel: 0161 694 9955 Dumitru Mujdar, Chair

29 March 2022

Ag OH123336

(4035541)

#### MAK POWER EUROPE LIMITED

(Company Number 09622554)

Registered office: Basepoint Business Centre c/o Meon Accounting Limited, Premier Way, Abbey Park Industrial Estate, Romsey, Hampshire SO51 9AQ to be changed to The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1 DH

Principal trading address: Basepoint Business Centre c/o Meon Accounting Limited, Premier Way, Abbey Park Industrial Estate, Romsey, Hampshire SO51 9AQ

Notice is hereby given that the following resolutions were passed on 29 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that David Meany (IP No. 9453) and Tauseef A Rashid (IP No. 9718) both of Quantuma Advisory Limited, The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH be and are hereby appointed Joint Liquidators for the purpose of such winding up. Any act required or authorised under any enactment to be done by a liquidator may be done by one of them.

Further details contact: Caroline Barratt, Tel: 01202 059 463 or Email: caroline.barratt@guantuma.com

David Streather, Director

30 March 2022

Aq OH123318

(4035574)

#### MAVENS PROPERTY SERVICES LTD

(Company Number 10819507)

Previous Name of Company: XPM LIMITED

In Creditors' Voluntary Liquidation ("the Company")

Registered office: 83 Galaxy Buildings Crews Street, London, E14 3SP

Principal trading address: 83 Galaxy Buildings Crews Street, London,

At a General Meeting of the members of the above named Company, duly convened and held at 5 Mercia Business Village, Torwood Close, Coventry CV4 8HX on 28 March 2022 the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution

"That the Company be wound-up voluntarily" and "that Brendan P. Hogan, of Cromwell & Co Insolvency Practitioners, 5 Mercia Business Village, Torwood Close, Coventry, CV4 8HX, be appointed Liquidator of the Company."

Office Holder Details: Brendan P. Hogan (IP No 13030) of Cromwell & Co Insolvency Practitioners, 5 Mercia Business Village, Torwood Close, Coventry, CV4 8HX. Date of Appointment: 28 March 2022. For further details contact Brendan P. Hogan on 0800 061 4002 or by Email: info@cromwellinsolvency.co.uk

Denise Joan Flacks, Chairperson

(4037799)

#### MICK WHEELER CONSTRUCTION LIMITED

(Company Number 07799624)

Registered office: Matrix House, 12-16 Lionel Road, Canvey Island, Essex, SS8 9DE

Principal trading address: 61 Brentwood Road, Ingrave, Brentwood, Essex, CM13 3RE

At a General Meeting of the members of the above named company, duly convened and held at 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ on 29 March 2022 the following resolutions were duly passed; as a Special Resolution and as an Ordinary Resolution respectively:-

1. That the Company be wound up voluntarily.

2. That Philip David Nunney and Joanne Louise Hammond of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of liquidator from time to time.

Joint Liquidator: *Philip David Nunney* (IP number 9507) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House 60 Charter Row Sheffield S1 3FZ.

Joint Liquidator: Joanne Louise Hammond (IP number 17030 ) of Begbies Traynor (SY) LLP , 3rd Floor, Westfield House 60 Charter Row Sheffield S1 3FZ .

Date of Appointment: 29 March 2022

For further details contact Shannon Jones on 0114 275 5033 or at sheffield.north@btguk.com

Michael Philip Wheeler, Chair (4036563)

# **OTI'S RUBBISH COLLECTION LIMITED**

(Company Number 11073215)

Registered office: Olympia House, Armitage Road, London, NW11 8RO

Principal trading address: N/A

At a General Meeting of the above named Company duly convened and held at Olympia House, Armitage Road, London, NW11 8RQ on 29 March 2022, the following resolutions were duly passed as a special resolution and ordinary resolution:

"That the Company be wound up voluntarily and that Jason Callender (IP No. 22650) of Panos Eliades Franklin & Co, Olympia House, Armitage Road, London, NW11 8RQ be appointed Liquidator of the Company for the purposes of the winding up."

Further details contact: Mr P Tomasino, Tel: 0208 731 6807.

Attila-Csaba Szeri, Chair

29 March 2022

Ag OH123258 (4035507)

#### PINEOAKIO FURNITURE LIMITED

(Company Number 07399873)

Registered office: M.R. Insolvency, 95 Pobox, Heckmondwike, West Yorkshire WF16 6AU

Principal trading address: Unit 6 Parkside Works, Parkwood Street, Keighley, BD21 4PJ

Notice is hereby given, that the following resolutions were passed on 30 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company cannot by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that *Michael Royce* (IP No. 9692) of M.R. Insolvency, PO Box 95, Heckmondwike, WF16 6AU be appointed as Liquidator for the purposes of such voluntary winding up."

Further details contact: Michael Royce, Tel: 07921 577752, Email: mr@mrinsolvency.co.uk

Michael Charles Bower, Director

30 March 2022

Ag OH123359

(4035403)

#### PROTECTION OF HUMAN RIGHTS IN PUBLIC LAW

(Company Number 07068071)

Registered office: City View House, 5 Union St, Manchester, M12 4JD to be changed to c/o Quantuma Advisory Limited, 14 Derby Road, Stapleford, Nottingham, NG9 7AA

Principal trading address: City View House, 5 Union St, Manchester, M12 4JD

Notice is hereby given that the following resolutions were passed on 28 March 2022, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Rehan Ahmed* (IP No. 20390) and *Tauseef A Rashid* (IP No. 9718) both of Quantuma Advisory Limited, 14 Derby Road, Stapleford, Nottingham, NG9 7AA be and are hereby appointed Joint Liquidators for the purpose of such winding up. Any act required or authorised under any enactment to be done by a Liquidator may be done by one of them."

Further details contact: Manraj Mand, Email: Manraj.Mand@quantuma.com, Tel: 0115 939 2454.

Mohammed Malik, Director

30 March 2022

Ag OH123238

(4035506)

# **QUINT SOLUTIONS LIMITED**

(Company Number 08719104)

Registered office: Caci House, 9 Spring Villa Park, Spring Villa Road, Edgware, HA8 7EB

Principal trading address: Caci House, 9 Spring Villa Park, Spring Villa Road, Edgware, HA8 7EB

Notice is hereby given that the following resolutions were passed on 28 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that Natasha Brodie (IP No. 19490) and Adam Harris (IP No. 15454) both of Mazars LLP, Tower Bridge House, St Katharine's Way, London, E1W 1DD be and they are hereby appointed Joint Liquidators for the purpose of such winding-up and that they be authorised to act jointly and severally." For further details contact: The Joint Liquidators, Tel: 0207 063 4384.

Kevin Kapinga, Director

30 March 2022

Ag OH123244 (4035407)

# RAVENBROOK (HOLDINGS) LIMITED

(Company Number 13089087)

Registered office: Units 8 - 10 Eyston Way, Abingdon, OX14 1TR and it is in the process of being changed to Office D, Beresford House, Town Quay, Southampton, SO14 2AQ

Principal trading address: Units 8 - 10 Eyston Way, Abingdon, OX14 1TR

Notice is hereby given that the following resolutions were passed on 24 March 2022, as a Special resolution and an Ordinary resolution respectively:

"That the Company be wound up voluntarily and that Andrew Watling (IP No. 15910) and Simon Campbell (IP No. 10150) both of Quantuma Advisory Limited, Office D, Beresford House, Town Quay, Southampton, SO14 2AQ be and is hereby appointed Liquidator for the purpose of such winding up. Any act required or authorised under any enactment to be done by a liquidator may be done by one of them."

Further details contact: Andrea Terraneo, Email: Andrea.Terraneo@quantuma.com, Tel: 02382 356 931.

Michael Geoffrey Nash, Director/Convener

29 March 2022

Ag OH123344 (4035391)

#### REMGENT LTD

(Company Number 11187146)

Registered office: Astral Bradshaw Lane, Bradshaw, Halifax, HX2 9XB Principal trading address: Astral Bradshaw Lane, Bradshaw, Halifax, HX2 9XB

At a General Meeting of the above-named Company, duly convened, and held at the offices of Poppleton & Appleby, The Media Centre, 7 Northumberland Street, Huddersfield, HD1 1RL on 29 March 2022 the following resolutions were passed as a Special resolution and as an Ordinary resolution:

"That the Company be wound up voluntarily, and that *Charles Michael Brook* (IP No. 9157) and *Allan Christopher Cadman* (IP No. 9522) both of Poppleton & Appleby, The Media Centre, 7 Northumberland Street, Huddersfield, HD1 1RL be and are hereby appointed Joint Liquidators for the purpose of such winding-up."

Further details contact: The Joint Liquidators, Tel: 01484 437 432.

Ross Gent, Chair 29 March 2022

Ag OH123255 (4035573)

#### **RENTONIT LIMITED**

(Company Number 08709485)

Registered office: 29 Headington Road, London, SW18 3PR Principal trading address: 29 Headington Road, London, SW18 3PR Nature of Business: Information technology consultancy activities.

Type of Liquidation: Creditors' Voluntary.

Place of meeting: The Offices of Marshall Peters Limited, Heskin Hall Farm, Heskin, Preston, PR7 5PA.

Date of meeting: 28 March 2022.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 28 March 2022

Liquidator's Name and Address: *Clive Morris* (IP No. 8820) of Marshall Peters, Heskin Hall Farm, Wood Lane, Heskin, Preston, PR7 5PA. Telephone: 01257 452021.

For further information contact Heather Dolan at the offices of Marshall Peters on 01257 452021, or heatherdolan@marshallpeters.co.uk.

28 March 2022 (4036376)

#### **RISING SUN HOTEL (UK) LTD**

(Company Number 11212134) Trading Name: The Rising Sun

Registered office: 138 Greenford Road, Sudbury Hill, Harrow, HA1 3QL

Principal trading address: 138 Greenford Road, Sudbury Hill, Harrow, HA1 3QL

At a General Meeting of the above-named Company, duly convened, and held at Suite 9, Amba House, 15 College Road, Harrow, Middlesex, HA1 1BA on 28 March 2022, the following resolutions were duly passed as a special resolution and as an ordinary resolution:

"That the Company be wound up voluntarily and that *Umang Patel* (IP No. 18790) of Neum Insolvency, Suite 9, Amba House, 15 College Road, Harrow, Middlesex, HA1 1BA be and he is hereby appointed Liquidator for the purposes of such winding-up."

Further details contact: Umang Patel, Tel: 020 3411 9598, Email: Umang@neuminsolvency.co.uk. Alternative contact: Harshin Soneji, Email: harshin@neuminsolvency.co.uk.

Mithilini Theepathas, Director

28 March 2022

Ag OH123356

(4035599)

#### SDI (OXFORD STREET HOF) LIMITED

(Company Number 09798784)

Registered office: Unit A, Brook Park East, Shirebrook, NG20 8RY Principal trading address: Unit A, Brook Park East, Shirebrook, NG20 8RY

Notice is hereby given that the following resolutions were passed on 23 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Michael Lennon* (IP No. 24650) of Kroll Advisory Ltd, The Chancery, 58 Spring Gardens, Manchester, M2 1EW and *Benjamin John Wiles* (IP No. 10670) of Kroll Advisory Ltd, The Shard, 32 London Bridge Street, London, SE1 9SG be and are hereby appointed Joint Liquidators of the Company for the purposes of the winding up and that they act jointly and severally."

Further details contact: The Joint Liquidators, Tel: 0161 827 9000. Alternative contact: George Reeves, Email: George.Reeves@kroll.com *Alastair Dick.* Chair

30 March 2022

Ag OH123386

(4035396)

(4035580)

#### SENTINEL ENTERPRISES LIMITED

(Company Number 07523567)

Registered office: The Greenhouse, Barnards Way, Lowestoft, NR32 2HD

Principal trading address: The Greenhouse, Barnards Way, Lowestoft, NR32 2HD

Notice is hereby given that the following resolutions were passed on 25 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the company be wound up voluntarily and that *Adam Henry Stephens* (IP No. 9748) and *Mark Christopher Ford* (IP No. 9521) both of Smith & Williamson LLP, 25 Moorgate, London, EC2R 6AY be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: Cameron Dalrymple-Rockett, Tel: 0207 131 4000, Email: Cameron.Dalrymple-Rockett@smithandwilliamson.com *Andrew Wilson-Sutter,* Director

25 March 2022

Ag OH123363 (4035389)

#### SENTINEL LEISURE TRUST

(Company Number 07521701)

Registered office: The Greenhouse, Barnards Way, Lowestoft, NR32

Principal trading address: The Greenhouse, Barnards Way, Lowestoft, NR32 2HD

Notice is hereby given that the following resolutions were passed on 25 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the company be wound up voluntarily and that *Adam Henry Stephens* (IP No. 9748) and *Kevin Ley* (IP No. 25090) both of Smith & Williamson LLP, 25 Moorgate, London, EC2R 6AY be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: Cameron Dalrymple-Rockett, Tel: 0207 131 4000, Email: Cameron.Dalrymple-Rockett@smithandwilliamson.com

Andrew Wilson-Sutter. Director

25 March 2022

Ag OH123360

#### **SLT ENTERPRISES LIMITED**

(Company Number 11116583)

Registered office: The Greenhouse, Barnards Way, Lowestoft, NR32 2HD

Principal trading address: The Greenhouse, Barnards Way, Lowestoft, NR32 2HD

Notice is hereby given that the following resolutions were passed on  $25\,$  March  $2022\,$  as a special resolution and an ordinary resolution respectively:

"That the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the company be wound up voluntarily and that *Adam Henry Stephens* (IP No. 9748) and *Mark Christopher Ford* (IP No. 9521) both of Smith & Williamson LLP, 25 Moorgate, London, EC2R 6AY be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: Cameron Dalrymple-Rockett, Tel: 0207 131 4000, Email: Cameron.Dalrymple-Rockett@smithandwilliamson.com Andrew Wilson-Sutter, Director

25 March 2022

Ag OH123362 (4035469)

#### **SLT GROUP**

(Company Number 11116012)

Registered office: The Greenhouse, Barnards Way, Lowestoft, NR32 2HD

Principal trading address: The Greenhouse, Barnards Way, Lowestoft, NR32 2HD

Notice is hereby given that the following resolutions were passed on 25 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the company be wound up voluntarily and that *Adam Henry Stephens* (IP No. 9748) and *Mark Christopher Ford* (IP No. 9521) both of Smith & Williamson LLP, 25 Moorgate, London, EC2R 6AY be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: Cameron Dalrymple-Rockett, Tel: 0207 131 4000, Email: Cameron.Dalrymple-Rockett@smithandwilliamson.com *Andrew Wilson-Sutter*, Director

25 March 2022

Ag OH123361 (4035563)

#### **SOUTHGATE CATERING LTD**

(Company Number 10901372) Trading Name: Southgate Kebabs

Registered office: 601 High Road Leytonstone, London, E11 4PA formerly, Southgate Kebabs, 8a Station Parade, London, N14 5BJ Principal trading address: Southgate Kebabs, 8a Station Parade, London, N14 5BJ

At a general meeting of the members of the above named company, duly convened and held at 32 Willoughby Rd, London, N8 0JG on 30 March 2022 the following special resolution was duly passed:

"That the Company be wound-up voluntarily and that George Michael and Harjinder Johal of Ashcrofts, 601 High Road Leytonstone, London, E11 4PA are hereby appointed joint liquidators and that they may act jointly and severally in this regard for the purpose of the winding up"

Joint Liquidator: *George Michael* (IP number 9230) of Ashcrofts, 601 High Road Leytonstone, London, E11 4PA.

Joint Liquidator: *Harjinder Johal* (IP number 9175) of Ashcrofts, 601 High Road Leytonstone, London, E11 4PA.

Date of Appointment: 30 March 2022

For further details contact Amrit Johal on 020 8556 2888 or at info@ashcrofts.co.uk (4037876)

#### TAPASWINI LIMITED

(Company Number 08423910) Trading Name: Chowki

Previous Name of Company: Tapaswani Limited

Registered office: 24 Barnehurst Avenue, Erith, Kent DA8 3NF Principal trading address: (Former) 580-582 Beverley Road, Hull, HU6

At a General Meeting of the members of the above named company, duly convened and held at 580-582 Beverley Road, Hull, HU6 7LH on 23 March 2022 at 10.00 am the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution

"That the Company be wound up voluntarily and that *Andrew Mackenzie* (IP No. 009581) and *Laura Baxter* (IP No. 25350) both of Begbies Traynor (Central) LLP, Unit 8B Marina Court, Castle Street, Hull, HU1 1TJ be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more persons holding the office of liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 01482 483060. Alternatively enquiries can be made to Laura Baxter by email at Laura.Baxter@btguk.com.

Tapan Mahapatra, Chair

23 March 2022

respectively:

Ag OH123285 (4035472)

#### TH GROUP LTD

Trading Name: T/A Taylor Higson (Company Number 11997415)

Registered office: Reedham House, 31 King Street West, Manchester,

M3 2PJ

Principal trading address: Centurion House, 129 Deansgate, Manchester. M3 3WR

Notice is given that by written resolutions, the members of the company passed a special resolution that the company be wound up voluntarily, and an ordinary resolution appointing the Joint Liquidators for the purposes of the winding-up. The requisite voting majority was received on 22 March 2022

Jonathan James Taylor, Director.

22 March 2022

Joint Liquidator's Name and Address: *James Fish* (IP No. 021390) of Cowgill Holloway Business Recovery LLP, Regency House, 45-53 Chorley New Road, Bolton, BL1 4QR. Telephone: 0161 827 1200.

Joint Liquidator's Name and Address: *Alan Coleman* (IP No. 009402) of Cowgill Holloway Business Recovery LLP, Regency House, 45-53 Chorley New Road, Bolton, BL1 4QR. Telephone: 0161 827 1200.

For further information contact Ben Gray at the offices of Cowgill Holloway Business Recovery LLP on 0161 672 5767, or Ben.Gray@cowgills.co.uk.

29 March 2022 (4036382)

# RESOLUTIONS

#### SPOTLESS CLEAN AND SHINE LTD

(Company Number 10866620)

Registered office: Rotunda Buildings, Montpellier Exchange, Cheltenham, GL50 1SX

Principal trading address: 5 Oswald Drive, Longford, GL2 9FW

At a General Meeting of the members of the above named company, duly convened and held at Rotunda Buildings, Montpellier Exchange, Cheltenham, GL50 1SX on 28 March 2022 the following special resolution was duly passed:

"That the Company be wound up voluntarily and that Richard Frank Simms of F A Simms & Partners Limited, Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB is hereby appointed liquidator for the purpose of the winding up"

Further Details: Maria Navratilova Tel No. 01455 555 444

Name of Office Holder: Richard Frank Simms

Office Holder Number: 9252

Address of Office Holder: Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB

Rhian Catherine Evans (4037803)

# RESOLUTION - CREDITORS' VOLUNTARY LIQUIDATION INSOLVENCY ACT 1986 - SECTION 84(1)(B) THE BEAR HOTEL AT WANTAGE LIMITED

(Company Number 07431339) Trading Name: The Bear Hotel

Registered office: The Bear Hotel Market Place, Wantage, OX12 8AB Principal trading address: The Bear Hotel Market Place, Wantage, OX12 8AB

Name of office holder: Peter Richard James Frost

Office holder IP number: 008935

Postal address of office holder: Staverton Court, Staverton, Cheltenham, GL51 0UX

Office holder's telephone no and email address: 01242 680000 and creditors@hazlewoods.co.uk

Alternative contact for enquiries on proceedings: Gina Clare

At a general meeting of the above name company, duly convened and held at Staverton Court, Staverton GL51 0UX on 22 March 2022 the subjoined Special Resolutions were passed "That the Company be wound up voluntarily" and "That Peter Richard James Frost of Hazlewoods LLP, Staverton Court, Staverton, Cheltenham, GL51 0UX, be appointed Liquidator of the Company" on 22 March 2022 for the purposes of such winding up.

Name of Director: Joanne Walkley-Pratt

(4037801)

#### THE BUSINESS PORTFOLIO UK LTD

(Company Number 06145179)

Registered office: Saxon House, Saxon Way, Cheltenham GL52 6QX Principal trading address: 15 Poplar Road, Oxton, Prenton, Mersevside, CH43 5TB

At a General Meeting of the members of the above named Company, duly convened and held at Saxon House, Saxon Way, Cheltenham GL52 6QX on 30 March 2022 the following resolutions were passed as a Special Resolution and as an Ordinary Resolution respectively:

"That it has been proved to the satisfaction of this meeting that the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that *Alisdair J Findlay*, of Findlay James, Saxon House, Saxon Way, Cheltenham GL52 6QX, (IP No. 008744) be and he is hereby appointed Liquidator for the purposes of such winding up."

For further details contact: Alisdair J Findlay, info@findlayjames.co.uk Tel: 01242 576555

Kingsley Taylor, Chairman

#### THE CASPI PUB COMPANY LIMITED

(Company Number 08316478)

Registered office: 59 Dexter Close, Luton, LU3 4DY

Principal trading address: 59 Dexter Close, Luton, LU3 4DY

Notice is given that by written resolutions, the sole member of the company passed a special resolution that the company be wound up voluntarily, and an ordinary resolution appointing the Joint Liquidators for the purposes of the winding-up. The requisite voting majority was received on 28/03/2022

Amer Bashlawi, Director,

28 March 2022

Joint Liquidator's Name and Address: *Alessandro Sidoli* (IP No. 14270) of Kay Johnson Gee Corporate Recovery Limited, 1 City Road East, Manchester, M15 4PN. Telephone: 0161 832 6221.

Joint Liquidator's Name and Address: *Peter Anderson* (IP No. 15336) of Kay Johnson Gee Corporate Recovery Limited, 1 City Road East, Manchester, M15 4PN. Telephone: 0161 832 6221.

Joint Liquidator's Name and Address: *Alan Fallows* (IP No. 9567) of Kay Johnson Gee Corporate Recovery Limited, 1 City Road East, Manchester, M15 4PN. Email: alanfallows@kjgcr.com. Telephone: 0161 832 6221.

For further information contact Lindsay Moore at the offices of Kay Johnson Gee Corporate Recovery Limited on 0161 212 8388, or lindsaymoore@kjgcr.com.

29 March 2022 (4036362)

#### THE CLOSED CIRCUIT TELEVISION COMPANY LIMITED

(Company Number 03264312)

Registered office: 75 Haverstock Hill, Hampstead, London, NW3 4SL Principal trading address: 47 Michleham Down, Woodside Park, London, N12 7JJ

Notice is hereby given that the following resolutions were passed on 30 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Mark Reynolds* (IP No. 8838) of Valentine & Co, 1st Floor Galley House, Moon Lane, Barnet, EN5 5YL be appointed Liquidator of the Company for the purposes of the voluntary winding-up."

The appointment of Mark Reynolds of Valentine & Co, Galley House, Moon Lane, Barnet, EN5 5YL as Liquidator was confirmed by the creditors on the same day.

Further details contact: Mark Reynolds, Tel: 020 8343 3710. Alternative contact: Natalie Ellis.

Graham Simon Chweidan, Director

30 March 2022

Ag OH123329 (4035419)

# THE CURTAIN LADY INTERIORS LIMITED

(Company Number 07356098)

Registered office: 18 Hyde Gardens, Eastbourne, BN21 4PT Principal trading address: 11-13 Revelstoke Road, London, SW18

5NJ

At a General Meeting of the above named Company, duly convened, and held at 11-13 Revelstoke Road, London, SW18 5NJ on 29 March 2022, the following resolutions were passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That it has been proved to the satisfaction of this meeting that the Company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that *Ian Yerrill* (IP No. 8924) of Yerrill Murphy, Gateway House, Highpoint Business Village, Henwood, Ashford, Kent, TN24 8DH be and is hereby appointed Liquidator for the purposes of such winding up."

For further details contact: Ian Yerrill, Email: mail@yerrillmurphy.co.uk. *Nigel Meehan.* Chair

(4035379)

29 March 2022

(4036176)

Ag OH123230

#### THE EDGBASTON HOTEL LIMITED

(Company Number 08151595)

Registered office: Highdown House, 11 Highdown Road, Learnington Spa. CV31 1XT

Principal trading address: 18 Highfield Road, Birmingham, B15 3DU At a General Meeting of the above named Company, duly convened and held at Wilton Drive, Tournament Fields, Warwick CV34 6RG on 23 March 2022 at 10.30 am the following resolutions were passed as a Special Resolution and Ordinary Resolution respectively:-

"That the Company be wound up voluntarily and that *Gavin Bates* (IP No. 8983) of Smart Business Recovery Limited, The Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG be appointed Liquidator of the Company."

Further details contact: Sharon Hill, Tel: 0116 2325117, Email: sharon@smartbusinessrecovery.co.uk.

D Insall, Chair 31 March 2022

Ag OH123385 (4035462)

# THE EPSOM EXPLORE COACH CO LTD.

(Company Number 10717959)

Trading Name: Epsom Coaches

Previous Name of Company: Epsom Coaches Limited

Registered office: c/o Harveys Insolvency & Turnaround, 2 Old Bath Road, Newbury, Berkshire RG14 1QL

Principal trading address: 145 Nathan Way, London, SE28 0AB

At a General Meeting of the Members of the above named Company, duly convened and held at 145 Nathan Way, London, SE28 0AB on 30 March 2022, at 12.00 noon the following resolutions were duly passed, as a special resolution and as an ordinary resolution:

"That it has been proved to the satisfaction of this meeting that the Company cannot by reason of its liabilities continue its business and that it is advisable to wind up the same and accordingly that the Company be wound up voluntarily and that *Debi Harvey* (IP No. 12150) of Harveys Insolvency & Turnaround Limited, 2 Old Bath Road, Newbury, Berkshire, RG14 1QL be appointed as Liquidator of the Company for the purposes of the voluntary winding up".

For further details contact: Tel: 01635 770941 or Email: info@harveyinsolvency.co.uk

Peter Lewis, Director 30 March 2022

Ag OH123300

23300 (4035445)

### THE GRAINSTORE BREWERY LIMITED

(Company Number 07241477)

Registered office: 110 Regent Road, Leicester LE1 7LT

Principal trading address: Station Approach, Oakham, Rutland, LE15 6RE

At a general meeting of the Members of the above-named company, duly convened, and held at 126 New Walk, Leicester, LE1 7JA on 30 March 2022, the following resolutions were passed by the Members: 1 as a Special resolution and 2 as an Ordinary resolution.

esolutions

"That the Company be wound up voluntarily" and

2 "That Neil Money of CBA Business Solutions, 126 New Walk, Leicester, LE1 7JA, be appointed Liquidator of the Company" Liquidator: *Neil Charles Money* (IP number 8900) of CBA Business

Solutions, 126 New Walk, Leicester, LE1 7JA.

Date of Appointment: 30 March 2022

For further details contact CBA Business Solutions on 0116 262 6804 (4037887)

### THE RETREAT NEWPORT LTD

(Company Number 09411765)

Trading Name: The Retreat Cafe (until January 2020)

Registered office: 41 St Michaels Road, St Helens, Ryde, Isle of Wight, PO33 1YJ

Principal trading address: 10 Scarrots Lane, Newport, Isle of Wight, PO33 1JD (until January 2020)

Notice is hereby given that the following resolutions were passed on 29 March 2022 as a special resolution and an ordinary resolution respectively:

"That the company be wound up voluntarily and that *Chris Tate* (IP No. 22910) and *Duncan Swift* (IP No. 8093) both of Azets, Secure House, Lulworth Close, Chandlers Ford, Southampton, SO53 3TL be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: The Joint Liquidators, Tel: 023 8024 7070. Alternative contact: Dawn Sherin

Janice Hansford, Director

29 March 2022

Ag OH123327

(4035595)

#### THE TEACHING ASSISTANT COLLEGE LIMITED

(Company Number 05840111)

Registered office: Unit 36 Staffordshire Business Village, Dyson Way, ST18 0TW

Principal trading address: Unit 36 Staffordshire Business Village, Dyson Way, ST18 0TW

At a general meeting of the members of the above named Company, duly convened and held at Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire ST5 9QF on 30 March 2022 the following resolutions were duly passed as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Michaela Daly* (IP No. 17570) and *Louise Longley* (IP No. 23874) both of Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF be and are hereby appointed Joint Liquidators of the Company for the purpose of the voluntary winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the person holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 01782 569510. Alternatively, enquiries can be made to Samantha Booth by e-mail at samantha.booth@btguk.com

Eva Jean Cartwright, Chair

30 March 2022

Ag OH123376

(4035579)

## **THOMPSON & SHELL TRANSPORT LIMITED**

(Company Number 09916737)

Registered office: 10 St Helen's Road Swansea SA1 4AW

Principal trading address: 1 Mariners Cottages South Shields NE33 2NG

At a meeting of the above named company duly convened and held on 29 March 2022, the following resolutions were passed:

That it has been proved to the satisfaction of this meeting that the company cannot, by reason of its current and/or impending liabilities, continue its business, and that the company be wound up the under a Creditors Voluntary Liquidation.

That Gareth Bishop and Helen Whitehouse of McAlister & Co Insolvency Practitioners Limited, 10 St Helen's Road, Swansea, SA1 4AW be appointed Joint Liquidators for the purposes of the voluntary winding up of the company and that they be authorised to act jointly and separately. The appointment of Gareth Bishop and Helen Whitehouse was confirmed by a deemed consent process which was completed on 29 March 2022.

Joint Liquidator: Gareth Bishop (IP number 17870) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1 4AW

Joint Liquidator: *Helen Whitehouse* (IP number 9680) of McAlister & Co Insolvency Practitioners Limited, 10 St Helens Road, Swansea, SA1 4AW.

Date of Appointment: 29 March 2022

For further details contact Gareth Bishop on 03300 563600 or at gareth@mcalisterco.co.uk

Alan Thompson, Director (4036577)

#### UNDER THE ROCK LTD

(Company Number 12360606)

Registered office: 33 Harrison Road, Halifax, HX1 2AF

Principal trading address: 13-15 Alexandra Street, Halifax, HX1 1BS Notice is hereby given that the following resolutions were passed on 28 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *J N Bleazard* (IP No. 09354) of XL Business Solutions, Premier House, Bradford Road, Cleckheaton, BD19 3TT be appointed as Liquidator for the purposes of such voluntary winding up."

Further details contact: Jeremy Bleazard, Tel: 01274 870101. Alternative contact: G Harsley.

B Adey, Director 30 March 2022

Ag OH123296 (4035486)

## **VIRTUAL DISTRIBUTION LIMITED**

(Company Number 07848768)

Registered office: C/O Ideal Corporate Solutions Limited, Lancaster House, 171 Chorley New Road, Bolton, BL1 4QZ

Principal trading address: Unit 1 Magnesium Way, Hapton, Burnley, BB12 7BF

Type of Liquidation: Creditors' Voluntary.

Place of meeting: Unit 1 Magnesium Way, Hapton, Burnley, BB12 7BF. Date of meeting: 23 March 2022.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Joint Liquidators for the purposes of the winding-up.

Date of Appointment: 23 March 2022

Joint Liquidator's Name and Address: *Andrew David Rosler* (IP No. 9151) of Ideal Corporate Solutions Limited, Lancaster House, 171 Chorley New Road, Bolton, BL1 4QZ.

Joint Liquidator's Name and Address: *Tom Bowes* (IP No. 17010) of Ideal Corporate Solutions Limited, Lancaster House, 171 Chorley New Road, Bolton, BL1 4QZ.

For further information contact Michael Walsh at the offices of Ideal Corporate Solutions Limited on 01204 663000, or michael.walsh@idealcs.co.uk.

25 March 2022 (4036369)

## XANTIVE LIMITED

(Company Number 03891544)

Previous Name of Company: PTCD Limited; Planthinkcheckdo Limited; Identinet Limited

Registered office: Lifford Hall, Lifford Lane, Kings Norton, Birmingham, West Midlands B30 3JN

Principal trading address: 1 Hollyfast Road, Coventry, West Midlands, CV6 2DP

Notice is hereby given that the following resolutions were passed on 28 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Adam Peter Jordan* (IP No. 9616) and *Nickolas Garth Rimes* (IP No. 009533) both of Rimes & Co, 3 The Courtyard, Harris Business Park, Hanbury Road, Stoke Prior, Bromsgrove B60 4DJ be appointed Joint Liquidators of the Company and that they be authorised to act either jointly or separately."

Further details contact: The Joint Liquidators, Email: info@rimesandco.co.uk. Alternative contact: Louise Jordan.

*lan Hippach,* Chair 30 March 2022 Ag OH123248

(4035354)

## Liquidation by the Court

## **APPOINTMENT OF LIQUIDATORS**

In the Leeds District Registry No 000255 of 2021

## **ALBION CONSULTANTS UK LIMITED**

(Company Number 04294741)

Registered office: 19 Grove Road, Gosport, England, PO12 4JH NOTICE IS HEREBY GIVEN pursuant to Rule 7.59 of the Insolvency (England & Wales) Rules 2016 that Ryan Davies of Moore Kingston Smith & Partners LLP was appointed Liquidator of the above named company on 23 March 2022.

Liquidator: Ryan Davies (IP number 15250) of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London EC1M 7AD.

Date of Appointment: 23 March 2022

For further details contact Michael Conway on 0207 566 4020 or at Mconway@mks.co.uk (4036333)

In the High Court of Justice Court Number: CR-2021-001156 AM2PM FELTHAM LTD

(Company Number 08097935)

Registered office: 24-26 The Centre, Feltham, TW13 4AU
Principal trading address: 24-26 The Centre, Feltham, TW13 4AU
Notice is hereby given, pursuant to Rule 7.59 of the Insolvency
(England and Wales) Rules 2016, that a Liquidator has been appointed to the Company by the Secretary of State.

Date of Appointment: 25 February 2022

Office Holder Details: *Laura Anne Walshe* (IP No. 26250) of Keywood Group Limited, Gill House, 140 Holyhead Road, Birmingham, West Midlands, B21 0AF

Further details contact: Email: team@keywoodgroup.co.uk, tel: 0121 201 0399. Alternative contact: Harinder Chahal

Laura Anne Walshe, Liquidator

30 March 2022 Ag OH123284

(4035374)

In the High Court of Justice Court Number: CR-2022-00037 ASHER HOLDINGS LTD

(Company Number 08866096)

Previous Name of Company: A&B Property Group Limited

Registered office: Vinters Business Park, New Cut Road, Maidstone, Kent, ME14 5NZ

Principal trading address: 42 Wolmer Gardens, Edgware, England, HA8 8OD

Notice is hereby given, pursuant to Rule 7.59 of the Insolvency (England and Wales) Rules 2016, that [a Liquidator has been appointed to the Company by the Secretary of State.

Date of Appointment: 24 March 2022

Office Holder Details: *Ruth Ellen Duncan* (IP No. 9246) of Maxwell Davies Limited, Vinters Business Park, New Cut Road, Maidstone, Kent MF14 5N7

Further details contact: Ruth Ellen Duncan, Email: ruth@maxwelldavies.com or Tel: 01622 764612. Alternative contact: Catherine Harrison, Email: catherine@maxwelldavies.com

Ruth Ellen Duncan, Liquidator

24 March 2022 Ag OH123317

(4035568)

In the High Court of Justice, No 000068 of 2022

#### YPG DEVELOPMENTS LTD

(Company Number 09883699)

Registered office: Cwg House, Gallamore Lane, Market Rasen, LN8 3HA

Principal trading address: 14 Columbus Quay, Liverpool, L3 4DB

Notice is hereby given of the appointment of the joint liquidators of the company by the Secretary of State for Business, Enterprise and Industrial Strategy.

Date of Appointment: 24 March 2022 By whom Appointed: Secretary of State

Joint Liquidator's Name and Address: *Craig Johns* (IP No. 013152) of Cowgill Holloway Business Recovery LLP, Regency House, 45-53 Chorley New Road, Bolton, BL1 4QR. Telephone: 0161 827 1200.

Joint Liquidator's Name and Address: Lucinda Clare Coleman (IP No. 10710) of Francis Clark, Centenary House, Peninsula Park, Rydon Lane. Exeter. EX2 7XE.

For further information contact Katie Parker at the offices of Cowgill Holloway Business Recovery LLP on 0161 672 5763, or Katie.Parker@cowgills.co.uk.

29 March 2022 (4036377)

## **NOTICES TO CREDITORS**

In the High Court of Justice Court Number: CR-2021-001156

AM2PM FELTHAM LTD

(Company Number 08097935)

Road, Birmingham, B21 0AF.

Registered office: 24-26 The Centre, Feltham, TW13 4AU Principal trading address: 24-26 The Centre, Feltham, TW13 4AU Notice is hereby given that creditors of the Company are required, on or before 30 April 2022 to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Liquidator at Gill House, 140 Holyhead

If so required by notice from the Liquidator, creditors must produce any document or other evidence which the Liquidator considers is necessary to substantiate the whole or any part of a claim.

Date of Appointment: 25 February 2022

Office Holder Details: *Laura Anne Walshe* (IP No. 26250) of Keywood Group Limited, Gill House, 140 Holyhead Road, Birmingham, West Midlands, B21 0AF

Further details contact: Email: team@keywoodgroup.co.uk, tel: 0121 201 0399. Alternative contact: Harinder Chahal

Laura Anne Walsh, Liquidator

30 March 2022

Ag OH123284 (4035397)

In the High Court of Justice Court Number: CR-2022-000037

**ASHER HOLDINGS LTD** 

(Company Number 08866096)

Previous Name of Company: A&B Property Group Limited

Registered office: Vinters Business Park, New Cut Road, Maidstone, Kent, ME14 5NZ

Principal trading address: 42 Wolmer Gardens, Edgware, England, HA8 8QD

Notice is hereby given that creditors of the Company are required, on or before 30 June 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Liquidator at Vinters Business Park, New Cut Road, Maidstone, Kent, ME14 5NZ. If so required by notice from the Liquidator, creditors must produce any document or other evidence which the Liquidator considers is necessary to substantiate the whole or any part of a claim.

Date of Appointment: 24 March 2022

Office Holder Details: *Ruth Ellen Duncan* (IP No. 9246) of Maxwell Davies Limited, Vinters Business Park, New Cut Road, Maidstone, Kent, ME14 5NZ

Further details contact: Ruth Ellen Duncan, Email: ruth@maxwelldavies.com or Tel: 01622 764612. Alternative contact: Catherine Harrison. Email: catherine@maxwelldavies.com

Ruth Ellen Duncan, Liquidator

30 March 2022

Ag OH123317 (4035533)

#### **PETITIONS TO WIND-UP**

In the High Court of Justice
No. CR-2022-000650 of 2022
In the matter of A1 SUPA SKIPS LTD

Trading As: A1 Supa Skips Ltd,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of A1 Supa Skips Ltd (03569203) of Stoneraise Quary, Great Salkeld, Penrith CA11 9NF, whose nature of business is Collection of non-hazardous waste, presented on Thursday 10 March 2022, at 15:47 by ABS RESOURCING LIMITED, of Little Garth, Tirley Lane, Tarporley CW6 0JZ claiming to be a Creditor of the Company, will be heard at the High Court of Justice, Rolls Building, Fetter Lane, London EC4A 1NL on Wednesday 27 April 2022, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 26 April 2022

The Petitioner's Solicitor is Stuart Cranston, COLTMAN WARNER CRANSTON LLP, Unit 3, Coventry Innovation Village, Cheetah Road, Coventry, West Midlands CV1 2TL,, Telephone: 02476 627280, Email: scranston@coltmanco.com (Reference number: C0186597.) (4036356)

In the High Court of Justice CR-2022-000544

In the matter of HARLEY STREET EYE CLINIC LIMITED

Trading As: Harley Street Eye Clinic Limited, and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of Harley Street Eye Clinic Limited (10854472) of 86 Harley Street, London W1G 7HP, (where service of the petition was effected), whose nature of business is Medical optical services, presented on Monday 28 February 2022, at 10:00 by AURORA LEASING LIMITED, of 1-2 Downland Close, London N20 9LZ claiming to be a Creditor of the Company, will be heard at the High Court of Justice, Rolls Building, Fetter Lane, London EC4A 1NL on Wednesday 13 April 2022, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 12 April 2022

The Petitioner's Solicitor is Karen Dobson, SHERRARDS SOLICITORS LLP, 4 Beaconsfield Road, St. Albans AL1 3RD,, Telephone: 01727 738906, Email: kd@sherrards.com (Reference number: KPD/ 1A4664-107.)

Friday 01 April 2022 (4036482)

In the High Court of Justice No. 000688 of 2022

In the matter of NBM CONTRACTING LIMITED

Trading As: NBM Contracting Limited,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of NBM Contracting Limited (10255788) of Unit 24 Hermes Road, Gilmoss Industrial Estate, Liverpool L11 0ED, whose nature of business is 41100 - Development of building projects 43210 - Electrical installation 43290 - Other construction installation 43390 - Other building completion and finishing, presented on Friday 11 March 2022, at 10:00 by REXEL UK LIMITED, of Eagle 2, Hatchford Way, Birmingham B26 3RZ claiming to be a Creditor of the Company, will be heard at the High Court of Justice, Business and Property Courts of England and Wales, Rolls Building, Fetter Lane, London EC4A 1NL on Wednesday 27 April 2022, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 26 April 2022

The Petitioner's Solicitor is Paul Bowden, WILKIN CHAPMAN LLP, Cartergate House, 26 Chantry Lane, Grimsby DN31 2LJ,, Telephone: 01472262626, Email: paul.bowden@wilkinchapman.co.uk (4035960)

In the High Court of Justice

No. CR-2022-000521 of 2022

In the matter of FRESH FOR ALL GROUP LIMITED

Trading As: Fresh For All Group Limited,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of Fresh For All Group Limited (11072403) of The Packing Shed, Cleeve Road, Evesham WR11 8JU, whose nature of business is Mixed farming, presented on Thursday 24 February 2022, at 15:40 by QUEENTANA TRADING COMPANY LIMITED, of 50 Vista Centre, Salisbury Road, Hounslow TW4 6JQ claiming to be a Creditor of the Company, will be heard at the High Court of Justice, Rolls Building, Fetter Lane, London EC4A 1NL on Wednesday 13 April 2022, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 12 April 2022

The Petitioner's Solicitor is Stuart Cranston, COLTMAN WARNER CRANSTON LLP, Unit 3, Coventry Innovation Village, Cheetah Road, Coventry, West Midlands CV1 2TL,, Telephone: 02476 627280, Email: scranston@coltmanco.com (Reference number: C0185507.) (4037354)

In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES INSOLVENCY AND COMPANIES LIST (ChD) No CR-2022-000842 of 2022

In the Matter of CIBE LIMITED

(Company Number 11178295)

and in the Matter of the INSOLVENCY ACT 1986

A petition to wind-up the above-named Company of 46 Goodhart Place London E14 8EG (Registered Office) presented on 22 March 2022 by Christopher Bates a director of the business will be heard at The Royal Courts of Justice, 7 Rolls Building, Fetter Lane, London EC4A 1NL on 17 June 2022 at 11:30am (or so soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioner's solicitors in accordance with Rule 7.14 of the Insolvency (England and Wales) Rules 2016, SI2016/1024 by 16:00 hours on 16 June 2022.

The Petitioner's details are *Christopher Bates*, 79 Cromwell Avenue Billericay, Essex CM120AG, email@christopherbates.co.uk

Dated: 23rd March 2022 (4037785)

In the High Court of Justice

Business and Property Courts in Leeds Insolvency and Companies List (ChD) No CR-2022-LDS-000095 of 2022

In the Matter of GRAHAM JONES CRANES LIMITED

(Company Number 05745437)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above named company having its Registered Office at 25 Grosvenor Road, Wrexham, LL11 1BT presented on 22 February 2022 by GRENKE LEASING LIMITED, 2 London Square, Cross Lanes, Guildford, GU1 1UN, claiming to be a creditor of the company, will be heard at Leeds Combined Court Centre, The Courthouse, 1 Oxford Row, LEEDS, LS1 3BG

on 12 April 2022, at 10:00 (or as soon thereafter as the petition can be heard)

Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of intention to do so to the petition or its solicitor in accordance with Rule 7.14 by 4 pm on 11 April 2022.

The Petitioner's Solicitor is Lester Aldridge LLP, Russell House Oxford Road Bournemouth BH8 8EX. (Ref: EFO.GR00143.693.)

30th March 2022 (4037781)

In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES INSOLVENCY AND COMPANIES LIST (ChD) No CR-2021-001264 of 2021

In the Matter of NORTHOP HALL COUNTRY HOUSE HOTEL LIMITED

(Company Number 05667278)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company (registered no 05667278) of Beck House, 77a King Street, Knutsford, England, WA16 6DX presented on 16 July 2021 by Mr. Michel Angelo Romano of 45 Carlton Avenue, Worksop, Nottinghamshire S81 7JY (the Petitioner) claiming to be a creditor of the company, will be heard at The Rolls Building, 7 Rolls Buildings, Fetter Lane, London EC4A 1NL Date: Wednesday 13 April 2022

Time: 10:30 hours (or as soon thereafter as the petition can be heard) Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16.00 hours on Tuesday 12 April 2022.

The Petitioner's Solicitor is Paul Muscutt (Partner) of Crowell & Moring, Tower 42, 25 Old Broad Street, London EC2N 1HQ. (Ref: 120288.0000002.)

Dated: 30 March 2022 (4037786)

A Petition to wind up the above-named company (registered no 05625483) of 9 Buckingham Square, Hurricane Way, Wickford, England, SS11 8YQ presented on 18 February 2022 by ORSTED SALES (UK) LIMITED of 5 Howick Place, London SW1P 1WG (the Petitioner) claiming to be a creditor of the company, will be heard at Business and Property Courts in Leeds, Leeds Combined Court Centre, The Courthouse, 1 Oxford Row, Leeds, LS1 3BG.

Date: 12 April 2022

Time: 10:00AM (or as soon thereafter as the petition can be heard) Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16.00 hours on 11 April 2022.

The Petitioner's Solicitor is Clarion Solicitors Limited, Elizabeth House, 13-19 Queen Street, Leeds, LS1 2TW. (Ref: JJLW/ALP/ 418163.6.)

Dated: 29 March 2022 (4037783)

#### In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS IN LEEDS INSOLVENCY AND COMPANIES LIST (ChD) No CR-2022-LDS-000114 of 2022

In the Matter of SUPREMEINDIANKITCHEN LTD

(Company Number 11721383)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company (registered no 11721383) of 3-7 Milsom Place, Bath, BA1 1BZ presented on 25 February 2022 by ORSTED SALES (UK) LIMITED of 5 Howick Place, London SW1P 1WG (the Petitioner) claiming to be a creditor of the company, will be heard at Business and Property Courts in Leeds, Leeds Combined Court Centre, The Courthouse, 1 Oxford Row, Leeds, LS1 3BG.

Date: 12 April 2022

Time: 10:00AM (or as soon thereafter as the petition can be heard) Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16.00 hours on 11 April 2022.

The Petitioner's Solicitor is Clarion Solicitors Limited, Elizabeth House, 13-19 Queen Street, Leeds, LS1 2TW. (Ref: JJLW/ALP/ 418163.3)

Dated: 29 March 2022 (4037780) In the High Court of Justice, Business and Property Courts of England

and Wales, Companies Court (ChD) Court Number: CR-2022-000552

In the Matter of ALLWOOD BUILDINGS LTD

(Company Number 02676458)

and In the Matter of the Insolvency Act 1986

A Petition to wind up the above named Company of Winchester House, Deane Gate Avenue, Taunton, Somerset TA1 2UH presented on 28 February 2022 by Falcon Panel Products Limited of Clock House, Station Approach, Shepperton, Middlesex, TW17 8AN claiming to be a Creditor of the Company will be heard at The Companies Court at The Rolls Building, 7 Rolls Buildings, Fetter Lane, London, EC4A 1NL on 13 April 2022 at 10.30 am (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the Hearing of the Petition (whether to support or oppose it) must give Notice of Intention to do so to the Petitioner or their Solicitor in accordance with Rule 7.14 by 16.00 hours on 12 April 2022.

The Petitioner's Solicitor is Abrahams Dresden LLP, 111 Charterhouse Street, London, EC1M 6AW, Email: ray.mansel@ad-solicitors.co.uk, Tel: 0207 251 3663, Ref: RM/RS/DB27970.

Ag OH123264 (4035510)

## In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS IN LEEDS INSOLVENCY AND COMPANIES LIST (ChD) No CR-2022-LDS-000084 of 2022

In the Matter of SURVEY CONFEDERATION LIMITED

(Company Number 10039875)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company (registered no 10039875) of Newspaper House, Churchgate, Bolton, Lancashire, BL1 1HL presented on 18 February 2022 by ORSTED SALES (UK) LIMITED of 5 Howick Place, London SW1P 1WG (the Petitioner) claiming to be a creditor of the company, will be heard at Business and Property Courts in Leeds, Leeds Combined Court Centre, The Courthouse, 1 Oxford Row, Leeds, LS1 3BG.

Date: 12 April 2022

Time: 10:00AM (or as soon thereafter as the petition can be heard) Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16.00 hours on 11 April 2022.

The Petitioner's Solicitor is Clarion Solicitors Limited, Elizabeth House, 13-19 Queen Street, Leeds, LS1 2TW. (Ref: JJLW/ALP/ 418163.12)

Dated: 29 March 2022 (4037787)

In the COUNTY COURT AT MANCHESTER No 2001 of 2022

In the Matter of BL 634 LIMITED

(Company Number 11146131)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up BL 634 LIMITED (company registered no 11146131) of Beck House, 77a King Street, Knutsford, Cheshire WA16 6DX presented on 14 March 2022 by MR SAMSON AKWAGBE (the Petitioner) of 21 Tanyard Place, Harlow CM20 2FF claiming to be a creditor of the company, will be heard at THE MANCHESTER COUNTY COURT, 1 Bridge Street West, Manchester M60 9DJ on 3 May 2022 at 10:00am (or as soon thereafter as the petition can be heard).

Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of the intention to do so to the Petitioner in accordance with rule 7.14 of The Insolvency (England and Wales) Rules 2016 by 4pm on 29 April 2022.

The Petitioner's contact details are Samson Akwagbe of 21 Tanyard CM20 07958464298, 2FF, Tel: sammy\_33uk@hotmail.com. (4037784)

In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS IN LEEDS INSOLVENCY AND COMPANIES LIST (ChD) No CR-2022-LDS-000088 of 2022

In the Matter of T L C INNS LIMITED

(Company Number 05625483)

and in the Matter of the INSOLVENCY ACT 1986

In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS IN LIVERPOOL INSOLVENCY AND COMPANIES LIST (ChD) No CR-2022-LIV-000049 of 2022

In the Matter of NAYLOR UTILITIES LIMITED

(Company Number 10814504)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company (registered no 10814504) of The Old Council Depot, Ings Road, Wombwell, Barnsley, S73 0BP presented on 18 FEBRUARY 2022 by UTILITIES DESIGN & PLANNING LIMITED Of Astor Road, Salford, Manchester, M50 1BB (the Petitioner) claiming to be a creditor of the company, will be heard in person at Business & Property Court at Liverpool, 35 Vernon Street, Liverpool

Date: 12 April 2022

Time: 10:30 hours (or as soon thereafter as the petition can be heard) Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16.00 hours on Monday 11 April 2022

The Petitioner's Solicitor is CKE Law, 53 King Street, Manchester, M2

4LQ. Tel: 0161 503 1717 - email: aci@ckelaw.com

Dated: 31 March 2022 (4037788)

In the High Court of Justice (Chancery Division)

Companies Court No 0482 of 2022

In the Matter of PARAGON PS UK LIMITED

(Company Number 11652918)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named Company, Registration Number 11652918, of ,OFFICE 6, UNIT 2 SYCAMORE TRADING ESTATE, SQUIRES GATE LANE, BLACKPOOL, ENGLAND, FY4 3RL, presented on 21 February 2022 by the COMMISSIONERS FOR HM REVENUE AND CUSTOMS, of 14 Westfield Avenue, Stratford, London, E20 1HZ,, claiming to be Creditors of the Company, will be heard at the High Court, Royal Courts of Justice, 7 Rolls Building, Fetter Lane, London, EC4A 1NL on 13 April 2022 at 1030 hours (or as soon thereafter as the Petition can be heard).

Any persons intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 1600 hours on 12 April 2022 .

The Petitioners' Solicitor is the Solicitor to, HM Revenue and Customs, Solicitor's Office & Legal Services, 14 Westfield Avenue, Stratford, London, E20 1HZ, telephone 03000 598642 . (Ref SLR2045980/U.)

1 April 2022 (4034699)

In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS IN LEEDS INSOLVENCY AND COMPANIES LIST (ChD) No CR-2022-LDS-000116 of 2022

In the Matter of **SAMBUCA (TYNE TUNNEL) RISTORANTE LIMITED** (Company Number 08791631)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company (registered no 08791631) of Northumberland Dock Road, Wallsend, NE28 6ST presented on 25 February 2022 by ORSTED SALES (UK) LIMITED of 5 Howick Place, London SW1P 1WG (the Petitioner) claiming to be a creditor of the company, will be heard at Business and Property Courts in Leeds, Leeds Combined Court Centre, The Courthouse, 1 Oxford Row, Leeds, LS1 3BG.

Date: 12 April 2022

Time: 10:00AM (or as soon thereafter as the petition can be heard) Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16.00 hours on 11 April 2022.

The Petitioner's Solicitor is Clarion Solicitors Limited, Elizabeth House, 13-19 Queen Street, Leeds, LS1 2TW. (Ref: JJLW/ALP/418163.26.)

Dated: 29 March 2022 (4037779)

In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES INSOLVENCY AND COMPANIES LIST (ChD) No CR-2022-000592 of 2022

In the Matter of **VAT BRIDGE LIMITED** 

(Company Number 10573364)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company (CRN: 10573364) (hereinafter the Company) of 20a High Street, Ruislip, HA4 7AN, presented on 3 March 2022 by TOKENHOUSE VB LIMITED – In Administration (CRN: 09711892) of 110 Cannon Street, London, EC4N 6EU, acting by Paul Allen and Glyn Mummery of FRP ADVISORY TRADING LIMITED as Joint Administrators (the Petitioner) claiming to be a creditor of the Company, will be heard at the Business and Property Courts of England and Wales, The Rolls Building, 7 Rolls Buildings, Fetter Lane, London, EC4A 1NL on:

Date: Monday 23 May 2022

Time: 11:30 hours (or as soon thereafter as the petition can be heard)

Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16:00 hours on Friday 20 May 2022.

The Petitioner's Solicitor is Walker Morris LLP, 33 Wellington Street,

Leeds, LS1 4DL. (Ref: ALA/RRD/FRP03724.8.)

Dated: 28 March 2022 (4037782)

## WINDING-UP ORDERS

#### **ALLIANCE BRICKWORK LIMITED**

(Company Number 10282783)

Registered office: Suite 73, 89 Commercial Road, BOURNEMOUTH,

BH2 5RR

In the Birmingham District Registry

No 000058 of 2022

Date of Filing Petition: 9 February 2022 Date of Winding-up Order: 28 March 2022

K Jackson 1st Floor, Spring Place, 105 Commercial Road,

Southampton, SO15 1EG, telephone: 03030031735

Capacity of office holder(s): Liquidator

28 March 2022 (4034034)

#### DOWMAN IMPORTS LIMITED

(Company Number 02237477)

Registered office: 39 Harlestones Road, Cottenham, CAMBRIDGE,

CB24 8TR

In the County Court at Peterborough

No 45 of 2021

Date of Filing Petition: 10 December 2021 Date of Winding-up Order: 10 February 2022

M Commins Eastbrook, Shaftesbury Road, Cambridge, CB2 8DR,

telephone: 03030031741

Capacity of office holder(s): Liquidator

29 March 2022 (4034030)

## **GRIMSBY CONSTRUCTION LIMITED**

(Company Number 12634803)

Registered office: Cleethorpes Business Centre, 145 Grimsby Road,

CLEETHORPES, DN35 7DG In the High Court Of Justice

No 000315 of 2022

Date of Filing Petition: 9 February 2022 Date of Winding-up Order: 23 March 2022

L Burgess 3rd Floor, 1 City Walk, Leeds, LS11 9DA, telephone: 0300

678 0016

Capacity of office holder(s): Liquidator

23 March 2022 (4034039)

## Members' voluntary liquidation

## APPOINTMENT OF LIQUIDATORS

Name of Company: STEEDMAN CVC LTD

Company Number: 07800123

Registered office: 9 Hollands Field, Broadbridge Heath, Horsham,

West Sussex, RH12 3HQ

Principal trading address: 9 Hollands Field, Broadbridge Heath,

Horsham, West Sussex, RH12 3HQ

Nature of Business: Management consultancy activities other than

financial management

Type of Liquidation: Members' Voluntary Liquidation

Liquidator's name and address: *Michael James Gregson* of TC Bulley Davey Limited, 4 Cyrus Way, Cygnet Park, Hampton, Peterborough, PET 8HD

Office Holder Number: 9339. Date of Appointment: 22 March 2022 By whom Appointed: Members

Alternative contact: Denise Gordon, denisegordon@tc-group.com - Telephone: 01733 569494 (4037806)

Name of Company: ANJTECH CONSULTANCY LTD

Company Number: 09431942

Nature of Business: Information technology consultancy

Registered office: c/o Neum Insolvency, Suite 9, Amba House, 15

College Road, Harrow, Middlesex, HA1 1BA

Type of Liquidation: Members Date of Appointment: 28 March 2022

Umang Patel (IP No. 18790) of Neum Insolvency, Suite 9, Amba

House, 15 College Road, Harrow, Middlesex, HA1 1BA

By whom Appointed: Members

Ag OH123358 (4035496)

Name of Company: ANNIEM MEDIA LIMITED

Company Number: 10324538

Nature of Business: Information technology consultancy activities Registered office: C/O Antony Batty & Co, Thames Valley, Innovation

Centre, 99 Park Drive, Milton Park, Oxfordshire, OX14 4RY

Type of Liquidation: Members Date of Appointment: 29 March 2022

Lawrence King (IP No. 10452) and Matthew Waghorn (IP No. 9432) both of Antony Batty & Co, Thames Valley, Innovation Centre, 99 Park

Drive, Milton Park, Oxford, OX14 4RY By whom Appointed: Members

Ag OH123378 (4035493)

Company Number: 08736879

Name of Company: AQUEL CONSULTING LIMITED

Nature of Business: Engineering Consultancy

Registered office: Stamford House, Northenden Road, Sale, Cheshire M33 2DH Previous registered office Bennett Brooks & Co Limited

Suite 345, 50 Eastcastle Street, Fitzrovia W1W 8EA Type of Liquidation: Members Voluntary Liquidation

Joint Liquidator: Steve Kenny (IP number 24030) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33

Joint Liquidator: Richard Cole (IP number 2 6070) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH.

Date of Appointment: 30 March 2022 By whom Appointed: The Company

For further details contact Adam Horan on 0161 637 8100 or at Adam@KBL-Advisory.com (4037849)

Name of Company: ASPINALL IT CONSULTING LIMITED

Company Number: 09347908

Nature of Business: Other information technology service activities Registered office: Crown Chambers, Princes Street, Harrogate, HG1 1NJ to be changed to FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18

Type of Liquidation: Members Date of Appointment: 23 March 2022

David Antony Willis (IP No. 9180) and Martyn James Pullin (IP No. 15530) both of FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18 3TX

By whom Appointed: Members

(4035385) Ag OH123351

Name of Company: AUTUMNTIGER LIMITED

Company Number: 04105391

Nature of Business: Non-trading company Name of Company: EF PARTICIPANTS LIMITED

Company Number: 03533476

Nature of Business: Non-trading company

Registered office: (Both) Ernst & Young LLP, 1 More London Place,

London, SE1 2AF

Type of Liquidation: (Both) Members Date of Appointment: 21 March 2022

Matthew Coomber (IP No. 24430) and Samantha Jane Keen (IP No. 9250) both of Ernst & Young, 1 More London Place, London, SE1 2AF

By whom Appointed: Members

Ag OH123287 (4035366) Company Number: 07504925 Name of Company: BHP IT LIMITED

Previous Name of Company: Wakeco (427) Limited

Nature of Business: Non-trading company Registered office: Unit 8 President Buildings, Savile Street East,

Sheffield, S4 7UQ

Principal trading address: Unit 8 President Buildings, Savile Street

East, Sheffield, S4 7UQ

Type of Liquidation: Members Voluntary Liquidation

Joint Liquidator: Claire Elizabeth Dowson (IP number 19272) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter

Row, Sheffield, S1 3FZ.

Joint Liquidator: Gareth David Rusling (IP number 9481) of Beabies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ.

Date of Appointment: 28 March 2022

By whom Appointed: Members For further details contact Gemma Anderson on 0114 275 5033 or at

sheffield.north@begbies-traynor.com (4037843)

Company Number: 09767352

Name of Company: BRILLIANT BASICS HOLDINGS LIMITED

Nature of Business: Information technology consultancy activities Registered office: The Bower 10/11th Floor, 207-211 Old Street,

London, EC1V 9NR

Type of Liquidation: Members Voluntary Liquidation

Liquidator: Ryan Davies (IP number 15250) of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London EC1M

Date of Appointment: 31 March 2022

By whom Appointed: The Members of the Company

For further details contact Tara Wessely on 0207 566 4020 or at (4037835)twessely@mks.co.uk

Company Number: 08011232

Name of Company: BRILLIANT BASICS LTD

Nature of Business: Other information technology service activities Registered office: The Bower 10th/11th Floor, 207-211 Old Street,

London EC1V 9NR

Type of Liquidation: Members Voluntary Liquidation

Liquidator: Ryan Davies (IP number 15250) of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London, EC1M 7AD.

Date of Appointment: 31 March 2022

By whom Appointed: The Sole Member of the Company

For further details contact Tara Wessely on 0207 566 4020 or at (4036328) twessely@mks.co.uk

Name of Company: C L J CONSULTANCY LIMITED

Company Number: 11883647

Nature of Business: Others - Not Reported

Registered office: 8th Floor, One Temple Row, Birmingham, B2 5LG

Type of Liquidation: Members Date of Appointment: 24 March 2022

Craig Povey (IP No. 9665) and Charles Turner (IP No. 9195) both of Begbies Traynor (Central) LLP, 8th Floor, One Temple Row,

Birmingham, B2 5LG By whom Appointed: Members

Ag OH123323

(4035350)

Name of Company: C4 PROJEX LIMITED

Company Number: 11314436

Nature of Business: Management consultancy

Registered office: Little Bourne, 2 Slate Way, Pitton, Salisbury, SP5

Type of Liquidation: Members Date of Appointment: 16 March 2022

Lauren Rachel Cullen (IP No. 18050) of Cullen & Co UK Limited, 197

Kingston Road, Epsom, Surrey, KT19 0AB

By whom Appointed: Members

Ag OH123251 (4035375) Name of Company: CALCON RAIL LTD

Company Number: 06904156

Registered office: 117 North Road, Crawley, West Sussex, RH10 1SQ Principal trading address: 117 North Road, Crawley, West Sussex,

**RH10 1SQ** 

Nature of Business: Other Business Support Service Activities N.E.C

Type of Liquidation: Member's Voluntary Date of Appointment: 21 March 2022

Joint Liquidator's Name and Address: Richard Keley (IP No. 18072) of BM Advisory, Arundel House, 1 Amberley Court, Whitworth Road, Crawley, RH11 7XL. Telephone: 01293 410333 .

Joint Liquidator's Name and Address: Andrew Pear (IP No. 9016) of BM Advisory, Arundel House, 1 Amberley Court, Whitworth Road, Crawley, RH11 7XL. Telephone: 01293 410333.

For further information contact Sue Markham at the offices of BM Advisory on 01293 452844, or Sue.Markham@bm-advisory.com.

By whom Appointed: The Members And Creditors

(4036372) 29 March 2022

Name of Company: CORFU CONSULTANCY LIMITED

Company Number: 08390402

Nature of Business: Management consultancy activities other than

financial management

Registered office: c/o KRE Corporate Recovery Limited, Unit 8 The

Aquarium, 1-7 King Street, Reading, Berks RG1 2AN

Type of Liquidation: Members Date of Appointment: 29 March 2022

Rob Keyes (IP No. 008841) and Paul Ellison (IP No. 7254) both of KRE Corporate Recovery Limited, Unit 8, The Aquarium, 1-7 King Street,

Reading, RG1 2AN

By whom Appointed: Members

Ag OH123365 (4035487)

Name of Company: COTTONBURY PLANNING LIMITED

Company Number: 11016106

Registered office: Menta Business Centre, 5 Eastern Way, Bury St

Edmunds IP32 7AB

Principal trading address: Field House, Bells Road, Belchamp Walter,

Sudbury, Suffolk CO10 7A

Nature of Business: Planning consultancy

Type of Liquidation: Members

Liquidators name, address, tel number and email address: Stephen Rout of Menta Business Centre, Bury St Edmunds, Suffolk IP32 7AB

Tel 01223 329392-email smrout@aol.com

Office Holder Number: 6062. Date of Appointment: 24 March 2022

(4037816) By whom Appointed: Members

Company Number: 03941275

Name of Company: CRAY U.K. LIMITED

Previous Name of Company: HACKREMCO (NO.1622) LIMITED

Nature of Business: 95110 - Repair of computers and peripheral

Registered office: Ground Floor, 210 Wharfedale Road, Winnersh

Triangle, Berkshire, United Kingdom, RG41 5TP Type of Liquidation: Members Voluntary Liquidation

Joint Liquidator: Laura Waters (IP number 9477) of PricewaterhouseCoopers LLP, 7 More London Riverside, London,

SE1 2RT.

Joint Liquidator: Steven Sherry (IP number 19752) PricewaterhouseCoopers LLP, 7 More London Riverside, London,

Date of Appointment: 31 March 2022

By whom Appointed: The Members of the Company

For further details contact Anne Louise Mccartney on 07802 659389

or at anne.louise.mccartney@pwc.com

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators' appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators.

(4037950)

Name of Company: CREATED EDUCATION LIMITED

Company Number: 10889843

Registered office: 8 C/O Hansel Henson Ltd, 3rd Floor, 8 Bloomsbury

Street, London, WC1B 3SR

Principal trading address: 8 C/O Hansel Henson Ltd, 3rd Floor, 8

Bloomsbury Street, London, WC1B 3SR

Nature of Business: Other business support service activities not

elsewhere classified

Type of Liquidation: Members' Voluntary Date of Appointment: 23 March 2022

Joint Liquidator's Name and Address: Vincent John Green (IP No. 009416) of Crowe U.K. LLP, 4 Mount Ephraim Road, Tunbridge Wells,

Kent, TN1 1EE. Telephone: 01892 700200.

Joint Liquidator's Name and Address: Steven Edwards (IP No. 26090) of Crowe U.K. LLP, 4 Mount Ephraim Road, Tunbridge Wells, Kent,

TN1 1EE, Telephone: 01892 700200.

For further information contact George Budd at the offices of Crowe U.K. LLP on 01892 700200, or recoverysolutions@crowe.co.uk.

By whom Appointed: Members

24 March 2022 (4036379)

Name of Company: DALMAN (IMPORT & EXPORT) LIMITED

Company Number: 01169794

Nature of Business: Wholesale of electronic and telecommunications

equipment and parts

Registered office: 47 Hill Farm Road, Marlow, Buckinghamshire, SL7 3LX and it is in the process of being changed to The Old Town Hall,

71 Christchurch Road, Ringwood, BH24 1DH

Type of Liquidation: Members Date of Appointment: 30 March 2022

Chris Newell (IP No. 13690) and David Meany (IP No. 9453) both of Quantuma Advisory Limited, The Old Town Hall, 71 Christchurch

Road, Ringwood, BH24 1DH By whom Appointed: The Company

Ag OH123265 (4035437)

Name of Company: **DEWDROP CONSULTING LTD** 

Company Number: 08893944

Nature of Business: Information Technology Consultancy

Registered office: 12 High Street, Stonehouse, Gloucestershlre, GL10

2NA

Type of Liquidation: Members Date of Appointment: 18 March 2022

Lisa Alford (IP No. 9723) and Chris Parkman (IP No. 9588) both of Purnells, 5a Kernick Industrial Estate, Penryn, Cornwall, TR10 9EP

By whom Appointed: Shareholders

Ag OH123308 (4035492)

Name of Company: EPIRIS CAPITAL PARTNERS LLP

Company Number: OC412420 Nature of Business: Dormant company

Registered office: 1 More London Place, London SE1 2AF

Type of Liquidation: Members Date of Appointment: 21 March 2022

Matthew Coomber (IP No. 24430) and Samantha Jane Keen (IP No. 9250) both of Ernst & Young, 1 More London Place, London, SE1 2AF

By whom Appointed: Members

Ag OH123288 (4035513)

Name of Company: ESTELIA LIMITED

Company Number: 09688132

Nature of Business: Information Technology Consultants Registered office: Lismore, The Purlieu, Malvern, WR14 4DJ

Type of Liquidation: Members Date of Appointment: 17 March 2022

Lisa Alford (IP No. 9723) and Chris Parkman (IP No. 9588) both of Purnells, 5a Kernick Industrial Estate, Penryn, Cornwall, TR10 9EP

By whom Appointed: Shareholders

Ag OH123278 (4035428) Name of Company: FIELDER GAMMON INVESTMENT COMPANY

LIMITED

Company Number: 00729537

Nature of Business: Property Investment Company

Registered office: WSM MBI Coakley LLP, 2nd Floor, Shaw House, 3

Tunsgate, Guildford, Surrey, GU1 3QT Type of Liquidation: Members Date of Appointment: 21 March 2022

Dermot Coakley (IP No. 6824) of WSM MBI Coakley LLP, 2nd Floor,

Shaw House, 3 Tunsgate, Guildford, Surrey, GU1 3QT

By whom Appointed: Members

Ag OH123339 (4035399)

Name of Company: FOX AND HOUNDS HOTEL LIMITED

Company Number: 06296746

Nature of Business: Hotels and similar accommodation Registered office: 138 High Street, Crediton, Devon, EX17 3DX

Type of Liquidation: Members
Date of Appointment: 23 March 2022

Steve Henson (IP No. 10652) and Adam Buck (IP No. 23070) both of Castle Hill Insolvency Limited, 1 Battle Road, Heathfield, Newton

Abbot, TQ12 6RY

By whom Appointed: Members

Ag OH123304 (4035458)

Name of Company: FRESH FRUIT DISTRIBUTION LIMITED

Company Number: 07209941

Nature of Business: Distribution of food

Registered office: 29 High Street, Blue Town, Sheerness, ME12 1RN

Type of Liquidation: Members Date of Appointment: 23 March 2022

Laurence Pagden (IP No. 9055) and Jonathan David Bass (IP No. 11790) both of Menzies LLP, Lynton House, 7-12 Tavistock Square,

London, WC1H 9LT

By whom Appointed: Members

Ag OH123364 (4035439)

Name of Company: G.S.N PROPERTY DEVELOPMENT LIMITED

Company Number: 12073368

Nature of Business: Others - Not reported

Registered office: Begbies Traynor, Suite WG3, The Officers' Mess

Business Centre, Royston Road, Duxford, Cambridge, CB22 4QH

Type of Liquidation: Members Date of Appointment: 21 March 2022

Louise Donna Baxter (IP No. 009123) of Begbies Traynor (Central) LLP, Suite WG3, The Officers' Mess Business Centre, Royston Road, Duxford, Cambridge, CB22 4QH and Lloyd Biscoe (IP No. 9141) of Begbies Traynor (Central) LLP, The Old Exchange, 234 Southchurch

Road, Southend-on-Sea, Essex, SS1 2EG

By whom Appointed: Members

Ag OH123348 (4035529)

PURSUANT TO SECTION 109 OF THE INSOLVENCY ACT 1986

Name of Company: GATEWAY ACCOUNTING LIMITED

Company Number: 11328283

Registered office: 2 Chestnut Close, Orpington, BR6 6LP Principal trading address: 2 Chestnut Close, Orpington, BR6 6LP

Nature of Business: Accounting Services

Type of Liquidation: Members' Voluntary Liquidation

Liquidator's name and address: Richard Frank Simms, Liquidator of F A Simms & Partners Limited, Alma Park, Woodway Lane, Claybrooke

Parva, Lutterworth, Leicestershire, LE17 5FB

Office Holder Number: 9252.

Date of Appointment: 25 March 2022

By whom Appointed: The Members

(4037811)

Name of Company: GEOFFREY BLUMENTHAL LIMITED

Company Number: 04233950 Nature of Business: Architects

Registered office: The Old Rectory, Main Street, Glenfield, Leicester,

LE3 8DG

Type of Liquidation: Members
Date of Appointment: 24 March 2022

Gavin Bates (IP No. 8983) of Smart Business Recovery Limited, The

Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG

By whom Appointed: Members

Ag OH123338 (4035463)

Name of Company: GOLD GROUP PROPERTIES LIMITED

Company Number: 06396536

Nature of Business: Activities of head offices

Previous Name of Company: Pimco 2705 Limited (From 11 October

2007 to 20 November 2007)

Registered office: Gold Group House, Godstone Road, Whyteleafe, CR3 0GG which is in the process of being changed to c/o Quantuma Advisory Limited, High Holborn House, 52-54 High Holborn, London, WC1V 6RL

Type of Liquidation: Members Date of Appointment: 22 March 2022

Simon Bonney (IP No. 9379) and Michael Kiely (IP No. 9617) both of Quantuma Advisory Limited, High Holborn House, 52-54 High

Holborn, London, WC1V 6RL By whom Appointed: Members

Ag OH123369 (4035398)

Company Number: 07838188

Name of Company: HAWORTH GROUP HOLDINGS (UK) LIMITED

Nature of Business: Provision of data storage facilities and computer

disaster recovery services.

Registered office: 4 Ground Floor, More London Riverside, London,

England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside,

London, England SE1 2AU Company Number: 02262052

Name of Company: HAWORTH GROUP LIMITED

Nature of Business: Holding company

Registered office: 4 Ground Floor, More London Riverside, London,

England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside,

London, England SE1 2AU Company Number: 02063535

Name of Company: **DISASTER RECOVERY SERVICES LIMITED** 

Previous Name of Company: Flightace Computing Limited (13/01/1987)

Nature of Business: Provision of secure data storage and

management services

Registered office: 4 Ground Floor, More London Riverside, London,

England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside,

London, England SE1 2AU Company Number: 03599730

Name of Company: GLOBAL LOGISTICS WORLDWIDE LIMITED

Previous Name of Company: Monocrest Limited (27/07/1998)

Nature of Business: Holding company

Registered office: 88 Crawford Street, London W1H 2EJ Principal trading address: 88 Crawford Street, London W1H 2EJ

Company Number: 02652784

Name of Company: SARACEN DATASTORE LIMITED

Nature of Business: Provision of archival document, data and media

storage and related records management services

Registered office: 4 Ground Floor, More London Riverside, London,

England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside,

London, England SE1 2AU

Company Number: 01331798 Name of Company: **RECALL LIMITED** 

Previous Name of Company: Security Archives Limited (Until 28/04/2000); Security Archives (Holdings) Plc (Until 01/08/1994);

Seacave Limited (Until 09/11/1984)

Nature of Business: Investment holding company

Registered office: 4 Ground Floor, More London Riverside, London,

England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside,

London, England SE1 2AU

Type of Liquidation: Members Voluntary Liquidation

Company Number: 01136000

Name of Company: RECALL EUROPE LIMITED

Previous Name of Company: Recall Total Information Management Limited (Until 23/07/2013); Brambles Records Management Limited (Until 09/04/1999); Security Archives Limited (Until 26/07/1994); Central Civil and Commercial Investments Limited (Until 31/12/1976)

Nature of Business: Investment holding company

Registered office: 4 Ground Floor, More London Riverside, London,

England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside,

London, England SE1 2AU

Type of Liquidation: All in Members Voluntary Liquidation

Joint Liquidator: Nicholas James Timpson (IP number 20610) of

Interpath Ltd, 10 Fleet Place, London EC4M 7RB.

Joint Liquidator: Blair Carnegie Nimmo (IP number 8208) of Interpath

Ltd, 10 Fleet Place, London EC4M 7RB. Date of Appointment: 23 March 2022 By whom Appointed: Members

For further details contact Precious Odiase on +44 (0) 203 989 2848 or at precious.odiase@interpathadvisory.com (4037912)

Name of Company: HYBRID GLOBAL LTD

Company Number: 11148283

Nature of Business: Information technology consultancy activities Registered office: Telecom House, 125-135 Preston Road, Brighton,

BN1 6AF

Type of Liquidation: Members Date of Appointment: 28 March 2022

Donald McNaught (IP No. 9359) of Johnston Carmichael LLP, 227

West George Street, Glasgow, G2 2ND By whom Appointed: The Company

Ag OH123393 (4035392)

Name of Company: INGENIUM CS LTD

Company Number: 08573721

Nature of Business: Other profesional, scientific and technical

activities not elsewhere classified

Registered office: First Floor, Telecom House, 125-135 Preston Road,

Brighton, BN1 6AF Type of Liquidation: Members Date of Appointment: 28 March 2022

Donald McNaught (IP No. 9359) of Johnston Carmichael LLP, 227

West George Street, Glasgow, G2 2ND By whom Appointed: The Company

Ag OH123389 (4035547)

Name of Company: J READHEAD SOLUTIONS LTD

Company Number: 11079915

Nature of Business: Information technology consultancy activities Registered office: Flat 2 Curzon House, 3 St. Peters Street, Ipswich,

Suffolk, IP1 1XF Type of Liquidation: Members Date of Appointment: 28 March 2022

Andrew Anderson Kelsall (IP No. 009555) and Lee Anthony Green (IP No. 015610) both of Larking Gowen LLP, King Street House, 15

Upper King Street, Norwich, Norfolk, NR3 1RB

By whom Appointed: Members

Ag OH123302 (4035561)

Name of Company: JES HOTELS LTD

Company Number: 12138027

Nature of Business: Hotels and similar accommodation Registered office: 17 Moor Park Avenue, Preston, PR1 6AS

Type of Liquidation: Members
Date of Appointment: 24 March 2022

John Paul Bell (IP No. 8608) and Toyah Marie Poole (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street,

Manchester, M2 4NG By whom Appointed: Members

Ag OH123391 (4035351)

Name of Company: KMJ ASSOCIATES LTD

Company Number: 07409030

Nature of Business: Financial Management

Registered office: First Floor, Telecom House, 125-135 Preston Road,

Brighton BN1 6AF

Type of Liquidation: Members

Date of Appointment: 21 March 2022

Matthew Fox (IP No. 9325) of Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP

By whom Appointed: Members

Ag OH123307 (4035605)

Name of Company: LA BAITA LIMITED

Company Number: 04595720 Nature of Business: Licensed cafe

Registered office: 2 Clarendon Road, Ashford, TW15 2QE

Type of Liquidation: Members Date of Appointment: 18 March 2022

John Paul Bell (IP No. 8608) and Toyah Marie Poole (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street,

Manchester, M2 4NG By whom Appointed: Members

Ag OH123375 (4035511)

Name of Company: LANCASHIRE TEXTILE MANUFACTURERS

**ASSOCIATION** 

Nature of Business: Trade Association

Registered office: N/A
Type of Liquidation: Members
Date of Appointment: 30 March 2022

Charles MacMillan (IP No. 6000) of Beever and Struthers, St George's

House, 215-219 Chester Road, Manchester, M15 4JE

By whom Appointed: Members

Ag OH123266 (4035550)

Name of Company: LANGLAND INVESTMENTS LIMITED

Company Number: 00565649

Registered office: 2-8 Parkstone Road Towngate House, Poole, BH15

2PW

Principal trading address: c/o Ellis Jones Solicitors, 14a Haven Road,

Poole, Dorset, BH13 7LP

Nature of Business: Property management Type of Liquidation: Member's Voluntary Date of Appointment: 16 March 2022

Joint Liquidator's Name and Address: Lucinda Clare Coleman (IP No. 10710) of Francis Clark LLP, Centenary House, Peninsula Park, Rydon

Lane, Exeter, EX2 7XE. Telephone: 01392 667000.

Joint Liquidator's Name and Address: Stephen James Hobson (IP No. 006473) of Francis Clark LLP, Centenary House, Peninsula Park,

Rydon Lane, Exeter, EX2 7XE. Telephone: 01392 667000.

For further information contact George Bird at the offices of Francis Clark LLP on 01392 667000, or George.Bird@pkf-francisclark.co.uk.

By whom Appointed: Members

28 March 2022 (4036365)

Name of Company: LONDON VET IMAGING LTD

Company Number: 10164271 Nature of Business: Veterinary activities

Registered office: 9 Ensign House, Admirals Way, Marsh Wall, London E14 9XQ (Formerly) Plaza 9 KD Tower, Cotterells, Hemel Hempstead,

Hertfordshire HP1 1FW Type of Liquidation: Members Date of Appointment: 25 March 2022

Richard Hunt (IP No. 21772) of SFP Restructuring Limited, 9 Ensign

House, Admirals Way, Marsh Wall, London E14 9XQ

By whom Appointed: Members

(4035353) Ag OH123240

Name of Company: LOST LAMBS PRODUCTIONS UK LIMITED

Company Number: 03667013

Registered office: 12th Floor, Brunel Building, 2 Canalside Walk,

London, W2 1DG

Principal trading address: 12th Floor, Brunel Building, 2 Canalside

Walk, London, W2 1DG

Nature of Business: Motion picture production activities

Type of Liquidation: Member's Voluntary Date of Appointment: 24 March 2022

Joint Liquidator's Name and Address: Thomas Edward Guthrie (IP No. 15012) of BRI Business Recovery and Insolvency, 2nd Floor, Elm House, Woodlands Business Park, Linford Wood West, Milton Keynes, MK14 6FG. Telephone: 01908 317387.

Joint Liquidator's Name and Address: Sukhvinder Kaur Bains (IP No. 9990) of BRI Business Recovery and Insolvency, 2nd Floor, Elm House, Woodlands Business Park, Linford Wood West, Milton Keynes, MK14 6FG. Telephone: 01908 317387.

For further information contact Kathryn Harper at the offices of BRI Business Recovery and Insolvency on 01908 771353, or

kharper@briuk.co.uk.

By whom Appointed: Sole Member

29 March 2022 (4036393)

Name of Company: LTR RECOVERY SERVICES LIMITED

Company Number: 07448855

Nature of Business: Maintenance and repair of motor vehicles Registered office: Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF

Type of Liquidation: Members Date of Appointment: 23 March 2022

Michaela Daly (IP No. 17570) of Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF and Louise Longley (IP No. 23874) of Begbies Traynor (Central) LLP, Fourth Floor, Toronto Square,

Toronto Street, Leeds, LS1 2HJ By whom Appointed: Members

Ag OH123267 (4035426)

Name of Company: MOCKINGBIRD INDUSTRIES LTD

Company Number: 07011367

Registered office: 2 Gallery Court, 1-7 Pilgrimage Street, London, SE1

Principal trading address: 2 Gallery Court, 1-7 Pilgrimage Street,

London, SE1 4LL

Type of Liquidation: Member's Voluntary Date of Appointment: 28 March 2022

Joint Liquidator's Name and Address: Michael Solomons (IP No. 9043) of Moorfields Advisory Limited, 82 St John Street, London, EC1M 4JN. Telephone: 020 7549 8050.

Joint Liquidator's Name and Address: Andrew Pear (IP No. 9016) of Moorfields Advisory Limited, 82 St John Street, London, EC1M 4JN.

Telephone: 020 7549 8050.

For further information contact Holly Thomas at the offices of Moorfields Advisory Limited on 020 7549 8042, or holly.thomas@bm-

By whom Appointed: The Members

28 March 2022 (4036387) Company Number: 06515716

Name of Company: MY PROPERTY PARTNERS LIMITED

Nature of Business: Consulting Business

Registered office: No. 1 The Octagon County Hall East, Forum

Magnum Square, London SE1 7GN

Principal trading address: No. 1 The Octagon County Hall East, Forum

Magnum Square, London SE1 7GN

Type of Liquidation: Members Voluntary Liquidation

Liquidator: Jeremy Charles Frost (IP number 9091) of Frost Group

Limited, One Elmfield Park, Bromley BR1 1LU. Date of Appointment: 22 March 2022

By whom Appointed: Members For further details contact Kelly Walford on 0845 260 0101 or at kellyw@frostbr.co.uk (4036469)

Name of Company: OH2M LIMITED

Company Number: 09810431

Nature of Business: Financial management

Registered office: Apartment 278 37 Millharbour, London, E14 9DL

Type of Liquidation: Members Date of Appointment: 30 March 2022

Richard Gardiner (IP No. 9488) of Thomson Cooper, 3 Castle Court,

Carnegie Campus, Dunfermline, Fife, KY11 8PB

By whom Appointed: Members

Aa OH123357 (4035499)

Name of Company: PROJECT PLANNING AND FORENSIC

**CONSULTING LTD** Company Number: 11242909

Nature of Business: Other engineering activities

Registered office: Collingham House, 10-12 Gladstone Road, Wimbledon, London SW19 1QT

Type of Liquidation: Members Date of Appointment: 18 March 2022

Matthew Fox (IP No. 9325) of Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP

By whom Appointed: Members

Ag OH123280 (4035362)

Name of Company: PSK CREDIT CONSULTANCY LTD

Company Number: 11962959

Registered office: Drake House, Gadbrook Park, Northwich, CW9 7RA Principal trading address: 37 Linnet Close, Winsford, Cheshire, CW7

3FA

Nature of Business: Management consultancy activities other than

financial management

Type of Liquidation: Members' Voluntary Date of Appointment: 18 March 2022

Liquidator's Name and Address: Clive Morris (IP No. 8820) of Marshall Peters, Heskin Hall Farm, Wood Lane, Heskin, Preston, PR7 5PA.

Telephone: 01257 452021.

For further information contact Daisy Charnley at the offices of Marshall Peters 01257 452021. on

daisycharnley@marshallpeters.co.uk. By whom Appointed: Members

29 March 2022 (4036383)

Name of Company: REALMCROFT LIMITED

Company Number: 03309576

Nature of Business: Engineering and manufacturing

Registered office: 49 Mead Road, Bury St Edmunds, Suffolk, IP32

Type of Liquidation: Members Date of Appointment: 30 March 2022

Gary Rupping (IP No. 16012) and Hayley Watson (IP No. 23450) both of McTear Williams & Wood Limited, Prospect House, Rouen Road,

Norwich NR1 1RE

By whom Appointed: Members

Ag OH123297 (4035370) Name of Company: RED SKY BUSINESS CONSULTING LIMITED

Company Number: 10081702 Nature of Business: Consultancy

Registered office: Rosewood House, Quarry Park Road, Pedmore,

Stourbridge, DY8 2RE Type of Liquidation: Members Date of Appointment: 30 March 2022

Margaret Carter (IP No. 020730) and James Martin (IP No. 008316) both of Azets, 6th Floor, Bank House, Cherry Street, Birmingham, B2

5AL

By whom Appointed: Members

Ag OH123262 (4035447)

Name of Company: SG & J CONSULTING LIMITED

Company Number: 06338207

Nature of Business: Other business support service activities not

elsewhere classified

Registered office: 40 Grange Lane, Wickham, Tyne & Wear, NE16 5AH

Type of Liquidation: Members Date of Appointment: 24 March 2022

Darren Brookes (IP No. 9297) of Milner Boardman & Partners, 187a

Ashley Road, Hale, Cheshire, WA15 9SQ By whom Appointed: Members

Ag OH123347 (4035502)

Name of Company: SHELHAR PROPERTIES LIMITED

Company Number: 04906816

Nature of Business: Other letting and operating of own or leased real

estate

Registered office: 1 Goodwood Court, Lancaster, LA1 4LY

Type of Liquidation: Members Date of Appointment: 29 March 2022

Richard Ian Williamson (IP No. 8013) and Christopher Brindle (IP No. 22092) both of Campbell, Crossley & Davis, Ground Floor, Seneca House, Links Point, Amy Johnson Way, Blackpool, Lancashire, FY4

2FF

By whom Appointed: The Member

Ag OH123353 (4035388)

Name of Company: SOLARA VENTURES LIMITED

Company Number: 06116800

Nature of Business: Information technology consultancy activities Registered office: Currently 1 Marybrook Street, Berkeley, GL13 9AA but shortly changing to c/o Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF Type of Liquidation: Members

Date of Appointment: 29 March 2022

Julie Anne Palmer (IP No. 008835) and Andrew Hook (IP No. 26150) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF

By whom Appointed: Members

Ag OH123373 (4035456)

Name of Company: **VALUAD LTD** Company Number: 06225633 Nature of Business: Business Support

Registered office: The Innovation Centre, Brunswick Street, Nelson,

BB9 0PQ

Type of Liquidation: Members
Date of Appointment: 23 March 2022

Christopher James Lawton (IP No. 23818) and Dean Watson (IP No. 009661) both of Begbies Traynor (Central) LLP, 2-3 Winckley Court,

Chapel Street, Preston, PR1 8BU By whom Appointed: Members

Ag OH123374 (4035515)

Name of Company: VICTORIA BELL PROJECT MANAGEMENT LIMITED

Company Number: 12172683

Nature of Business: Management consultancy activities other than

financial management

Registered office: Bury Lodge, Bury Road, Stowmarket, Suffolk, IP14

1JA

Type of Liquidation: Members Date of Appointment: 28 March 2022

Andrew Anderson Kelsall (IP No. 009555) and Lee Anthony Green (IP No. 015610) both of Larking Gowen LLP, King Street House, 15

Upper King Street, Norwich, Norfolk, NR3 1RB

By whom Appointed: Members

Ag OH123312 (4035446)

Name of Company: VIMIAC LIMITED

Company Number: 10919125

Nature of Business: Information technology consultancy activities Registered office: 71 - 75 Shelton Street, Covent Garden, London,

WC2H 9JQ

Type of Liquidation: Members Date of Appointment: 28 March 2022

Donald McNaught (IP No. 9359) of Johnston Carmichael LLP, 227

West George Street, Glasgow, G2 2ND By whom Appointed: The Company

Ag OH123390 (4035564)

Name of Company: VIPAMEDIA LIMITED

Company Number: 04663044

Nature of Business: Information technology consultancy activities Registered office: 2 Mount Parade, Harrogate, North Yorkshire, HG1 1BX to be changed to FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18 3TX

Principal trading address: 2 Mount Parade, Harrogate, North

Yorkshire, HG1 1BX

Type of Liquidation: Members

Date of Appointment: 17 March 2022

Date of Appointment: 17 March 2022

David Antony Willis (IP No. 9180) and Martyn James Pullin (IP No. 15530) both of FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18 3TX

By whom Appointed: Members

Ag OH123281 (4035465)

Name of Company: WARMA HEATING AND PLUMBING SUPPLIES LIMITED

Company Number: 01158765

Registered office: 79A High Road, Willesden, London NW10 2SU Principal trading address: 308-312 Slade Road, Birmingham B23 7LU Nature of Business: Sale of plumbing and heating supplies

Type of Liquidation: Members

Liquidator's name and address: *K S Tan* of K S Tan & Co, 79A High Road, Willesden, London NW10 2SU, Email: mail@kstan.co.uk Tel: 020 7586 1280

Office Holder Number: 8032. Date of Appointment: 24 March 2022

By whom Appointed: Members (4037814)

Company Number: 02965839

Name of Company: **WILLIAM COLE LIMITED**Nature of Business: Letting & operating of real estate

Registered office: MB Insolvency, 11 Roman Way, Berry Hill,

Droitwich, WR9 9AJ

Type of Liquidation: Members Voluntary Liquidation

Liquidator: *Mark Bowen* (IP number 8711) of MB Insolvency , 11 Roman Way, Berry Hill, Droitwich Worcestershire WR9 9AJ.

Date of Appointment: 30 March 2022 By whom Appointed: The Members

For further details contact Sophie Murcott on 01905 776 771 or at sophiemurcott@mb-i.co.uk (4037378)

Name of Company: YORK ENDOCRINE SERVICES LIMITED

Company Number: 04635337

Nature of Business: Consultancy services to healthcare sector Registered office: 11 Clifton Moor Business Village, James Nicolson

Link, Clifton Moor, York, YO30 4XG Type of Liquidation: Members Date of Appointment: 29 March 2022

Michael Jenkins (IP No. 20114) and David Adam Broadbent (IP No. 009458) both of Begbies Traynor (Central) LLP, 11 Clifton Moor Business Village, James Nicolson Link, Clifton Moor, York, YO30 4XG

By whom Appointed: Members

Ag OH123355 (4035503)

## **NOTICES TO CREDITORS**

#### THE INSOLVENCY ACT 1986 STEEDMAN CVC LTD

(Company Number 07800123)

("the Company")

Registered office: 9 Hollands Field, Broadbridge Heath, Horsham, West Sussex, RH12 3HQ

Principal trading address: 9 Hollands Field, Broadbridge Heath, Horsham, West Sussex, RH12 3HQ

NOTICE IS HEREBY GIVEN that the creditors of the above named company, which Is being voluntarily wound up, are required, on or before the 4 May 2022 to send In their names and addresses and particulars of their debts or claims and of any security held by them, and the names and addresses of their Solicitors (If any) to the undersigned Michael James Gregson, of TC Bulley Davey Limited, 4 Cyrus Way, Cygnet Park, Hampton, Peterborough PE7 8HP (Office Holder number: 9339), the Liquidator of the said Company, and If so required by notice in writing from the Liquidator, are, by their Solicitors or personally, to come In and prove their debts or claims and establish any title they may have to priority, at such time and place as shall be specified in such Notice or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved, or such priority Is established, or as the case may be, from objecting to such distribution.

Alternative contact: Denise Gordon, denisegordon@tc-group.com - Telephone: 01733 569494.

Note: This Notice Is purely formal. All creditors of the company have been, or will be, paid In full.

MICHAEL JAMES GREGSON

Liquidator

23 March 2022 (4037813)

## ANJTECH CONSULTANCY LTD

(Company Number 09431942)

Registered office: c/o Neum Insolvency, Suite 9, Amba House, 15 College Road, Harrow, Middlesex, HA1 1BA

Principal trading address: Flat 1, 24 Onyx Crescent, Leicester, LE4 9A.I

Notice is hereby given that creditors of the Company are required on or before 17 May 2022 to send their names and addresses with particulars of their debts or claims (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016), and the names and addresses of their solicitors, if any, to the Liquidator at Neum Insolvency, Suite 9, Amba House, 15 College Road, Harrow, Middlesex, HA1 1BA and, if so required by notice in writing from the said Liquidator, either personally or by their solicitors, to come in and prove their debts or in default thereof they be excluded from the benefit of any distribution made before such debts are proved.

Note: The Director of the Company has made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 28 March 2022

Office Holder Details: *Umang Patel* (IP No. 18790) of Neum Insolvency, Suite 9, Amba House, 15 College Road, Harrow, Middlesex, HA1 1BA

Further details contact: Umang Patel, Email: Umang@neuminsolvency.co.uk or Tel: 020 3411 9598. Alternative contact: Jwalant Patel, Email: jwalant@neuminsolvency.co.uk

Umang Patel, Liquidator

28 March 2022

Ag OH123358 (4035481)

## ANNIEM MEDIA LIMITED

(Company Number 10324538)

Registered office: C/O Antony Batty & Co, Thames Valley, Innovation Centre, 99 Park Drive, Milton Park, Oxfordshire, OX14 4RY

Principal trading address: 47 Brookdene Avenue, Watford, WD19 4LG Notice is hereby given that creditors of the Company are required, on or before 7 May 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Joint Liquidators at Antony Batty & Co, Thames Valley, Innovation Centre, 99 Park Drive, Milton Park, Oxford, OX14 4RY.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 29 March 2022

Office Holder Details: Lawrence King (IP No. 10452) and Matthew Waghorn (IP No. 9432) both of Antony Batty & Co, Thames Valley, Innovation Centre, 99 Park Drive, Milton Park, Oxford, OX14 4RY Further details contact: The Liquidators, Tel: 01235 856321 or Email: contact@kwr.antonybatty.com. Alternative contact: Adam Judd.

Lawrence King, Joint Liquidator

31 March 2022

Ag OH123378

(4035497)

## **AQUEL CONSULTING LIMITED**

(Company Number 08736879)

Registered office: Stamford House, Northenden Road, Sale, Cheshire M33 2DH Previous registered office Bennett Brooks & Co Limited Suite 345, 50 Eastcastle Street, Fitzrovia W1W 8EA

Steve Kenny (IP No.24030) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH and Richard Cole (IP No. 26070) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH give notice that we were appointed joint liquidators of the above-named company on 30 March 2022 by a resolution of members.

NOTICE IS HEREBY GIVEN that the creditors of the above named company which is being voluntarily wound up, are required, on or before 30 April 2022 to prove their debts by sending to the undersigned Steve Kenny of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH the joint liquidator of the company, written statements of the amounts they claim to be due to them from the company and, if so requested, to provide such further details or produce such documentary evidence as may appear to the joint liquidator to be necessary.

Please note that this is a solvent liquidation and therefore the joint liquidator is entitled to make the distribution without regard to the claim of any person in respect of a debt not proved.

Joint Liquidator: Steve Kenny (IP number 24030) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH.

Joint Liquidator: *Richard Cole* (IP number 2 6070) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33

Date of Appointment: 30 March 2022

For further details contact Adam Horan on 0161 637 8100 or at Adam@KBL-Advisory.com (4037848)

## ASPINALL IT CONSULTING LIMITED

(Company Number 09347908)

Registered office: Crown Chambers, Princes Street, Harrogate, HG1 1NJ

Principal trading address: N/A

Notice is hereby given that the creditors of the above named Company, over which I was appointed Joint Liquidator on 23 March 2022 are required, on or before 2 May 2022 to send their full names, their addresses and descriptions, full particulars of their debts or claims and the names and addresses of their solicitors (if any) to the undersigned David Antony Willis of FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18 3TX, the Joint Liquidator of the said Company, and, if so required by notice in writing from the said Joint Liquidator, are, personally or by their solicitors, to come in and prove their debts or claims at such time and place as shall be specified in such notice.

A distribution may be made without regard to the claim of any person in respect of a debt not proved.

The winding up is a members' voluntary winding up and it is anticipated that all debts will be paid.

Date of Appointment: 23 March 2022

Office Holder Details: *David Antony Willis* (IP No. 9180) and *Martyn James Pullin* (IP No. 15530) both of FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18 3TX

Further details contact: The Joint Liquidators, Tel: 01642 917555. Alternative contact: Kelly Mullen.

David Antony Willis, Joint Liquidator

23 March 2022 Ag OH123351

(4035453)

## **AUTUMNTIGER LIMITED**

(Company Number 04105391)

#### **EF PARTICIPANTS LIMITED**

(Company Number 03533476)

Registered office: (Both) Ernst & Young LLP, 1 More London Place, London SE1 2AF

Principal trading address: (Both) Forum St Pauls, 33 Gutter Lane, London, EC2V 8AS

As Joint Liquidators of the Companies, we hereby give notice that we intend to make a final distribution to their creditors. The last date for proving is 29 April 2022 and creditors of the Companies should by that date send their full names and addresses and particulars of their debts or claims to me, Matthew Coomber of Ernst & Young LLP, 1 More London Place, London SE1 2AF.

In accordance with Rule 14.38(1)(c) of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, we may thereafter make the proposed distribution without regard to the claim of any person in respect of a debt not yet proved.

Date of Appointment: 21 March 2022

Office Holder Details: *Matthew Coomber* (IP No. 24430) and *Samantha Jane Keen* (IP No. 9250) both of Ernst & Young, 1 More London Place, London, SE1 2AF

Further details contact: The Joint Liquidators, Tel: 020 7783 0293.

Alternative contact: Jack Park.

Matthew Coomber, Joint Liquidator

30 March 2022 Ag OH123287

(4035408)

## BHP IT LIMITED

(Company Number 07504925)

Previous Name of Company: Wakeco (427) Limited

Registered office: Unit 8 President Buildings, Savile Street East, Sheffield, S4 7UQ

Principal trading address: Unit 8 President Buildings, Savile Street East, Sheffield, S4 7UQ

The Company was placed into members' voluntary liquidation on 28 March 2022 and on the same date, Claire Elizabeth Dowson (IP Number: 19272) and Gareth David Rusling (IP Number: 9481), both of Begbies Traynor (SY) LLP, of 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ were appointed as Joint Liquidators of the Company.

NOTICE IS HEREBY GIVEN that the Creditors of the Company are required on or before 6 May 2022 to send in their names and addresses, particulars of their debts or claims and the names and addresses of their Solicitors (if any) to the undersigned of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ the Joint Liquidator of the Company and, if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Joint Liquidator: Claire Elizabeth Dowson (IP number 19272) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ.

Joint Liquidator: *Gareth David Rusling* (IP number 9481) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ.

Date of Appointment: 28 March 2022

For further details contact Gemma Anderson on 0114 275 5033 or at sheffield.north@begbies-traynor.com (4037844)

#### **BRILLIANT BASICS HOLDINGS LIMITED**

(Company Number 09767352)

Registered office: The Bower 10/11th Floor, 207-211 Old Street, London, EC1V 9NR

NOTICE IS HEREBY GIVEN that the creditors of the above-named company, which is being voluntarily wound up, must send their full names and addresses (and those of their Solicitors, if any), together with full particulars of their debts or claims to the Liquidator at Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London, EC1M 7AD by 22 April 2022.

The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: It is anticipated that all known Creditors will be paid in full.

Liquidator: Ryan Davies (IP number 15250) of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London EC1M 7AD.

Date of Appointment: 31 March 2022

For further details contact Tara Wessely on 0207 566 4020 or at twessely@mks.co.uk (4037836)

#### BRILLIANT BASICS LTD

(Company Number 08011232)

Registered office: The Bower 10th/11th Floor, 207-211 Old Street, London EC1V 9NR

NOTICE IS HEREBY GIVEN that the creditors of the above-named company, which is being voluntarily wound up, must send their full names and addresses (and those of their Solicitors, if any), together with full particulars of their debts or claims to the Liquidator at Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London, EC1M 7AD by 22 April 2022.

The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: It is anticipated that all known Creditors will be paid in full.

Liquidator: Ryan Davies (IP number 15250) of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London, EC1M 7AD.

Date of Appointment: 31 March 2022

For further details contact Tara Wessely on 0207 566 4020 or at twessely@mks.co.uk (4036327)

## C L J CONSULTANCY LIMITED

(Company Number 11883647)

Registered office: 8th Floor, One Temple Row, Birmingham B2 5L6 Principal trading address: 5 Pauline Crescent, Whitton, Twickenham, TW2 8JD

The Company was placed into members' voluntary liquidation on 24 March 2022 and on the same date, *Craig Povey* (IP No. 9665) and *Charles Turner* (IP No. 9195) both of Begbies Traynor (Central) LLP, 8th Floor, One Temple Row, Birmingham, B2 5LG were appointed as Joint Liquidators of the Company.

Notice is hereby given that the Creditors of the Company are required on or before 25 April 2022 to send in their names and addresses, particulars of their Debts or Claims and the names and addresses of their Solicitors (if any) to the undersigned Craig Povey of Begbies Traynor (Central) LLP, 8th Floor, Temple Point, 1 Temple Row, Birmingham, B2 5LG and if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Any person who requires further information may contact Katie Shingler by telephone on 0121 200 8162 or alternatively by email at birmingham@btguk.com.

Craig Povey, Joint Liquidator

30 March 2022

Ag OH123323

(4035339)

## **C4 PROJEX LIMITED**

(Company Number 11314436)

Registered office: Little Bourne, 2 Slate Way, Pitton, Salisbury, SP5

Principal trading address: Little Bourne, 2 Slate Way, Pitton, Salisbury, SP5 1EE

Notice is hereby given that the Creditors of the Company are required, on or before 29 April 2022 to send their names and addresses and particulars of their debts or claims and the names and addresses of their solicitors (if any) to Lauren Rachel Cullen of Cullen & Co UK Limited, 197 Kingston Road, Epsom, Surrey KT19 0AB, the Liquidator of the company, and, if so required by notice in writing from the Liquidator, by her solicitors or personally, to come in and prove their debts or claims at such time and place as shall be specified in any such notice, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.

The Director has made a Declaration of Solvency, and the Company is being wound up for the purposes of distribution of surplus assets to shareholders.

NOTE: This notice is purely formal. All known creditors have been or will be paid in full.

Date of Appointment: 16 March 2022

Office Holder Details: Lauren Rachel Cullen (IP No. 18050) of Cullen & Co UK Limited, 197 Kingston Road, Epsom, Surrey, KT19 0AB

Further details contact: Lauren Rachel Cullen, Tel: 0203 8877 200.

Alternative contact: Gemma Lawrence. Lauren Rachel Cullen, Liquidator

29 March 2022 Ag OH123251

(4035360)

#### **CALCON RAIL LTD**

(Company Number 06904156)

Registered office: 117 North Road, Crawley, West Sussex, RH10 1SQ Principal trading address: 117 North Road, Crawley, West Sussex, **RH10 1SQ** 

Nature of Business: Other Business Support Service Activities N.E.C. Final Date For Submission: 2 May 2022.

Notice is hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the joint liquidators of the Company named above (in members' voluntary liquidation) intend to make final distributions to creditors. Creditors are required to prove their debts on or before the final date for submission specified in this notice by sending full details of their claims to the joint liquidators. Creditors must also, if so requested by the joint liquidators, provide such further details and documentary evidence to support their claims as the joint liquidators deem necessary.

The intended distributions are final distributions and may be made without regard to any claims not proved by the final date for submission specified in this notice. Any creditor who has not proved their debt by that date, or who increases the claim in their proof after that date, will not be entitled to disturb the intended final distributions. The joint liquidators intend that, after paying or providing for final distributions in respect of creditors who have proved their claims, all funds remaining in the joint liquidators' hands following the final distributions to creditors shall be distributed to the shareholders of the Company absolutely.

The directors have made a statutory declaration that the Company is able to pay all known liabilities in full.

Date of Appointment: 21 March 2022

Joint Liquidator's Name and Address: Richard Keley (IP No. 18072) of BM Advisory, Arundel House, 1 Amberley Court, Whitworth Road, Crawley, RH11 7XL. Telephone: 01293 410333 .

Joint Liquidator's Name and Address: Andrew Pear (IP No. 9016) of BM Advisory, Arundel House, 1 Amberley Court, Whitworth Road, Crawley, RH11 7XL. Telephone: 01293 410333 .

For further information contact Sue Markham at the offices of BM Advisory on 01293 452844, or Sue.Markham@bm-advisory.com.

29 March 2022 (4036371)

## **CORFU CONSULTANCY LIMITED**

(Company Number 08390402)

Registered office: c/o KRE Corporate Recovery Limited, Unit 8 The Aquarium, 1-7 King Street, Reading, Berks RG1 2AN

Principal trading address: c/o Hurst Hill, Churt Road, Hindhead, Surrey, GU26 6HT (previous Registered Office)

Notice is hereby given, that creditors of the Company are required, on or before 29 April 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Joint Liquidators at KRE Corporate Recovery Limited, Unit 8, The Aquarium, 1-7 King Street, Reading, RG1 2AN or info@krecr.co.uk.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 29 March 2022

Office Holder Details: Rob Keyes (IP No. 008841) and Paul Ellison (IP No. 7254) both of KRE Corporate Recovery Limited, Unit 8, The Aquarium, 1-7 King Street, Reading, RG1 2AN

For further details contact: Lee Lloyd, Email: lee.lloyd@krecr.co.uk, Tel: 01189 479090.

Rob Keyes, Joint Liquidator

31 March 2022

Ag OH123365

(4035569)

## NOTICE TO CREDITORS TO SUBMIT CLAIMS **COTTONBURY PLANNING LIMITED**

(Company Number 11016106)

Registered office: Menta Business Centre, 5 Eastern Way, Bury St Edmunds IP32 7AB

Principal trading address: Field House, Bells Road, Belchamp Walter, Sudbury, Suffolk CO10 7A

Nature of company's business: Planning consultancy

NOTICE IS HEREBY GIVEN that the creditors of the above named company, which is being voluntarily wound up, are required on or before 3 May 2022, to send in their names and addresses and full particulars of their debts or claims, and the names and addresses of their solicitors (if any), to Stephen Rout of Menta Business Centre, 5 Eastern Way, Bury St Edmunds IP32 7AB and, if so requested by me, to provide such further details or produce such documentary or other evidence as may appear to be necessary.

A creditor who has not provided details of their debt before the date specified above will be excluded from any dividend paid after that date and is not entitled to disturb by reason that they have not participated in it, the dividend so declared.

Note: The liquidation is a Members Voluntary Liquidation and it is anticipated that all debts will be paid in full. Stephen Rout (IP No. 6062) of Menta Business Centre, 5 Eastern Way, Bury St Edmunds, Suffolk IP32 7AB (email smrout@aol.com/tel 01223 329392) was appointed liquidator on 24 March 2022.

Stephen Rout

Liquidator

30 March 2022

(4037815)

## **CRAY U.K. LIMITED**

(Company Number 03941275)

Previous Name of Company: HACKREMCO (NO.1622) LIMITED

Registered office: Ground Floor, 210 Wharfedale Road, Winnersh Triangle, Berkshire, United Kingdom, RG41 5TP

NOTICE IS HEREBY GIVEN that the creditors of the above named company, which is being voluntarily wound up, must send their full names and addresses (and those of their Solicitors, if any), together of their particulars debts or laura.m.waters@pwc.com at PricewaterhouseCoopers LLP, 7 More London Riverside, London, SE1 2RT by 05 May 2022.

The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: It is anticipated that all known Creditors will be paid in full.

Joint Liquidator: Laura Waters (IP number 9477) PricewaterhouseCoopers LLP, 7 More London Riverside, London, SE1 2RT.

Joint Liquidator: Steven Sherry (IP number 19752) PricewaterhouseCoopers LLP, 7 More London Riverside, London,

Date of Appointment: 31 March 2022

For further details contact Anne Louise Mccartney on 07802 659389 or at anne.louise.mccartney@pwc.com

Email:

(4035363)

Hayter,

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators' appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators.

Hators

Further

30 March 2022

Aa OH123265

**CREATED EDUCATION LIMITED** 

(Company Number 10889843)

Registered office: 8 C/O Hansel Henson Ltd, 3rd Floor, 8 Bloomsbury Street, London, WC1B 3SR

Principal trading address: 8 C/O Hansel Henson Ltd, 3rd Floor, 8 Bloomsbury Street, London, WC1B 3SR

Nature of Business: Other business support service activities not elsewhere classified.

Final Date For Submission: 30 April 2022.

Notice is hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the joint liquidators of the Company named above (in members' voluntary liquidation) intend to make final distributions to creditors. Creditors are required to prove their debts on or before the final date for submission specified in this notice by sending full details of their claims to the joint liquidators. Creditors must also, if so requested by the joint liquidators, provide such further details and documentary evidence to support their claims as the joint liquidators deem necessary.

The intended distributions are final distributions and may be made without regard to any claims not proved by the final date for submission specified in this notice. Any creditor who has not proved their debt by that date, or who increases the claim in their proof after that date, will not be entitled to disturb the intended final distributions. The joint liquidators intend that, after paying or providing for final distributions in respect of creditors who have proved their claims, all funds remaining in the joint liquidators' hands following the final distributions to creditors shall be distributed to the shareholders of the Company absolutely.

The directors have made a statutory declaration that the Company is able to pay all known liabilities in full.

Date of Appointment: 23 March 2022

Joint Liquidator's Name and Address: *Vincent John Green* (IP No. 009416) of Crowe U.K. LLP, 4 Mount Ephraim Road, Tunbridge Wells, Kent, TN1 1EE. Telephone: 01892 700200.

Joint Liquidator's Name and Address: Steven Edwards (IP No. 26090) of Crowe U.K. LLP, 4 Mount Ephraim Road, Tunbridge Wells, Kent, TN1 1EE. Telephone: 01892 700200.

For further information contact George Budd at the offices of Crowe U.K. LLP on 01892 700200, or recovery solutions@crowe.co.uk.

24 March 2022 (4036380)

## **DALMAN (IMPORT & EXPORT) LIMITED**

(Company Number 01169794)

Registered office: 47 Hill Farm Road, Marlow, Buckinghamshire, SL7 3LX

Principal trading address: Unit 11 Thomas Road, Wooburn Industrial Park, Wooburn Green, High Wycombe, HP10 0PE

Notice is hereby given that creditors of the Company are required, on or before 27 April 2022 to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Quantuma Advisory Limited, The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 30 March 2022

Office Holder Details: *Chris Newell* (IP No. 13690) and *David Meany* (IP No. 9453) both of Quantuma Advisory Limited, The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH

# DEWDROP CONSULTING LTD

details

Chris Newell, Joint Liquidator

(Company Number 08893944)

Registered office: Suite 4 Portfolio House, 3 Princes Street, Dorchester, Dorset, DT1 1TP

Kirsty

contact:

Kirsty.Hayter@quantuma.com, Tel: 01202 970467.

Principal trading address: 12 High Street, Stonehouse, Gloucestershlre, GL10 2NA

Notice is hereby given pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016 that the Joint Liquidators of the above named company intend to declare and pay a first and final dividend/distribution to creditors.

The creditors of the Company are required, by 13 May 2022 to deliver their proofs of debt to the undersigned of Purnells, 5a Kernick Industrial Estate, Penryn, Cornwall, TR10 9EP, the Joint Liquidators of the company, and if so requested, to provide such further details or produce such documentary or other evidence as may appear to the Joint Liquidators to be necessary.

A creditor who has not proved his debt before the date mentioned above will not be entitled to disturb, by reason that he has not participated in it. As this is a sole and final distribution it will be made without regard to the claim of any person in respect of a debt not proved.

The dividend distribution to creditors will be paid within two months of the last date for proving as indicated above.

The final distribution to the shareholders may be made without regard to the claim of any person in respect of a debt not proved.

Date of Appointment: 18 March 2022

Office Holder Details: *Lisa Alford* (IP No. 9723) and *Chris Parkman* (IP No. 9588) both of Purnells, 5a Kernick Industrial Estate, Penryn, Cornwall, TR10 9EP

For further details contact: The Joint Liquidators, Tel: 01326 340579, Fax: 01326 340461, Email: lisa@purnells.co.uk or helen@purnells.co.uk

Lisa Alford, Joint Liquidator

18 March 2022

Ag OH123308 (4035355)

## **EJO HOLDINGS LIMITED**

(Company Number 12188219)

Registered office: Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA

Principal trading address: Unit 15a, Norman Way, Over, Cambridge, CB24 5QE

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that further to the appointment of the Joint Liquidators on 3 February 2022, they intend to declare a first and final dividend to creditors of the above company within two months of the last date for proving, specified below.

Notice is hereby given that creditors of the Company are required, on or before 29 April 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Joint Liquidators at Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Creditors who have not yet done so must prove their debts by sending their full names and addresses, particulars of their debts or claims and the names and addresses of their solicitors (if any), to the Joint Liquidators at Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA by no later than 29 April 2022 (the last date for proving).

As the distribution will be a final distribution, it may be made without regard to the claim of any person in respect of a debt not proved.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 3 February 2022

Office Holder Details: Steve Markey (IP No. 14912) and Mark Colman (IP No. 9721) both of Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA

For further details contact: The Joint Liquidators, Tel: 0161 413 0930. Alternative contact: Usman Marghoob.

Steve Markey, Joint Liquidator

30 March 2022 Ag OH123387

(4035460)

#### **EPIRIS CAPITAL PARTNERS LLP**

(Company Number OC412420)

Registered office: 1 More London Place, London SE1 2AF

Principal trading address: Forum St Pauls, 33 Gutter Lane, London, EC2V 8AS

As Joint Liquidators of the Partnership, we hereby give notice that we intend to make a final distribution to its creditors. The last date for proving is 29 April 2022 and creditors of the Partnership should by that date send their full names and addresses and particulars of their debts or claims to me, Matthew Coomber of Ernst & Young LLP, 1 More London Place, London SE1 2AF.

In accordance with Rule 14.38(1)(c) of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, we may thereafter make the proposed distribution without regard to the claim of any person in respect of a debt not yet proved.

Date of Appointment: 21 March 2022

Office Holder Details: *Matthew Coomber* (IP No. 24430) and *Samantha Jane Keen* (IP No. 9250) both of Ernst & Young, 1 More London Place, London, SE1 2AF

For further details contact: The Joint Liquidators, Tel: 020 7783 0293. Alternative contract: Jack Park.

Matthew Coomber, Joint Liquidator

30 March 2022

Ag OH123288 (4035356)

## **EQUITY IN FINANCE HOLDINGS LIMITED**

(Company Number 08966368)

Registered office: Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Greater Manchester, M45 7TA

Principal trading address: Hilton Hall, Hilton Lane, Essington, Wolverhampton, WV11 2BQ

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that further to the appointment of the Joint Liquidators on 3 December 2021, they intend to declare a first and final dividend to creditors of the above company within two months of the last date for proving, specified below.

Notice is hereby given that creditors of the Company are required, on or before 29 April 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Joint Liquidators at Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Creditors who have not yet done so must prove their debts by sending their full names and addresses, particulars of their debts or claims and the names and addresses of their solicitors (if any), to the Joint Liquidators at Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA by no later than 29 April 2022 (the last date for proving).

As the distribution will be a final distribution, it may be made without regard to the claim of any person in respect of a debt not proved.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 3 December 2021

Office Holder Details: Steve Markey (IP No. 14912) and Conrad Beighton (IP No. 9556) both of Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA

For further details contact: The Joint Liquidators, Tel: 0161 413 0930. Alternative contact: Usman Marghoob.

Steve Markey, Joint Liquidator

31 March 2022

Ag OH123388 (4035604)

## **ESTELIA LIMITED**

(Company Number 09688132)

Registered office: Purnells, Suite 4 Portfolio House, 3 Princes Street, Dorchester, Dorset, DT1 1TP

Principal trading address: Lismore, The Purlieu, Malvern, WR14 4DJ Notice is hereby given pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016 that the Joint Liquidators of the above named company intend to declare and pay a first and final dividend/distribution to creditors.

The creditors of the Company are required by 12 May 2022 to deliver their proofs of debt to the undersigned of Purnells, 5a Kernick Industrial Estate, Penryn, Cornwall, TR10 9EP, the Joint Liquidators of the company, and if so requested, to provide such further details or produce such documentary or other evidence as may appear to the Joint Liquidators to be necessary.

A creditor who has not proved his debt before the date mentioned above will not be entitled to disturb, by reason that he has not participated in it. As this is a sole and final distribution it will be made without regard to the claim of any person in respect of a debt not proved.

The dividend distribution to creditors will be paid within two months of the last date for proving as indicated above.

The final distribution to the shareholders may be made without regard to the claim of any person in respect of a debt not proved.

Date of Appointment: 17 March 2022

Office Holder Details: *Lisa Alford* (IP No. 9723) and *Chris Parkman* (IP No. 9588) both of Purnells, 5a Kernick Industrial Estate, Penryn, Cornwall, TR10 9EP

Further details contact: The Liquidators, Email: lisa@purnells.co.uk or helen@purnells.co.uk, Tel: 01326 340579, Fax: 01326 340461.

Lisa Alford, Joint Liquidator

17 March 2022

Ag OH123278 (4035411)

#### FIELDER GAMMON INVESTMENT COMPANY LIMITED

(Company Number 00729537)

Registered office: WSM MBI Coakley LLP, 2nd Floor, Shaw House, 3 Tunsgate, Guildford, Surrey, GU1 3QT

Principal trading address: 8 Knoll Quarry, Godalming, Surrey, GU7 3FS

Notice is hereby given that Creditors of the Company are required, on or before 10 May 2022 (the last date for proving), to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016), including particulars of their debts or claims, and the names and addresses of their solicitors (if any) to the Liquidator at WSM MBI Coakley LLP, 2nd Floor, Shaw House, 3 Tunsgate, Guildford, Surrey, GU1 3QT.

If so required by notice from the Liquidator, creditors must produce any document or other evidence which the Liquidator considers to be necessary to substantiate the whole or any part of a claim.

Notice is also hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the Liquidator intends to declare a first and final distribution to unsecured creditors of the Company within the period of two months from the last date for proving as specified above.

The distribution may be made without regard to the claim of any person in respect of a debt not proved by the date of last proving.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 21 March 2022

Office Holder Details: *Dermot Coakley* (IP No. 6824) of WSM MBI Coakley LLP, 2nd Floor, Shaw House, 3 Tunsgate, Guildford, Surrey, GU1 3QT

For further details contact: The Joint Liquidators, Tel: 01483 405160, Email: forum@mbicoakley.co.uk. Alternative contact: Cameron Harrison, Tel: 01483 405160, Email: cameron.harrison@mbicoakley.co.uk.

Dermot Coakley, Liquidator

30 March 2022

Ag OH123339 (4035477)

#### **FOX AND HOUNDS HOTEL LIMITED**

(Company Number 06296746)

Registered office: 138 High Street, Crediton, Devon, EX17 3DX

Principal trading address: Fox & Hounds Hotel, Eggesford, Chulmleigh, EX18 7JZ

Notice is hereby given that creditors of the Company are required, on or before 11 May 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Joint Liquidators at Castle Hill Insolvency Limited, 1 Battle Road, Heathfield, Newton Abbot, TQ12 6RY.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 23 March 2022

Office Holder Details: Steve Henson (IP No. 10652) and Adam Buck (IP No. 23070) both of Castle Hill Insolvency Limited, 1 Battle Road, Heathfield, Newton Abbot, TQ12 6RY

Further details contact: The Liquidator, Tel: 01626 510020 or E-mail: adam.buck@castlehillinsolvency.com. Alternative contact: Tina Grimes.

Steve Henson, Joint Liquidator 30 March 2022 Ag OH123304

(4035592)

The Company was placed into members' voluntary liquidation on 21 March 2022 and on the same date, Office Holder Details: Louise Donna Baxter (IP No. 009123) and Lloyd Biscoe (IP No. 9141) both of Begbies Traynor (Central) LLP, Suite WG3, The Officers' Mess Business Centre, Royston Road, Duxford, Cambridge, CB22 4QH were appointed as Joint Liquidators of the Company.

Notice is hereby given that the Creditors of the Company are required on or before 13 May 2022 to send in their names and addresses, particulars of their debts or claims and the names and addresses of their solicitors (if any) to the undersigned Louise Baxter of Begbies Traynor (Central) LLP, Suite WG3, The Officers' Mess Business Centre, Royston Road, Duxford, Cambridge, CB22 4QH the Joint Liquidator of the Company and, if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Any person who requires further information may contact the Joint Liquidator by telephone on 01223 495660. Alternatively enquiries can be made to Carol Wilson on 01223 495660 or by e-mail at carol.wilson@btguk.com

Louise Baxter, Joint Liquidator 30 March 2022

Ag OH123348 (4035532)

## FRESH FRUIT DISTRIBUTION LIMITED

(Company Number 07209941)

Registered office: 29 High Street, Blue Town, Sheerness, ME12 1RN Principal trading address: The Old Pumping Station, Breille Way, Sheerness, Kent, ME12 1YW

Notice is hereby given under Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, by Laurence Pagden and Jonathan David Bass both of Menzies LLP, Lynton House, 7-12 Tavistock Square, London, WC1H 9LT, the Joint Liquidators, to the creditors of Fresh Fruit Distribution Limited, that we intend declaring a first and final dividend to the unsecured creditors of the Company within two months of the last date for proving specified below.

Every person claiming to be a creditor of the Company is required on or before 11 May 2022, which is the last date for proving, to submit a proof of debt to us at Menzies LLP, Lynton House, 7-12 Tavistock Square, London, WC1H 9LT and, if so requested by us, to provide such further details or produce such documentary or other evidence as may appear to be necessary.

A creditor who has not proved his debt before the last date for proving mentioned above is not entitled to disturb, by reason that he has not participated in the dividend, the distribution of that dividend or any other dividend declared before his debt is proved. The winding up of the Company is a members' voluntary winding up. The distribution proposed to be made is to be the final distribution in the winding up of the above named Company and, accordingly, the Joint Liquidators may make the distribution without regard to the claim of any person in respect of a debt not already proved.

Date of Appointment: 23 March 2022

Office Holder Details: Laurence Pagden (IP No. 9055) and Jonathan David Bass (IP No. 11790) both of Menzies LLP, Lynton House, 7-12 Tavistock Square, London, WC1H 9LT

Further details contact: The Joint Liquidators, Email: JLe@menzies.co.uk, or Tel: 0207 465 1905. Alternative contact: Jessica Le.

Laurence Pagden, Joint Liquidator 23 March 2022

Ag OH123364

(4035576)

## **G.S.N PROPERTY DEVELOPMENT LIMITED**

(Company Number 12073368)

Registered office: Begbies Traynor, Suite WG3, The Officers' Mess Business Centre, Royston Road, Duxford, Cambridge, CB22 4QH Principal trading address: Pertenhall Road, The Town, Great Staughton, St Neots, PE19 5BE

# NOTICE TO CREDITORS GATEWAY ACCOUNTING LIMITED

(Company Number 11328283)

Registered office: 2 Chestnut Close, Orpington, BR6 6LP

Principal trading address: 2 Chestnut Close, Orpington, BR6 6LP

NOTICE IS HEREBY GIVEN that Richard Frank Simms of F A Simms & Partners Limited, Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB was appointed liquidator of the above company by the members on 25 March 2022.

NOTICE IS ALSO HEREBY GIVEN that the creditors of the above named company are required on or before 20 April 2022 to send their names and addresses with particulars of their debt to Richard Frank Simms of F A Simms & Partners Limited, Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB the liquidator of the said company. Failure to submit a claim by this date will result in you being excluded from any distributions paid. Please note that this is a solvent liquidation and all known creditors have been or will be paid in full.

Name of Office Holders: Richard Frank Simms

Office Holder Number: 9252

Address of Office Holder: Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire

Capacity: Liquidator

Date of Appointment: 25 March 2022

Further Details: Michelle Collier Tel No. 01455 555 444 (4037812)

## GEOFFREY BLUMENTHAL LIMITED

(Company Number 04233950)

Registered office: 22 Ensign Business Centre Westwood Way, Westwood Business Park, Coventry, CV4 8JA

Principal trading address: 22 Ensign Business Centre Westwood Way, Westwood Business Park, Coventry, CV4 8JA

Notice is hereby given that the creditors of the above named Company, which was voluntarily wound up on 24 March 2022 are required, on or before 27 April 2022 to send their full names and addresses together with full particulars of their debts or claims to Gavin Bates of Smart Business Recovery Limited, The Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG, and, if so requested by me, to provide such further details or produce such documentary or other evidence as may appear to be necessary, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.

Note: This is a solvent liquidation and all known creditors have been or will be paid in full.

Date of Appointment: 24 March 2022

Office Holder Details: Gavin Bates (IP No. 8983) of Smart Business Recovery Limited, The Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG

Further details contact Sharon Hill on telephone 0116 2325117 or by email at sharon@smartbusinessrecovery.co.uk

Gavin Bates, Liquidator 30 March 2022

Aq OH123338

(4035348)

#### **GOLD GROUP PROPERTIES LIMITED**

(Company Number 06396536)

Previous Name of Company: Pimco 2705 Limited (From 11 October 2007 to 20 November 2007)

Registered office: Gold Group House, Godstone Road, Whyteleafe, CR3 0GG which is in the process of being changed to c/o Quantuma Advisory Limited, High Holborn House, 52-54 High Holborn, London,

Principal trading address: Gold Group House, Godstone Road, Whyteleafe, CR3 0GG

Notice is hereby given that creditors of the Company are required, on or before 4 May 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Joint Liquidators at Quantuma Advisory Limited, High Holborn House, 52-54 High Holborn, London, WC1V

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 22 March 2022

Office Holder Details: Simon Bonney (IP No. 9379) and Michael Kiely (IP No. 9617) both of Quantuma Advisory Limited, High Holborn House, 52-54 High Holborn, London, WC1V 6RL

details contact: Stokes. Sian.Stokes@guantuma.com, Tel: 020 3856 6745.

Simon Bonney, Joint Liquidator

29 March 2022

Aq OH123369

(4035593)

## J READHEAD SOLUTIONS LTD

(Company Number 11079915)

Registered office: Flat 2 Curzon House, 3 St. Peters Street, Ipswich, Suffolk, IP1 1XF

Principal trading address: Flat 2 Curzon House, 3 St. Peters Street, Ipswich, Suffolk, IP1 1XF

Notice is hereby given pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Liquidators intend to declare a first and final distribution to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by sending their full names and addresses, particulars of their debts or claims, and the names and addresses of their solicitors (if any), to the Joint Liquidators at King Street House, 15 Upper King Street, Norwich, NR3 1RB by no later than 25 April 2022 (the last date for

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 28 March 2022

Office Holder Details: Andrew Anderson Kelsall (IP No. 009555) and Lee Anthony Green (IP No. 015610) both of Larking Gowen LLP, King Street House, 15 Upper King Street, Norwich, Norfolk, NR3 1RB

For further details contact: Lee Green, Email: lee.green@larkinggowen.co.uk, Tel: 01603 624181. or Alternative contact: Rebecca Bloomfield, Email: rebecca.bloomfield@larking-gowen.co.uk

Andrew Anderson Kelsall, Joint Liquidator

30 March 2022

Ag OH123302 (4035602)

## JES HOTELS LTD

(Company Number 12138027)

Registered office: 17 Moor Park Avenue, Preston, PR1 6AS

Principal trading address: 61 South Promenade, Lytham St Annes,

FY8 1LZ

Notice is hereby given that Creditors of the Company are required, on or before 26 April 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG. If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full. Date of Appointment: 24 March 2022.

Office Holder Details: John Paul Bell (IP No. 8608) and Toyah Marie Poole (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey

John Paul Bell, Joint Liquidator

31 March 2022

Ag OH123391

(4035484)

## **KMJ ASSOCIATES LTD**

(Company Number 07409030)

Registered office: First Floor, Telecom House, 125-135 Preston Road, Brighton BN1 6AF

Principal trading address: 1 Coles Farm, Chequers Lane, Watford, WD25 0GL

Notice is hereby given that creditors of the Company are required, on or before 13 May 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Liquidator at Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP.

If so required by notice from the Liquidator, creditors must produce any document or other evidence which the Liquidator considers is necessary to substantiate the whole or any part of a claim.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 21 March 2022

Office Holder Details: Matthew Fox (IP No. 9325) of Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP

For further details contact: Matthew Fox, Email: mfox@beaconllp.com or Tel: 02380 651441. Alternative contact: Joanne Hayward.

Matthew Fox, Liquidator

30 March 2022

Ag OH123307

(4035434)

## LA BAITA LIMITED

(Company Number 04595720)

Registered office: 2 Clarendon Road, Ashford, TW15 2QE

Principal trading address: Bandstand Cafe, Windmill Drive, Clapham Common, London, SW4 9DE

Notice is hereby given that Creditors of the Company are required, on or before 26 April 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG. If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full. Date of Appointment: 18 March 2022.

Office Holder Details: John Paul Bell (IP No. 8608) and Toyah Marie Poole (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Elizabeth Scott

John Paul Bell, Joint Liquidator

31 March 2022

Aq OH123375

(4035603)

## LANCASHIRE TEXTILE MANUFACTURERS ASSOCIATION

Registered office: N/A

Principal trading address: 4 St Andrew's Street, Blackburn, BB1 8AE I, *Charles MacMillan* (IP No. 6000) of Beever and Struthers, St George's House, 215-219 Chester Road, Manchester, M15 4JE give notice that I was appointed as liquidator of the above-named Association on 30 March 2022 by a resolution of members.

Notice is hereby given that the creditors of the above named Association which is being voluntarily wound up, are required on or before 29 April 2022 to prove their debts by sending to the undersigned Charles MacMillan of Beever and Struthers, 215-219 Chester Road, Manchester, M15 4JE, the Liquidator of the Association written statements of the amounts they claim to be due to them from the Association and, if so requested, to provide such further details or produce such documentary evidence as may appear to the Liquidator to be necessary.

Please note that this is a solvent liquidation and therefore the Liquidator is entitled to make the distribution without regard to the claim of any person in respect of a debt not proved.

For further details contact: Neil Clarke, Email: insolvency@beeverstruthers.co.uk, Tel: 0161 832 4901.

Charles MacMillan, Liquidator

30 March 2022

Ag OH123266

(4035567)

#### LANGLAND INVESTMENTS LIMITED

(Company Number 00565649)

Registered office: 2-8 Parkstone Road Towngate House, Poole, BH15 2PW

Principal trading address: c/o Ellis Jones Solicitors, 14a Haven Road, Poole, Dorset, BH13 7LP

Nature of Business: Property management.

Final Date For Submission: 11 May 2022.

Notice is hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the joint liquidators of the Company named above (in members' voluntary liquidation) intend to make final distributions to creditors. Creditors are required to prove their debts on or before the final date for submission specified in this notice by sending full details of their claims to the joint liquidators. Creditors must also, if so requested by the joint liquidators, provide such further details and documentary evidence to support their claims as the joint liquidators deem necessary.

The intended distributions are final distributions and may be made without regard to any claims not proved by the final date for submission specified in this notice. Any creditor who has not proved his debt by that date, or who increases the claim in his proof after that date, will not be entitled to disturb the intended final distributions. The joint liquidators intend that, after paying or providing for final distributions in respect of creditors who have proved their claims, all funds remaining in the joint liquidators' hands following the final distributions to creditors shall be distributed to the shareholders of the Company absolutely.

The directors have made a statutory declaration that the Company is able to pay all known liabilities in full.

Date of Appointment: 16 March 2022

Joint Liquidator's Name and Address: *Lucinda Clare Coleman* (IP No. 10710) of Francis Clark LLP, Centenary House, Peninsula Park, Rydon Lane, Exeter, EX2 7XE. Telephone: 01392 667000.

Joint Liquidator's Name and Address: Stephen James Hobson (IP No. 006473) of Francis Clark LLP, Centenary House, Peninsula Park, Rydon Lane, Exeter, EX2 7XE. Telephone: 01392 667000.

For further information contact George Bird at the offices of Francis Clark LLP on 01392 667000, or George.Bird@pkf-francisclark.co.uk. 28 March 2022 (4036364)

## **LONDON VET IMAGING LTD**

(Company Number 10164271)

Registered office: 9 Ensign House, Admirals Way, Marsh Wall, London E14 9XQ (formerly Plaza 9 KD Tower, Cotterells, Hemel Hempstead, Hertfordshire, HP1 1FW)

Principal trading address: (Former) Plaza 9 KD Tower, Cotterells, Hemel Hempstead, Hertfordshire, HP1 1FW

Notice is hereby given under Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that I, the Liquidator of the above-named Company, Richard Hunt of SFP Restructuring Limited, 9 Ensign House, Admirals Way, Marsh Wall, London E14 9XQ intend declaring a first and final dividend to unsecured creditors.

Creditors who have not already proved are required, on or before 10 May 2022, to submit their proofs of debt to me at SFP Restructuring Limited, 9 Ensign House, Admirals Way, Marsh Wall, London E14 9XQ and, if so requested by me, to provide such further details or produce such documentary or other evidence as may appear to be necessary. Please note that this is a solvent liquidation and therefore I am entitled to make the distribution and any further distribution to creditors or shareholders without regard to the claim of any person in respect of a debt not proved.

Please note: The last date for submitting a proof of debt is 10 May 2022. A proof of debt can be downloaded at https://www.gov.uk/government/publications/rule-144-proof-of-debt-general-form

Date of Appointment: 25 March 2022

Office Holder Details: *Richard Hunt* (IP No. 21772) of SFP Restructuring Limited, 9 Ensign House, Admirals Way, Marsh Wall, London E14 9XQ

For further details contact: Richard Hunt or Claire Harris, Tel: 020 7538 2222.

Richard Hunt, Liquidator

29 March 2022

Aa OH123240

(4035444)

#### LOST LAMBS PRODUCTIONS UK LIMITED

(Company Number 03667013)

Registered office: 12th Floor, Brunel Building, 2 Canalside Walk, London, W2 1DG

Principal trading address: 12th Floor, Brunel Building, 2 Canalside Walk, London, W2 1DG

Nature of Business: Motion picture production activities.

Final Date For Submission: 22 April 2022.

Notice is hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the joint liquidators of the Company named above (in members' voluntary liquidation) intend to make final distributions to creditors. Creditors are required to prove their debts on or before the final date for submission specified in this notice by sending full details of their claims to the joint liquidators. Creditors must also, if so requested by the joint liquidators, provide such further details and documentary evidence to support their claims as the joint liquidators deem necessary.

The intended distributions are final distributions and may be made without regard to any claims not proved by the final date for submission specified in this notice. Any creditor who has not proved their debt by that date, or who increases the claim in their proof after that date, will not be entitled to disturb the intended final distributions. The joint liquidators intend that, after paying or providing for final distributions in respect of creditors who have proved their claims, all funds remaining in the joint liquidators' hands following the final distributions to creditors shall be distributed to the shareholders of the Company absolutely.

The directors have made a statutory declaration that the Company is able to pay all known liabilities in full.

Date of Appointment: 24 March 2022

Joint Liquidator's Name and Address: *Thomas Edward Guthrie* (IP No. 15012) of BRI Business Recovery and Insolvency, 2nd Floor, Elm House, Woodlands Business Park, Linford Wood West, Milton Keynes, MK14 6FG. Telephone: 01908 317387.

Joint Liquidator's Name and Address: Sukhvinder Kaur Bains (IP No. 9990) of BRI Business Recovery and Insolvency, 2nd Floor, Elm House, Woodlands Business Park, Linford Wood West, Milton Keynes, MK14 6FG. Telephone: 01908 317387.

For further information contact Kathryn Harper at the offices of BRI Business Recovery and Insolvency on 01908 771353, or kharper@briuk.co.uk.

29 March 2022 (4036395)

## LTR RECOVERY SERVICES LIMITED

(Company Number 07448855)

Registered office: Unit 1-2 Florida Close, Hot Lane Industrial Estate, Stoke on Trent, ST6 2DJ

Principal trading address: N/A

The Company was placed into members' voluntary liquidation on 23 March 2022 when *Michaela Daly* (IP No. 17570) of Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF and *Louise Longley* (IP No. 23874) of Begbies Traynor (Central) LLP, Fourth Floor, Toronto Square, Toronto Street, Leeds, LS1 2HJ were appointed as Joint Liquidators of the Company.

Notice is hereby given that the creditors of the company are required on or before 27 April 2022, to send in their names and addresses, particulars of their debts or claims and the names and addresses of their solicitors (if any) to the undersigned Michaela S Daly of Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF, the Joint Liquidator of the Company, and if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

The Company is able to pay all its known creditors in full.

Any person who requires further information may contact the Liquidators on 01782 569510 or Abigail Spruce by telephone on 01782 569516 or alternatively by email at abigail.spruce@btguk.com *Michaela S Daly*, Joint Liquidator

29 March 2022

Ag OH123267

(4035393)

#### LTR RECOVERY SERVICES LIMITED

(Company Number 07448855)

Registered office: Unit 1-2 Florida Close, Hot Lane Industrial Estate, Stoke on Trent, ST6 2DJ

Principal trading address: N/A

The Company was placed into members' voluntary liquidation on 23 March 2022 when *Michaela Daly* (IP No. 17570) of Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF and *Louise Longley* (IP No. 23874) of Begbies Traynor (Central) LLP, Fourth Floor, Toronto Square, Toronto Street, Leeds, LS1 2HJ were appointed as Joint Liquidators of the Company. The Company is able to pay all its known creditors in full.

Notice is hereby given, pursuant to Part 14 of the Insolvency (England and Wales) Rules 2016, that the Joint Liquidators of the Company intend to make a first and final distribution to creditors. Creditors of the Company are required, on or before 27 April 2022, to prove their debts by sending to Michaela Daly of Begbies Traynor (Central) LLP, Unit 15, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF, the Joint Liquidator of the Company, written statements of the amount they claim to be due to them from the Company. They must also, if so requested, provide such further details or produce such documentary or other evidence as may appear to the Joint Liquidators to be necessary. A creditor who has not proved his debt before 27 April 2022, or who increases the claim in his proof after that date, will not be entitled to disturb, by reason that he has not participated in it, the intended distribution or any other distribution declared before his debt is proved.

The Joint Liquidators intend that, after paying or providing for a first and final distribution in respect of the claims of all creditors who have proved their debts by the above date, the funds remaining in the hands of the Joint Liquidators shall be distributed to shareholders absolutely.

Proof of debts must be delivered to the Joint Liquidators and can be sent by post to Begbies Traynor (Central) LLP, Unit 15, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF or alternatively by email to abigail.spruce@btguk.com. Any person who requires further information may contact the Joint Liquidator by telephone on 01782 569510. Alternatively enquiries can be made to Abigail Spruce by telephone on 01782 569510.

Michaela S Daly, Joint Liquidator

29 March 2022

Ag OH123268

(4035536)

## MARLOWS HOME & GARDEN LTD

(Company Number 09470464)

Registered office: Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA

Principal trading address: Hollow Road, Bury St. Edmunds, Suffolk, IP32 7AP

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that further to the appointment of the Joint Liquidators on 4 February 2022, they intend to declare a first and final dividend to creditors of the above company within two months of the last date for proving, specified below.

Notice is hereby given that creditors of the Company are required, on or before 29 April 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Joint Liquidators at Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Creditors who have not yet done so must prove their debts by sending their full names and addresses, particulars of their debts or claims and the names and addresses of their solicitors (if any), to the Joint Liquidators at Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA by no later than 29 April 2022 (the last date for proving).

As the distribution will be a final distribution, it may be made without regard to the claim of any person in respect of a debt not proved.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 4 February 2022

Office Holder Details: Steve Markey (IP No. 14912) and Sean Williams (IP No. 11270) both of Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Manchester M45 7TA

For further details contact: The Joint Liquidators, Tel: 0161 413 0930. Alternative contact: Usman Marghoob

Steve Markey, Joint Liquidator

31 March 2022

Ag OH123382

(4035560)

## **MOCKINGBIRD INDUSTRIES LTD**

(Company Number 07011367)

Registered office: 2 Gallery Court, 1-7 Pilgrimage Street, London, SE1 4LL

Principal trading address: 2 Gallery Court, 1-7 Pilgrimage Street, London, SE1 4LL

Final Date For Submission: 30 April 2022.

Notice is hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the joint liquidators of the Company named above (in members' voluntary liquidation) intend to make final distributions to creditors. Creditors are required to prove their debts on or before the final date for submission specified in this notice by sending full details of their claims to the joint liquidators. Creditors must also, if so requested by the joint liquidators, provide such further details and documentary evidence to support their claims as the joint liquidators deem necessary.

The intended distributions are final distributions and may be made without regard to any claims not proved by the final date for submission specified in this notice. Any creditor who has not proved their debt by that date, or who increases the claim in their proof after that date, will not be entitled to disturb the intended final distributions. The joint liquidators intend that, after paying or providing for final distributions in respect of creditors who have proved their claims, all funds remaining in the joint liquidators' hands following the final distributions to creditors shall be distributed to the shareholders of the Company absolutely.

The directors have made a statutory declaration that the Company is able to pay all known liabilities in full.

Date of Appointment: 28 March 2022

Joint Liquidator's Name and Address: *Michael Solomons* (IP No. 9043) of Moorfields Advisory Limited, 82 St John Street, London, EC1M 4JN. Telephone: 020 7549 8050.

Joint Liquidator's Name and Address: *Andrew Pear* (IP No. 9016) of Moorfields Advisory Limited, 82 St John Street, London, EC1M 4JN. Telephone: 020 7549 8050.

For further information contact Holly Thomas at the offices of Moorfields Advisory Limited on 020 7549 8042, or holly.thomas@bm-advisory.com.

28 March 2022 (4036386)

## MY PROPERTY PARTNERS LIMITED

(Company Number 06515716)

Registered office: No. 1 The Octagon County Hall East, Forum Magnum Square, London SE1 7GN

Principal trading address: No. 1 The Octagon County Hall East, Forum Magnum Square, London SE1 7GN

The Company was placed into Members' Voluntary Liquidation on 22 March 2022 and is able to pay all its known creditors in full. Pursuant to Rule 4.182A of the Insolvency Rules 1986, NOTICE IS HEREBY GIVEN that the Liquidator intends to make a first and final distribution to remaining creditors of the above-named Company and that the last date for proving debts against the above-named Company, which is being voluntarily wound up, is 26 April 2022. Claims must be sent to the undersigned, Jeremy Charles Frost (IP number: 9091) of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU, the Liquidator of the Company.

After 26 April 2022, the Liquidator may make that distribution without regard to the claim of any person in respect of a debt not already proved.

The Liquidator intends that after paying or providing for a final distribution in respect of the claims of all creditors who have proved their debts by the above date, the assets remaining in the hands of the Liquidator shall be distributed to the shareholders absolutely.

For further information contact: Kelly Walford, E-mail: kellyw@frostbr.co.uk, Telephone: 0845 260 0101. (4036470)

#### **OH2M LIMITED**

(Company Number 09810431)

Registered office: Apartment 278 37 Millharbour, London, E14 9DL Principal trading address: Flat 1004 Houblon Apartments, 6 Tyne St, London, E1 7AN

Notice is hereby given that the Creditors of the above named Company are required, on or before 13 May 2022 the last day for proving, to send in their names and addresses and to submit their proof of debt to Richard Gardiner, Liquidator, Thomson Cooper, 3 Castle Court, Carnegie Campus, Dunfermline, Fife, KY11 8PB.

If so required by notice from the Liquidator, creditors must produce any document or other evidence which the Liquidator considers is necessary to substantiate the whole or any part of a claim.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 30 March 2022

Office Holder Details: *Richard Gardiner* (IP No. 9488) of Thomson Cooper, 3 Castle Court, Carnegie Campus, Dunfermline, Fife, KY11 8PB

Further details contact: Derek Simpson, Email: corporate@thomsoncooper.com or telephone 01383 628800.

Richard Gardiner, Liquidator

30 March 2022

Ag OH123357

(4035474)

## PROJECT PLANNING AND FORENSIC CONSULTING LTD

(Company Number 11242909)

Registered office: Collingham House, 10-12 Gladstone Road, Wimbledon, London SW19 1QT

Principal trading address: 73 Bywater Place, London SE16 5NE

Notice is hereby given that creditors of the Company are required, on or before 13 May 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Liquidator at Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP.

If so required by notice from the Liquidator, creditors must produce any document or other evidence which the Liquidator considers is necessary to substantiate the whole or any part of a claim.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 18 March 2022

Office Holder Details: *Matthew Fox* (IP No. 9325) of Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP

For further details contact: Matthew Fox, Email: mfox@beaconllp.com or Tel: 02380 651441. Alternative contact: Joanne Hayward.

Matthew Fox, Liquidator

30 March 2022

Ag OH123280

(4035546)

#### **PSK CREDIT CONSULTANCY LTD**

(Company Number 11962959)

Registered office: Drake House, Gadbrook Park, Northwich, CW9 7RA Principal trading address: 37 Linnet Close, Winsford, Cheshire, CW7 3FA

Nature of Business: Management consultancy activities other than financial management.

Final Date For Submission: 29 April 2022.

Notice is hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the liquidator of the Company named above (in members' voluntary liquidation) intends to make final distributions to creditors. Creditors are required to prove their debts on or before the final date for submission specified in this notice by sending full details of their claims to the liquidator. Creditors must also, if so requested by the liquidator, provide such further details and documentary evidence to support their claims as the liquidator deems necessary.

The intended distributions are final distributions and may be made without regard to any claims not proved by the final date for submission specified in this notice. Any creditor who has not proved their debt by that date, or who increases the claim in their proof after that date, will not be entitled to disturb the intended final distributions. The liquidator intends that, after paying or providing for final distributions in respect of creditors who have proved their claims, all funds remaining in the liquidator's hands following the final distributions to creditors shall be distributed to the shareholders of the Company absolutely.

The directors have made a statutory declaration that the Company is able to pay all their known liabilities in full.

Date of Appointment: 18 March 2022

Liquidator's Name and Address: *Clive Morris* (IP No. 8820) of Marshall Peters, Heskin Hall Farm, Wood Lane, Heskin, Preston, PR7 5PA. Telephone: 01257 452021.

For further information contact Daisy Charnley at the offices of Marshall Peters on 01257 452021, or daisycharnley@marshallpeters.co.uk.

29 March 2022 (4036384)

## REALMCROFT LIMITED

(Company Number 03309576)

Registered office: 49 Mead Road, Bury St Edmunds, Suffolk, IP32 7PF

Principal trading address: 49 Mead Road, Bury St Edmunds, Suffolk, IP32 7PE

The above Company was placed into members' voluntary liquidation on 30 March 2022 when *Gary Rupping* (IP No. 16012) and *Hayley Watson* (IP No. 23450) both of McTear Williams & Wood Limited, Prospect House, Rouen Road, Norwich, NR1 2RE were appointed Joint Liquidators.

The Company is able to pay all its known creditors in full.

Notice is hereby given pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016 that the Joint Liquidators of the Company intend making a first and final distribution to creditors within two months of the last date for proving specified below. Creditors of the Company are required to prove their debts before 20 April 2022 by sending to Hayley Watson of McTear Williams & Wood Limited, Prospect House, Rouen Road, Norwich NR1 1RE written statements of the amount they claim to be due to them from the Company. They must also, if so requested, provide such further details or produce such documentary or other evidence as may appear to the Joint Liquidators to be necessary. A creditor who has not proved his debt before 6 March 2020 or who increases the claim in his proof after that date, will not be entitled to disturb the intended final distribution. The Joint Liquidators may make the intended distribution without regard to the claim of any person in respect of a debt not proved or claim increased by that date.

The Joint Liquidators intend that, after paying or providing for a final distribution in respect of the claims of all creditors who have proved their debts, the funds remaining in the hands of the Joint Liquidators shall be distributed to shareholders absolutely.

Enquiries should be made to McTear Williams & Wood Limited, Prospect House, Rouen Road, Norwich NR1 1RE (Tel: 01603 877540, Fax: 01603 877549, Email: info@mw-w.com).

Hayley Watson, Joint Liquidator

30 March 2022

Ag OH123297

(4035585)

#### **RED SKY BUSINESS CONSULTING LIMITED**

(Company Number 10081702)

Registered office: Rosewood House, Quarry Park Road, Pedmore, Stourbridge, DY8 2RE

Principal trading address: Rosewood House, Quarry Park Road, Pedmore, Stourbridge, DY8 2RE

Notice is hereby given that creditors of the Company are required, on or before 1 May 2022 to prove their debts by delivering their proofs (in the format specified in Rule 14.4 OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Azets, 6th Floor, Bank House, Cherry Street, Birmingham, B2 5AL.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 30 March 2022

Office Holder Details: *Margaret Carter* (IP No. 020730) and *James Martin* (IP No. 008316) both of Azets, 6th Floor, Bank House, Cherry Street, Birmingham, B2 5AL

Further details contact: The Joint Liquidators, Tel: 0121 374 0180. Alternative contact: Sam Forman, Email: sam.forman@azets.co.uk *Margaret Carter,* Joint Liquidator

30 March 2022

Ag OH123262

(4035571)

## **SG & J CONSULTING LIMITED**

(Company Number 06338207)

Registered office: 40 Grange Lane, Wickham, Tyne & Wear, NE16 5AH Principal trading address: 40 Grange Lane, Wickham, Tyne & Wear, NE16 5AH

Notice is hereby given that the Creditors of the above named Company, which was voluntarily wound up on 24 March 2022 are required, on or before 29 April 2022 to send their full names and addresses together with full particulars of their debts or claims to Milner Boardman & Partners, The Old Bank, 187a Ashley Road, Hale, Cheshire, WA15 9SQ and, if so requested by me, to provide such further details or produce such documentary or other evidence as may appear to be necessary, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved. Note: This is a solvent liquidation and it is envisaged that all known creditors will be paid in full.

Date of Appointment: 24 March 2022

Office Holder Details: *Darren Brookes* (IP No. 9297) of Milner Boardman & Partners, 187a Ashley Road, Hale, Cheshire, WA15 9SQ For further details contact Sally Baggs on telephone 0161 927 7788, or by email at sallyb@milnerboardman.co.uk

Darren Brookes, Liquidator

30 March 2022

Ag OH123347

(4035438)

## SHELHAR PROPERTIES LIMITED

(Company Number 04906816)

Registered office: 1 Goodwood Court, Lancaster, LA1 4LY

Principal trading address: 1 Goodwood Court, Lancaster, LA1 4LY

Notice is hereby given that Creditors of the Company are required, on or before 4 May 2022, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Campbell, Crossley & Davis, Ground Floor, Seneca House, Links Point, Amy Johnson Way, Blackpool, Lancashire, FY4 2FF.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Director of the Company has made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 29 March 2022.

Office Holder Details: *Richard Ian Williamson* (IP No. 8013) and *Christopher Brindle* (IP No. 22092) both of Campbell, Crossley & Davis, Ground Floor, Seneca House, Links Point, Amy Johnson Way, Blackpool, Lancashire. FY4 2FF

For further details contact: The Joint Liquidators, Email: r.ianwilliamson@crossleyd.co.uk or chris.brindle@crossleyd.co.uk or on Tel: 01253 349331. Alternative contact: Email: melissa.mcphee@crossleyd.co.uk.

Ian Williamson, Joint Liquidator

29 March 2022

Ag OH123353

(4035441)

## **SOLARA VENTURES LIMITED**

(Company Number 06116800)

Registered office: Currently 1 Marybrook Street, Berkeley, GL13 9AA but shortly changing to c/o Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF Principal trading address: N/A

The Company was placed into members' voluntary liquidation on 29 March 2022 and on the same date, *Julie Anne Palmer* (IP No. 008835) and *Andrew Hook* (IP No. 26150) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF were appointed as Joint Liquidators of the Company.

Notice is hereby given that the Creditors of the Company are required on or before 29 April 2022 to send in their names and addresses with particulars of their Debts or Claims and the names and addresses of their Solicitors (if any) to the undersigned Julie Palmer of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF the Joint Liquidator of the Company and, if so required by notice in writing to prove their Debts or Claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Any person who requires further information may contact the Joint Liquidators by telephone on 01722 435190. Alternatively enquiries can be made to Ryan Cullinane by telephone on 023 8021 9820 or by email at ryan.cullinane@btguk.com.

J A Palmer, Joint Liquidator

30 March 2022

Ag OH123373

(4035491)

## **VALUAD LTD**

(Company Number 06225633)

Registered office: The Innovation Centre, Brunswick Street, Nelson, BB9 0PQ

Principal trading address: The Innovation Centre, Brunswick Street, Nelson, BB9 0PQ

The Company was placed into members' voluntary liquidation on 23 March 2022 and on the same date *Christopher James Lawton* (IP No. 23818) and *Dean Watson* (IP No. 009661) both of Begbies Traynor (Central) LLP, 2-3 Winckley Court, Chapel Street, Preston, PR1 8BU were appointed as Joint Liquidators of the Company.

Notice is hereby given that the Creditors of the above-named Company are required on or before 2 May 2022 to send in their names and addresses, particulars of their debts or claims and the names and addresses of their Solicitors (if any) to the undersigned, Chris Lawton of Begbies Traynor (Central) LLP, 1 Winckley Court, Chapel Street, Preston, PR1 8BU the Joint Liquidator of the Company and, if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Any person who requires further information may contact the Joint Liquidator by telephone on 01772 202000. Alternatively enquiries can be made to Keeley Lord by e-mail at keeley.lord@btguk.com or by telephone on 01772 565 221.

Chris Lawton, Joint Liquidator 30 March 2022

Ag OH123374

(4035414)

#### **VICTORIA BELL PROJECT MANAGEMENT LIMITED**

(Company Number 12172683)

Registered office: Bury Lodge, Bury Road, Stowmarket, Suffolk, IP14 1.IA

Principal trading address: 19A Copenhagen Street, London, N1 0JB Notice is hereby given pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Liquidators intend to declare a first and final distribution to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by sending their full names and addresses, particulars of their debts or claims, and the names and addresses of their solicitors (if any), to the Joint Liquidators at Larken Gowen, King Street House, 15 Upper King Street, Norwich, NR3 1RB by no later than 25 April 2022 (the last date for proving).

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 28 March 2022

Office Holder Details: Andrew Anderson Kelsall (IP No. 009555) and Lee Anthony Green (IP No. 015610) both of Larking Gowen LLP, King Street House, 15 Upper King Street, Norwich, Norfolk, NR3 1RB

For further details contact: Andrew Kelsall, Tel: 01603 624181; Email: Andrew.kelsall@larking-gowen.co.uk, or Alternative contact: Rebecca Bloomfield, Email: rebecca.bloomfield@larking-gowen.co.uk

Andrew Anderson Kelsall, Joint Liquidator

30 March 2022

Ag OH123312 (4035454)

## VIPAMEDIA LIMITED

(Company Number 04663044)

Registered office: 2 Mount Parade, Harrogate, North Yorkshire, HG1 1RX

Principal trading address: 2 Mount Parade, Harrogate, North Yorkshire, HG1 1BX

Notice is hereby given that the creditors of the above named Company, over which I was appointed Joint Liquidator on 17 March 2022 are required, on or before 4 May 2022 to send their full names, their addresses and descriptions, full particulars of their debts or claims and the names and addresses of their solicitors (if any) to the undersigned David Antony Willis of FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18 3TX, the Joint Liquidator of the said Company, and, if so required by notice in writing from the said Joint Liquidator, are, personally or by their solicitors, to come in and prove their debts or claims at such time and place as shall be specified in such notice.

A distribution may be made without regard to the claim of any person in respect of a debt not proved.

The winding up is a members' voluntary winding up and it is anticipated that all debts will be paid.

Date of Appointment: 17 March 2022

Office Holder Details: *David Antony Willis* (IP No. 9180) and *Martyn James Pullin* (IP No. 15530) both of FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees. TS18 3TX

Further details contact: The Joint Liquidators, Tel: 01642 917555. Alternative contact: Kelly Mullen

David Antony Willis, Joint Liquidator

30 March 2022

Ag OH123281 (4035562)

NOTICE IS HEREBY GIVEN that the creditors of the Company are required, on or before Tuesday, 3 May 2022 to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the Insolvency (England and Wales) Rules 2016) to the Liquidator at K S Tan & Co, 79A High Road, Willesden, London NW10 2SU or via email to mail@kstan.co.uk.

If so required by notice from the Liquidator, creditors must produce any document or other evidence which the Liquidator considers is necessary to substantiate the whole or any part of a claim.

Note: The directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 24 March 2022

Liquidator, IP number, firm and address: K S Tan, IP No: 8032 of K S Tan & Co, 79A High Road, Willesden, London NW10 2SU

Office Holder's email address or telephone number Email: mail@kstan.co.uk Tel: 020 7586 1280 (4037809)

#### WILLIAM COLE LIMITED

(Company Number 02965839)

Registered office: MB Insolvency, 11 Roman Way, Berry Hill, Droitwich, WR9 9AJ

NOTICE IS HEREBY GIVEN that the creditors of the above named Company, which was voluntarily wound up on 30 March 2022, are required, on or before 28 April 2022, to send their full names and addresses together with full particulars of their debts or claims to MB Insolvency, 11 Roman Way, Berry Hill, Droitwich, Worcestershire WR9 9AJ and, if so requested by me, to provide such further details or produce such documentary or other evidence as may appear to be necessary, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.

Note: It is anticipated that all known Creditors will be paid in full.

Liquidator: Mark Bowen (IP number 8711) of MB Insolvency , 11 Roman Way,, Berry Hill Droitwich Worcestershire WR9 9AJ.

Date of Appointment: 30 March 2022

For further details contact Sophie Murcott on 01905 776 771 or at sophiemurcott@mb-i.co.uk (4037379)

## YORK ENDOCRINE SERVICES LIMITED

(Company Number 04635337)

Registered office: 11 Clifton Moor Business Village, James Nicolson Link, Clifton Moor, York, YO30 4XG

Principal trading address: Club Chambers, Museum Street, York, YO1 7DN

The Company was placed into members' voluntary liquidation on 29 March 2022 and on the same date, *Michael Jenkins* (IP No. 20114) and *David Adam Broadbent* (IP No. 009458) both of Begbies Traynor (Central) LLP, 11 Clifton Moor Business Village, James Nicolson Link, Clifton Moor, York, YO30 4XG were appointed as Joint Liquidators of the Company.

Notice is hereby given that the Creditors of the Company are required on or before 27 April 2022 to send in their names and addresses, particulars of their debts or claims and the names and addresses of their Solicitors (if any) to the undersigned Michael Jenkins of Begbies Traynor (Central) LLP, 11 Clifton Moor Business Village, James Nicolson Link, Clifton Moor, York, YO30 4XG the Joint Liquidator of the Company and, if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Any person who requires further information may contact the Joint Liquidators by telephone on 01904 479801. Alternatively enquiries can be made to Jo Hudson by email at jo.hudson@btguk.com or by telephone on 01904 479801.

Michael Jenkins, Joint Liquidator

30 March 2022

Ag OH123355 (4035535)

# NOTICE TO CREDITORS TO SUBMIT CLAIMS WARMA HEATING AND PLUMBING SUPPLIES LIMITED

(Company Number 01158765)

Registered office: 79A High Road, Willesden, London NW10 2SU Principal trading address: 308-312 Slade Road, Birmingham B23 7LU

## **RESOLUTION FOR VOLUNTARY WINDING-UP**

# THE INSOLVENCY ACT 1986 STEEDMAN CVC LTD

(Company Number 07800123)

("the Company")

Registered office: 9 Hollands Field, Broadbridge Heath, Horsham, West Sussex, RH12 3HQ

Principal trading address: 9 Hollands Field, Broadbridge Heath, Horsham, West Sussex, RH12 3HQ

Pursuant to Chapter 2 of Part 13 of the Companies Act 2006, the following resolutions were passed by the shareholders of the Company on 22 March 2022 as special and ordinary written resolutions, respectively:

1. That the Company be wound up voluntarily.

2. That Michael James Gregson, Licensed Insolvency Practitioner, of TC Bulley Davey Limited, 4 Cyrus Way, Cygnet Park, Hampton, Peterborough, PE7 8HP, be and is hereby appointed Liquidator for the purposes of Winding Up the Company.

Michael James Gregson (IP No 9339) Liquidator, TC Bulley Davey Limited, 4 Cyrus Way, Cygnet Park, Hampton, Peterborough, PE7 8HP

Contact: Denise Gordon, denisegordon@tc-group.com - Telephone: 01733 569494

Mr John Robert Findlay Steedman

Director (4037817)

## ANJTECH CONSULTANCY LTD

(Company Number 09431942)

Registered office: c/o Neum Insolvency, Suite 9, Amba House, 15 College Road, Harrow, Middlesex, HA1 1BA

Principal trading address: Flat 1, 24 Onyx Crescent, Leicester, LE4 9AJ

At a General Meeting of the above named Company, duly convened, and held at Apartment 1, 24 Onyx Crescent, Leicester, LE4 9AJ on 28 March 2022 the following resolutions were duly passed as Special and Ordinary Resolutions:

"That the Company be wound up voluntarily and that *Umang Patel* (IP No. 18790) of Neum Insolvency, Suite 9, Amba House, 15 College Road, Harrow, Middlesex, HA1 1BA be and he is hereby appointed Liquidator for the purposes of such winding up."

Further details contact: Umang Patel, Email: Umang@neuminsolvency.co.uk or Tel: 020 3411 9598. Alternative contact: Jwalant Patel, Email: jwalant@neuminsolvency.co.uk

Vivek Radhakrishna Bhat, Chair

28 March 2022

Ag OH123358 (4035464)

## ANNIEM MEDIA LIMITED

(Company Number 10324538)

Registered office: C/O Antony Batty & Co, Thames Valley, Innovation Centre, 99 Park Drive, Milton Park, Oxfordshire, OX14 4RY

Principal trading address: 47 Brookdene Avenue, Watford, WD19 4LG Notice is hereby given that the following resolutions were passed on 29 March 2022, as a special resolution and an ordinary resolution respectively:

"That the company be wound up voluntarily and that Lawrence King (IP No. 10452) and Matthew Waghorn (IP No. 9432) both of Antony Batty & Co, Thames Valley, Innovation Centre, 99 Park Drive, Milton Park, OX14 4RY be appointed Joint Liquidators of the Company and that they be authorised to act either jointly or separately."

Further details contact: The Liquidators, Tel: 01235 856321 or Email: contact@kwr.antonybatty.com. Alternative contact: Adam Judd. *Jiten Mistry*, Director

31 March 2022

Ag OH123378 (4035387)

## AQUEL CONSULTING LIMITED

(Company Number 08736879)

Registered office: Stamford House, Northenden Road, Sale, Cheshire M33 2DH Previous registered office Bennett Brooks & Co Limited Suite 345. 50 Eastcastle Street, Fitzrovia W1W 8EA

Notice is hereby given that by written resolution of the members of the above named Company on 30 March 2022.:

The Company be wound up voluntarily.

Steve Kenny (IP no 24030) and Richard Cole (IP no 26070) both of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH be and are hereby appointed Joint Liquidators of the Company to act on a joint and several basis.

Joint Liquidator: Steve Kenny (IP number 24030) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH.

Joint Liquidator: *Richard Cole* (IP number 2 6070) of KBL Advisory Limited, Stamford House, Northenden Road, Sale, Cheshire, M33 2DH

Date of Appointment: 30 March 2022

For further details contact Adam Horan on 0161 637 8100 or at Adam@KBL-Advisory.com

Dean Benson, Director (4037847)

#### **ASPINALL IT CONSULTING LIMITED**

(Company Number 09347908)

Registered office: Crown Chambers, Princes Street, Harrogate, HG1 1NJ

Principal trading address: N/A

At a General Meeting of the above named Company duly convened and held at 4.45 pm on 23 March 2022 at Crown Chambers, Princes Street, Harrogate, HG1 1NJ the following resolutions were passed as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *David Antony Willis* (IP No. 9180) and *Martyn James Pullin* (IP No. 15530) both of FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18 3TX be and are hereby appointed Joint Liquidators for the purpose of the voluntary winding up."

Further details contact: The Joint Liquidators, Tel: 01642 917555. Alternative contact: Kelly Mullen.

David Antony Willis, Joint Liquidator

23 March 2022

Ag OH123351 (4035624)

## **AUTUMNTIGER LIMITED**

(Company Number 04105391)

## **EF PARTICIPANTS LIMITED**

(Company Number 03533476)

Registered office: (Both) Ernst & Young LLP, 1 More London Place, London, SE1 2AF

Principal trading address: (Both) Forum St Pauls, 33 Gutter Lane, London, EC2V 8AS

On 21 March 2022 the following written resolutions were passed by the shareholders of the Companies, as a special resolution and as an ordinary resolution respectively:

"That the Companies be wound up voluntarily and that *Matthew Coomber* (IP No. 24430) and *Samantha Jane Keen* (IP No. 9250) both of Ernst & Young, 1 More London Place, London, SE1 2AF be and they are hereby appointed Joint Liquidators for the purposes of the windings up."

Further details contact: The Joint Liquidators, Tel: 020 7783 0293. Alternative contact: Jack Park.

Stephen Daryl Ozin, Director

30 March 2022

Ag OH123287 (4035390)

## BHP IT LIMITED

(Company Number 07504925)

Previous Name of Company: Wakeco (427) Limited

Registered office: Unit 8 President Buildings, Savile Street East, Sheffield, S4 7UQ

Principal trading address: Unit 8 President Buildings, Savile Street East, Sheffield, S4  $7\mathrm{UQ}$ 

At a General Meeting of the members of the above named Company held on 28/03/2022 the following Resolutions were passed as a special resolution and as an ordinary resolution respectively: "That the Company be wound up voluntarily and that Claire Elizabeth Dowson and Gareth David Rusling both of Begbies Traynor (SY) LLP of 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ be and are hereby appointed as joint liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Joint Liquidator: *Claire Elizabeth Dowson* (IP number 19272) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ.

Joint Liquidator: *Gareth David Rusling* (IP number 9481) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ.

Date of Appointment: 28 March 2022

For further details contact Gemma Anderson on 0114 275 5033 or at sheffield.north@begbies-traynor.com (4037842)

## **BRILLIANT BASICS HOLDINGS LIMITED**

(Company Number 09767352)

Registered office: The Bower 10/11th Floor, 207-211 Old Street, London, EC1V 9NR

The following Written Resolutions were passed pursuant to the provisions of section 288 of the Companies Act 2006 as a Special and Ordinary Resolution respectively:

That the Company be wound up voluntarily and that Ryan Michael Davies of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London, EC1M 7AD, be appointed Liquidator of the Company for the purposes of the voluntary winding-up.

Liquidator: Ryan Davies (IP number 15250) of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London EC1M 7AD.

Date of Appointment: 31 March 2022

For further details contact Tara Wessely on 0207 566 4020 or at twessely@mks.co.uk

Benjamin Wiener, Director / Chair (4037834)

## **BRILLIANT BASICS LTD**

(Company Number 08011232)

Registered office: The Bower 10th/11th Floor, 207-211 Old Street, London EC1V 9NR

The following Written Resolutions were passed pursuant to the provisions of section 288 of the Companies Act 2006 as a Special and Ordinary Resolution respectively:

That the Company be wound up voluntarily and that Ryan Michael Davies of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London, EC1M 7AD, be appointed Liquidator of the Company for the purposes of the voluntary winding-up.

Liquidator: Ryan Davies (IP number 15250) of Moore Kingston Smith & Partners LLP, Devonshire House, 60 Goswell Road, London, EC1M 7AD.

Date of Appointment: 31 March 2022

For further details contact Tara Wessely on 0207 566 4020 or at twessely@mks.co.uk

Benjanmin Wiener, Director / Chair (4036326)

## **C L J CONSULTANCY LIMITED**

(Company Number 11883647)

Registered office: 5 Pauline Crescent, Whitton, Twickenham, TW2 8JD Principal trading address: 5 Pauline Crescent, Whitton, Twickenham, TW2 8JD

At a General Meeting of the members of C L J Consultancy Limited held on 24 March 2022, the following Resolutions were passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Craig Povey* (IP No. 9665) and *Charles Turner* (IP No. 9195) both of Begbies Traynor (Central) LLP, 8th Floor, One Temple Row, Birmingham, B2 5LG be and are hereby appointed as Joint Liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Any person who requires further information may contact Katie Shingler by telephone on 0121 200 8162 or alternatively by email at birmingham@btquk.com.

Clare Theresa Jones, Chair

24 March 2022

Ag OH123323

(4035587)

(4035530)

#### **C4 PROJEX LIMITED**

(Company Number 11314436)

Registered office: Little Bourne, 2 Slate Way, Pitton, Salisbury, SP5 1FF

Principal trading address: Little Bourne, 2 Slate Way, Pitton, Salisbury, SP5 1EE

Notice is hereby given that the following resolutions were passed on 16 March 2022, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that Lauren Rachel Cullen (IP No. 18050) of Cullen & Co UK Limited, 197 Kingston Road, Epsom, Surrey, KT19 0AB be and is hereby appointed Liquidator of the Company for the purposes of such winding up."

Further details contact: Lauren Rachel Cullen, Tel: 0203 8877 200. Alternative contact: Gemma Lawrence.

Wilson Cramer, Director

16 March 2022

Ag OH123251

#### **CALCON RAIL LTD**

(Company Number 06904156)

Registered office: 117 North Road, Crawley, West Sussex, RH10 1SQ Principal trading address: 117 North Road, Crawley, West Sussex, RH10 1SQ

Place of meeting: 117 North Road, Crawley, West Sussex, RH10 1SQ. Date of meeting: 21 March 2022.

Time of meeting: 4:00 pm.

At a general meeting of the members of the above-named Company, duly convened and held at the place, time and on the date specified above, the following resolutions were passed as a special resolution, and an ordinary resolution respectively: that the Company be wound up voluntarily, and the Joint Liquidators be appointed.

Date of Appointment: 21 March 2022

Joint Liquidator's Name and Address: Richard Keley (IP No. 18072) of BM Advisory, Arundel House, 1 Amberley Court, Whitworth Road, Crawley, RH11 7XL. Telephone: 01293 410333.

Joint Liquidator's Name and Address: Andrew Pear (IP No. 9016) of BM Advisory, Arundel House, 1 Amberley Court, Whitworth Road, Crawley, RH11 7XL. Telephone: 01293 410333.

For further information contact Sue Markham at the offices of BM Advisory on 01293 452844, or Sue.Markham@bm-advisory.com.

29 March 2022 (4036370)

## CORFU CONSULTANCY LIMITED

(Company Number 08390402)

Registered office: c/o KRE Corporate Recovery Limited, Unit 8 The Aquarium, 1-7 King Street, Reading, Berks RG1 2AN

Principal trading address: c/o Hurst Hill, Churt Road, Hindhead, Surrey, GU26 6HT (previous Registered Office)

At a General Meeting of the Members of the above named Company, duly convened and held at Hurst Hill, Churt Road, Hindhead, Surrey, GU26 6HT on 29 March 2022 the following resolutions were duly passed as special and ordinary resolutions:

"That the Company be wound up voluntarily and that *Rob Keyes* (IP No. 008841) and *Paul Ellison* (IP No. 7254) both of KRE Corporate Recovery Limited, Unit 8, The Aquarium, 1-7 King Street, Reading, RG1 2AN be and are hereby appointed Joint Liquidators of the Company"

For further details contact: Lee Lloyd, Email: lee.lloyd@krecr.co.uk, Tel: 01189 479090.

Richard Bennett, Chair

29 March 2022

Ag OH123365

(4035583)

## **COTTONBURY PLANNING LIMITED**

(Company Number 11016106)

Registered office: Menta Business Centre, 5 Eastern Way, Bury St Edmunds IP32 7AB

Principal trading address: Field House, Bells Road, Belchamp Walter, Sudbury, Suffolk CO10 7A

At an Extraordinary General Meeting of the members of the above named company, duly convened and held at Eldo House, Kempson Way, Bury St Edmunds IP32 7AR on 24 March 2022 the following Resolutions were passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the company be wound up voluntarily, and that Stephen Mark Rout of Stephen M Rout & Company, Menta Business Centre, 5 Eastern Way, Bury St Edmunds IP32 7AB be and is hereby appointed Liquidator for the purposes of such winding up".

Further details contact S Rout on 01223 329392 or email smrout@aol.com

Andrew Michael Aves

Director (4037808)

#### **CRAY U.K. LIMITED**

(Company Number 03941275)

Previous Name of Company: HACKREMCO (NO.1622) LIMITED

Registered office: Ground Floor, 210 Wharfedale Road, Winnersh Triangle, Berkshire, United Kingdom, RG41 5TP

By written resolution of the sole Member of the above-named company passed on 31 March 2022, the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution:

1 THAT the Company be wound up voluntarily.

Ordinary resolution

2 THAT Laura Waters and Steven Sherry of PricewaterhouseCoopers LLP, 7 More London Riverside, London, SE1 2RT be and are hereby appointed Joint Liquidators of the Company for the purposes of such winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators is to be done by all or any one or more of the persons for the time being holding office.

Liquidator: Laura Waters (IP number PricewaterhouseCoopers LLP, 7 More London Riverside, London, SE1 2RT.

Joint Liquidator: Steven Sherry (IP number 19752) PricewaterhouseCoopers LLP, 7 More London Riverside, London, SE1 2RT.

Date of Appointment: 31 March 2022

For further details contact Anne Louise Mccartney on 07802 659389 or at anne.louise.mccartney@pwc.com

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators' appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators. (4037949)

## **CREATED EDUCATION LIMITED**

(Company Number 10889843)

Registered office: 8 C/O Hansel Henson Ltd, 3rd Floor, 8 Bloomsbury Street, London, WC1B 3SR

Principal trading address: 8 C/O Hansel Henson Ltd, 3rd Floor, 8 Bloomsbury Street, London, WC1B 3SR

Place of meeting: Online. Date of meeting: 23 March 2022.

Time of meeting: 10:00 am.

At an extraordinary general meeting of the members of the abovenamed Company, duly convened and held at the place, time and on the date specified above, the following resolutions were passed as a special resolution, and an ordinary resolution respectively: that the Company be wound up voluntarily, and the Joint Liquidators be appointed.

Date of Appointment: 23 March 2022

Joint Liquidator's Name and Address: Vincent John Green (IP No. 009416) of Crowe U.K. LLP, 4 Mount Ephraim Road, Tunbridge Wells, Kent, TN1 1EE. Telephone: 01892 700200.

Joint Liquidator's Name and Address: Steven Edwards (IP No. 26090) of Crowe U.K. LLP, 4 Mount Ephraim Road, Tunbridge Wells, Kent, TN1 1EE. Telephone: 01892 700200.

For further information contact George Budd at the offices of Crowe U.K. LLP on 01892 700200, or recoverysolutions@crowe.co.uk.

24 March 2022 (4036378)

## **DALMAN (IMPORT & EXPORT) LIMITED**

(Company Number 01169794)

Registered office: Unit 11 Thomas Road, Wooburn Industrial Park, Wooburn Green, High Wycombe, HP10 0PE

Principal trading address: Unit 11 Thomas Road, Wooburn Industrial Park, Wooburn Green, High Wycombe, HP10 0PE

The following written resolutions were duly passed by the members of the Company on 30 March 2022, as special and ordinary resolutions: "That the Company be wound up voluntarily and that Chris Newell (IP No. 13690) and David Meany (IP No. 9453) both of Quantuma Advisory Limited, The Old Town Hall, 71 Christchurch Road,

Ringwood, BH24 1DH be appointed Joint Liquidators of the Company and that they be authorised to act jointly and severally.' Further details contact: Lisa Glock, Email: lisa.glock@quantuma.com,

Tel: 01202 970438. Jessica Ann Nugent, Director

30 March 2022

(4035478) Aa OH123265

#### **DEWDROP CONSULTING LTD**

(Company Number 08893944)

Registered office: 12 High Street, Stonehouse, Gloucestershlre, GL10 2NA

Principal trading address: 12 High Street, Gloucestershire, GL10 2NA

At a General Meeting of the Company, duly convened and held at 12 High Street, Stonehouse, Gloucestershlre, GL10 2NA on 18 March 2022 at 4.00 pm, the following Resolutions were passed as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that Lisa Alford (IP No. 9723) and Chris Parkman (IP No. 9588) both of Purnells, 5a Kernick Industrial Estate, Penryn, Cornwall, TR10 9EP be and are hereby appointed Joint Liquidators for the purpose of such winding up."

For further details contact: The Joint Liquidators, Tel: 01326 340579, Fax: 01326 340461, Email: lisa@purnells.co.uk. helen@purnells.co.uk

Lorna Sambrook, Director

18 March 2022

Ag OH123308 (4035383)

## **EPIRIS CAPITAL PARTNERS LLP**

(Company Number OC412420)

Registered office: 1 More London Place, London SE1 2AF

Principal trading address: Forum St Pauls, 33 Gutter Lane, London, EC2V 8AS

The following written determinations were passed on 21 March 2022, by the members of the Partnership:

"That the Partnership be wound up voluntarily and that Matthew Coomber (IP No. 24430) and Samantha Jane Keen (IP No. 9250) both of Ernst & Young LLP, 1 More London Place, London, SE1 2AF be and they are hereby appointed Joint Liquidators for the purposes of the winding up."

For further details contact: The Joint Liquidators, Tel: 020 7783 0293. Alternative contract: Jack Park.

Stephen Daryl Ozin, Designated Member

29 March 2022

Ag OH123288

(4035364)

## **ESTELIA LIMITED**

(Company Number 09688132)

Registered office: Lismore, The Purlieu, Malvern, WR14 4DJ Principal trading address: Lismore, The Purlieu, Malvern, WR14 4DJ At a General Meeting of the Company duly convened and held at Lismore, The Purlieu, Malvern, WR14 4DJ on 17 March 2022 at 4.00 pm, the following resolutions were passed as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Lisa Alford* (IP No. 9723) and *Chris Parkman* (IP No. 9588) both of Purnells, 5a Kernick Industrial Estate, Penryn, Cornwall, TR10 9EP be and are hereby appointed Joint Liquidators for the purpose of such winding up."

Further details contact: The Liquidators, Email: lisa@purnells.co.uk, Tel: 01326 340579, Fax: 01326 340461.

Caroline Downs, Director

17 March 2022

Ag OH123278 (4035555)

#### FIELDER GAMMON INVESTMENT COMPANY LIMITED

(Company Number 00729537)

Registered office: WSM MBI Coakley LLP, 2nd Floor, Shaw House, 3 Tunsgate, Guildford, Surrey, GU1 3QT

Principal trading address: 8 Knoll Quarry, Godalming, Surrey, GU7 3ES

Notice is hereby given that the following resolutions were passed on 21 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Dermot Coakley* (IP No. 6824) of WSM MBI Coakley LLP, 2nd Floor, Shaw House, 3 Tunsgate, Guildford, Surrey, GU1 3QT is appointed as Liquidator for the purposes of such voluntary winding up."

For further details contact: The Liquidator, Tel: 01483 405160, Email: forum@mbicoakley.co.uk. Alternative contact: Cameron Harrison, Tel: 01483 405160, Email: cameron.harrison@mbicoakley.co.uk.

Kate Gammon, Director

21 March 2022

Ag OH123339 (4035443)

#### FOX AND HOUNDS HOTEL LIMITED

(Company Number 06296746)

Registered office: 138 High Street, Crediton, Devon, EX17 3DX

Principal trading address: Fox & Hounds Hotel, Eggesford, Chulmleigh, EX18 7JZ

Notice is hereby given that the following resolutions were passed on 23 March 2022, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that Steve Henson (IP No. 10652) and Adam Buck (IP No. 23070) both of Castle Hill Insolvency Limited, 1 Battle Road, Heathfield, Newton Abbot, TQ12 6RY be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: The Liquidator, Tel: 01626 510020 or E-mail: adam.buck@castlehillinsolvency.com. Alternative contact: Tina Grimes.

Nicholas Culverhouse, Director

30 March 2022

Ag OH123304

(4035457)

## FRESH FRUIT DISTRIBUTION LIMITED

(Company Number 07209941)

Registered office: 29 High Street, Blue Town, Sheerness, ME12 1RN Principal trading address: The Old Pumping Station, Breille Way, Sheerness, Kent, ME12 1YW

Notice is hereby given that the following resolutions were passed on 23 March 2022, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Laurence Pagden* (IP No. 9055) and *Jonathan David Bass* (IP No. 11790) both of Menzies LLP, Lynton House, 7-12 Tavistock Square, London, WC1H 9LT be appointed Joint Liquidators of the Company for the purposes of winding up the Company and that they be authorised to act either jointly or separately."

Further details contact: The Joint Liquidators, Email: JLe@menzies.co.uk, or Tel: 0207 387 5868. Alternative contact: Jessica Le.

Heather Isobel June Whalley, Director

23 March 2022

Ag OH123364 (4035508)

## **G.S.N PROPERTY DEVELOPMENT LIMITED**

(Company Number 12073368)

Registered office: Pertenhall Road, The Town, Great Staughton, St Neots, PE19 5BE

Principal trading address: Pertenhall Road, The Town, Great Staughton, St Neots, PE19 5BE

At a General Meeting of the members of the above named Company held on 21 March 2022 the following Resolutions were passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that Louise Donna Baxter (IP No. 009123) and Lloyd Biscoe (IP No. 9141) both of Begbies Traynor (Central) LLP, Suite WG3, The Officers' Mess Business Centre, Royston Road, Duxford, Cambridge, CB22 4QH be and are hereby appointed as joint liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Any person who requires further information may contact Carol Wilson on 01223 495660 or alternatively by e-mail at carol.wilson@btguk.com Stephen Homer, Chair

21 March 2022

Ag OH123348

(4035417)

# RESOLUTIONS GATEWAY ACCOUNTING LIMITED

(Company Number 11328283)

Registered office: 2 Chestnut Close, Orpington, BR6 6LP

Principal trading address: 2 Chestnut Close, Orpington, BR6 6LP

At a general meeting of the members of the above named company, duly convened and held at 2 Chestnut Close, Orpington, BR6 6LP on 25 March 2022 the following special resolution was duly passed:

'That the Company be wound-up voluntarily and that Richard Frank Simms of F A Simms & Partners Limited, Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB is hereby appointed liquidator for the purpose of winding up'

Further Details: Michelle Collier, Tel No 01455 555 444

Name of Office Holder: Richard Frank Simms

Office Holder: 9252

Address of Office Holder: Alma Park, Woodway Lane, Claybrooke

Parva, Lutterworth, Leicestershire, LE17 5FB

Capacity: Liquidator

Rajesh Kumar (4037807)

## GEOFFREY BLUMENTHAL LIMITED

(Company Number 04233950)

Registered office: 22 Ensign Business Centre Westwood Way, Westwood Business Park, Coventry, CV4 8JA

Principal trading address: 22 Ensign Business Centre Westwood Way, Westwood Business Park, Coventry, CV4 8JA

On 24 March 2022 on the following written resolutions were passed as a Special Resolution and Ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Gavin Bates* (IP No. 8983) of Smart Business Recovery Limited, The Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG be appointed Liquidator of the Company."

Further details contact Sharon Hill on telephone 0116 2325117 or by email at sharon@smartbusinessrecovery.co.uk

H J Blumenthal, Director

24 March 2022

Ag OH123338

(4035405)

## GOLD GROUP PROPERTIES LIMITED

(Company Number 06396536)

Registered office: Gold Group House, Godstone Road, Whyteleafe, CR3.0GG

Principal trading address: Gold Group House, Godstone Road, Whyteleafe, CR3 0GG

At a General Meeting of the members of the above named company duly convened and held at Gold Group House, Godstone Road, Whyteleafe, CR3 0GG on 22 March 2022 at 3.15 pm, the following resolutions were duly passed as special and ordinary resolutions:

"That the Company be wound up voluntarily and that Simon Bonney (IP No. 9379) and Michael Kiely (IP No. 9617) both of Quantuma Advisory Limited, High Holborn House, 52-54 High Holborn, London, WC1V 6RL be and are hereby appointed Joint Liquidators of the Company and that they be authorised to act jointly and severally."

For further details contact: James Varney, Email: James.Varney@quantuma.com, Tel: 02038 566 725.

John Boyle, Chair 22 March 2022 Ag OH123369

(4035559)

## HAWORTH GROUP HOLDINGS (UK) LIMITED

(Company Number 07838188)

Registered office: 4 Ground Floor, More London Riverside, London, England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside, London, England SE1 2AU

## HAWORTH GROUP LIMITED

(Company Number 02262052)

Registered office: 4 Ground Floor, More London Riverside, London, England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside, London, England SE1 2AU

## **DISASTER RECOVERY SERVICES LIMITED**

(Company Number 02063535)

Previous Name of Company: Flightace Computing Limited (13/01/1987)

Registered office: 4 Ground Floor, More London Riverside, London, England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside, London, England SE1 2AU

## **GLOBAL LOGISTICS WORLDWIDE LIMITED**

(Company Number 03599730)

Previous Name of Company: Monocrest Limited (27/07/1998)
Registered office: 88 Crawford Street, London W1H 2EJ

Principal trading address: 88 Crawford Street, London W1H 2EJ

## SARACEN DATASTORE LIMITED

(Company Number 02652784)

Registered office: 4 Ground Floor, More London Riverside, London, England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside, London, England SE1 2AU

## **RECALL LIMITED**

(Company Number 01331798)

Previous Name of Company: Security Archives Limited (Until 28/04/2000); Security Archives (Holdings) Plc (Until 01/08/1994); Seacave Limited (Until 09/11/1984)

Registered office: 4 Ground Floor, More London Riverside, London, England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside, London, England SE1 2AU

## RECALL EUROPE LIMITED

(Company Number 01136000)

Previous Name of Company: Recall Total Information Management Limited (Until 23/07/2013); Brambles Records Management Limited (Until 09/04/1999); Security Archives Limited (Until 26/07/1994); Central Civil and Commercial Investments Limited (Until 31/12/1976) Registered office: 4 Ground Floor, More London Riverside, London, England SE1 2AU

Principal trading address: 4 Ground Floor, More London Riverside, London, England SE1 2AU  $\,$ 

Pursuant to chapter 2 of part 13 of the Companies Act 2006, the following written resolutions were passed on 23 March 2022 Special resolution

That the Companies be wound up voluntarily in accordance with Chapter III of Part IV of the Insolvency Act 1986.

Ordinary resolution

That Nicholas James Timpson and Blair Nimmo of Interpath Ltd, 10 Fleet Place, London EC4M 7RB, United Kingdom, be and are hereby appointed as Joint Liquidators of the Companies and that any act required or authorised under any enactment to be done by the Liquidator is to be done by them jointly or by any one of them.

Joint Liquidator: *Nicholas James Timpson* (IP number 20610) of Interpath Ltd, 10 Fleet Place, London EC4M 7RB.

Joint Liquidator: Blair Carnegie Nimmo (IP number 8208) of Interpath Ltd, 10 Fleet Place, London EC4M 7RB.

Date of Appointment: 23 March 2022

For further details contact Precious Odiase on +44 (0) 203 989 2848 or at precious.odiase@interpathadvisory.com (4037913)

## **HYBRID GLOBAL LTD**

(Company Number 11148283)

Registered office: Telecom House, 125-135 Preston Road, Brighton, BN1 6AF

Principal trading address: N/A

The following written resolutions of the sole member of the above named Company were passed on 28 March 2022 as Special and Ordinary Resolutions:

"That pursuant to section 84(1)(b) of the Insolvency Act 1986 the Company be wound up voluntarily and that pursuant to sections 84(1) and 91 of the Insolvency Act 1986 *Donald McNaught* (IP No. 9359) of Johnston Carmichael LLP, 227 West George Street, Glasgow, G2 2ND be appointed Liquidator of the Company for the purposes of winding up the Company's affairs and distributing its assets."

Further details contact: Sarah Houston, Email: sarah.houston@icca.co.uk

Saima Sultan, Sole shareholder

30 March 2022

Ag OH123393 (4035433)

## **INGENIUM CS LTD**

(Company Number 08573721)

Registered office: First Floor, Telecom House, 125-135 Preston Road, Brighton, BN1 6AF

Principal trading address: N/A

Special and Ordinary Resolutions of Ingenium CS Ltd ("the Company") were passed on 28 March 2022, by Written Resolution of the sole member of the Company:

"That pursuant to Section 84(1)(b) of the INSOLVENCY ACT 1986 the Company be wound up voluntarily. That pursuant to sections 84(1) and 91 of the Insolvency Act 1986 *Donald McNaught* (IP No. 9359) of Johnston Carmichael LLP, 227 West George Street, Glasgow, G2 2ND be appointed Liquidator of the Company for the purposes of winding up the Company's affairs and distributing its assets."

Further details contact: Angela Hindmarch, Email: Angela.Hindmarch@jcca.co.uk.

Stuart Haigh, Shareholder

28 March 2022

Ag OH123389 (4035557)

## J READHEAD SOLUTIONS LTD

(Company Number 11079915)

Registered office: Flat 2 Curzon House, 3 St. Peters Street, Ipswich, Suffolk, IP1 1XF

Principal trading address: Flat 2 Curzon House, 3 St. Peters Street, Ipswich, Suffolk, IP1  $1\mbox{XF}$ 

Notice is hereby given that the following resolutions were passed on 28 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Andrew Anderson Kelsall* (IP No. 009555) and *Lee Anthony Green* (IP No. 015610) both of Larking Gowen LLP, King Street House, 15 Upper King Street, Norwich, Norfolk, NR3 1RB be appointed Joint Liquidators of the Company and that they be authorised to act either jointly or separately."

For further details contact: Lee Green, Email: lee.green@larking-gowen.co.uk, Tel: 01603 624181. or Alternative contact: Rebecca Bloomfield, Email: rebecca.bloomfield@larking-gowen.co.uk

Jessica Redhead, Director

30 March 2022

Ag OH123302 (4035429)

## **JES HOTELS LTD**

(Company Number 12138027)

Registered office: 17 Moor Park Avenue, Preston, PR1 6AS

Principal trading address: 61 South Promenade, Lytham St Annes, FY8 1LZ

Notice is hereby given that the following resolutions were passed on 24 March 2022, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG be appointed Joint Liquidators for the purposes of the voluntary winding up of the Company."

For further details contact: The Joint Liquidators, Email myl@clarkebell.com. Alternative contact: Thomas Bailey

Roger Paul Banks, Director

24 March 2022

Ag OH123391

(4035476)

#### **KMJ ASSOCIATES LTD**

(Company Number 07409030)

Registered office: First Floor, Telecom House, 125-135 Preston Road, Brighton BN1 6AF

Principal trading address: 1 Coles Farm, Chequers Lane, Watford, WD25 0GL

Notice is hereby given that the following resolutions were passed on 21 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Matthew Fox* (IP No. 9325) of Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP be and is hereby appointed as Liquidator of the company for the purposes of such voluntary winding up."

For further details contact: Matthew Fox, Email: mfox@beaconllp.com or Tel: 02380 651441. Alternative contact: Joanne Hayward.

Neil Michael Rigby, Director

30 March 2022

Ag OH123307

(4035361)

## LA BAITA LIMITED

(Company Number 04595720)

Registered office: 2 Clarendon Road, Ashford, TW15 2QE

Principal trading address: Bandstand Cafe, Windmill Drive, Clapham Common, London, SW4 9DE

Notice is hereby given that the following resolutions were passed on 18 March 2022, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG be appointed Joint Liquidators for the purposes of the voluntary winding up of the Company."

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Elizabeth Scott

Lucio Campo, Director

18 March 2022

Ag OH123375

(4035538)

## LANCASHIRE TEXTILE MANUFACTURERS ASSOCIATION

Registered office: N/A

Principal trading address: 4 St Andrew's Street, Blackburn, BB1 8AE The following written resolutions were duly passed on 30 March 2022 as a special and an ordinary resolution by the members of the association:

"That the Association be wound up voluntarily and that *Charles MacMillan* (IP No. 6000) of Beever and Struthers, St George's House, 215-219 Chester Road, Manchester, M15 4JE be hereby appointed Liquidator of the Association."

For further details contact: Neil Clarke, Email: insolvency@beeverstruthers.co.uk, Tel: 0161 832 4901.

John Morrison, Director

30 March 2022

Ag OH123266 (4035549)

## LANGLAND INVESTMENTS LIMITED

(Company Number 00565649)

Registered office: 2-8 Parkstone Road Towngate House, Poole, BH15 2PW

Principal trading address: c/o Ellis Jones Solicitors, 14a Haven Road, Poole, Dorset, BH13 7LP

Notice is hereby given that pursuant to Chapter 2 of Part 13 of the Companies Act 2006, the following resolutions were passed by the sole member on 16 March 2022 as a special resolution, and an ordinary resolution respectively: that the Company be wound up voluntarily, and the Joint Liquidators specified below be appointed Joint Liquidators of the Company for the purposes of the voluntary winding up.

Timothy Sharpley, Director

Date of Appointment: 16 March 2022

Joint Liquidator's Name and Address: *Lucinda Clare Coleman* (IP No. 10710) of Francis Clark LLP, Centenary House, Peninsula Park, Rydon Lane, Exeter, EX2 7XE. Telephone: 01392 667000.

Joint Liquidator's Name and Address: Stephen James Hobson (IP No. 006473) of Francis Clark LLP, Centenary House, Peninsula Park, Rydon Lane, Exeter, EX2 7XE. Telephone: 01392 667000.

For further information contact George Bird at the offices of Francis Clark LLP on 01392 667000, or George.Bird@pkf-francisclark.co.uk. 28 March 2022 (4036366)

#### **LONDON VET IMAGING LTD**

(Company Number 10164271)

Registered office: Plaza 9 KD Tower, Cotterells, Hemel Hempstead, Hertfordshire, HP1 1FW

Principal trading address: (Former) Plaza 9 KD Tower, Cotterells, Hemel Hempstead, Hertfordshire, HP1 1FW

At a General Meeting of the members of the above named company, duly convened and held at Plaza 9 KD Tower, Cotterells, Hemel Hempstead, Hertfordshire, HP1 1FW on 25 March 2022, the following resolutions were duly passed, as a special resolution and as an ordinary resolution:

"That the Company be wound up voluntarily and that *Richard Hunt* (IP No. 21772) of SFP Restructuring Limited, 9 Ensign House, Admirals Way, Marsh Wall, London E14 9XQ be, and he is hereby, appointed as Liquidator for the purpose of the voluntary winding up."

For further details contact: Richard Hunt or Grace Burton, Tel: 020 7538 2222.

Roberto Tapia, Director

30 March 2022

Ag OH123240

(4035544)

## LOST LAMBS PRODUCTIONS UK LIMITED

(Company Number 03667013)

Registered office: 12th Floor, Brunel Building, 2 Canalside Walk, London, W2 1DG

Principal trading address: 12th Floor, Brunel Building, 2 Canalside Walk, London, W2 1DG

Notice is hereby given that pursuant to Chapter 2 of Part 13 of the Companies Act 2006, the following resolutions were passed by the sole member on 24 March 2022 as a special resolution, and an ordinary resolution respectively: that the Company be wound up voluntarily, and the Joint Liquidators specified below be appointed Joint Liquidators of the Company for the purposes of the voluntary winding up.

Darren Nigel Hopgood, Director

Date of Appointment: 24 March 2022

Joint Liquidator's Name and Address: *Thomas Edward Guthrie* (IP No. 15012) of BRI Business Recovery and Insolvency, 2nd Floor, Elm House, Woodlands Business Park, Linford Wood West, Milton Keynes, MK14 6FG. Telephone: 01908 317387.

Joint Liquidator's Name and Address: Sukhvinder Kaur Bains (IP No. 9990) of BRI Business Recovery and Insolvency, 2nd Floor, Elm House, Woodlands Business Park, Linford Wood West, Milton Keynes, MK14 6FG. Telephone: 01908 317387.

For further information contact Kathryn Harper at the offices of BRI Business Recovery and Insolvency on 01908 771353, or kharper@briuk.co.uk.

29 March 2022 (4036392)

## LTR RECOVERY SERVICES LIMITED

(Company Number 07448855)

Registered office: Unit 1-2 Florida Close, Hot Lane Industrial Estate, Stoke on Trent, ST6 2DJ

Principal trading address: N/A

At a General Meeting of the members of LTR Recovery Services Limited held on 23 March 2022 the following Resolutions were passed as a special resolution and as an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Michaela Daly* (IP No. 17570) of Begbies Traynor (Central) LLP, Lymedale Business Centre, Lymedale Business Park, Hooters Hall Road, Newcastle, Staffordshire, ST5 9QF and *Louise Longley* (IP No. 23874) of Begbies Traynor (Central) LLP, Fourth Floor, Toronto Square, Toronto Street, Leeds, LS1 2HJ be and are hereby appointed as Joint Liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by then jointly or by each of them alone."

Any person who requires further information may contact Abigail Spruce by telephone on 01782 569510 or alternatively by email at abigail.spruce@btquk.com

Lawrence Albert Colton, Chair

23 March 2022

Ag OH123267

(4035373)

## **MOCKINGBIRD INDUSTRIES LTD**

(Company Number 07011367)

Registered office: 2 Gallery Court, 1-7 Pilgrimage Street, London, SE1 4I I

Principal trading address: 2 Gallery Court, 1-7 Pilgrimage Street, London SF1 4LL

Notice is hereby given that pursuant to Chapter 2 of Part 13 of the Companies Act 2006, the following resolutions were passed by the sole member on 28 March 2022 as a special resolution, and an ordinary resolution respectively: that the Company be wound up voluntarily, and the Joint Liquidators specified below be appointed Joint Liquidators of the Company for the purposes of the voluntary winding up.

Alexander Hewson, Director

Date of Appointment: 28 March 2022

Joint Liquidator's Name and Address: *Michael Solomons* (IP No. 9043) of Moorfields Advisory Limited, 82 St John Street, London, EC1M 4JN. Telephone: 020 7549 8050.

Joint Liquidator's Name and Address: *Andrew Pear* (IP No. 9016) of Moorfields Advisory Limited, 82 St John Street, London, EC1M 4JN. Telephone: 020 7549 8050.

For further information contact Holly Thomas at the offices of Moorfields Advisory Limited on 020 7549 8042, or holly.thomas@bm-advisory.com.

28 March 2022 (4036388)

## MY PROPERTY PARTNERS LIMITED

(Company Number 06515716)

Registered office: No. 1 The Octagon County Hall East, Forum Magnum Square, London SE1 7GN

Principal trading address: No. 1 The Octagon County Hall East, Forum Magnum Square, London SE1 7GN

At a General Meeting of the Company, duly convened and held at 1 Bow Churchyard, London EC4M 9DQ on 22nd March 2022 at 11:45am the following Resolutions were passed:

"That the Company be wound up voluntarily"

"That Jeremy Charles Frost (IP number: 9091) of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU, be appointed Liquidator of the Company"

Liquidator: Jeremy Charles Frost (IP number 9091) of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU.

Date of Appointment: 22 March 2022

For further details contact Kelly Walford on 0845 260 0101 or at kellyw@frostbr.co.uk (4036468)

#### **OH2M LIMITED**

(Company Number 09810431)

Registered office: Apartment 278 37 Millharbour, London, E14 9DL Principal trading address: Flat 1004 Houblon Apartments, 6 Tyne St, London, E1 7AN

At a General Meeting of the company duly convened and held at 8 Chester Street, Edinburgh, EH3 7RA, on 30 March 2022 at 11.30 am, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Richard Gardiner* (IP No. 9488) of Thomson Cooper, 3 Castle Court, Carnegie Campus, Dunfermline, Fife, KY11 8PB be and is hereby appointed Liquidator for the purpose of such winding up."

Further details contact: Derek Simpson, Email: corporate@thomsoncooper.com or telephone 01383 628800.

Oliver Henry Maxwell Musto, Chair

30 March 2022

Ag OH123357

(4035586)

#### PROJECT PLANNING AND FORENSIC CONSULTING LTD

(Company Number 11242909)

Registered office: Collingham House, 10-12 Gladstone Road, Wimbledon, London SW19 1QT

Principal trading address: 73 Bywater Place, London SE16 5NE

Notice is hereby given that the following resolutions were passed on 18 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Matthew Fox* (IP No. 9325) of Beacon, BizSpace, Steel House, Plot 4300, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7FP be and is hereby appointed as Liquidator of the company for the purposes of such voluntary winding up."

For further details contact: Matthew Fox, Email: mfox@beaconllp.com or Tel: 02380 651441. Alternative contact: Joanne Hayward.

Lizanne Linde, Director

18 March 2022

Ag OH123280

(4035566)

## **PSK CREDIT CONSULTANCY LTD**

(Company Number 11962959)

Registered office: Drake House, Gadbrook Park, Northwich, CW9 7RA Principal trading address: 37 Linnet Close, Winsford, Cheshire, CW7 3FA

Place of meeting: The Offices of Marshall Peters Limited, Heskin Hall Farm, Heskin, Preston, PR7 5PA.

Date of meeting: 18 March 2022.

Time of meeting: 10:00 am.

At a general meeting of the members of the above-named Company, duly convened and held at the place, time and on the date specified above, the following resolutions were passed as a special resolution, and an ordinary resolution respectively: that the Company be wound up voluntarily, and the Liquidator specified below be appointed Liquidator of the Company for the purposes of the voluntary winding up.

Date of Appointment: 18 March 2022

Liquidator's Name and Address: *Clive Morris* (IP No. 8820) of Marshall Peters, Heskin Hall Farm, Wood Lane, Heskin, Preston, PR7 5PA. Telephone: 01257 452021.

For further information contact Daisy Charnley at the offices of Marshall Peters on 01257 452021, or daisycharnley@marshallpeters.co.uk.

29 March 2022 (4036385)

## REALMCROFT LIMITED

(Company Number 03309576)

Registered office: 49 Mead Road, Bury St Edmunds, Suffolk, IP32 7PE

Principal trading address: 49 Mead Road, Bury St Edmunds, Suffolk, IP32 7PF

Notice is hereby given that written resolutions were passed on 30 March 2022 by the members to wind up the above named Company and appoint Joint Liquidators as follows:

"That the Company be wound up voluntarily and that *Gary Rupping* (IP No. 16012) and *Hayley Watson* (IP No. 23450) both of McTear Williams & Wood Limited, Prospect House, Rouen Road, Norwich, NR1 2RE be and are hereby appointed joint liquidators of the Company for the purpose of the voluntary winding up with authority to act either jointly or separately in all matters."

Enquiries should be made to McTear Williams & Wood Limited, Prospect House, Rouen Road, Norwich NR1 1RE (Tel: 01603 877540, Fax: 01603 877549, Email: info@mw-w.com).

Amanda Meredith, Director 30 March 2022

Ag OH123297

(4035352)

## **RED SKY BUSINESS CONSULTING LIMITED**

(Company Number 10081702)

Registered office: Rosewood House, Quarry Park Road, Pedmore, Stourbridge, DY8 2RE

Principal trading address: Rosewood House, Quarry Park Road, Pedmore, Stourbridge, DY8 2RE

Notice is hereby given that the following resolutions were passed on 30 March 2022 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that Margaret Carter (IP No. 020730) and James Martin (IP No. 008316) both of Azets, 6th Floor, Bank House, Cherry Street, Birmingham, B2 5AL be appointed as Joint Liquidators for the purposes of such voluntary winding up." Further details contact: The Joint Liquidators, Tel: 0121 374 0180. Alternative contact: Sam Forman, Email: sam.forman@azets.co.uk Caroline Ann Cronin-Brunt, Director

30 March 2022

Ag OH123262 (4035421)

#### **SG & J CONSULTING LIMITED**

(Company Number 06338207)

Registered office: 40 Grange Lane, Wickham, Tyne & Wear, NE16 5AH Principal trading address: 40 Grange Lane, Wickham, Tyne & Wear, NE16 5AH

Written Resolutions of the above-named Company were duly passed on 24 March 2022 as a Special resolution and Ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Darren Brookes* (IP No. 9297) of Milner Boardman & Partners, 187a Ashley Road, Hale, Cheshire, WA15 9SQ be appointed Liquidator of the Company."

For further details contact Sally Baggs on telephone 0161 927 7788, or by email at sallyb@milnerboardman.co.uk

Stephen Gary Falkous, Director

24 March 2022

Ag OH123347 (4035556)

## SHELHAR PROPERTIES LIMITED

(Company Number 04906816)

Registered office: 1 Goodwood Court, Lancaster, LA1 4LY Principal trading address: 1 Goodwood Court, Lancaster, LA1 4LY Notice is hereby given that the following resolutions were passed on 29 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that the Joint Liquidators be authorised to distribute to the member of the company in specie part or the whole of the assets of the company and may for that purpose, value any assets and that *Richard Ian Williamson* (IP No. 8013) and *Christopher Brindle* (IP No. 22092) both of Campbell, Crossley & Davis, Ground Floor, Seneca House, Links Point, Amy Johnson Way, Blackpool, Lancashire, FY4 2FF be appointed as Joint Liquidators for the purposes of such voluntary winding up."

For further details contact: The Joint Liquidators, Email: r.ianwilliamson@crossleyd.co.uk or chris.brindle@crossleyd.co.uk or on Tel: 01253 349331. Alternative contact: Email: melissa.mcphee@crossleyd.co.uk.

Sharon Elizabeth Price, Director

29 March 2022

Ag OH123353 (4035553)

## **SOLARA VENTURES LIMITED**

(Company Number 06116800)

Registered office: 1 Marybrook Street, Berkeley, GL13 9AA

Principal trading address: N/A

At a general meeting of the members of Solara Ventures Limited, held on 29 March 2022 the following resolutions were passed as a special resolution and as an ordinary resolution respectively:

"That the Company be wound up voluntarily and that Julie Anne Palmer (IP No. 008835) and Andrew Hook (IP No. 26150) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF be and are hereby appointed as Joint Liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Any person who requires further information may contact Ryan Cullinane by telephone on 023 8021 9820 or by email at ryan.cullinane@btguk.com.

Nicholas James Martlew, Director

29 March 2022

Ag OH123373 (4035436)

#### VALUAD LTD

(Company Number 06225633)

Registered office: The Innovation Centre, Brunswick Street, Nelson, BB9 0PO

Principal trading address: The Innovation Centre, Brunswick Street, Nelson, BB9 0PQ  $\,$ 

Notice is hereby given that a General Meeting of the shareholders of the Company will be held at The Innovation Centre, Brunswick Street, 889 0PQ on 23 March 2022, at 3.45 pm for the purpose of considering and, if thought fit, passing the following resolutions as a special resolution and as an ordinary resolution:

"That the Company be wound up voluntarily and that *Christopher James Lawton* (IP No. 23818) and *Dean Watson* (IP No. 009661) both of Begbies Traynor (Central) LLP, 2-3 Winckley Court, Chapel Street, Preston, PR1 8BU be and are hereby appointed as joint liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Any person who requires further information may contact the Joint Liquidator by telephone on 01772 202000. Alternatively enquiries can be made to Keeley Lord by e-mail at keeley.lord@btguk.com or by telephone on 01772 565 221.

Lianne Colclough, Director

23 March 2022

Ag OH123374 (4035542)

## VICTORIA BELL PROJECT MANAGEMENT LIMITED

(Company Number 12172683)

Registered office: Bury Lodge, Bury Road, Stowmarket, Suffolk, IP14 1JA

Principal trading address: 19A Copenhagen Street, London, N1 0JB Notice is hereby given that the following resolutions were passed on 28 March 2022, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Andrew Anderson Kelsall* (IP No. 009555) and *Lee Anthony Green* (IP No. 015610) both of Larking Gowen LLP, King Street House, 15 Upper King Street, Norwich, Norfolk, NR3 1RB be appointed Joint Liquidators of the Company and that they be authorised to act either jointly or separately."

For further details contact: Andrew Kelsall, Tel: 01603 624181; Email: Andrew.kelsall@larking-gowen.co.uk, or Alternative contact: Rebecca Bloomfield, Email: rebecca.bloomfield@larking-gowen.co.uk

Victoria Bell, Director

30 March 2022

Ag OH123312 (4035395)

## **VIMIAC LIMITED**

(Company Number 10919125)

Registered office: 71 - 75 Shelton Street, Covent Garden, London,

WC2H 9JQ

Principal trading address: N/A

The following written resolutions of the sole member of the above named Company were passed on 28 March 2022 as Special and Ordinary Resolutions:

"That pursuant to section 84(1)(b) of the Insolvency Act 1986 the Company be wound up voluntarily and that pursuant to sections 84(1) and 91 of the Insolvency Act 1986 *Donald McNaught* (IP No. 9359) of Johnston Carmichael LLP, 227 West George Street, Glasgow, G2 2ND be appointed Liquidator of the Company for the purposes of winding up the Company's affairs and distributing its assets."

Further details contact: Angela Hindmarch, Email: angela.hindmarch@jcca.co.uk

David Wooldridge, Sole shareholder

30 March 2022

Ag OH123390 (4035442)

## **VIPAMEDIA LIMITED**

(Company Number 04663044)

Registered office: 2 Mount Parade, Harrogate, North Yorkshire, HG1 1RX

Principal trading address: 2 Mount Parade, Harrogate, North Yorkshire, HG1 1BX

At a General Meeting of the above named Company duly convened and held at 2.15 pm on 17 March 2022 at 2 Mount Parade, Harrogate, North Yorkshire, HG1 1BX the following resolutions were passed as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *David Antony Willis* (IP No. 9180) and *Martyn James Pullin* (IP No. 15530) both of FRP Advisory Trading Limited, 1st Floor, 34 Falcon Court, Preston Farm Business Park, Stockton on Tees, TS18 3TX be and are hereby appointed Joint Liquidators for the purpose of the voluntary winding up."

Further details contact: The Joint Liquidators, Tel: 01642 917555. Alternative contact: Kelly Mullen

David Antony Willis, Joint Liquidator

17 March 2022

Ag OH123281 (4035516)

# NOTICE OF RESOLUTION WARMA HEATING AND PLUMBING SUPPLIES LIMITED

(Company Number 01158765)

Registered office: 79A High Road, Willesden, London NW10 2SU Principal trading address: 308-312 Slade Road, Birmingham B23 7LU At a general meeting of the Company, duly convened and held at Unit 3A Wing Yip Centre, 278 Thimble Mill Lane, Nechells, Birmingham, West Midlands B7 5HD on Thursday 24 March 2022, the following Resolutions were passed as a Special Resolution and an Ordinary Resolution respectively: "That the Company be wound up voluntarily"; and "That K S Tan of K S Tan & Co, 79A High Road, Willesden, London NW10 2SU, be and is hereby appointed Liquidator of the Company for the purposes of such winding up."

Date on which Resolutions were passed: 24th March 2022

Liquidator, IP number, firm and address K S Tan, IP No: 8032, of K S Tan & Co, 79A High Road, Willesden, London NW10 2SU

Office Holder's email address or telephone number Email: mail@kstan.co.uk Tel: 020 7586 1280

Susan Boys, Director (4037810)

## **WILLIAM COLE LIMITED**

(Company Number 02965839)

Registered office: MB Insolvency, 11 Roman Way, Berry Hill, Droitwich, WR9 9AJ

At a meeting of the above named company duly convened and held on 30 March 2022, the following resolutions were passed:

That the Company be wound up voluntarily.

That Mark Bowen be appointed as Liquidator for the purposes of such winding up.

Liquidator: Mark Bowen (IP number 8711) of MB Insolvency , 11 Roman Way, Berry Hill, Droitwich Worcestershire WR9 9AJ.

Date of Appointment: 30 March 2022

For further details contact Sophie Murcott on 01905 776 771 or at sophiemurcott@mb-i.co.uk

William Cole, Director (4037377)

## YORK ENDOCRINE SERVICES LIMITED

(Company Number 04635337)

Registered office: 11 Clifton Moor Business Village, James Nicolson

Link, Clifton Moor, York, YO30 4XG

Principal trading address: Club Chambers, Museum Street, York, YO1

At a General Meeting of the members of the above named Company held on 29 March 2022, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily, and that *Michael Jenkins* (IP No. 20114) and *David Adam Broadbent* (IP No. 009458) both of Begbies Traynor (Central) LLP, 11 Clifton Moor Business Village, James Nicolson Link, Clifton Moor, York, YO30 4XG be and are hereby appointed as joint liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Any person who requires further information may contact the Joint Liquidators by telephone on 01904 479801. Alternatively enquiries can be made to Jo Hudson by email at jo.hudson@btguk.com or by telephone on 01904 479801.

Paul Edward Jennings, Chair

29 March 2022 Ag OH123355

H123355 (4035480)

# **Partnerships**

## **DISSOLUTION OF PARTNERSHIP**

## MANSFORD HELIOS LIMITED PARTNERSHIP

LP013835

Limited Partnerships Act 1907

The Partnership was dissolved with effect from 30 March 2022.

Mansford Helios General Partner Limited as the general partner of the Partnership

For and on behalf of

MANSFORD HELIOS GENERAL PARTNER LIMITED

30 March 2022

Ag OH123325 (4035337)

## TRANSFER OF INTEREST

# STATEMENT BY GENERAL PARTNER INNISFREE PFI CONTINUATION FUND

(Registered No. LP10938)

Notice is hereby given, pursuant to section 10 of the Limited Partnership Act 1907, of an arrangement under which 21.38% of the share of Caisse de dépôt et placement du Québec as a limited partner of Innisfree PFI Continuation Fund (a limited partnership registered in England with registered number LP10938) has been assigned to Falkirk Council (as administering authority of Falkirk Council Pension Fund), 29.15% of the share of Caisse de dépôt et placement du Québec has been assigned to Fife Council (as administering authority of Fife Pension Fund), 46.65% of the share of Caisse de dépôt et placement du Québec has been assigned to City of Edinburgh Council (as administering authority of Lothian Pension Fund) and 2.82% of the share of Caisse de dépôt et placement du Québec has been assigned to Lincolnshire Council with effect from the date of the publication of this notice.

Signed for and on behalf of Innisfree Continuation

Partners LLP as General Partner of Innisfree

PFI Continuation Fund acting by David Burton

being a director of one of its designated members

Date: 29 March 2022 (4036507)

(4034044)

# **PEOPLE**

# Personal insolvency

## **APPOINTMENT AND RELEASE OF TRUSTEES**

**RULE 10.74(1)** 

NOTICE OF APPOINTMENT OF TRUSTEE

In the High Court of Justice

No 102 of 2020

MOHAMMED KAMAL AHMED

(in bankruptcy)

197 Plashnet Grove, London, E6 IBX

Birth details: 28/12/1975

Bankrupt's occupation: Systems Analyst

Notice is hereby given, pursuant to Rule 10.74 (1) of the Insolvency Act (England and Wales) 2016 that Chris Parkman has been appointed as Trustee of the above named by a decision of creditors, with effect from 14/03/2022.

Appointment Date: 14/03/2022

Chris Parkman BSC HONS, FABRP, MIPA, FCCA (IP Number: 9500) of Purnells, 5a Kernick Industrial Est, Penryn, Cornwall TR10 9EP, was appointed as Trustee on 14/03/2022.

For further information please contact chris@purnells.co.uk or alan@purnells.co.uk.

Chris Parkman, BSC HONS, FABRP, MIPA, FCCA, Trustee, Purnells, 5a Kernick Industrial Est, Penryn, Cornwall TR10 9EP. Tel: 01326 340579; Email: chris@purnells.co.uk; Website: www.purnells.co.uk Date: 28 March 2022 (4036549)

### **BROWN, BETHANY ANNE**

19 Holden Crescent, Walsall, WS3 1PY Birth details: 27 November 1993

Bethany Anne Brown, Currently not working, of 19 Holden Crescent, Walsall, West Midlands, WS3 1PY, formerly of 25 Stanley Street,

Walsall, West Midlands, WS3 3EL In the Office of the Adjudicator

No 5135214 of 2022

Date of Filing Petition: 26 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 10:00

Whether Debtor's or Creditor's PetitionDebtor's

K Read 3rd Floor, Cannon House, 18 The Priory Queensway,

Birmingham, B4 6FD, telephone: 0300 678 0016

Capacity of office holder(s): Trustee

28 March 2022 (4034056)

## CASS, TRACEY VANESSA

17 Brunel Court, Truro, TR1 3AE Birth details: 5 October 1971

Tracey Vanessa Cass, Employed, of 17 Brunel Court, Truro, Cornwall, TR1 3AE, formerly of 2 Trelander South, Truro, Cornwall, TR1 1PH

In the Office of the Adjudicator

No 5135216 of 2022

Date of Filing Petition: 27 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

M Mace 3rd Floor, Senate Court, Southernhay Gardens, Exeter, EX1

1UG, telephone: 0300 678 0016 Capacity of office holder(s): Trustee

28 March 2022 (4034046)

## **BANKRUPTCY ORDERS**

## AMPHLETT, JULIE

17A St. Pauls Road, Newcastle, ST5 2PQ

Birth details: 12 April 1965

Julie Amphlett, Employed, of 17A St. Pauls Road, Newcastle, Staffordshire, ST5 2PQ, formerly of 29 Rosendale Avenue, Newcastle-under-Lyme, Staffordshire, ST5 7NG and formerly of 14 Pennine Way,

Newcastle-under-Lyme, Staffordshire, ST5 6JA

In the Office of the Adjudicator No 5135196 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022

Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's

L Cook 11th Floor, Southern House, Wellesley Grove, Croydon, CR0

1XN, telephone: 03030031736 Capacity of office holder(s): Trustee

28 March 2022 (4034050)

## **BICKNELL, LESLEY JANE**

16 Willders Garth, Holbeach, Spalding, PE12 7RB

Birth details: 23 July 1970

Lesley Jane Bicknell, Currently not working, of 16 Willders Garth, Holbeach, Spalding, Lincolnshire, PE12 7RB, formerly of 37 Norburn,

Peterborough, PE3 8NS In the Office of the Adjudicator

No 5135162 of 2022

Date of Filing Petition: 25 March 2022

Bankruptcy order date: 28 March 2022

Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

M Commins Eastbrook, Shaftesbury Road, Cambridge, CB2 8DR,

telephone: 03030031741

Capacity of office holder(s): Trustee

28 March 2022 (4034040)

DOBBINS, TOBIAS NICHOLAS

1 Oak Close, West Parley, Ferndown, BH22 8UA

Birth details: 6 November 1982

Tobias Nicholas Dobbins, Employed, of 1 Oak Close, West Parley, Ferndown, Dorset, BH22 8UA, formerly of 256 Columbia Road,

Bournemouth, Dorset, BH10 4DR In the Office of the Adjudicator

No 5135220 of 2022

Date of Filing Petition: 27 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

J Sullivan 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ,

telephone: 0300 678 0016 Capacity of office holder(s): Trustee

28 March 2022

## DOVER, COLIN

Lightoaks Cottage, Mill Green Road, Mill Green, INGATESTONE, Essex. CM4 0HX

COLIN DOVER occupation unknown, residing at Lightoaks Cottage, Mill Green Road, Mill Green, Ingatestone, Essex, CM4 OHX

In the County Court at Chelmsford

No 38 of 2021

Date of Filing Petition: 31 August 2021 Bankruptcy order date: 15 March 2022 Time of Bankruptcy Order: 10:05

Whether Debtor's or Creditor's PetitionCreditor's

Name and address of petitioner: TOWER TRADING GROUP LIMITED3rd Floor Tower 42, 25 Old Broad Street, LONDON, EC2N 1HO

J Sullivan 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ,

telephone: 0300 678 0016

Capacity of office holder(s): Trustee

15 March 2022 (4034047)

## **DELLAR, SADE ZARA**

35 Heasman Place, Southwater, Southwater, 35 Heasman Place, Horsham, RH13 9FT

Birth details: 21 February 1985

Sade Zara Dellar, Currently not working, of 35 Heasman Place, Southwater, Horsham, West Sussex, RH13 9FT, formerly of 51 Staples Barn, Henfield, West Sussex, BN5 9PP, and formerly of 25 Saltings Way, Upperbeeding, West Sussex, BN44 3JH, and formerly of 14 Highbridge Walk, Aylesbury, Buckinghamshire, HP21 7SE

In the Office of the Adjudicator

No 5135156 of 2022

Date of Filing Petition: 24 March 2022 Bankruptcy order date: 25 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

K Read 3rd Floor, Cannon House, 18 The Priory Queensway,

Birmingham, B4 6FD, telephone: 0300 678 0016

Capacity of office holder(s): Trustee

25 March 2022 (4034048)

83 Maltby Drive, ENFIELD, EN1 4EL Birth details: 22 August 1962

Mr Ian Hackett, OCCUPATION UNKNOWN OF 83 MALTBY

DRIVE,ENFIELD,EN1 4EL

In the County Court at Central London

No 10 of 2022

HACKETT, IAN

Date of Filing Petition: 20 January 2022 Bankruptcy order date: 24 March 2022 Time of Bankruptcy Order: 12:53

Whether Debtor's or Creditor's PetitionCreditor's

Name and address of petitioner: LONDON BOROUGH OF ENFIELDPO BOX 63, CIVIC CENTRE, SILVER STREET, ENFIELD, EN1 3XW

 $\it L\ Cook$  11th Floor, Southern House, Wellesley Grove, Croydon, CR0

1XN, telephone: 03030031736 Capacity of office holder(s): Trustee

24 March 2022 (4034045)

## **EMERY, KYLE LEIGH**

30 Sidebrook Court, Thorplands, Thorplands, Northampton, NN3 8UT Birth details: 13 July 1988

Kyle Leigh Emery, Self Employed, of 30 Sidebrook Court, Thorplands, Northampton, NN3 8UT and carrying on business as Rooted

Landscapes

In the Office of the Adjudicator

No 5135170 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

K Read 3rd Floor, Cannon House, 18 The Priory Queensway,

Birmingham, B4 6FD, telephone: 0300 678 0016

Capacity of office holder(s): Trustee

28 March 2022 (4034036)

#### **HUDAKOVA, MARIA**

121 Knypersley Road, Stoke-On-Trent, ST6 8JA

Birth details: 18 February 1969

Maria Hudakova, A full-time carer, of 121 Knypersley Road, Stoke-on-Trent, Staffordshire, ST6 8JA, formerly of 74 Burton Manor Road,

Stafford, Staffordshire, ST17 9PR In the Office of the Adjudicator

No 5135178 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

K Read Level 1, Apex Court, City Link, NOTTINGHAM, NG2 4LA, telephone: 0300 678 0016. email:

Nottingham.OR@insolvency.gsi.gov.uk Capacity of office holder(s): Trustee

28 March 2022 (4034054)

## FRANCO, EDUARDO CISNEROS

Flat 18, Braithwaite House, Bunhill Row, London, EC1Y 8NE

Birth details: 7 July 1982

Eduardo Cisneros Franco, Employed, of Flat 18, Braithwaite House,

Bunhill Row, London, EC1Y 8NE In the Office of the Adjudicator

No 5135192 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

K Jackson 1st Floor, Spring Place, 105 Commercial Road,

Southampton, SO15 1EG, telephone: 03030031735

Capacity of office holder(s): Trustee

28 March 2022 (4034041)

## **JONES, ROBERT DANIEL**

10 The Terrace, Chawton, ALTON, Hampshire, GU34 1SE

Birth details: 25 April 1980

Robert Daniel Jones, Self Employed, of 10 The Terrace, Chawton, Alton, Hampshire, GU34 1SE, formerly of 95 Hazel Road, Four Marks, Alton, Hampshire, GU34 5EX, and carrying on business as The Jones

And Jones Company

In the Office of the Adjudicator

No 5135174 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

L Cook 11th Floor, Southern House, Wellesley Grove, Croydon, CR0

1XN, telephone: 03030031736 Capacity of office holder(s): Trustee

28 March 2022 (4034038)

## **GOLDA, MARTA MARZENA**

5-7 Commercial Street, Cwmbran, NP44 3LR

Birth details: 26 August 1983

Marta Marzena Golda, a director, of 5-7 Commercial Street, Cwmbran, Torfaen, NP44 3LR, formerly of Greenways, Caerleon road, Llanfrechfa, NP44 8TY, formerly of 30 Chepstow Road, Newport, NP19 8EA, formerly of 53 Pitt's Road, Lincolnshire, LN4 1BH

In the Office of the Adjudicator

No 5135204 of 2022

Date of Filing Petition: 26 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

S Baxter 3rd Floor, Companies House, Crown Way, CARDIFF, CF14

3ZA, telephone: 0300 678 0016

Capacity of office holder(s): Trustee 28 March 2022 (4034042)

## **KOZMA, ZOLTAN**

20 Richmond Road, Mangotsfield, Bristol, BS16 9HB

Birth details: 8 July 1979

Zoltan Kozma, Currently not working, of 20 Richmond Road, Mangotsfield, Bristol, BS16 9HB, formerly of 6 Lyndale Avenue, Bristol, BS9 1BS and formerly of The Clocktower Flat, Marlborough, Wiltshire, SN8 1PN

In the Office of the Adjudicator

No 5135210 of 2022

Date of Filing Petition: 26 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

M Mace 3rd Floor, Senate Court, Southernhay Gardens, Exeter, EX1

1UG, telephone: 0300 678 0016 Capacity of office holder(s): Trustee

28 March 2022 (4034035)

## MCLEISH, KIERAN FRANK

Aristomenous 37, Athina, 104 40 Birth details: 24 January 1996

Kieran Frank Mcleish, Employed, of Aristomenous 37, Athina, 104 40, Greece formerly of Fidiou 2, Athens, 10678, Greece and formerly of 31

Westfield Avenue, Fleetwood, Lancashire, FY7 7NB

In the Office of the Adjudicator

No 5135180 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

C Megram 2nd Floor, Rosebrae Court, Woodside Ferry Approach,

Birkenhead, CH41 6DU, telephone: 03030031738

Capacity of office holder(s): Trustee

28 March 2022 (4034027)

## PRIME, DAVID

113 Howbeck Road, Arnold, Nottingham, NG5 8QB

Birth details: 28 September 1971

David Prime, Employed, of 113 Howbeck Road, Arnold, Nottingham,

Nottinghamshire, NG5 8QB In the Office of the Adjudicator

No 5135194 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

K Read Level 1, Apex Court, City Link, NOTTINGHAM, NG2 4LA, telephone: 0300 678 0016. email:

Nottingham.OR@insolvency.gsi.gov.uk Capacity of office holder(s): Trustee

28 March 2022 (4034029)

### MUKTADIR, ABDUL

72 Milne Street, Chadderton, Oldham, OL9 0HR

Birth details: 5 July 1967

Abdul Muktadir, Currently not working, of 72 Milne Street, Chadderton, Oldham, Greater Manchester, OL9 0HR, formerly of Ground Rear Studio, 24 Springcroft Road, Birmingham, West Midland, P11 2EI

In the Office of the Adjudicator

No 5135190 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

S Baxter 3rd Floor, Companies House, Crown Way, CARDIFF, CF14

3ZA, telephone: 0300 678 0016 Capacity of office holder(s): Trustee

29 March 2022 (4034057)

#### RADIM, KAULSOOM

16 Breamore Court, Breamore Road, Ilford, IG3 9NP

Birth details: 27 May 1975

Kaulsoom Radim, Currently not working, of 16 Breamore Court, Breamore Road, Ilford, IG3 9NP, formerly of 145 Breamore Road, Ilford, IG3 9LU

In the Office of the Adjudicator

No 5135160 of 2022

Date of Filing Petition: 24 March 2022 Bankruptcy order date: 25 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

J Sullivan 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ,

telephone: 0300 678 0016

Capacity of office holder(s): Trustee

25 March 2022 (4034033)

## **OSMAN, TARTEEL**

Flat 4, Treeby Court, 18 George Lovell Drive, ENFIELD, EN3 6WD

Birth details: 30 May 1992

OCCUPATION UNKNOWN OF 4 TREEBY COURT,18 GEORGE

LOVELL DRIVE, ENFIELD, MIDDLESEX, EN3 6WD

In the County Court at Central London

No 231 of 2021

Date of Filing Petition: 23 December 2021 Bankruptcy order date: 24 March 2022 Time of Bankruptcy Order: 11:30

Whether Debtor's or Creditor's PetitionCreditor's

Name and address of petitioner: LONDON BOROUGH OF ENFIELDPO BOX 63, CIVIC CENTRE, SILVER STREET, ENFIELD MIDDLESEX, EN1 3XA

M Commins Eastbrook, Shaftesbury Road, Cambridge, CB2 8DR,

telephone: 03030031741 Capacity of office holder(s): Trustee

24 March 2022 (4034051)

## RAHMAN, MUSTAFA

10A Merton Park Parade, LONDON, SW19 3NT

Mr Mustafa Rahman Occupation unknown, residing at 10a Merton

Park Parade, London, SW19 3NT In the County Court at Croydon

No 63 of 2021

Date of Filing Petition: 8 October 2021 Bankruptcy order date: 15 March 2022 Time of Bankruptcy Order: 11:37

Whether Debtor's or Creditor's PetitionCreditor's

Name and address of petitioner: Markerstudy Insurance Services Limited45 Westerham Road, Bessels Green, Sevenoaks, Kent, TN13 2QB

L Cook 11th Floor, Southern House, Wellesley Grove, Croydon, CR0

1XN, telephone: 03030031736 Capacity of office holder(s): Trustee

15 March 2022 (4034052)

## PARR, YVONNE DAPHNE

188 Humber Way, Slough, SL3 8SU Birth details: 12 October 1959

Yvonne Daphne Parr, A full-time carer, of 188 Humber Way, Slough,

Berkshire, SL3 8SU

In the Office of the Adjudicator

No 5135212 of 2022

Date of Filing Petition: 26 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

L Cook 11th Floor, Southern House, Wellesley Grove, Croydon, CR0

1XN, telephone: 03030031736 Capacity of office holder(s): Trustee

28 March 2022 (4034049)

## RAJARATNAM, ANTHONY MANOJ

28 Cranleigh Gardens, Luton, LU3 1LT

Birth details: 13 June 1978

Anthony Manoj Rajaratnam, Employed, of 28 Cranleigh Gardens,

Luton, Bedfordshire, LU3 1LT In the Office of the Adjudicator

No 5135198 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

A Cassidy The Civic Centre, Barras Bridge, NEWCASTLE UPON

(4034032)

TYNE, NE1 8QF, telephone: 03030031737

Capacity of office holder(s): Trustee 28 March 2022 **ROBERTS, LORRAINE** 

1 Greenways, Eaton Bray, DUNSTABLE, Bedfordshire, LU6 2BG

Birth details: 29 January 1977

Lorraine Roberts, Employed, of 1 Greenways, Eaton Bray, Dunstable,

Bedfordshire, LU6 2BG In the Office of the Adjudicator

No 5135164 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

S Baxter 3rd Floor, Companies House, Crown Way, CARDIFF, CF14

3ZA, telephone: 0300 678 0016 Capacity of office holder(s): Trustee

29 March 2022 (4034053)

WHITE, SIMON PAUL 8 Courtenay Road, Dur

8 Courtenay Road, Dunkirk, Faversham, ME13 9LH

Birth details: 5 April 1964

Simon Paul White, Currently not working, of 8 Courtenay Road,

Dunkirk, Faversham, Kent, ME13 9LH In the Office of the Adjudicator

No 5135218 of 2022

Date of Filing Petition: 27 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

L Burgess 3rd Floor, 1 City Walk, Leeds, LS11 9DA, telephone: 0300

678 0016

Capacity of office holder(s): Trustee

28 March 2022 (4034055)

**ROBINSON, TRAVIS CRAIG** 

17 Melton Street, Stockport, SK5 7QN Birth details: 22 February 1985

Travis Craig Robinson, Employed, of 17 Melton Street, Stockport,

Greater Manchester, SK5 7QN In the Office of the Adjudicator No 5135228 of 2022

Date of Filing Petition: 27 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

J Sullivan 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ,

telephone: 0300 678 0016

Capacity of office holder(s): Trustee

28 March 2022 (4034028)

WILLIAMS, MARTIN

Flat 4, 27 Albert Road, Hythe, CT21 6BP

Birth details: 18 February 1969

Martin Williams, Employed, also known as Martin Wigg, of Flat 4, 27

Albert Road, Hythe, Kent, CT21 6BP In the Office of the Adjudicator No 5135200 of 2022

Date of Filing Petition: 25 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 17:17

Whether Debtor's or Creditor's PetitionDebtor's

J Sullivan 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ,

telephone: 0300 678 0016

Capacity of office holder(s): Trustee

28 March 2022 (4034061)

**ULETT, KATHERINE JANE** 

67 Hoxton Road, Scarborough, YO12 7SX

Birth details: 6 February 1981

Katherine Jane Ulett, Self Employed, of 67 Hoxton Road, Scarborough, North Yorkshire, YO12 7SX, formerly of Greenview,

Kirkbymoorside, York, YO62 6AG In the Office of the Adjudicator

No 5135148 of 2022

Date of Filing Petition: 24 March 2022 Bankruptcy order date: 25 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

L Burgess 3rd Floor, 1 City Walk, Leeds, LS11 9DA, telephone: 0300

678 0016

Capacity of office holder(s): Trustee

25 March 2022 (4034026)

ZEESHAN, JILL NEGILL

30 Anson Crescent, Reading, RG2 8JT Birth details: 20 November 1965

Jill Negill Zeeshan, Currently not working, also known as Jill Callcut, of 30 Anson Crescent, Reading, Berkshire, RG2 8JT, Formerly trading as Little People Childcare at 30 Anson Crescent, Reading, Berkshire, RG2 8JT.

In the Office of the Adjudicator

No 5135146 of 2022

Date of Filing Petition: 24 March 2022 Bankruptcy order date: 25 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

M Mace 3rd Floor, Senate Court, Southernhay Gardens, Exeter, EX1

1UG, telephone: 0300 678 0016 Capacity of office holder(s): Trustee

25 March 2022 (4034043)

WALDEN, DOMINIC REX

19 Stafford Road, Huntington, Cannock, WS12 4PD

Birth details: 11 February 1973

Dominic Rex Walden, Currently not working, of 19 Stafford Road,

Huntington, Cannock, Staffordshire, WS12 4PD

In the Office of the Adjudicator

No 5135206 of 2022

Date of Filing Petition: 26 March 2022 Bankruptcy order date: 28 March 2022 Time of Bankruptcy Order: 00:00

Whether Debtor's or Creditor's PetitionDebtor's

K Read Level 1, Apex Court, City Link, NOTTINGHAM, NG2 4LA, telephone: 0300 678 0016, email:

Nottingham.OR@insolvency.gsi.gov.uk Capacity of office holder(s): Trustee

28 March 2022 (4034031)

DISCHARGE FROM BANKRUPTCY

**DEMETRIADES, STEWART JOHN** 

74 Hammerton Close, Elland, HX5 0BD

Birth details: 17 November 1966

Stewart John Demetriades, Currently not working, of 74 Hammerton Close, Elland, West Yorkshire, HX5 0BD, formerly of 10 Grey Court Close, Kingcross, Halifax, West Yorkshire, HX1 3LN and formerly of 1

Pembroke Street, Littleborough, Lancashire, OL15 8BG In the Office of the Adjudicator

No 5120784 of 2021

Bankruptcy order date: 10 March 2021 Date of certificate of discharge: 14 March 2022

L Burgess 3rd Floor, 1 City Walk, Leeds, LS11 9DA, telephone: 0300

678 0016

Capacity of office holder(s): Trustee

10 March 2021 (4034037)

(4036552)

#### **NOTICES OF DIVIDENDS**

In the County Court at Nottingham No 1160 of 2010

#### MISS GEMMA COOLING

(Formerly in bankruptcy)

84 Broxtowe Drive, Mansfield, Nottinghamshire, NG18 2JE and lately residing at10 Clipstone Drive, Forest Town, Nottinghamshire, NG19 0JJand lately residing of 14 Saddlers Close, Forest Town, Nottinghamshire, NG19 0QG.

Birth details: 13/04/1982

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, LTADT, Dividend Team, 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, 0161 2348500, RTLU.NW@insolvency.gov.uk. 21/07/2010

29th March 2022 (4036561)

In the County Court at Ipswich

No 44 of 2003

#### MR ADRIAN JOHN BATTLE

(in bankruptcy)

30 Heath View, Kesgrave, Ipswich, lately residing at Kingsfleet Bunglow, Ferry Road Felixstowe, formerly residing at 18 Winchester Close Stowmarket all in the county of Suffolk

Birth details: 11/09/1961

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, LTADT Chatham, P.O Box 16665, Birmingham, B2 2JX

30/03/2022

(4036553)

In the County Court at Gloucester

No 71 of 2003

## MR CHRISTOPHER LAWRENCE BROWN

(in bankruptcy)

19 Chapel Street, Cam, Dursley, Gloucestershire, GL11 5NX Birth details: 26/10/1984

Bankrupt's occupation: Part-time Computer Technician and Student Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, LTADT Chatham, P.O Box 16665, Birmingham, B2 2JX

30/03/2022 (4036548)

In the Wrexham Court

No 75 of 1997

### FRASER STANLEY DIGGORY

(Previously in bankruptcy)

Ivy Cottage, Cloy Lane, Overton, Wrexham, LL13 0HR

Birth details: 08/02/1957

Notice is hereby given that I, the Official Receiver and Trustee, intend to declare a First and Final Dividend of 1p/£ to Unsecured creditors within two months of 04/05/2022 , being the last day for proving, failing which they will be excluded from the dividend.

Mr M Commins, ORS Ipswich, Dividend Team, PO Box 16663, Birmingham, B2 2NG, 0303 003 1741,

RTLU.lpswich@insolvency.gov.uk

30/03/2022 (4036540)

In the County Court at Dewsbury

No 427 of 2006

#### MISS KAY JANETTE DAWSON

(Formerly in bankruptcy)

15A Corfe Close, Howden Clough, Birstall, BATLEY, WF17 0JW, and lately residing at 52A Carr Road, Calverley, LEEDS, LS28 5RHand previously residing at 3 Low House Fold, LIVERSEDGE,WF15 6HY, and formerly residing at 10 Dewsbury Road, Gomersal, CLECKHEATON, BD19 4LD, all in the County of West Yorkshire

Birth details: 12/12/1972

Bankrupt's occupation: Customer Services Advisor

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms Lindsay Haselgrove, LTADT, Dividend Team, 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, 0300 6780 015, RTLU.NW@insolvency.gov.uk -

30/03/2022 (4036538)

In the High Court of Justice

No 7130 of 2007

#### MR PEDRO FERREIRA

(Formerly in bankruptcy)

54 Beresford Road, London N8 0AJ; Flat 1, 77 Benwell Road, London N7 7BW . NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 13/08/1969

Bankrupt's occupation: A Group Worker

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, LTADT Manchester, PO Box 16665, BIRMINGHAM, B2 2JX, 0161 234 8500, RTLU.NW@insolvency.gsi.gov.uk

30/03/2022

In the County Court at Macclesfield

No 144 of 2007

## MISS JULIE SARAH GREGORY

Formerly in bankruptcy, not deceased

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. 56 Hornby Drive, Congleton, Cheshire, CW12 4WB; 41 Boothbed Lane, Goostrey, Cheshire, CW4 8NA

Birth details: 29 December 1978

Bankrupt's occupation: Self employed domestic cleaner

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

L Gandy, Po Box 16665, Birmingham B2 2JX. Date of appointment: 26 July 2007

29th March 2022 (4036544)

In the County Court at Bristol No 558 of 2007

#### MISS JOYCE HOOPER

(Formerly in bankruptcy)

9 Birch Road, Yate, Bristol, BS37 5EP

Birth details: 02/12/1959

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms Lindsay Haselgrove, LTADT, Dividend Team, 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, 0300 6780 015, RTLU.NW@insolvency.gov.uk -

30/03/2022 (4036542)

In the County Court at Kings Lynn No 181 of 2003

#### MR ROBERT FREDRICK PAUL MANN

(Formerly in bankruptcy)

54 Station Road, Heacham, KING'S LYNN PE31 7AP; 31 Dunns Lane, NORTH CREAKE, NR21 9JX. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 22/08/1957

Bankrupt's occupation: Unemployed

(t/a BROADWATER BOATS at Ludham Marine Boatyard, Horsefen Road, LUDHAM, NR29 5QG)

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/05/2022 , being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX. Date of Appointment – 28/11/2003

(4036547) 29th March 2022

In the County Court at St Albans

No 214 of 2004

#### MR CHRISTOPHER JOHN PIKE

(Formerly in bankruptcy)

6 Bell Lane, London Colney, Hertfordshire, AL2 1DE

Birth details: 06/04/1955

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove; LTADT, Dividend Team, 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, 0161 2348500, RTLU.NW@insolvency.gov.uk. BO DATE 29/07/2004

(4036559) 30/03/2022

In the County Court at Weymouth and Dorchester

No 22 of 2008

#### DR ALAN NOEL PEELING

Formerly in bankruptcy, not deceased

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. 84 Weston Road, Portland, Dorset, DT5 2BZ

Birth details: 12 August 1953

Bankrupt's occupation: Software Developer

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

L Gandy, Po Box 16665, Birmingham B2 2JX. Date of appointment: 05 February 2008

29th March 2022 (4036560)

In the DONCASTER Court

No 482 of 2010 MRS ALISON ROWE

(in bankruptcy)

who at date of the bankruptcy order on 06/07/2010 residing at Clay Bank Farm, Clay Bank Lane, Everton, Doncaster, South Yorkshire, DN10 5BY trading in partnership as RS Motorhomes from 30 Commercial Road, Goldthorpe Industrial Estate, Goldthorpe, South Yorkshire S63 9BL as a manufacturer of motorhomes

Birth details: 23/01/1971

Bankrupt's occupation: EMPLOYED

(t/a RS Motorhomes)

Notice is hereby given that I, the Official Receiver and Trustee, intend to declare a First and Final Dividend of 0.01p/£ to Unsecured creditors within two months of 03/05/2022, being the last day for proving, failing which they will be excluded from the dividend.

MR M COMMINS (OFFICIAL RECEIVER & TRUSTEE), PO Box 16653,

Birmingham B2 2JX. Tel: 0303 003 1741

Contact: Mr F Kelly 29th March 2022

(4036536)

In the County Court at Stockport

No. 156 of 2003

## **PAUL TRACY BRIGGS**

In Bankruptcy

Date of bankruptcy order: 09 December 2003 2 Powicke Drive, Romiley, Stockport SK6 3EF

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

14 June 1963

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 09 December 2003 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 01:00 on 02 May 2022

ppi bkt01037233

Office holder details: Ms L Haselgrove, Po Box 16665, Birmingham B2 2JX, 0303 003 1540, stephen.hetterley@insolvency.gov.uk. (4031907)

In the County Court at Bath

No. 28 of 1998

#### STEPHEN CHARLES MORGAN

Formerly in bankruptcy

Date of bankruptcy order: 02 March 1998

Hartley Farm Cottage, Hartley, Bradford-on-avon BA15 2JB

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

23 December 1965

STEPHEN CHARLES MORGAN residing at Hartley Farm Cottage, Winsley, Bradford Upon Avon, WILTSHIRE. Occupation Stockman Notice is hereby given, pursuant to Rule 11.2 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 02 March 1998 intend(s) to declare a(n) First dividend to all Unsecured creditors of the Bankrupt's estate within 4 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by 17:00 on 01 June 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is

Office holder details: S Baxter, The Insolvency Service, Companies Crown Way, Cardiff CF14 3ZA, +443003046117, House, clare.thompson@insolvency.gov.uk.

Friday 01 April 2022

(4033810)

In the County Court at Leicester

No. 2 of 2002

YOUSUF ALI

Formerly in bankruptcy

Date of bankruptcy order: 08 January 2002 27 Shanti Margh, Leicester LE4 6SN

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

01 January 1942

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 08 January 2002 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Senate Court, Southernhay Gardens, Exeter EX1 1UG, 01612405000, RTLU.NW@insolvency.gov.uk, (PPI/BKT00573623). (4034563)

In the High Court of Justice

No. 4032 of 2002

#### PATRICIA ADETOLA

In Bankruptcv

Date of bankruptcy order: 29 May 2002 Flat 45, Martlesham, London N17 6HT

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

21 April 1942 Unknown

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 29 May 2002 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 17:00 on 06 May 2022

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, +443003045862, Andrew.Page@insolvency.gov.uk. (4036422)

In the County Court at Bath No. 200 of 2003

## ANNE MARIA ALTEN

Formerly in bankruptcy

Date of bankruptcy order: 12 September 2003

60 Meare Road, Bath BA2 5PX

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

15 May 1954

Unknown

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 12 September 2003 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 17:00 on 06 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 0303 003 1949, Nathan.Greensill@insolvency.gov.uk. (4037292)

In the County Court at Stafford

No 257 of 2007

## MR RICHARD WILLIAM ARBLASTER

(Formerly in bankruptcy)

94 Peter James Court, Stafford, ST16 3YX. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 20/03/1974

Bankrupt's occupation: Customer Services Advisor

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022 , being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, LTADT Manchester, PO Box 16665, BIRMINGHAM, B2 2JX, 0300 678 0015 RTLU.NW@insolvency.gov.uk 30/03/2022 (4036555)

In the County Court at Kings Lynn

No 54 of 2001

#### MR STEVEN DERRICK BOCKING

(Formerly in bankruptcy)

Flat G6 The Maltings Swiss Terrace KINGS LYNN Norfolk PE30 2QS Birth details: 11/05/1961

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, LTADT, Dividend Team, 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, 0161 2348500, RTLU.NW@insolvency.gov.uk. BO DATE 24/04/2001

30/03/2022 (4036546)

In the High Court of Justice No. 8283 of 2008

#### MONIA BEGUM

Formerly in bankruptcy

Date of bankruptcy order: 21 August 2008 Flat 39, Yates House, London E2 6NU

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

04 May 1964

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 21 August 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:01 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Senate Court, Southernhay Gardens, Exeter EX1 1UG, 01612405400, rtlu.nw@insolvency.gov.uk, (PPI/BKT00716838). (4034739)

In the High Court of Justice Court Number: BR-2019-462

## ANTHONY DALE CUMING

Formerly in Bankruptcy

3 Farriers Gate, Bassaleg, Newport NP10 8FX.

Birth details: 14 May 1959

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Trustees intend to declare a first and final dividend to unsecured creditors of the bankruptcy estate within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Trustees at 13-14 Orchard Street, Orchard Street Business Centre, Bristol, BS1 5EH by no later than 28 April 2022 (the last date for proving).

Creditors who have not yet proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

Date of Appointment: 20 December 2019

Office Holder Details: *Neil John Maddocks* (IP No. 9239) and *Rob Coad* (IP No. 11010) both of South West and Wales Business Recovery, Orchard St Business Centre, 13-14 Orchard Street, Bristol, BS1 5EH

Further details contact: Lynn Henderson, Email: lynn.henderson@undebt.co.uk Alternative contact: Email: julie.hamblin@undebt.co.uk

Neil Maddocks, Joint Trustee

Neil Maddocks, Joint Trusto 30 March 2022

Ag OH123331

(4035584)

In the County Court at Cheltenham No 77 of 1995

### MR ANDREW CUTTS

(in bankruptcy)

52 Churchill Grove, Tewkesbury, and lately residing at 9 Freemans Close, Twyning all in the County of Gloucester.

Birth details: 11/01/1958

Bankrupt's occupation: Not known

Notice is hereby given that I, the Official Receiver and Trustee, intend to declare a First and Final Dividend of  $0.04p/\Sigma$  to Unsecured creditors within two months of 03/05/2022, being the last day for proving, failing which they will be excluded from the dividend.

Mr M Commins, ORS Ipswich, Dividend Team, PO Box 16663, Birmingham, B2 2NG, 0300 678 0015,

RTLU.lpswich@insolvency.gov.uk

29th March 2022 (4036556)

In the County Court at Truro

No. 900 of 2008

#### **CHRISTOPHER JOHN DAVEY**

In Bankruptcy

Date of bankruptcy order: 28 November 2008

3 Marine Terrace, Penzance TR18 4DL

Formerly of: Jack Lane, Newlyn, Penzance TR18 5HZ

who at the date of the bankruptcy order was trading as: Dolphin Inn NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

18 May 1957 Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 28 November 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, The Insolvency Service, Po Box 16665, Birmingham B2 2JX, 03003045616, amindeep.bains@insolvency.gov.uk, (PPI/BKT00504264). (4033865)

In the County Court at Haverfordwest

No 18 of 1991

#### RICHARD WILLIAM DALZIEL AND JACQUELINE L DALZIEL

(Partnership Member Estate Of) Richard William Dalziel

Principal trading address: 65 Bush Street, Pembroke Dock, Dyfed, :

Birth details: 30/3/1944

Nature of Business: Public House

(t/a The Bush Tavern) (our ref: BKT00026216)

Please quote all of the above in any correspondence with the Insolvency Service

Partnership Bankruptcy Order: 20/3/1991

Notice is hereby given that I, S Baxter, Official Receiver and Trustee of Richard William Dalziel, intend to declare a first and final dividend of  $100 p/\Sigma$  to proven unsecured creditors within a period of two months from 1/6/2022, being the last day for proving, failing which they will be excluded from the dividend.

Unsecured creditors intending to have their claims taken into consideration in the above referenced dividend process must submit their formal claims to me by the above stated last date for proving at the address stated below. A blank proof of debt form for completion and submission as a formal claim is obtainable from the Insolvency Service website: https://www.gov.uk/government/publications/rule-144-proof-of-debt-general-form and clicking on either the PDF or ODT shortcuts to a 'Rule 14.4 Proof Of Debt (General Form)' listed.

Alternatively, a blank proof of debt form is obtainable upon request by contacting my office at the address stated below.

Note: the above named individuals and/or partnership may no longer have a connection with the above mentioned address.

Date 30/03/2022

S Baxter, Official Receiver and Trustee, OR Cardiff Distributions, PO Box 16665, BIRMINGHAM B2 2JX; Tel: 0303 003 1625; e-mail: tim.strange@insolvency.gov.uk (4036550)

In the County Court at Plymouth

No. 195 of 2006

#### MELISSA EDGE

Formerly in bankruptcy

Date of bankruptcy order: 10 March 2006

21 Grove Park, Torpoint PL11 2PP

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

21 July 1980

Sales Assistant

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 10 March 2006 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 05 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy , Po Box 16665, Birmingham B2 2JX, 03003047201, kathy.scrase1@insolvency.gov.uk. (4035332)

In the County Court at Hereford

No. 108 of 2008

#### SHIRLEY ELIZABETH EDWARDS

Formerly in bankruptcy

Date of bankruptcy order: 13 May 2008

21 Penhaligon Way, Hereford HR4 9YB

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

08 October 1934

Part Time Domestic

Notice is hereby given, pursuant to Rule 11.2 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 13 May 2008 intend(s) to declare a(n) Dividend to all Creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by 17:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 0303 003 1197, richard.fletcher@insolvency.gov.uk. (4034537)

In the County Court at Scarborough

No. 179 of 2008

## LUKE GENE FLETCHER

Formerly in bankruptcy

Date of bankruptcy order: 25 June 2008 36 Buckrose Grove, Bridlington YO16 4EQ

Formerly of: 70 Rectory Walk, Bridlington YO16 4DG

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

03 January 1979

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 25 June 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Insolvency Service, Po Box 16665, Birmingham B2 2JX, 03030031471, RTLU.NW@insolvency.gov.uk, (PPI/BKT00891334). (4034000)

In the County Court at Northampton

No 396 of 1993

#### MARTIN JOSEPH FINN

Principal trading address: 1 Faraday Close, off Sopwith Way, Drayton Fields Industrial Estate, Daventry, Northamptonshire NN11 5RD

Birth details: 29/6/1957 Nature of Business: Haulier (t/a Morlex Transport Agents) our ref: BKT00868871

Please quote all of the above in any correspondence with the

Insolvency Service

Partnership Bankruptcy Order: 13/10/1993

Unsecured creditors intending to have their claims taken into consideration in the above referenced dividend process must submit their formal claims to me by the above stated last date for proving at the address stated below. A blank proof of debt form for completion and submission as a formal claim is obtainable from the Insolvency Service website: https://www.gov.uk/government/publications/rule-144-proof-of-debt-general-form and clicking on either the PDF or ODT shortcuts to a 'Rule 14.4 Proof Of Debt (General Form)' listed.

Alternatively, a blank proof of debt form is obtainable upon request by contacting my office at the address stated below.

Note: the above named individuals and/or partnership may no longer have a connection with the above mentioned address.

S Baxter, Official Receiver and Trustee, OR Cardiff Distributions, PO Box 16665, BIRMINGHAM B2 2JX; Tel: 03030 031174; e-mail: rachel.vaughan@insolvency.gov.uk

29th March 2022 (4036558)

In the County Court at Norwich

No. 421 of 2010

#### PATRICK EDWARD GARRETT (DECEASED)

Formerly in bankruptcy

Date of bankruptcy order: 08 March 2010 Poplar Farm, High Common Road, Diss IP22 2HP

Formerly of: Thorpe Farmhouse, Shadwell, Thetford IP24 2RX

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

17 February 1938

RETIRED

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 08 March 2010 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, The Senate, Southernhay Gardens, Exeter EX1 1UG, +441612405000, RTLU.NW@insolvency.gov.uk. (4034879)

In the County Court at Newcastle upon Tyne

No. 1489 of 2008

#### **LESLIE GREENWELL (DECEASED)**

Formerly in bankruptcy

Date of bankruptcy order: 09 September 2008

16 The Croft, Stanley DH9 8NQ

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

22 April 1948

Retired

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 09 September 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 17:00 on 03 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 03003046571, Folashade.Olubakin@insolvency.gov.uk, (BKT00602259).

Tuesday 29 March 2022

(4033399)

In the County Court at Bury St Edmunds

## YVONNE FRANCES AGNES MAUD GOULD

also known as: YVONNE FRANCES AGNES MAUD LAKE

Formerly in bankruptcy

Date of bankruptcy order: 10 July 2008 7 Middle Way, Long Melford, Sudbury CO10 9EF

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

19 December 1933

Retired

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 10 July 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:01 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Senate Court, Southernhay Gardens, Exeter EX1 1UG, 01612405400, rtlu.nw@insolvency.gov.uk, (PPI/BKT00702664). (4033806)

In the County Court at Northampton

No 83 of 1993

#### RICHARD DAVID JOHN GREEN

Principal trading address: 1 Faraday Close, off Sopwith Way, Drayton Fields Industrial Estate, Daventry, Northamptonshire NN11 5RD

Birth details: 23/1/1949 Nature of Business: Haulier (t/a Morlex Transport Agents)

our ref: BKT00272717

Please quote all of the above in any correspondence with the

Insolvency Service
Partnership Bankruptcy Order: 24/6/1993

Notice is hereby given that I, S Baxter, Official Receiver and Trustee of Richard David John Green, intend to declare a first and final dividend of  $4.78p/\Sigma$  to proven unsecured creditors within a period of two months from 1/6/2022, being the last day for proving, failing which they will be excluded from the dividend.

Unsecured creditors intending to have their claims taken into consideration in the above referenced dividend process must submit their formal claims to me by the above stated last date for proving at the address stated below. A blank proof of debt form for completion and submission as a formal claim is obtainable from the Insolvency Service website: https://www.gov.uk/government/publications/rule-144-proof-of-debt-general-form and clicking on either the PDF or ODT shortcuts to a 'Rule 14.4 Proof Of Debt (General Form)' listed.

Alternatively, a blank proof of debt form is obtainable upon request by contacting my office at the address stated below.

Note: the above named individuals and/or partnership may no longer have a connection with the above mentioned address.

S Baxter, Official Receiver and Trustee, OR Cardiff Distributions, PO Box 16665, BIRMINGHAM B2 2JX; Tel: 03030 031174; e-mail: rachel.vaughan@insolvency.gov.uk

29th March 2022 (4036551)

In the County Court at Yeovil

No. 210 of 2005

PHILIP HARRY
In Bankruptcy

Date of bankruptcy order: 05 July 2005

Home Mead Cottage, North Brewham, Bruton BA10 0JG

Formerly of: 4 Chandos Road, London NW2 4LU

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

19 August 1973

Teacher

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 05 July 2005 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, The Insolvency Service, Po Box 16665, Birmingham B2 2JX, 03003045616, amindeep.bains@insolvency.gov.uk, (PPI/BKT00699034). (4033804)

In the County Court at Romford

No. 781 of 2008

**JEAN BLANCHE HEWITT** 

Formerly in bankruptcy

Date of bankruptcy order: 25 November 2008 6 Cherrydown Walk, Romford RM7 8HJ

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

09 July 1934 A Dance Teacher Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 25 November 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Miss L Haselgrove, The Insolvency Service, Po Box 16665, Birmingham B2 2JX, 01612405000, RTLU.NW@insolvency.gov.uk.

Wednesday 30 March 2022

(4034996)

In the County Court at Leicester

No. 95 of 2002

## **NAZMA KHATUN HAMID**

Formerly in bankruptcy

Date of bankruptcy order: 04 April 2002

40 Saxby Street, Leicester LE2 0NE

who at the date of the bankruptcy order was trading as: HQ Supermarket

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

28 June 1975

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 04 April 2002 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 01:00 on 04 May 2022

Office holder details: L Gandy, Po Box 16665, Birmingham B2 2JX, 0303 003 1934, Harry.Richardson@insolvency.gov.uk. (4034836)

In the County Court at Oxford

No. 269 of 2006

### MICHELLE HAMMERSLEY

In Bankruptcy

Date of bankruptcy order: 12 April 2006 20 Galingale Close, Bicester OX26 3FD

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

23 April 1979 Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 12 April 2006 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 17:00 on 06 May 2022

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, +443003045862, Andrew.Page@insolvency.gov.uk. (4037846)

In the County Court at Norwich

No. 587 of 2007

#### **LESTER JOHN HARGREAVES**

Formerly in bankruptcy

Date of bankruptcy order: 17 April 2007 18 Hearne Court, Norwich NR5 9NY

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

06 May 1970

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 17 April 2007 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003047201, kathy.scrase1@insolvency.gov.uk. (4034085)

In the County Court at Bedford

No 18 of 2017

#### MRS JULIE ANNE HARTARD

(in bankruptcy)

15 York Close, Bedford, MK45 4QB.

Birth details: 21/04/1959

Bankrupt's occupation: Manager at NHS

Notice is hereby given that I, the Official Receiver and Trustee, intend to declare a First and Final Dividend of  $0.04p/\mathfrak{L}$  to Unsecured creditors within two months of 03/05/2022, being the last day for proving, failing which they will be excluded from the dividend.

Mr M Commins, ORS Ipswich, Dividend Team, PO Box 16663, 0015. Birmingham. B2 0300 678 2NG.

RTLU.lpswich@insolvency.gov.uk

29th March 2022 (4036539)

In the County Court at Norwich

No. 2172 of 2009

#### **SOKOL HODAJ**

Formerly in bankruptcy

Date of bankruptcy order: 21 December 2009 48 Bevan Street West, Lowestoft NR32 2AD

Formerly of: 4 Stanley Road, Lowestoft NR33 9HD Suffolk

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

20 July 1973 **FACTORY OPERATIVE** 

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 21 December 2009 intend(s) to declare a(n) First and final dividend of £00.08 to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003047143, Julie.walker1@insolvency.gov.uk, (BKT00450321/

Wednesday 30 March 2022 (4034806)

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 03003048520, michele.chappell@insolvency.gov.uk, (PPI/ BKT00530395).

In the County Court at Barnsley

No. 36 of 2011

#### **GARY WILLIAM KIDGELL**

Formerly in bankruptcy

Date of bankruptcy order: 10 February 2011 86 Lilac Crescent, Hoyland, Barnsley S74 9PW

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

25 December 1962

Driver

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 10 February 2011 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy , Po Box 16665, Birmingham B2 2JX, 03003047201, kathy.scrase1@insolvency.gov.uk.

In the Bath Court

No 115 of 2006

## MS JEMMA LOUISE LATHAM

(in bankruptcy)

107 Rodden Road, Frome, Somerset, BA11 2AJ; 76 CourtsBarton, Frome, Somerset, BA11 4QB; 7 DairyFlats, Back Lane, Sherbourne, Dorset, DT9 4BQ

Birth details: 04/06/1982

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver appointed 02/03/2006

N C Sawdon

29th March 2022

(4036545)

In the County Court at Leicester

No. 298 of 2003 **SOHAL KASUJI** 

In Bankruptcv

Date of bankruptcy order: 26 September 2003

7 Mortoft Road, Leicester LE4 7QQ

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

30 April 1978

unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 26 September 2003 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

In the High Court of Justice

No. 3943 of 2005

## MARIA JOSE ALMEIDA LORY

Formerly in bankruptcy

Date of bankruptcy order: 25 April 2005

Flat 6, Avenue House, Willesden, London NW10 5AP

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

08 March 1957

A Nanny

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 25 April 2005 intend(s) to declare a(n) First and final dividend of £00.11 to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003047143, Julie.walker1@insolvency.gov.uk, (BKT00653824/PPI)

Wednesday 30 March 2022

(4034020)

In the County Court at Bristol No. 585 of 2011

#### PLERN-TA LYNN

Formerly in bankruptcy

Date of bankruptcy order: 12 May 2011 71 Wades Road, Filton, Bristol BS34 7EB Formerly of: 18 South Hayes, Bristol BS5 6UB

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

12 May 2011 Shop Worker

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 12 May 2011 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 01612405000, RTLU.NW@insolvency.gov.uk, (PPI/BKT3287757). (4034807)

In the County Court at Slough No. 21 of 2003

#### **CARL ANTHONY MCEVOY**

Formerly in bankruptcy

Date of bankruptcy order: 24 January 2003 14 Bramble Close, Uxbridge UB8 3QE

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

07 April 1967

unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 24 January 2003 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 0333 241 4613, Adam.Smith@insolvency.gov.uk. (4034525)

In the Shrewsbury Court

No 165 of 1999

#### MR JONATHAN BRYAN MATTHEWS

(Previously in bankruptcy)

36 Brandsfarm Way, Randlay, Telford, Shropshire, Warehouse Manager, lately residing at 11 Fosters Foel, Aqueduct, Telford.

Birth details: 25/08/1965

Notice is hereby given that I, the Official Receiver and Trustee, intend to declare a First and Final Dividend of  $3p/\mathfrak{L}$  to Unsecured creditors within two months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend.

Mr M Commins, ORS Ipswich, Dividend Team, PO Box 16663, Birmingham, B2 2NG, 0303 003 1741,

RTLU.lpswich@insolvency.gov.uk

30/03/2022 (4036537)

In the County Court at Hereford

No. 49 of 2002

#### **JUDITH LORRAINE MILES**

Formerly in bankruptcy

Date of bankruptcy order: 06 March 2002 51 Viaduct Way, Welwyn Garden City AL7 1UB

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

25 March 1954

Unknown

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 06 March 2002 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 17:00 on 06 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 0303 003 1949, Nathan.Greensill@insolvency.gov.uk. (4037858)

In the County Court at Stockport

No. 87 of 2003

#### **GARY GEORGE MILLINGTON**

Formerly in bankruptcy

Date of bankruptcy order: 01 July 2003

3 Wells Close, Heald Green, Cheadle SK8 3NA

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

09 May 1960

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 01 July 2003 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 01:00 on 04 May 2022

Office holder details: L Gandy, Po Box 16665, Birmingham B2 2JX, 0303 003 1934, Harry.Richardson@insolvency.gov.uk. (4034892)

In the County Court at Slough

No. 252 of 1997

#### **SEAN ROBERT MOSDALL**

Formerly in bankruptcy

Date of bankruptcy order: 16 October 1997 Lisbeth High Street, Colnbrook, Slough SL3 0LX

NOTE: the above-named was discharged from the proceedings and

may no longer have a connection with the addresses listed.

11 May 1966

SEAN ROBERT MOSDALL, Lisbeth, High Street, Colnbrook, Slough SL3 8LX, a Builder, lately residing at8 lbbotson Court, Poyle Road, Colnbrook, Slough SL3 0HP, lately carrying on business as a Building Contractortrading as WINDSOR CONSTRUCTION/SRM DECORATORS at8 lbbotson Court, Poyle Road, Colnbrook, Slough SL3 0HP.

Notice is hereby given, pursuant to Rule 11.2 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 16 October 1997 intend(s) to declare a(n) First dividend to all Unsecured creditors of the Bankrupt's estate within 4 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by 17:00 on 01 June 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: S Baxter, The Insolvency Service, Companies House, Crown Way, Cardiff CF14 3ZA, +443003046117, clare.thompson@insolvency.gov.uk.

Friday 01 April 2022 (4033783)

In the Shrewsbury Court

No 41 of 2007

#### MR SIMON NOCK

(in bankruptcy)

3 AnselmCourt, Telford, Shropshire TF3 1AT; 27 Willowfield, Telford,

Shropshire TF7 5NS Birth details: 20/06/1972

Bankrupt's occupation: a Technical Support Programmer

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver appointed 02/02/2007

N C Sawdon 29th March 2022

(4036554)

In the Birmingham County Court Court Number: BR-2019-56 JEREMIAH O'CONNOR

In Bankruptcy

71 Woolacombe Lodge Road, Selly Oak, Birmingham, B29 6QA.

Birth details: 12 May 1964 Occupation: Self employed builder

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Trustees intend to declare a first dividend to unsecured creditors of the bankruptcy estate within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Trustees at c/o Moorfields Advisory, 20 Old Bailey, London EC4M 7AN by no later than 25 April 2022 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

Date of Appointment: 25 April 2017

Office Holder Details: *Tom Straw* (IP No. 23850) and *Simon Robert Thomas* (IP No. 8920) both of Moorfields Advisory Limited, 20 Old Bailey, London EC4M 7AN

Further details contact: The Joint Trustees, Tel: 0207 186 1157. Alternative contact: Jamie Gallagher, Tel: 0207 186 1157, Email: jamie.gallagher@moorfieldscr.com

Tom Straw, Joint Trustee

30 March 2022

Ag OH123310 (4035449)

No. 134 of 2003

#### WAYNE COLIN PRICE

Formerly in bankruptcy

Date of bankruptcy order: 22 April 2003 2 Tovey Crescent, Leicester LE2 9DF

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

17 July 1964 Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 22 April 2003 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 05 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Miss L Haselgrove, The Insolvency Service, Po Box 16665, Birmingham B2 2JX, 01612405000,

RTLU.NW@insolvency.gov.uk, (PPI/BKT00645768).

Thursday 31 March 2022 (4036153)

In the County Court at Caernarfon

No. 60 of 2005

DAVID PRESCOTT

Formerly in bankruptcy

Date of bankruptcy order: 13 June 2005 9 Roslyn Gardens, Llandudno LL30 1BS

Formerly of: The Flat, Premier House, Lloyd Street, Llandudno LL30

2YA

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

29 April 1944

Insurance Introducer

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 13 June 2005 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003047201, kathy.scrase1@insolvency.gov.uk. (4034553)

In the County Court at Walsall

No. 653 of 2008

### **MURIEL EDITH PRICE**

Formerly in bankruptcy

Date of bankruptcy order: 01 August 2008

10 Grange Crescent, Walsall WS4 1NS

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

06 December 1940

Retired

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 01 August 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

In the County Court at Leicester

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:01 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Senate Court, Southernhay Gardens, Exeter EX1 1UG, 01612405400, rtlu.nw@insolvency.gov.uk, (PPI/BKT00454860). (4034074)

In the County Court at Sheffield

No. 77 of 2013

#### VIRGINIA JANE RICHMOND

Formerly in bankruptcy

Date of bankruptcy order: 19 February 2013

28 Mitchell Road, Sheffield S8 0GQ

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

18 April 1961

Self Employed Medical Adviser

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 19 February 2013 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 05 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003046118, Sara.Hawkins@insolvency.gov.uk. (4035093)

In the Northampton Court

No 358 of 1995

### MR WILLIAM JOHN RILEY

(Previously in bankruptcy)

24 Hatton Avenue, Wellingborough, Northants NN8 3AP, formerly a companydirector of The Cottage, Zion Hill, Walgrave, NorthantsNN6 9PN

Birth details: 02/04/1963

Bankrupt's occupation: Factory Worker

Notice is hereby given that I, the Official Receiver and Trustee, intend to declare a First and Final Dividend of 1p/£ to Unsecured creditors within two months of 04/05/2022 , being the last day for proving, failing which they will be excluded from the dividend.

Mr M Commins, ORS Ipswich, Dividend Team, PO Box 16663, Birmingham, B2 2NG, 0303 003 1741,

RTLU. Ipswich@insolvency. gov. uk

30/03/2022 (4036541)

In the County Court at Plymouth

No. 628 of 2006

## **REBECCA ANN ROBINSON**

In Bankruptcy

Date of bankruptcy order: 05 September 2006

2 Rose Cottages, Old Warleigh Lane, Plymouth PL5 4NE

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

30 May 1984

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 05 September 2006 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 17:00 on 06 May 2022

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, +443003045862, Andrew.Page@insolvency.gov.uk. (4037856)

In the County Court at Barnstaple

No 299 of 2005

### MISS LILIAN BARBARA RUMSON

(Formerly in bankruptcy)

2 Gaiety Flats, Somers Crescent, Ilfracombe, Devon, EX34 9DP

Birth details: 29/01/1939

Bankrupt's occupation: Unemployed

(t/a BARBARAS DRESS AGENCY, 76High Street, Ilfracombe, Devon) Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 04/05/2022, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy – Official Receiver, PO BOX 16665, Birmingham, B2 2JX. Date of Appointment – 15/12/2005

30/03/2022 (4036557)

In the County Court at Durham

No. 150 of 2008

#### JANETTE SCOTT

Formerly in bankruptcy

Date of bankruptcy order: 28 April 2008

13 Brickgarth, Easington Lane, Houghton Le Spring DH5 0LA

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

08 February 1962

Carer

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 28 April 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 0161 240 5000, RTLU.NW@insolvency.gsi.gov.uk, (PPI/BKT00140177). (4034540)

In the County Court at Wolverhampton

No. 70025 of 2010

## **ELZBIETA SMIGIELSKA**

Formerly in bankruptcy

Date of bankruptcy order: 14 January 2010 142 Powell Street, Wolverhampton WV10 0BN

Formerly of: 107 Nine Elms Lane, Wolverhampton WV10 9AW

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

13 July 1954 Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 14 January 2010 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, The Senate, Southernhay Gardens, Exeter EX1 1UG, +441612405000, RTLU.NW@insolvency.gov.uk. (4034554)

In the County Court at Great Yarmouth

No 57 of 1995

#### ANTHONY ALBERT EDWARD STORY AND PAUL ALLINGTON

Partnership Member Estate Of PAUL ALLINGTON

Principal trading address: 143 London Road South, Lowestoft, Suffolk

Birth details: 30/01/1964 Nature of Business: TICKET AGENTS (t/a 'LOWESTOFT CONCERT TRAVEL')

(our ref: BKT00037867)

Please quote all of the above in any correspondence with the

Insolvency Service

Partnership Bankruptcy Order: 18/05/1995

Notice is hereby given that I, S Baxter, Official Receiver and Trustee of PAUL ALLINGTON, intend to declare a first and final dividend of  $3.61p/\Sigma$  to proven unsecured creditors within a period of two months from 01/06/2022, being the last day for proving, failing which they will be excluded from the dividend.

Unsecured creditors intending to have their claims taken into consideration in the above referenced dividend process must submit their formal claims to me by the above stated last date for proving at the address stated below. A blank proof of debt form for completion and submission as a formal claim is obtainable from the Insolvency Service website: https://www.gov.uk/government/publications/rule-144-proof-of-debt-general-form and clicking on either the PDF or ODT shortcuts to a 'Rule 14.4 Proof Of Debt (General Form)' listed.

Alternatively, a blank proof of debt form is obtainable upon request by contacting my office at the address stated below.

Note: the above named individuals and/or partnership may no longer have a connection with the above mentioned address. 30/03/2022

S Baxter, Official Receiver and Trustee, Cardiff Distributions, PO Box 16665, BIRMINGHAM B2 2JX, Tel: 03003 046349; e-mail: desmond.kavanagh@insolvency.gov.uk (4036543)

In the County Court at Hastings

No. 27 of 2002

#### **ELIZABETH MARIANNE SCOVIL**

In Bankruptcy

Date of bankruptcy order: 27 March 2002 2 Fernlea Court, Lydd Road, Rye TN31 7RS Formerly of: 45 Egerton Gardens, London SW3 2DD

NOTE: the above-named was discharged from the proceedings and

may no longer have a connection with the addresses listed.

05 November 1944

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 27 March 2002 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 17:00 on 06 May 2022

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, +443003045862, Andrew.Page@insolvency.gov.uk. (4036340)

In the County Court at Brighton

No. 987 of 2004

## CHRISTINA MARIA SPARKS

also known as: Christina Bowler

Formerly in bankruptcy

Date of bankruptcy order: 01 September 2004

53 Hartfield Road, Seaford BN25 4PP

Formerly of: 25 South Coast Road, Peacehaven BN10 8SZ

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

18 December 1940

Retired

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 01 September 2004 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy , Po Box 16665, Birmingham B2 2JX, 03003047201, kathy.scrase1@insolvency.gov.uk. (4033863)

In the County Court at Bradford

No. 122 of 2000

## JOHN EDWARD STEWARD

Formerly in bankruptcy

Date of bankruptcy order: 11 May 2000 10 Irwell Street, Bradford BD4 7EQ

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

10 June 1965

JOHN EDWARD STEWARD a WAGON DRIVER of 10 Irwell Street, EastBowling, Bradford, West Yorkshire BD4 7EQ.

Notice is hereby given, pursuant to Rule 11.2 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 11 May 2000 intend(s) to declare a(n) First dividend to all Unsecured creditors of the Bankrupt's estate within 4 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by 17:00 on 01 June 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: S Baxter, The Insolvency Service, Companies House, Crown Way, Cardiff CF14 3ZA, +443003046117, clare.thompson@insolvency.gov.uk.

Friday 01 April 2022 (4034814)

In the County Court at Plymouth

No. 489 of 2008

#### **ROBERT JOHN STRAWBRIDGE (DECEASED)**

Formerly in bankruptcy

Date of bankruptcy order: 28 July 2008 36 St. Dunstans Road, Salcombe TQ8 8AN

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

19 October 1950

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 28 July 2008 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003046118, Sara.Hawkins@insolvency.gov.uk. (4034577)

In the County Court at Nottingham

No. 107 of 2006

#### **DEAN ANDREW THOMPSON**

Formerly in bankruptcy

Date of bankruptcy order: 18 January 2006 18 Tilford Gardens, Stapleford, Nottingham NG9 7FJ

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

24 April 1984 Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 18 January 2006 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 0161 240 5000, RTLU.NW@insolvency.gsi.gov.uk, (PPI/BKT00625007). (4033781)

In the County Court at Aylesbury

No. 192 of 2003

## **ALAN GRAHAM TIPPING**

In Bankruptcy

Date of bankruptcy order: 21 July 2003

13 Harroell, Long Crendon, Aylesbury HP18 9AY

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

04 December 1965 unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 21 July 2003 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 03003048520, michele.chappell@insolvency.gov.uk, (PPI/BKT00497055). (4033790)

In the County Court at Nottingham

No. 431 of 2002

#### **LEE JOHN TALBOT**

Formerly in bankruptcy

Date of bankruptcy order: 06 August 2002

27 Ringrose Close, Newark NG24 2JL

Formerly of: Flat 3, Prawles Court, Ewhurst Green, Robertsbridge, East Sussex TN32 5RG

who at the date of the bankruptcy order was trading as: Talbot Bricklaying Specialist

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

17 August 1971

Self Employed Bricklayer

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 06 August 2002 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 01:00 on 04 May 2022

Office holder details: L Gandy, Po Box 16665, Birmingham B2 2JX, 0303 003 1934, Harry.Richardson@insolvency.gov.uk. (4034565)

In the County Court at Kingston-upon-Thames

No. 251 of 2008

#### THOMAS JOHN TRAYLEN

Formerly in bankruptcy

proving specified below.

Date of bankruptcy order: 17 April 2008

6a Meadow View, Hithermoor Road, Staines-upon-thames TW19 6BH NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

20 March 1942 Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 17 April 2008 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:01 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Senate Court, Southernhay Gardens, Exeter EX1 1UG, 01612405400, rtlu.nw@insolvency.gov.uk, (PPI/BKT00913210). (4034087)

In the Chelmsford County Court Court Number: BR-2020-30

PIETRO TURONE

Also known as: Peter Turone

Compton Brow, Coppice Row, Theydon Bois, Essex CM16 7DP.

Date of Birth: UnknownOccupation: Unknown

Notice is hereby given by the Joint Trustee, Michael Wellard of RBW Restructuring Limited, Unit 4 Limes Court, Conduit Lane, Hoddesdon, Hertfordshire EN11 8EP, pursuant to Rule 14.29 of the Insolvency (England and Wales) Rules 2016 that I intend to declare a first dividend to the creditors of the Bankrupt within 2 months of the last date for proving set out below.

The creditors of the Bankrupt are required, on or before 22 April 2022 (the last date for proving), to prove their debts by sending to the undersigned, Michael Wellard of RBW Restructuring Limited, Unit 4 Limes Court, Conduit Lane, Hoddesdon, Hertfordshire EN11 8EP, the joint trustee of the bankruptcy estate, written statements of the amounts they claim to be due to them from the Bankrupt and, if so requested, to provide such further details or produce such documentary evidence as may appear to the joint trustee to be necessary.

A creditor who has not proved this debt before the declaration of any dividend is not entitled to disturb, by reason that he has not participated in it, the distribution of that dividend or any other dividend declared before his debt was proved.

Date of Appointment: 24 November 2020

Office Holder Details: *Michael James Wellard* (IP No. 9670) of RBW Restructuring Limited, Unit 4, Limes Court, Conduit Lane, Hoddesdon, Hertfordshire, EN11 8EP

Further details contact: Michael Wellard, Email: info@rbwr.co.uk, Telephone No. 01992 392030.

Michael Wellard, Joint Trustee

30 March 2022 Ag OH123333

(4035512)

In the High Court of Justice No. 1746 of 2008

HASAN UNEK

Formerly in bankruptcy

Date of bankruptcy order: 29 February 2008

140 Sirdar Road, London N22 6RD

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

01 October 1966

Part Time Meat Packer

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 29 February 2008 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate within 2 months of the last date for proving

Creditors who have not vet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003046118, Sara.Hawkins@insolvency.gov.uk. (4033874)

In the County Court at Romford No. 76 of 2006

#### **JAIME LOUISA WAKER**

Formerly in bankruptcy

Date of bankruptcy order: 16 February 2006 172 Dagnam Park Drive, Romford RM3 9RT

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

24 June 1979 Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 16 February 2006 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 05 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy , Po Box 16665, Birmingham B2 2JX, 03003047201, kathy.scrase1@insolvency.gov.uk. (4035757)

In the County Court at Slough No. 109 of 2006

## JULIE ELIZABETH WALLACE

also known as: Julie Elizabeth Ansell

In Bankruptcy

Date of bankruptcy order: 09 March 2006 58a High Street, Sunninghill, Ascot SL5 9NF

Formerly of: Oakview, Bonseys Lane, Chobham, Surrey GU24 8JJ NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

13 February 1975 Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 09 March 2006 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 17:00 on 06 May 2022

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, +443003045862, Andrew.Page@insolvency.gov.uk. (4037826)

In the County Court at Warrington

No. 85 of 2002 TRACEY JANE WHITTAKER

also known as: Tracey Wilkes Formerly in bankruptcy

Date of bankruptcy order: 31 July 2002 55 Poplars Avenue, Warrington WA2 9LJ

Formerly of: 107 Engaine Drive, Shenley Church End, Milton Keynes

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

10 June 1966

Cleaning Supervisor

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 31 July 2002 intend(s) to declare a(n) First and final dividend of £00.19 to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 03 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003047143, Julie.walker1@insolvency.gov.uk, (BKT00439760/

Wednesday 30 March 2022

(4033798)

In the County Court at Plymouth

No. 380 of 2008

#### **GREG HUGH WILLIAMS**

Formerly in bankruptcy

Date of bankruptcy order: 09 June 2008 10 Poplar Close, Plymouth PL7 2GE

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

23 May 1962

Taxi Driver

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 09 June 2008 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 May 2022 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Laura Gandy, Po Box 16665, Birmingham B2 2JX, 03003046118, Sara.Hawkins@insolvency.gov.uk. (4034077)

# Wills & probate

## **DECEASED ESTATES - LONDON EDITION**

Notice is hereby given pursuant to s. 27 of the Trustee Act 1925, that any person having a claim against or an interest in the estate of any of the deceased persons whose names and addresses are set out below is hereby required to send particulars in writing of his claim or interest to the person or persons whose names and addresses are set out below, and to send such particulars before the date specified in relation to that deceased person displayed below, after which date the personal representatives will distribute the estate among the persons entitled thereto having regard only to the claims and interests of which they have had notice and will not, as respects the property so distributed, be liable to any person of whose claim they shall not then have had notice

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
ADELAJA , Miatjorie Elma	314 Katherine Road, London E7 8PG. 7 May 2021	Anderson Browne Solicitors, Sworders Yard, 19e North Street, Bishops Stortford, Hertfordshire CM23 2LD. (.)	2 June 2022	(4036506)
ELDEN, LILY ((formerly known as Lily Prince))	39 Courtway Drive Sneyd Green Stoke-On-Trent Staffordshire ST1 6DX. Pottery Presser (Retired). 8 March 2022	Nowell Meller Solicitors Limited of Windsor House, 5a King Street, Newcastle under Lyme, Staffordshire ST5 1EH. (Stephen Paul Kirwan and Jane Alison Matthewman.)	6 June 2022	(4035175)
VERMA, Mr Om Prakash	44 MORNINGTON CRESCENT, HOUNSLOW TW5 9SS. MAGISTRATE (RETIRED). 2 February 2021	44 MORNINGTON CRESCENT, HOUNSLOW TW5 9SS. (Sushila Wanti Verma.)	2 June 2022	(4035172)
A`COURT, Dennis George	1 Miners Close, Long Ashton, Bristol, BS41 9DE. 23 May 2021	PowellsLaw, 7-13 Oxford Street Weston-super-Mare North Somerset BS23 1TE	2 June 2022	(4030971)
BELCHER, MR KEITH	582 Pilsley Road, Morton, ALFRETON, DE55 6HP. MACHINE OPERATOR - RETIRED. 21 January 2022	SMALLEYS SOLICITORS, MARCIA CLARE PETCHER, 2a Cross Street, Arnold, NOTTINGHAM, NG5 7BL.	3 June 2022	(4036411)
BAILEY, Leslie Arthur John (John Bailey)	2 Lambert Walk, Thame, OX9 3DT . 15 October 2020	Lightfoots LLP, 1-3 High Street, Thame, Oxon, OX9 2BX (Ref: SLE/ BAl67-1) (Tarquin James May.)	2 June 2022	(4031853)
BAILEY, Mary Beryl ( (Beryl Bailey))	25 The Limes, Halesworth, Suffolk IP19 8LT. Medical Librarian (Retired). 30 November 2021	Butterworths Solicitors, 100a Stricklandgate, Kendal LA9 4PU. Ref: B01130-0069.	9 June 2022	(4035384)
BANFIELD, Pamela Mary	Two Trees Over Compton, Sherborne, Dorset, DT9 4QS. 3 February 2022	Battens Solicitors Limited, Battens The Bank House, Long Street, Sherborne, Dorset, DT9 3BU (Sophie Louise Walshe.)	2 June 2022	(4027914)
BANFIELD, Leslie Stewart	Two Trees Over Compton, Sherborne, Dorset, DT9 4QS. 27 January 2022	Battens Solicitors Limited, Battens, The Bank House, Long Street, Sherborne, Dorset, DT9 3BU (Sophie Louise Walshe.)	2 June 2022	(4027918)
BANKS, William	443 Trelawney Avenue, Langley, Slough, Berkshire, SL3 7TT. 15 March 2022	Barrett & Thomson, One Pegasus Court, 25 Herschel Street, Slough, Berkshire, SL1 1TQ (Catherine Anne Matharoo & Claire Angela Fitzgerald.)	2 June 2022	(4029428)
BARRETT, Winifred Irene	6 Austen Close, East Grinstead, RH19 1RZ. 6 November 2021	Mayo Wynne Baxter LLP, 3 Bell Lane, Lewes, East Sussex, BN7 1JU ( Fiona Margaret Dodd, Nicholas Charles Garrish.)	2 June 2022	(4028833)
BARRINGTON, Joan Mary	Avalon Nursing Home, 5 Nevill Avenue, Eastbourne, BN22 9PR. 18 October 2021	Mayo Wynne Baxter LLP, 3 Bell Lane, Lewes BN7 1JU (Fiona Margaret Dodd and Dean Orgill.)	2 June 2022	(4027920)
BARRY, David John	13 Tank House Road, Winslow, Bucks, MK18 3LT. 28 February 2021	Rachel Stafford Legal Services, 22 Whittle Road Thame, OX9 3PF (Rachel Louise Stafford.)	2 June 2022	(4026754)
BEEKEN, Kenneth Wilfred	Tadworth Grove Care Home, The Avenue, TADWORTH, Surrey, KT20 5AT. 9 February 2022	Michelle Ann Beeken, The London Gazette (24684), PO Box 3584, Norwich, NR7 7WD.	3 June 2022	(4037726)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
BETTS, George Patrick	Sycamore Care Centre Boldon, New Road, Boldon Colliery, NE35 9DR. 13 August 2021	PGS Law LLP, Law Court Chambers, 22 Denmark Centre, Fowler Street, South Shields NE33 2LR (Ref: SMG/BET2297/1) (William Joseph Dryden.)	2 June 2022	(4033518)
BHASIN, Mr Ashok Kumar	37 Fullers Road, South Woodford, London, E18 2QB, and t/a Computer Accounting and Payroll Services (CAPS), 263A Chingford Mount Road,, London, E4 8LP. 15 December 2020	Aarti Bhasin, The London Gazette (24690), PO Box 3584, Norwich, NR7 7WD.	2 June 2022	(4037795)
BOORMAN, Ruby (Newell)	Kerswell Cottage, Creedy Park, Crediton, Devon, EX17 2DH. 12 April 2020	Charlesworth Nicholl & Co, 31 High Street, Crediton, Devon EX17 3AJ (Philip Murray Hutchens.)	2 June 2022	(4029862)
BRANCHFLOWER, Keith	42 Amy Street Leicester, LE3 2FB. 3 December 2021	Hawley & Rodgers Solicitors, 19/23 Granby Street, Loughborough, Leicestershire LE11 3DY (Lisa Christina Corcoran.)	2 June 2022	(4027303)
BRAY, Belinda Catherine	Beechcroft Manor Nursing Home, 1 Beechcroft Road, Gosport, Hampshire, PO12 2EP. 25 February 2022	Lloyds Bank Estate Administration Service, PO BOX 5005 Lancing BN99 8AZ (Lloyds Bank Plc.)	2 June 2022	(4037285)
BRIAN, Mrs Shelagh (Shelagh Adkins)	Wythall Home, 241 Station Road, Wythall, B97 6ET. 1 February 2022	Haroon Qayum c/o Blackhams, Lancaster House, 67 Newhall Street, Birmingham, B3 1NR.	2 June 2022	(4037969)
BROOKS, Sylvia	16 Orchard Close, Ramsgate, Kent, CT12 6QR. 22 September 2021	Boys & Maughan Solicitors LLP, India House Hawley Street Margate Kent CT9 1PZ (Yvonne Edwards.)	2 June 2022	(4028054)
CALWAY , Mrs Angela Josephine (Angela Calway, Angela Josephine Higgins)	48 Foxfield Avenue, Bristol, Gloucestershire, BS32 0BW. 5 September 2021	Wards SolicitorsJenny Pierce and Rebecca Parkman, 6-8 Fountain Court, New Leaze, Woodlands Lane, Bradley Stoke, BS32 4LA.	2 June 2022	(4037363)
CASEMORE, Carol Diane	Rodwell House, Rowtown, Addlestone, Surrey, KT15 1HH. 20 May 2020	Owen White & Catlin LLP, North Lodge, Esher Park Avenue, Esher, Surrey, KT10 9NP (Lisa Carol Kirtland.)	2 June 2022	(4030260)
CHILD, Margaret Jean (Jean Child)	2 New Farm Cottages, Hatherden, Andover, SP11 0HQ. 19 February 2022	Middletons Solicitors, East Gate House, 94 East Street Warminster, BA12 9BG (Caroline Anne Dingley.)	2 June 2022	(4028887)
CLARK, Mr Frank Ernest (Frank Clark, Ernest Frank Clark)	3 Church View, Andover, Hampshire, SP11 7QF. 6 March 2022	Talbot Walker LLPExecutors of Frank Ernest Clark, 16 Bridge Street, Andover, SP10 1BJ.	2 June 2022	(4037970)
CLEGHORN, Michael	4 Nightingales Court, Cokes Lane, Amersham, Buckinghamshire, . 8 August 2021	Inheritance Planning Company, c/o A R Adams, 18 Walsworth Road, Hitchin, Herts SG4 9SP (18 Walsworth Road Hitchin, Herts SG4 9SP.)	2 June 2022	(4030198)
COOMBER, David Arthur James	17 Hollands Way, East Grinstead, RH19 3EN. 13 August 2021	Mayo Wynne Baxter LLP, 3 Bell Lane, Lewes, BN7 1JU (Jennifer Ruby Paice.)	2 June 2022	(4026459)
COOPER, Katharine Enid	14 Frys Lane Yateley GU46 7TJ, . 8 August 2021	New Patch, 11 Haydon Place Yateley GU46 7UA (Carole Wall.)	2 June 2022	(4037275)
COPE, James Roger	Flat 1, 47 Marmion Road, Southsea, Hampshire PO5 2AT. Civil Servant. 4 January 2021	Wedlake Bell LLP, 71 Queen Victoria Street, London EC4V 4AY. Ref: DFT/F343-326. (Andrew Stephen Fraser and Neil Anthony Fraser)	6 June 2022	(4035358)
COPE, Angela Maud	4 Chalkridge Road, Portsmouth PO6 2BE. Nurse (Retired). 16 July 2020	Wedlake Bell LLP, 71 Queen Victoria Street, London EC4V 4AY. Ref: DFT/F343-326. (Andrew Stephen Fraser and Neil Anthony Fraser)	6 June 2022	(4035545)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives		
CORBET, Christopher John	44 Westleigh, Warminster, Wiltshire, BA12 8NH. 3 December 2021	Middletons Solicitors, East Gate House, 94 East Street Warminster, Wiltshire, BA12 9BG (Charles John Barrington Goodbody.)	2 June 2022	(4027917)
CROSS, Yvonne Sally	Surrey Hills Nursing Home, Brook Road, Wormley, Godalming Surrey, GU8 5UA. 9 August 2021	Neale Turk Rochfort, 12 Kings Ride Camberley Surrey GU15 4JG (Raymond Cross and Keith Cross.)	2 June 2022	(4027319)
CROWTHER, Miss Ann Lilias Margaret	Flat 22, Purchace Manor, BURGESS HILL, RH15 0GP. Retired Special Needs Teacher. 1 January 2022	Bennett Oakley Solicitors, 19 Keymer Road, HASSOCKS, BN6 8AD.	2 June 2022	(4037754)
CUMMINS, Mr Kenneth John (Kenneth Cummins)	The Eyebrow, 8 London Road, Andover, Hampshire, SP10 2PU. 25 January 2022	Executor of Kenneth Cummins c/o Talbot Walker LLP, 16 Bridge Street, Andover, SP10 1BJ.	2 June 2022	(4037971)
CUYPERS, Bernarda Alida	The Malthouse Bay Road, Gillingham, Dorset, SP8 4EW. 20 January 2021	Rutters Solicitors, 2 Bimport, Shaftesbury, Dorset, SP7 8AY (Duncan Peter Weir & James George Wood.)	2 June 2022	(4027964)
DALLING, Brian John	Palm Court Guest House, 371 Kingsway, Hove, BN3 4QD. 12 February 2022	Burt Brill & Cardens, 30 Old Steyne Brighton, BN1 1FL	2 June 2022	(4027819)
DEAN, Mavis	Ford Place Nursing Home, Ford Street, Thetford, IP24 2EP. 20 September 2021	Ashtons Legal, Fornham Business, Court Fornham St. Martin, Bury St Edmunds, Suffolk, IP31 1SL (Alan Michael Brown, Matthew Robert Gregor Cameron.)	2 June 2022	(4036109)
DOOLEY, Eileen Bernadette	21 Connaught Court, Alma Road, Windsor, SL4 3GZ. 16 May 2021	NewLaw Solicitors, Helmont House, Churchill Way, Cardiff, CF10 2HE (Mo Hussain.)	2 June 2022	(4028854)
EBURNE, Myrtle Allison	Retired Nurses National Home, Riverside Avenue, Bournemouth, BH7 7EE, . 11 December 2021	Luff Brook Carter, 521 Ringwood Road Ferndown Dorset BH22 9AQ (Nicholas Stanley Ward.)	2 June 2022	(4036306)
EDWARDS, Jill Christine	57 Mayflower Close Dartmouth Devon, TQ6 9JN. 13 October 2021	Wollens, 67 The Terrace Torquay Devon TQ1 1DP (K R Edwards.)	2 June 2022	(4026910)
EDWARDS , Douglas Llewellyn	Prestemede Home for the Elderley Presteigne Formerly of Dol-Y-Fan Farm, Llangunllo, Tref-y-clawdd, Powys, LD7 1SU. 9 September 2021	PCB Solicitors LLP, 7 Wylcwm Place, Knighton, Powys, LD7 1AE (Aaron James Powis & Francis Nock .)	2 June 2022	(4029410)
ELLINS, Florence Emma	Moorfield Road, Harefield, UB9 5NF. 6 January 2022	Frazine Johnson Solicitors, Old Bank Chambers, 32 Station Parade, Denham, UB9 5ET (Margaret Thomas.)	2 June 2022	(4029417)
ENGLISH, Mrs Ruth	Ford Cottage, Gatehead, Alston, Cumbria, CA9 3EB. 21 June 2021	Arnison Heelis Solicitors, 1 St Andrew's Place, Penrith, CA11 7AW.	2 June 2022	(4037380)
ETHERINGTON, Mr Victor John (Victor Etherington)	Tymaen House, Maesteg, Pen-y-bont ar Ogwr, CF34 0EH14 Elmhurst Road, Thatcham, Berkshire, RG18 3DG. 7 December 2021	Mary McCutchan c/o Hugh James, Two Central Square, Cardiff, CF10 1FS.	2 June 2022	(4037367)
FRANCES, RODNEY STANHOPE LYTTLETON	57 Gainsborough Drive, Lawford, Manningtree, Essex, CO11 2LF. 11 January 2022	SPARLINGS SOLICITORS, 13 High Street Manningtree, Essex, CO11 1AQ (SPARLINGS SOLICITORS.)	2 June 2022	(4029269)
FALCINI, Vivian Florence Mary	Abundant Grace Nursing Home, Firle Road, Seaford, BN25 2JD - previously of 48 Hartfield Road, Seaford, BN25 4PW. 19 January 2022	Mayo Wynne Baxter LLP, 3 Bell Lane, Lewes East Sussex BN7 1JU (Viviana Zelide Maria Rovelli.)	2 June 2022	(4028056)
FANCOURT, Mrs Kathleen Grace (Kathleen Fancourt)	88 Little Breach, Chichester, West Sussex, PO19 5TZ. 16 September 2021	Emma Grundy c/o Kenny Solicitors, The Old Rectory, Old Rectory Drive, Eastergate, PO20 3XH.	2 June 2022	(4037371)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
FARMERY, Stephen Charles	23 Clarkfield, Mill End, Rickmansworth, Hertfordshire, . 23 July 2021	Sanders Witherspoon LLP, Unit 7, Radford Crescent, Billericay Essex CM12 0DU (Joanna Louise Farmery and Fiona Mary Farmery.)	2 June 2022	(4028433)
FIRTH, Pauline Mary	High Wadsworth, Branthwaite, Workington, Cumbria, CA14 4SZ . 7 January 2022	Brockbanks Solicitors, 44 Duke Street, Whitehaven, Cumbria, CA28 7NP. Ref: OSW/AM/MAT29298. (Tom Corban)	8 June 2022	(4035372)
FOTHERGILL, Mrs Joyce	1 Abbots Close, Aylsham, NORWICH, NR11 6HQ. 5 December 2021	Rachael Drew, The London Gazette (24660), PO Box 3584, Norwich, NR7 7WD.	1 July 2022	(4035015)
GIBBONS, Sylvia Lorna	61 Mirador Place, Plymouth, PL4 9HE. 31 January 2022	Portcullis Legals Ltd, New Creaven House, 3 Sandy Court, Plymouth PL7 5JX (Trevor Winston Worth.)	2 June 2022	(4030387)
GONZALEZ, Valerie Constance (Valerie Constance Thomas)	Middle Farm, Kellaton, Kingsbridge, Devon, TQ7 2ES. 1 November 2021	Beers LLP, 29 Fore Street, Kingsbridge, Devon TQ7 1AA	2 June 2022	(4031733)
GONZALEZ, Jose Luis (Jose Luis Gonzalez Bellido)	Middle Farm, Kellaton, Kingsbridge, Devon, TQ7 2ES. 20 November 2021	Beers LLP, 29 Fore Street, Kingsbridge, Devon TQ7 1AA	2 June 2022	(4031924)
GOOCH, William Lawrence	1 St. Margarets Avenue, Aspley, Nottingham, NG8 5GD. 5 November 2021	ELM Legal Services, Kingsland House, Gas Lane, Bristol BS2 0QL (James David Emsley.)	2 June 2022	(4026793)
GOWEN, Dorothy Marion	3 Shrublands Way Gorleston Great Yarmouth Norfolk, NR31 8DU. 6 February 2022	Mears Hobbs LLP, 92 High Street, Gorleston Great, Yarmouth, Norfolk NR31 6RH	2 June 2022	(4030629)
GRAHAM, Andrew Alexander Kenny	Fell End, Butterwick, Penrith, Cumbria, CA10 2QQ. 9 May 2021	Bishop & Sewell LLP, 59-60 Russell Square London, WC1B 4HP (Nicholas Hugh Barlow.)	3 June 2022	(4027992)
GRAHAM , Richard Norman	Parkfield Residential Home, 256 London Road, Carlisle, CA1 2QS and formerly of Hetherside, Kirklinton, Carlisle, CA6 6AZ. 11 December 2021	Burnetts Solicitors, Victoria House, Wavell Drive, Rosehill Industrial Estate, Carlisle, CA1 2ST (Richard William Miller and Julius Robert Beresford Deane.)	2 June 2022	(4030279)
GRIFFITHS, Mrs. Jeanette	12 Townsend Court, Priory Way, MALMESBURY, SN16 0FB. Nursing Officer (Retired). 16 February 2022	Brevitts Solicitors, Brevitt Jane, 5 Oak Tree Lane, Selly Oak, BIRMINGHAM, B29 6JE.	23 June 2022	(4035635)
GRIFFITHS, Mr. Peter George	Upton Mill Care Home, Mercer Way, TETBURY, GL8 8FH12 Townsend Court, Priory Way, MALMESBURY, SN16 0FB. Local Government Officer (Retired). 17 February 2022	Brevitts Solicitors, Brevitt Jane, 5 Oak Tree Lane, Selly Oak, BIRMINGHAM, B29 6JE.	23 June 2022	(4035647)
HARRIS, MISS BETTY JUNE (BETTY HARRIS)	9 Tower Road, Paignton, Devon, TQ3 3EP. 7 October 2021	Raymond Patterson c/o Knapman & Co, 16 DENDY ROAD, PAIGNTON, TQ4 5DB.	2 June 2022	(4037967)
HALL, Irene Annie Juliette	Vicarage Cottage, Awre, Newnham, Gloucestershire, GL14 1EL. 6 March 2022	Okells Francis Law, Oakfield Hill Street Lydney, Gloucestershire, GL15 5HE (Andrew David Morris.)	2 June 2022	(4029411)
HARKER, Joseph Derek	Scar Barn, Langthwaite, Richmond, DL11 6RE. 28 May 2021	Tilly Bailey & Irvine LLP, Newgate, Barnard Castle, DL12 8NG (Denys Richard Harker & Helen Clare Dexter.)	2 June 2022	(4028871)
HARRIS, Mrs Hazel Doreen	62 Ramsey Drive, Arnold, NOTTINGHAM, NG5 6QQ. 20 October 2021	Janet Moon, Sherwood House, 7 Gregory Boulevard, NOTTINGHAM, NG7 6LB.	2 June 2022	(4036409)
HARVEY, Edward Phillip	194 Harley Shute Road St Leonards- on-Sea East Sussex, TN38 9JH. 16 January 2021	NewLaw Solicitors, Helmont House, Churchill Way, Cardiff CF10 2HE (Newlaw Trustees Limited.)	2 June 2022	(4030566)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given		
HAVAS , Anne Veronica	4 Church Close, Inkberrow, Worcester, WR7 4EA. 25 January 2022	Hallmark Whatley Hulme, 3-5 Sansome Place, Worcester WR1 1UQ (Stephanie Anne Bucknall.)	2 June 2022	(4028359)	
HAYWARD, Hilary Frances	2 Garth Terrace Knighton Powys, LD7 1HB. 4 July 2021	K Anderson & Co., 27 Station Road, Knighton, Powys LD7 1DT (Karen Anderson, Sally Hyde.)	2 June 2022	(4030142)	
HEAD, Mr Thomas	Flat 35, Homeholly House, WICKFORD, SS11 7DS. 13 August 2021	Scott Sutherland, The London Gazette (24682), PO Box 3584, Norwich, NR7 7WD.	2 June 2022	(4036573)	
HENNEVELD, Maureen Grace	Llwyncelyn, Cefnypant, Whitland, Carmarthenshire, SA34 0TS. 9 March 2021	Redkite Law LLP, Disgwylfa, St. John Street, Whitland, Carmarthenshire SA34 0AN (Ref: ges/w/hen00089/2) (Nicola Jane Ratnett & Petra Anne Henneveld.)	2 June 2022	(4028401)	
HOLLAND, John Edward	15 Livingstone Close Oakham Rutland, LE15 6FN. 21 January 2021	SHB Solicitors Ltd, 1 Berridge Street, Leicester LE1 5JT (Clive Roy Cook and Janice Heather Allen.)	2 June 2022	(4028343)	
HUGHES, Madge	Hollymount Care Home, 3 West Park Road, Blackburn, BB2 6DE. Machinist - Textile Industry (Retired). 13 February 2022	Whiting & Mason Solicitors, 24 Union Road, New Mills, High Peak, SK22 3ES. (Joanne Catherine Potts and Janis Elftesi)	6 June 2022	(4035594)	
JACKSON, Anthony William Frank	7 Eveline Road Mitcham, CR4 3LF. 7 February 2021	Preuveneers LLP, 103-105 London Road, Mitcham, Surrey CR4 2JA (Kelly Anne Cirillo.)	2 June 2022	(4026950)	
JARIWALA, Bharati Angela	29 Fruen Road, Feltham, Middlesex, TW14 9NT. 25 December 2021	NewLaw Solicitors, Helmont House, Churchill Way, Cardiff, CF10 2HE (Jaimesh Patel.)	2 June 2022	(4029152)	
JOHNSON, Maureen Joan	37 Holly Close, Speedwell, Bristol, BS5 7XR. 1 January 2022	Star Legal incorporating Davis Wood Solicitors, 884 Fishponds Road, Fishponds, Bristol BS16 3XB (Alexander Johnson.)	2 June 2022	(4026755)	
KILLICK, SHIRLEY CLARE	24a Bullsland Lane, Chorleywood, Rickmansworth, WD3 5BQ. 26 November 2020	ARCHER RUSBY SOLICITORS, 3 Norfolk Court Norfolk Road, Rickmansworth, WD3 1LA	2 June 2022	(4026743)	
KANE, Mrs Joan Shirley	2 Colmers Walk, The Roundabout, BIRMINGHAM, West Midlands, UK, B31 2UR. Retired. 21 September 2021	Louise Nancy Wise, The London Gazette (23778), PO Box 3584, Norwich, NR7 7WD.	2 June 2022	(3998397)	
KEMP, Edwin	3 Linton Close, Bishopgarth, Stockton- on-Tees, TS19 8XL. 15 February 2022	Appleby, Hope & Matthews, 35 High Street Normanby, Middlesbrough, TS6 0LE (Gordon Sisson.)	2 June 2022	(4025786)	
KENNEDY, June	Flat 54 Stratheden Court, Esplande, Seaford, BN25 1JP. 20 February 2022	Barwells Solicitors, 10 Sutton Park Road, Seaford, BN25 1RB (Stephen John Ash, Margaret Kennedy.)	2 June 2022	(4028881)	
KINGHAM, John Neville	17 Mossfield Cobham Surrey, KT111 1DF. 25 March 2021	Carter Bells LLP, Kings Stone House, 12 High Street, Kingston upon Thames KT1 1HD (Sally Katherine Benthall, Timothy John Ranger, Edward Anfrid Pringle Benthall.)	2 June 2022	(4028036)	
KNIGHTS, Nadine Carole (Carole Knights)	Fair Green Farms, Hill Road, Middleton, King's Lynn, Norfolk, PE32 1RN. 27 March 2021	Kenneth Bush Solicitors, 11 New Conduit Street, King's Lynn, Norfolk, PE30 1DG (Kenneth Bush.)	2 June 2022	(4028132)	
LAHIFF, MAUREEN EVELYN	6 Stourview Close, Mistley, Manningtree, Essex, CO11 1LZ. 12 January 2022	SPARLINGS SOLICITORS, 13 High Street Manningtree, Essex, CO11 1AQ (SPARLINGS SOLICITORS.)	2 June 2022	(4029355)	
LYES, BETTY	1 Elianore Road Colchester, Essex, CO3 3RX. 3 October 2021	BIRKETT LONG LLP, 1 Amphora Place, Sheepen Road Colchester, Essex, CO3 3WG (PAMELA ANN HARROD, JOHN WILLIAM HARROD.)	2 June 2022	(4029508)	

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
LEE, David Edward	88 Fairlee Road, Newport, Isle of Wight, PO30 2EL. 21 January 2022	Glanvilles Damant Solicitors, The Courtyard, St Cross Business Park, Monks Brook, Isle of Wight, PO30 5BF (Patricia Carol Butt.)	2 June 2022	(4026998)
LEREGO, Susan	Flat 1 Albermarle Court, Cranleigh Rise, Rumney, Cardiff CF3 4AL. General Domestic Assistant (Retired). 26 November 2021	Spicketts Battrick Law Practice, 54 Albany Road, Roath, Cardiff CF24 3RR. Ref: ST/SB0356/33. (Richard James Battrick)	6 June 2022	(4035524)
LEWIS, Patricia (Pat)	15 Long Ridings Avenue, Hutton, BRENTWOOD, CM13 1DZ. 20 November 2021	Daniel Lewis, 577 Broadgate, Weston Hills, SPALDING, PE12 6DP.	2 June 2022	(4036225)
LITTLE, Margaret	Ingleby Nursing Home, Lamb Lane, Ingleby Barwick, STOCKTON-ON- TEES, TS17 0QP. 15 February 2022	Countrywide Tax & Trust 3 June 2022 Corporation Ltd, GABLES HOUSE, 62 KENILWORTH ROAD, LEAMINGTON SPA, WARWICKSHIRE, CV32 6JX.		(4037837)
LOVELY, Dean Adrian	29 West Street, Royston, Barnsley, S71 4DG . 29 May 2021	Bury & Walkers LLP, Britannic House, Regent Street, Barnsley, S70 2EQ (Luke Michael Reynolds .)	2 June 2022	(4030140)
LUNEY, Mrs Ruth Watkins	125 Forefield Lane, LIVERPOOL, L23 9SL. 28 September 2021	Richard Luney, The London Gazette (24679), PO Box 3584, Norwich, NR7 7WD.	2 June 2022	(4036184)
LYNN, Constance	2 Alleyn Park, Dulwich, London, SE21 8AE. 31 January 2021	Grant Saw Solicitors Llp, Grant Saw House, 8 Tranquil Passage, Blackheath, London, SE3 0BJ (Raymond Howard Crudgington, Michael Edward William Clary, Jeremy David Lynn And Judith Lynn.)	2 June 2022	(4030377)
MORRIS, MRS AUDREY MARY	29 Rowleys Court, Sandhurst Street, LEICESTER, LE2 5AS. 19 January 2022	Philip J Hammond & Sons, Richard David Hammond, 47 Friar Lane, LEICESTER, LE1 5QX. (Katherine Mary Cohoon, )	2 June 2022	(4031916)
MACLEAN, Peter Thomas	Flat 7, 23-29 Warrior Square, St Leonards on Sea, East Sussex, TN37 6BG. 16 August 2021	Heringtons LLP, 37-39 Gildredge Road, Eastbourne, BN21 4RX (Sally Louise Kinsey.)	2 June 2022	(4031025)
MANN , Marcus Anthony ((otherwise Marcus Mann))	Flat 8, 105 Myddleton Road, Wood Green, London, N22 8NE (previously of Flat 7, 2 - 4 Broadlands Road, London, N6 4AN and 17E Bishops Road, London, N6 4HP). Retired. 22 May 2020	Dilwyns Solicitors, Oxford Chambers, Temple Street, Llandrindod Wells, Powys, LD1 5DL. (Patricia Ann Mills.)	2 June 2022	(4035173)
MANT, Janie Lilian	Millbrook House, Child Okeford, Blandford Forum, Dorset, DT11 8EY. 3 March 2022	Rutters Solicitors, 2 Bimport, Shaftesbury, Dorset, SP7 8AY (Derek John Rowntree.)	2 June 2022	(4027931)
MARTIN, Sandra Lynne	Elburton Heights Nursing Home, 33 Springfield Road Plymouth, Devon, . 24 October 2021	Windeatts Solicitors LLP, 48 Fore Street Kingsbridge Devon TQ7 1PE (Christopher Kendall and Matthew James Clarke .)	2 June 2022	(4036304)
MATHIAS, Catherine Mary (Catherine Mary Graham)	8 George Street Place, Warminster, Wiltshire, BA12 8SD. 18 December 2021	Farnfields Solicitors, 37 High Street, Warminster, Wiltshire BA12 9AJ (Nigel Vaughan Edwards, Claire Nicola Edwards.)	2 June 2022	(4031751)
MCMILLAN, Joseph	110 Stonald Road, Whittlesey, PETERBOROUGH, PE7 1QW. 21 August 2021	Glenn McMillan, The London Gazette (24672), PO Box 3584, Norwich, NR7 7WD.	2 June 2022	(4035833)
MERRETT, Ursula Pauline	79 Tutnalls Street, Lydney, Gloucestershire, GL15 5PQ. 5 September 2021	Okells Francis Law, Oakfield Hill Street Lydney, Gloucestershire, GL15 5HE (Andrew David Morris.)	2 June 2022	(4028875)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
MILLOTT, Anthony Arthur	Flat 1a, 22 Grove Park London, SE5 8LH. 3 April 2021	IWC ESTATE PLANNING & Management Ltd, Suite 3, 9-13 Bocking End, Braintree, Essex, CM7 9AE (Carol Ann Millott.)	2 June 2022	(4028356)
MOORE, Alan	New House Farm, Hill Wootton, Warwick, CV35 7PP. 11 April 2021	Cocks Lloyd, Riversley House, Coton Road, Nuneaton, Warwickshire CV11 5TX (John William Gibbs and Brian Thomas Baker.)	2 June 2022	(4030548)
MOORE, Elsie Joyce	Cedar House, 249-251 Southend Road, Stanford Le Hope, Essex, SS17 7AB. 7 November 2021	Fisher Jones Greenwood, 104-106 Kings Avenue, Holland On Sea, Essex CO15 5EP (Ref: JG/ 217698/1/MOORE) (Susanne Louise Grimwade.)	2 June 2022	(4033533)
MORAN, Peter Thomas	53 Queensway, Penwortham, PR1 0DT. 4 December 2021	Vincents Solicitors, 7 Rope Walk, Garstang PR3 1NS (Catherine Elizabeth McIntyre.)	2 June 2022	(4027912)
MORRIS, Olive Edith	Belmore Lodge, 20 Milford Road, Lymington, Hampshire, SO41 8DJ . 27 February 2022	Heppenstalls Solicitors Ltd, 49 South Efford House, Milford Road, Lymington, Hampshire SO41 0JD	2 June 2022	(4029880)
MORTIMER, James Claude	31 Mathews Close, Halstead, CO9 2BJ. 4 July 2021	Birkett Long, 1 Amphora Place, Sheepen Road, Colchester, Essex, CO3 3WG (Peter John Game.)	2 June 2022	(4031646)
MORTON, Doreen Harriet (Doreen Harriett Morton)	Penvearne, Pennance Lane, REDRUTH, TR16 5TL. 18 October 2021	Countrywide Tax & Trust Corporation Ltd, GABLES HOUSE, 62 KENILWORTH ROAD, LEAMINGTON SPA, WARWICKSHIRE, CV32 6JX.	3 June 2022	(4037840)
MORTON, Mr John Leslie	3 Morris Orchard, Cam, DURSLEY, GL11 5LZ. Retired Airline Captain. 23 October 2021	Mark George Morton, 3 Morris Orchard, Cam, DURSLEY, GL11 5LZ.	3 June 2022	(4037854)
MOTT, Enid Dora	3 Brookside Dalton Thirsk, North Yorkshire, YO7 3HY. 5 March 2022	Eccles Heddon LLP, 5 South End Bedale, North Yorkshire, DL8 2BJ (The Partners of Eccles Heddon LLP.)	2 June 2022	(4027815)
MUSSON, Roger Thomas	Sandlers End, 8 Coombe Park Wroxall Isle Of Wight, PO38 3PH. 4 March 2022	A J Careless Solicitors, 19 Church Street Ventnor, Isle of Wight, PO38 1SN (Anthony John Careless.)	2 June 2022	(4027322)
NAYLOR, Eric	33 Applegate House, Seymour Road, Trowbridge, Wiltshire, BA14 8FT. 11 January 2022	McCloy Legal, 4 The Shambles, Bradford On Avon, Wiltshire BA15 1JS (Alison De Del Valle, Sally Thomson.)	2 June 2022	(4029268)
NEWTON, Sheila Rose (Sheila Rose Charter Sheila Rose Webb)	25 Essex Road, Romford, Essex, RM7 8BE. 26 November 2021	Mullis & Peake LLP, 194 Hutton 2 June 2022 Road, Shenfield, Essex, CM15 8NR (Manzurul Islam.)		(4026923)
NICHOLS, Diana Lesley	56 Heathlands Westfield, Hastings, East Sussex, TN35 4QZ. 3 February 2022	Wykehnm-Hurford Sheppard & Son 2 June 2022 LTD, 6 High Street, Battle, East Sussex TN33 0AE (Sally Reeve, Kathryn Anne Steele, Peter Robert Steele.)		(4027821)
NORRIS-JONES, Sheila Elise	137 Broad Street Guildford, GU3 3BJ. 30 June 2021	DMH Stallard LLP, 1st FLoor Wonersh House, The Guildway, Old Portsmouth Road, Guildford, Surrey GU3 1LR (Timothy Charles Norris-Jones, Charles Henry Pfister.)	2 June 2022	(4028071)
OLUBODUN , Gilbert Olatunde Dongas	51 Browning Street, London, SE17 1LU . 9 October 2021	Co-op Legal Services Limited, Aztec 650 Aztec West, Almondsbury Bristol, BS32 4SD (Ref: CTS / 5978144P / Olubodun ) (The Co-operative Trust Corporation .)	2 June 2022	(4028337)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
PRESTEDGE, Mr Michael John	47 Albany Court, Robertson Terrace, Hastings, East Sussex, TN34 1JH. 22 December 2019	PERCY WALKER & CO LLP, Aileen Stacey FRANCIS, Robertson Chambers, The Memorial, Hastings, East Sussex, TN34 1JB.	2 June 2022	(4033768)
PARKES, Mary Catherine	Pixham Firs Cottage, Pixham Lane, DORKING, RH4 1PH. 20 December 2020	Martin Surrey Goddard, The London Gazette (24619), PO Box 3584, Norwich, NR7 7WD.	2 June 2022	(4031690)
PATTINSON, Pamela Maud	32 Scotts Farm Road, EPSOM, KT19 9LJ. 14 December 2021	John Rolfe, 32 Scotts Farm Road, EPSOM, KT19 9LJ.	5 June 2022	(4036297)
PAYNE, Jill Valerie	8 Roxburgh Croft, Leamington Spa, CV32 7HT. 17 October 2021	Ward & Rider, 2 Manor Yard, New Union Street, Coventry CV1 2PF (Sofia Eileen Gryparis & Gareth Eastgate & Jason Claridge.)	2 June 2022	(4029220)
PELLEGRINO, Mr Guiseppe	32 Waverley Road, Bristol, Somerset, BS48 3LQ. 24 February 2022	Philippa Jones c/o Right Legal Group, 16 Stanier Way, Wyvern Business Park, Derby, DE21 6BF.	2 June 2022	(4037962)
PERRY, Ronald	48 Castle Rise, Rumney, Cardiff, CF3 9BB. 28 December 2001	Howells Solicitors, Fitzalan House Fitzalan Court Fitzalan Road Cardiff CF24 0EL (John Vincent O'Connell.)	2 June 2022	(4031560)
PFISTERMULLER, Barbara Joyce	30 Aplins Close, Harpenden, AL5 2QB. 19 April 2021	Harold Bell Infields & Co, Devon House, 174 Kingston Road, Ewell, Surrey KT19 0SD (Katherine Louise Maxie Pfistermuller, David John Worrall.)	2 June 2022	(4030184)
PICKLES, Miss Frances Ruth	Harp Hill, Station Road, CARLISLE, CA4 9PL5 Archer Street, Harrington, WORKINGTON, CA14 5QJ. 26 July 2021	Arnison Heelis Solicitors, Matthew Rogers, 1 St. Andrews Place, PENRITH, CA11 7AW.	2 June 2022	(4036337)
PLUNKETT, Russell Miles Oliver	24 Stephenson Drive, Frome, Somerset, BA11 2XD. Estate Agent. 11 August 2021	Thatcher & Hallam LLP Solicitors, Island House, Midsomer Norton, Radstock, BA3 2HJ. (Karen Amanda Plunkett-Jones)	6 June 2022	(4035448)
POXON, Christina Mary	2-4 Watcombe Circus Nottingham Nottinghamshire, . 9 August 2021	Gladstone Solicitors, 22 Market Place Arnold Nottingham NG5 6ND (Catherine Poxon .)	2 June 2022	(4035045)
RAMSAY, Edwin Ewan (Lord Edwin Ewan Ramsay)	54 New Hall, Lodge Park Low Road Harwich, Essex, CO12 3TS. 28 August 2021	Birkett Long, 1 Amphora Place, Sheepen Road Colchester, Essex, CO3 3WG (Andrew Baker.)	2 June 2022	(4031745)
REDFERN, Anne	Flat 6 Jays Court, 6 Montagu Road, Highcliffe, Christchurch, Dorset, BH23 5JZ. 1 October 2021	Kiteleys Solicitors Limited, 40 Haven Road, Canford Cliffs, Poole Dorset BH13 7LP (Eileen Coles, Robert Coles, Alison Barbara Nicholls .)	2 June 2022	(4026994)
RENSHAW, Evelyn	White House Residential Home, 5-7 Grove Road, Wrexham, LL11 1DY formerly of 166 Pen y Maes Road, Holywell, CH8 7HL, . 20 September 2021	Swayne Johnson Solicitors, High Street Church House LL17 0RD (Graham Malcolm Renshaw and Janet Evelyn Mytton.)	2 June 2022	(4031445)
RIAZ, Najeeba Sultana	27 Aldborough Spur, Slough, SL1 3EN formerly of 512 Broad Oak Court, Farnham Road, Slough, SL2 1HY. 4 March 2022	barrett and thomson, one pegasus court, 25 herschel street, slough SL1 1TQ (Catherine Anne Matharoo .)	2 June 2022	(4030200)
RICHARDS, Mrs Lilian Marguerite	21 Seabank, The Esplanade, PENARTH, Vale of Glamorgan, CF64 3AR. 26 July 2021	Alison Marguerite Cooper, c/o Lucas Law Solicitors, Unit B Off Edge Station Approach, PENARTH, Vale of Glamorgan, CF64 3EE.	3 June 2022	(4034867)
RIGDEN, Martin Peter	3 Porlock Court, Oakwood, Derby, DE21 2PG. 7 October 2021	Straw and Pearce, 18 Rectory Place, Loughborough, Leicestershire LE11 1UU (Christopher Martin Rigden, Kim Louise Williams.)	2 June 2022	(4026552)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	ions of Persons to whom notice of claims to be of claims are to be given given les, in parentheses, of	
ROCHE, Susan Rose	Tudor Cottage, 33 Greenways, Hinchley Wood, Surrey, KT10 0QH. 20 May 2021	BDB Pitmans LLP, One Bartholomew Close London EC1A 7BL (Penelope Jane Chapman, Elizabeth Ann Neale.)	2 June 2022	(4026999)
SCHOFIELD , Paul Anthony	75 Granville Road Audenshaw Manchester, M34 5TA. 14 January 2021	Sleigh, Son & Booth, 1 Market Street, Denton, Manchester M34 2BN (Philip Lawrence Schofield.)	2 June 2022	(4036301)
SCOTT, Mr Terence Alan (Terence Scott)	6 Maidenhead Court Park, Maidenhead, Berkshire, SL6 8HN. 1 June 2021	Laura Quinn c/o Penningtons Manches Cooper LLP, Matrix House, Basing View, Basingstoke, RG21 4DZ.	2 June 2022	(4037963)
SHAH, Vijyaben Premchand Jeshang ( (Vijyaben Jeshang Shah))	PO Box 42869-00100, Nairobi, Kenya. Housewife. 15 January 2021	Alder Wills & Probate Ltd, 377-399 London Road, Camberley, Surrey GU15 3HL. (Peter Arthur Hoare)	6 June 2022	(4035430)
SIGGS, Doris Edith	Timberley Eastbourne Road Halland Lewes, BN8 6PU. 13 February 2022	Dawson Hart Solicitors Limited, The Old Grammar School Church Street Uckfield East Sussex TN22 1BH (Mrs Sonia Elaine Townley.)	2 June 2022	(4030861)
SIMPSON, Roydon Eric (Royden)	37 Herbert Road, Bearwood, West Midlands, B67 5DD. 18 December 2021	Nicholls Brimble Bhol Solicitors, Grays Court, 11 High Street, Harborne, Birmingham B17 9NT	6 June 2022	(4031722)
SIMPSON, Hilda Grace	129 Casewick Road, West Norwood London, SE27 0TA. 13 December 2021	Co-op Legal Services Limited, Aztec 650, Aztec West Almondsbury, Bristol, BS32 4SD (Ref: KRI / 6025125P / Simpson) (The Co-operative Trust Corporation .)	2 June 2022	(4029105)
SLONE, Anthony Frederick	12 School Close, Broughton Moor, Maryport, Cumbria, CA15 7SA. 5 March 2022	Milburns Solicitors, 21-23 Oxford Street, Workington, Cumbria CA14 2AL	2 June 2022	(4031644)
SMART, Roger Eric	23 Cloudesdale Road, London, SW17 8ET. 4 September 2021	McGlennons Solicitors, 157 Arthur Road, London SW19 8AD (Ref: WE/ 00171/21/Smart) (William Allan Essex.)	2 June 2022	(4031889)
SMITH, Olive Ivy	Bucklesham Grange Care Home, 141 Bucklesham Road, Ipswich IP3 8UB formerly of 16 Ruskin Road, Ipswich, IP4 1PT. 12 February 2022	Birketts LLP, Providence House, 141-145 Princes Street, Ipswich IP1 1QJ (Louise Mary Long, James Maynard Dinwiddy.)	2 June 2022	(4026470)
SMITH, Marion Frances Dorothy	15 Carrington Road, High Wycombe, Buckinghamshire, HP12 3HY. 6 December 2020	Blaser Mills Law, Liston Exchange, Cromwell Gardens, Marlow, Buckinghamshire, SL7 1BG (Colin George Smith & Alan Francis Smith.)	2 June 2022	(4030388)
SMITH, Mrs Rosemary Ann (Rosemary Smith)	14 Hall Road, Chatham, Kent, ME5 8PL. 4 September 2020	Claire Dewhirst c/o Kingsguard Legal, The Manse, 33 New Road, Stourbridge, DY8 1PA.	2 June 2022	(4037369)
SOANES, Patricia Ethel	11 Station Hill, Harleston, Norfold, IP209EX. 16 September 2021	Banner Jones Solicitors, Corner House, Union Street, Mansfield, NG18 1RP (Deborah Laura Soanes.)	2 June 2022	(4030504)
SPROSTON, Cynthia	5 Ilam Close, Silverdale, Newcastle- under-Lyme, Staffordshire, ST5 6RX. 28 February 2022	Tinsdills Solicitors, 15-19 Marsh Parade, Newcastle under Lyme, Staffordshire ST5 1BT (Ref: SL/ 110748/0002) (Peter James Charles Hamilton.)	2 June 2022	(4033549)
STEWART, Jean Isobel Mary (Jean Isobel Mary Stewart Patterson Alexieva)	162 Rayleigh Road, Hutton, Brentwood, Essex, CM13 1PN. 13 January 2022	Claire Flood Solicitors, 162 Rayleigh Road, Hutton, Brentwood, Essex, CM13 1PN (Annabella Nikolaeva Fortune, Louisa Nikolaeva Felicity Freer, Dorothy Ann Ellen Catchpole.)	2 June 2022	(4031571)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
STILWELL, Sheila Margaret	8 Walmsley Road Broadstairs, CT10 2BH. 4 June 2021	Barnes Marsland Solicitors Limited, 103 High Street, Broadstairs, Kent CT10 1JS (Jacqueline Miles, Jane Mary Mackenzie.)	2 June 2022	(4030596)
STOREY, Mrs Irene	30 Croft Avenue, Penrith, Cumbria, CA11 7RQ. 12 June 2021	Arnison Heelis Solicitors, 1 St Andrew's Place, Penrith, CA11 7AW.	2 June 2022	(4037373)
TARR, ALAN JOHN	24 Barns Close Dulverton, Somerset, TA22 9EA. 21 February 2021	Risdon Hosegood, Bank Chambers, 10 Bank Square Dulverton, Somerset, TA22 9BU	2 June 2022	(4027966)
TOWNEND , Anne	40 Long Meadow, Abermule, Montgomery, Powys SY15 6JS. Housewife. 4 October 2021	MILWYN JENKINS SOLICITORS, Cranford House, 1 Severn Square, Newtown, Powys SY16 2AG. (Janet Lesley Cook.)	2 June 2022	(4035176)
TAAFE, Georgina	251 Bolton Road, Worsley, MANCHESTER, United Kingdom, M28 3QB. 20 December 2021	Linda Chmielewska, Little Meadow, Paul, PENZANCE, TR19 6UY.	2 June 2022	(4024533)
TAYLOR, Susan Catherine	1 Abraham Close Great Gonerby, Grantham, NG31 8PS. 23 July 2021	Chattertons Legal Services Limited, 30 Avenue Road Grantham, NG31 6TH (Chattertons Trustee Corporation Limited, Susan Jane Westlake.)	2 June 2022	(4030356)
TEASDALE, Mildred	49 Viceroy Street, Seaham, SR7 7HY. 16 December 2021	Mortons Law Limited, 11 Marlborough, Seaham SR7 7SA (Linda Charlton and Susan Adamson.)	2 June 2022	(4030623)
THORNS, Marjorie Alice	Greville House Care Home, 1 Greville Road, Richmond, TW10 6HR. 20 January 2022	Dixon Ward Solicitors, 16 The Green, Richmond, Surrey TW9 1QD (Ref: GBW/THO173.003) (Partners of Dixon Ward.)	2 June 2022	(4031902)
VAISEY, Clive Russell	Flat 39 Vanbrugh Court, Wincott Street, London, SE11 4NS. 21 February 2022	Judge & Priestley LLP, 108-110 Beckenham High Street, Beckenham, BR3 1EB (Prashanth Chandra.)	2 June 2022	(4030122)
WALKER, Vasant Lata	12 Wellington Street, St Ives, Cambridgeshire PE27 5AZ. Registered Nurse (Retired). 6 October 2020	Hugh James Solicitors, Two Central Square, Cardiff CF10 1FS. Ref: SMO/WAL2213/1. (Hugh James Trust Corporation Limited)	7 June 2022	(4035570)
WALKER, Donald	12 Wellington Street, St Ives, Cambridgeshire PE27 5AZ. Model Maker (Retired). 30 July 2020	Hugh James Solicitors, Two Central Square, Cardiff, CF10 1FS. Ref: SMO/WAL2216/1. (Hugh James Trust Corporation Limited)	7 June 2022	(4035451)
WALTON, Jean Thompson	'Rosedale' The Old Vicarage Catterick Garrison North Yorkshire (formerly of 20 Ashdowne Close Little Crakehall Bedale North Yorkshire), DL8 1LQ. 10 February 2022	Eccles Heddon LLP, 5 South End, Bedale, North Yorkshire, DL8 2BJ (David Alisdair Shackleton and Nicholas James White.)	2 June 2022	(4028040)
WARNER, Yvonne Ann	17 Neighbrook Close, Redditch, B97 5YA. 19 October 2021	FBC Manby Bowdler, 7 Church Road, Redditch, B97 4AD	2 June 2022	(4027859)
WARREN, Ann	4 Macdonough Cottages, Station Road, Harwich, CO12 3LX. 7 December 2021	Irwin Mitchell (Sheffield), Riverside East, 2 Millsands, Sheffield, GB, S3 8DT (Ref: ED/05420146)	2 June 2022	(4029124)
WATERHOUSE , Isobel Laleah	Church Park Cottage East Portlemouth, Salcombe Devon, TQ8 8PF. 25 March 2021	Windeatts Solicitors LLP, 48 Fore Street, Kingsbridge Devon TQ7 1PE (Crispin Waterhouse, Fiona Hughes .)	2 June 2022	(4029116)
WHELAN, Freda (Freda White)	Flat 36 Thirlmere Court ! Harding Close Liverpool, L5 6SX. 26 September 2021	Paul Crowley & Co Solicitors Limited, 232 Breck Road Everton L5 6PX (Paul Francis Crowley .)	2 June 2022	(4026545)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives  Date before which notice of claims to be given given		
WHITE, Mr Brian Alexander (Brian White)	19 Dean Gardens, Brighton, East Sussex, BN41 2FW. 20 January 2022	Philippa Jones c/o Right Legal Group, 16 Stanier Way, Wyvern Business Park, Derby, DE21 6BF.	2 June 2022	(4037372)
WHITTLEY, Marion Joan	2 Ash Lane Winsford, Minehead, TA24 7JH. 28 October 2021	RISDON HOSEGOOD, 18 Fore Street Williton, Taunton, TA4 4QD	2 June 2022	(4026711)
WILLIAMS, Ellaline (Ella Williams)	Mae House, Tintern, Cas-Gwent, NP16 6SZ. 15 January 2021	Margraves Ltd, Old Court Chambers Spa Road, Llandrindod Wells, Powys, LD15EY (Helen Edwina Hayward .)	2 June 2022	(4028034)
WITCHELL, Mr Douglas	43 Cedar Lodge, Lynwood Village, Rise Road, Ascot, Berkshire, SL5 0FG. 17 February 2022	Charlotte Woods c/o Herrington Carmichael LLP, Building 2, Riverside Way, Watchmoor Park, Camberley, GU15 3YL.	2 June 2022	(4037965)
WOODFORD, Mr Richard Arthur	Flat 1, Wraycroft, REIGATE, RH2 0NR. Marketing Procurement Director. 1 December 2021	Deborah Dawn Woodford, The London Gazette (24585), PO Box 3584, Norwich, NR7 7WD.	2 June 2022	(4030764)
YOUNG, Audrey Ivy Gladys	South View Station Road, Over Wallop Stockbridge, Hampshire, SO20 8JA. 17 February 2022	Preston Redman, Hinton House, Hinton Road, Bournemouth BH1 2EN (Paul Thomas Boulton Smith and Adrian Clive Falck.)	2 June 2022	(4030524)
YOUNG, Mr Hilary Alistair (Hilary Young)	The Old Cottage, South Church Street, Bakewell, Derbyshire, DE45 1FD. 10 October 2021	Suzanne Porter c/o Wake Smith Solicitors, No.1 Velocity, 2 Tenter Street, Sheffield, S1 4BY.	2 June 2022	(4037366)
MAGUIRE, Rita	13 Gowy Close, SANDBACH, CW11 1YF. 22 November 2021	Paul maguire, The London Gazette (24663), PO Box 3584, Norwich, NR7 7WD.	6 July 2022	(4035036)

## Terms and conditions relating to submission of notices

The Gazette (which includes the London, Belfast and Edinburgh Gazette) is an official public record and the United Kingdom's longest continuously published newspaper. It is managed by The National Archives (a non-ministerial government department) under a concessionary contract with The Stationery Office Limited ('TSO' or the "Publisher", as defined below). Any capitalised terms referred to in these terms and conditions relating to submission of notices are defined below.

By placing a Notice in The Gazette you are consenting to put official information permanently on the public record and in the public domain, online (in The Gazette website or via The Gazette mobile app), in print, and via a data service (rather than by having to search for notices on The Gazette website, customers can either create a pdf of the Notices that they are interested in, or subscribe to an electronic version of The Gazette (in full or in part) which is provided as a data service).

These terms should be read in conjunction with:

- 1 The Publisher's privacy policy
- 2 The Publisher's policies relating to submission of notice

which together govern the submission of Notices.

Notice Placers, as defined below, may place a Notice in The Gazette either because there is a statutory requirement to do so, or to do so voluntarily to put information in The Gazette in order to create an official record of fact. All Notice Placers must have the authority to place the notice that they submit for publishing. TSO, as the Publisher, is required to verify the authority of Notice Placers who place Notices and has the authority to refuse to publish Notices from Notice Placers whose authority cannot be effectively verified.

Notices received for publication usually fall under the following broad headings:

Church, Companies, Environment and Infrastructure, Health and Medicine, Honours and awards, Money, Parliament and Assemblies, People, Royal Family and State. Further information can be found at <a href="https://www.thegazette.co.uk">www.thegazette.co.uk</a>.

These terms and conditions ( "Terms and Conditions") govern submission of Notices (as defined below) to The Gazette. By submitting Notices, howsoever communicated, whether at the website <a href="www.thegazette.co.uk">www.thegazette.co.uk</a> (the "Website") or by email, post and/or facsimile, the Notice Placer (as defined below) agrees to be bound by these Terms and Conditions. Where the Notice Placer is acting as an agent or as a representative of a principal, the Notice Placer warrants that the principal agrees to be bound by these Terms and Conditions.

The Publisher reserves the right to modify these Terms and Conditions at any time. Such modifications shall be effective immediately upon publication. By submitting Notices to The Gazette after the Publisher has published such modifications, the Notice Placer, including any principal, agrees to be bound by the revised Terms and Conditions.

#### 1 Definitions

1.1 In these Terms and Conditions: "Authorised Scale of Charges" means the scale of charges set out at in the printed copy of the Gazette or at www.thegazette.co.uk/place-notice/pricing, as modified from time to time; "Charges" means the payment due for the acceptance of a Notice by the Publisher payable by the Notice Placer as set out in the Authorised Scale of Charges; "Forwarding Service" means the service provided to use The Gazette's postal box for correspondence in order to prevent a personal address from being publicly and permanently available on the official public record. "Local Newspaper Notice" means any notice placed in a local newspaper other than The Gazette; "Notice" means all advertisements and state, public, legal or other notices (without limitation) submitted for potential publication in The Gazette by the Notice Placer, save in respect of any Local Newspaper Notice, to which other terms may apply where indicated in these Terms and Conditions; "Notice Placer" means any agency, company, firm, organisation or person who has requested to place a Notice in The Gazette, whether acting on their own account or as agent or representative of a principal; "Publisher" means The Stationery Office Limited or TSO, with registered company number 03049649, acting in accordance with the concessionary contract awarded by The National Archives.

1.2 the singular includes the plural and vice-versa; and

- 1.3 any reference to any legislative provision shall be deemed to include any subsequent re-enactment or amending provision.
- 2 By submitting a Notice to the Publisher, the Notice Placer agrees to be bound by these Terms and Conditions which, unless stated otherwise in these Terms and Conditions, represent the entire terms agreed between the parties in relation to the publication of Notices in The Gazette and which every Notice shall be subject to. For the avoidance of doubt, these Terms and Conditions shall prevail over any other terms or conditions (whether or not inconsistent with these Terms and Conditions) contained or referred to in any correspondence or documentation submitted by the Notice Placer or implied by custom, practice or course of dealing which the parties agree shall not apply, unless otherwise expressly agreed in writing by the
- 3 The Publisher reserves the right, to be exercised at its sole and absolute discretion, to make reasonable efforts to verify the validity of the Notice Placer.
- 4 The Publisher may, at its sole and absolute discretion edit the Notice, subject to the following restrictions:
- 4.1 the sense of the Notice submitted by the Notice Placer will not be altered;
- 4.2 Notices shall be edited for house style only, not for content;
- 4.3 Notices can be edited to remove obvious duplications of information;
- 4.4 Notices can be edited to re-position material for style;
- 4.5 any additions, amendments or deletions required in order to include the minimum necessary information set out in any Notice guidelines shall be confirmed with the Notice Placer; and
- 4.6 subject to clause 5 below, no amendments to the text (other than those made as a consequence of 4(i) (v) above) shall be made without confirmation from the Notice Placer.

For the avoidance of doubt, the Notice Placer agrees and accepts that, subject to the limited rights to edit any Notice referred to above, it is the Notice Placer that shall be solely responsible for the content of any Notice, including its validity and accuracy and that the Publisher shall not be responsible for, nor shall have any liability in respect of such content in any way whatsoever.

5 The Notice Placer accepts that it submits a Notice entirely at its own risk and that the Publisher shall have the sole and absolute discretion whether to accept a Notice for publication; whether to publish it (including after acceptance); the timing of any publication of a Notice; or whether to remove or withdraw the Notice after publication, such decision to be final. The Notice Placer must satisfy itself as to the legal, statutory and/or procedural requirements and accuracy relating to any Notice and, for the avoidance of doubt, the Publisher shall have the sole and absolute discretion to refuse to publish or withdraw from publication (if already published) any Notices where the content of the Notice, in the Publisher's sole and absolute opinion, may not comply with any such requirements. In instances where publication has not yet taken place, the Publisher shall notify the Notice Placer of any action required to remedy any deficiency and publication shall not take place until the Publisher is satisfied that such action has been taken by the Notice Placer. Where publication has taken place the Notice placer will be contacted with the proposed remedy which may include, but is not limited to, removal, reinsertion, retraction or substitution notice.

6 Save for any liability that cannot be excluded or restricted by law, The National Archives or the Publisher's (including any successor organisations, affiliates, officers, directors, agents, subcontractors and/or employees) total aggregate liability whether arising from the acts and/or omissions of The National Archives or the Publisher arising out of or made in connection with any Notice or otherwise for any and all liabilities, losses, damages, expenses, costs (including all interest, penalties, legal costs and other professional costs and/or expenses), whether in contract (including under an indemnity or warranty), tort (including negligence), misrepresentation, equity, breach of statutory duty, strict liability or otherwise incurred shall be limited to one hundred and fifty per cent (150%) of the value of the Charge paid for such Notice under these Terms and Conditions.

6A Other than as set out in clause 6 above, neither the Publisher, nor The National Archives, shall have any liability in respect of any act and/or omission of the Notice Placer and/or any third party or in respect of any Notice submitted by any Notice Placer for potential publication in The Gazette, which the Notice Placer warrants and accepts is solely its responsibility.

7 For the avoidance of doubt, subject to clause 6 above and save for the payment of the Charges, in no circumstances shall either party be liable for any losses including, without limitation, loss of revenues, profits, contracts, business or savings or anticipated savings, any loss of goodwill or reputation, or any special, indirect or consequential damages (however arising, including negligence).

8 Where the Publisher is responsible for a failure to publish, or has published a Notice in error or with an error, the Publisher shall, at no charge to the Notice Placer, either publish the Notice at the next suitable opportunity, or in the event of an error, remove the Notice or publish a reinsertion, retraction or substitution Notice as appropriate at the next suitable opportunity. Both parties agree (including on behalf of any principal, if applicable) that this shall be the sole remedy of the Notice Placer (including any principal, if applicable) and the full extent of the limit of the Publisher's liability in these circumstances.

9 In the event that the Publisher believes, in its sole opinion, a Notice Placer is submitting Notices in bad faith, is in breach of clause 11 below, or has dealings with Notice Placers who are in breach of these Terms and Conditions or has breached such Terms and Conditions previously, the Publisher may require further verification of information to be provided by the Notice Placer and may, at its sole and absolute discretion, delay publication of those Notices until it is satisfied that the Notice it has received is based on authentic information.

10 The location of the Notice in The Gazette shall be at the discretion of the Publisher. For the avoidance of doubt, the Notice shall be published in the house style of The Gazette.

#### 11 The Notice Placer warrants:

11.1 that it has the right, power and authority to submit the Notice; 11.2 the Notice is true and accurate in all respects and does not mislead or contain potentially fraudulent information;

11.3 the Notice is submitted in good faith, does not contravene any law (statutory or otherwise), nor is it in any way illegal, defamatory or an infringement of any other party's rights or of any applicable advertising regulations, guidelines or codes of practice, nor is it subject to any court order prohibiting such publication.

12 To the extent permissible by law the Publisher excludes all warranties, conditions or other terms, whether implied by statute or otherwise, relating to the placing of any Notices.

13 The Notice Placer agrees to fully indemnify (as a debt) and hold the Publisher and The National Archives or any successor organisation, including any affiliates, officers, directors, agents, subcontractors and employees harmless from all liabilities, costs, expenses, damages and losses (including, without limitation) any direct, indirect, consequential and/or special losses and/or damage, loss of profit, loss of reputation and/or goodwill and all interest, penalties and legal costs (calculated on a full indemnity basis) and all other professional costs and/or expenses (including legal costs) suffered or incurred (including negligence) in respect of any matter arising out of, in connection with or relating to any Notice (including, without limitation, clause 11 of the Terms and Conditions) and including (without limitation) in respect of any claim and/or demand (including threatened and/or potential claims or demands) made by any third party which may constitute a breach and/or potential breach by the Notice Placer (or their principal) of these Terms and Conditions or of any law and/or any of the rights of a third party. The Publisher shall consult with the Notice Placer as to the way in which such applicable claims, demands or potential claims or demands are handled but the Publisher shall retain the sole, absolute and final decision on all aspects of any matter arising from the aforementioned indemnity, including the choice of instructing legal representatives, steps taken in or related litigation and/or decisions to settle the case. The Notice Placer shall use best endeavours to provide, at its own expense, such co-operation and assistance as the Publisher may reasonably request including in respect of any principal (if applicable) and including, without limitation, the provision of and/or access to witnesses, access to premises and delivery up of documents and/or any evidence, including supporting any associated litigation and/or dispute resolution process.

14 The Notice Placer shall promptly notify the Publisher in writing of any actual, threatened or suspected claim made by a third party or parties against the Notice Placer and/or the Publisher in relation to a Notice. The Publisher reserves the right, following a claim or threatened claim, to immediately remove the Notice which is the subject of the complaint from the website at <a href="www.thegazette.co.uk">www.thegazette.co.uk</a> and all other websites controlled by the Publisher containing the

Notice, as well as from any other medium in which the Notice has been placed that is controlled by The Gazette, where possible. The Publisher may (at its sole and absolute discretion) require the Notice Placer to amend the Notice at its own cost before it agrees to republish the Notice if it is capable of rectification to avoid the claim, threatened or suspected claim. Any reinstatement of the Notice shall be at the sole and absolute discretion of the Publisher, whose decision in respect of such matter shall be final.

In exceptional circumstances, for example if the Notice was found by the Publisher to have been submitted by an unauthorised Notice Placer, the Publisher (upon receiving written approval from The National Archives) will remove a Notice in its entirety from <a href="https://www.thegazette.co.uk">www.thegazette.co.uk</a>, but will retain the Notice identification online and include explanatory text saying why and when the Notice was removed.

15 The Notice Placer acknowledges that the Publisher may re-use Notices and/or allow third parties to re-use Notices accepted for publication in The Gazette for ongoing Gazette-related activity – for example to create a company profile page on The Gazette website and where you have also opted for a newspaper advertisement as well as a Gazette notice – and hereby assigns to the Publisher for and on behalf of the Crown, all rights, including but not limited to, copyright and/or other such intellectual property rights (as applicable) in all Notices, and warrants that any such activity in respect of any Notice (including any activity in the preparation of such Notice for publication in The Gazette) by the Publisher and/or third parties does not and will not infringe any legal right of the Notice Placer or any third party. For the avoidance of doubt, all Notices and any content therein shall be Crown copyright and may be subject to the Open Government Licence (or any variation thereof).

16 The Notice Placer accepts that the purpose of The Gazette is to disseminate information of interest to the public as widely as possible in the public interest. The Notice Placer accepts that the information contained in the Notices published in The Gazette may be used for additional purposes by the Publisher or third parties after publication as stated above and that such use may be beyond the control of The Gazette. In such instances, the Notice Placer agrees that the Publisher shall have no liability whatsoever in respect of such use by the Publisher or third parties. The Gazette will at all times act with confidentiality, discretion and adhere to any legislative requirements.

17 The Notice Placer acknowledges and agrees that the publication of any Notice is subject to any court order and/or direction of the court or such other regulatory and/or enforcement authorities including the Information Commissioner's Office, the police, the Financial Conduct Authority (and such other related regulatory organisations), the Solicitors Regulation Authority and such other authorities as may be applicable (without limitation) and that the Publisher may delay, refuse to publish or withdraw from publication if it has received evidence to that effect and may not publish such notice until it has received written evidence from the court (as the Publisher may reasonably require from time to time) that demonstrates that any previous order and/or direction has been withdrawn and/or is no longer applicable (as the Publisher may reasonably require from time to time) and/or, subject to any statutory and/or applicable laws, The Gazette may share information and/or data related to the Notice and/or the Notice Placer's account related to such authorities and the Notice Placer hereby consents to such

18 In respect of any Local Newspaper Notice, this clause 18 shall apply. For the avoidance of doubt, all other terms of the Terms and Conditions shall apply to Local Newspaper Notices only to the extent that they do not conflict with the terms set out below. In the event of any conflict, the terms set out in this clause 18 shall prevail:

18.1 The Local Newspaper Notice may be placed in a local newspaper by any subcontractor and/or a third party organisation at the Publisher's sole and absolute discretion and the Notice Placer hereby consents to such use (including any activity that is ancillary and/or reasonably necessary to such use). For the avoidance of doubt, this may include the processing of personal data in accordance with the EU General Data Protection Regulation (Regulation 2016/679), and any legislation which implements, amends, re-enacts or replaces it in England and Wales, Northern Ireland or Scotland ("GDPR"), by the Publisher, any subcontractor and/or third party organisation, together with the local newspaper and related organisations;

18.2 The placement of a Local Newspaper Notice shall be upon the standard terms and conditions of the local newspaper in question in

addition to these Terms and Conditions. The Notice Placer expressly agrees to such local newspaper terms and by submitting a Local Newspaper Notice to The Gazette, expressly consents to the Publisher, its subcontractors and/or any applicable third party organisation agreeing to such terms on behalf of the Notice Placer; 18.3 To the extent that such local newspaper and the applicable terms allow, where the Publisher, any subcontractor, any third party acting on behalf of the Publisher and/or the local newspaper is responsible for any error, the Publisher shall arrange for the local newspaper to publish the corrected Local Newspaper Notice at no additional cost to the Notice Placer. Both parties agree (including on behalf of any principal, if applicable) that this shall be the sole remedy of the Notice Placer (including any principal, if applicable) and the full extent of the limit of liability in these circumstances;

In the event that a corrected Local Newspaper Notice is not published for whatever reason, the total aggregate liability of the Publisher and The National Archives, whether direct or indirect, and including (without limitation) all liabilities, losses, damages, expenses, costs (including all interest, penalties, legal costs and/or other professional costs and/or expenses) suffered or incurred, howsoever arising (including negligence), whether arising from the acts and/or omissions of the Publisher, The National Archives and/or the Notice Placer and/or any third party (including, without limitation, any principal of the Notice Placer) or arising out of or made in connection with the Notice or otherwise shall be limited to the value of the Local Newspaper Notice placed through The Gazette except that nothing in these Terms and Conditions shall limit or exclude any liability for fraudulent misrepresentation, or for death or personal injury resulting from the Publisher's or The National Archives' negligence or the negligence of the their agents, subcontractors and/or employees or third parties acting on behalf of the Publisher.

19 In respect of the use of the Forwarding Service, The Gazette will replace the Notice Placer or executor's address with The Gazette's postal box address in the Notice, and Local Newspaper Notice if applicable. All correspondence received will be sent on from The Gazette to the Notice Placer or executor (if different).

Please be aware that correspondence received by The Gazette's postal box may be opened (for example in circumstances where the intended recipient of the correspondence is not clear) and in using this service you are consenting to this, however The Gazette will at all

times act with confidentiality, discretion and adhere to any legislative requirements.

The Forwarding Service will remain in place for a period of 10 months from publication of the Notice in The Gazette, or 10 months from publication of the Notice in a Local Newspaper, in line with the Inheritance (Provision for Family and Dependants) Act 1975. After the 10 months period has lapsed any correspondence received will be returned to the sender where possible or securely destroyed, and the Notice Placer or executor's name and address details will be removed from the Forwarding Service.

20 The Notice Placer accepts that the Charges may be amended from time to time and will be payable at the rate in force at the time of invoicing unless otherwise agreed by the Publisher in writing. The Charges must be paid in full by the Notice Placer in advance of publication unless other requirements of the Publisher in respect of the payment of such Charges (as determined from time to time) are notified to the Notice Placer.

21 Please note that we use PayPage by Verifone on The Gazette to process credit or debit card transactions. This involves a £1.00 preauthorised transaction by your bank to check that funds are available and that sum is then reserved, although no actual deduction or charge takes place. The reserved funds then become available to the customer after the pre-authorisation expires usually between 3-7 days (as a guideline). We use this process to validate all card orders and this allows us to hold orders for goods that are out of stock or not yet published as we do not charge for these until the goods are in stock. By placing the order, you agree to such pre-authorisation processes.

22 If the Notice Placer wishes to make a complaint, all such complaints shall be submitted in writing to customer.services@thegazette.co.uk

23 Save in respect of The National Archives (or any successor organisation), a person who is not a party to these Terms and Conditions has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce any term of these Terms and Conditions but this does not affect any right or remedy of a party specified in these Terms and Conditions or which exists or is available apart from that Act.

24 These Terms and Conditions and all other express terms of the contract shall be governed and construed in accordance with the laws of England and the parties hereby submit to the exclusive jurisdiction of the English courts.

All communications on the business of The London Gazette should be addressed to The Gazette, PO Box 3584, Norwich NR7 7WD

Telephone: +44 (0)333 200 2434 Fax: +44 (0)333 202 5080

Email: london@thegazette.co.uk



	AUTHORISED SCALE OF CHARGES From 1 January 2022		Public sector placing All mandatory notices or state notices		All other advertisers	
	All charges are exclusive of VAT at the prevailing rate, currently 20%	XML, webform, Gazette template	Other	XML, webform, Gazette template	Other	
	No VAT is payable on printed copies	Ex VAT	Ex VAT	Ex VAT	Ex VAT	
	Corporate and Personal Insolvency Notices	20.00	£24.60	£75.90	£103.60	
	(2 - 5 Related Companies/Individuals charged at double the single rate)	20.00	£49.20	£151.80	£207.20	
1	(6 - 10 Related Companies charged at treble the single rate)	20.00	£73.80	£227.70	£310.80	
	[Pursuant to the Insolvency Act 1986, the Insolvency (England and Wales) Rules 2016, Companies (Forms) (Amendment) Regulations 1987 and any subsequent amending legislation]					
2	Deceased Estates Notices Pursuant to s.27 Trustee Act 1925			£75.90	£103.60	
	All other Notices - charged by event	20.00	£24.60	£75.90	£103.60	
3	(2 - 5 Related events will be charged at double the single rate)	20.00	£49.20	£151.80	£207.20	
	(6 - 10 Related events will be charged at treble the single rate)	20.00	£73.80	£227.20	£310.80	
	If you are unsure how to price your notice or your notice contains more than 40 events please contact london@thegazette.co.uk					
4	Offline proofing		£44.50		£47.20	
5	Late advertisements - accepted after 11.30am, two days prior to publication		£44.50		£47.20	
6	Withdrawal of Notices - after 11.30 am, two days prior to publication		£24.60	£75.90	£103.60	
7	Other services					
	A brand, logo, map, signature image	£63.45	£63.45	£68.85	£68.85	
	Forwarding service for Deceased Estates	£63.45	£63.45	£68.85	£68.85	
	Newspaper placement for Deceased Estates (webform and template only)	£215.00		£215.00		
	Redaction of information within a published notice	£216.40	£216.40	£228.35	£228.35	
	Reinsertion of notice	£24.60	£24.60	£75.90	£103.60	

- A single edition of the printed copy is available to notice placers for £4.50 and non-notice placers for £9.00 (VAT exempt)
- An annual subscription to the printed copy is available to notice placers for £1,100.00 and non-notice placers for £2,200.00(VAT exempt)
- An annual subscription to the pdf copy is available to all customers for £690.00 (VAT exempt)
- A commemorative edition of this issue, or any past issue of The Gazette, is available for £70.00 (VAT exempt)
- Take advantage of The Gazette's research service: £30.00 for an individual/company name

For more information or to purchase a subscription or a commemorative edition, please telephone +44 (0)333 200 2434 or email customer.services@thegazette.co.uk, or visit www.thegazette.co.uk/shop

For more information and pricing for our data service please telephone +44 (0)1603 985949 or email data@thegazette.co.uk



a Williams Lea company

Published by TSO (The Stationery Office), a Williams Lea company, and available from:

#### Online

www.tsoshop.co.uk

Mail, Telephone, Fax & E-mail

TSO

PO BOX 29, Norwich, NR3 1GN

Telephone orders/General enquiries: +44 (0)333 202 5070

Fax orders: +44 (0)333 202 5080 E-mail: customer.services@tso.co.uk Textphone: +44 (0)333 202 5077

## TSO@Blackwell and other Accredited Agents



Published and printed in the UK by The Stationery Office Limited under the authority and superintendence of Jeff James, Controller of Her Majesty's Stationery Office and Queen's Printer of Acts of Parliament